A Critical Evaluation of the Application of Marine Citizenship in Sustainable Marine Management in the UK

EMMA MCKINLEY

Thesis submitted for the degree of Doctor of Philosophy

Bournemouth University

December 2010

This copy of the thesis has been supplied on condition that anyone who consults it is understood to recognise that its copyright rests with its author and due acknowledgement must always be made of the use of any material contained in, or derived from, this thesis.

ABSTRACT

Traditionally, governance of the marine¹ environment has been state driven dominated by a top-down approach to management. Recently, however, management has evolved into a more participatory, bottom up regime in an aim to address the historical failures associated with traditional marine management. This study seeks to establish the potential role of citizenship specific to the marine environment. The founding rationale of the research is the suggestion that with a greater level of public involvement and responsibility, marine management could be developed at a more sustainable, long-term level.

Following an extensive systematic literature review examining the role of citizenship in environmental management, and its potential applicability for the marine environment, a theoretical conceptual model of marine citizenship was generated. Through telephone interviews, the potential role of marine citizenship in marine management and policy delivery in the UK was evaluated. Results identified numerous factors with the potential to influence public expression of marine citizenship and these were further categorised into the key themes of education and personal attachment to the marine environment. In order to examine these issues further, two thematic case studies were designed to further evaluate their role in the inculcation of marine citizenship. It was found that current levels of marine education are insufficient, while sense of public connection with the marine environment was found to be low. Both factors were observed to be in need of improvement in order to engender a societal sense of marine citizenship in the UK. Synthesis of the results furthered the generation of the first working model of marine citizenship and established the necessary enabling factors required for successful expression of marine citizenship. In addition, specific management measures and recommendations for successful promotion of marine citizenship were defined.

The research concluded given that a movement towards enhanced public engagement in the process has been identified as the ideal situation for marine management, encouraging a sense of marine citizenship could be an effective delivery mechanism. Further to this, it was determined that, central to successful inculcation of marine citizenship is the recognition that its promotion will require considerable effort on the part of marine managers and governance bodies to address the current capacity issues associated with public engagement.

¹ For this research, 'marine' encompasses coastal, inter-tidal and the undersea environments of the UK.

Acknowledgements

I would first like to extend my continuing gratitude to my supervisors Professor Matthew Bennett and Dr. Stephen Fletcher for their support and advice throughout my PhD. In particular I would like to thank Steve for his patience with the roller coaster that has been my life over the past three years or so, and would like to express my sincerest thanks for his understanding and general concern for my personal well-being, as well as my research.

In addition to my supervisors, I would like to thank other members of Bournemouth University's academic team, namely, Ralph Clarke, Roger Herbert, Kate Welham, Ellie Hambelton, Lorraine Brown and Chris Wood for their help and support throughout my time at Bournemouth. I would also like to thank all of the administrative staff, particularly Louise Pearson, in Christchurch House for their help at various times and for generally just humouring me when I came in to distract them!

With regards to my research itself, I would like to thank the 400+(!) individuals who allowed me to pick their brains in the context of my research - without their participation, my work would not have been possible. Thank you to Duncan and Mark (undergrad students) for their aid in the collection of data and, in particular to my brother, Mark, who gave up his holiday time to come to Arran and help me in my case study research and obtained Excellent Brother brownie points!

Without a doubt my greatest thanks has to go to my countless friends and family who have been there during the tears and tantrums that go along with completing your PhD. First and foremost I would like to thank my four grandparents for their love and support over the last 9 years, although they gave us all a few scares at different points throughout my time at Bournemouth. Next I would like to thank my parents: To my long suffering mum and Leo who have both had to listen to the bad days and the good days, thank you for everything you have done over the last 10 years to get me here. I couldn't have done this without you. Lots of love xxxx And thank you to my Dad and Antoinette for their support across the miles, and in particular for sending me PhD inspirational shoes! X To my brothers and sister, Mark, Caroline, Iain and Shane: thank you all for helping me when you could (Mark), cheering me up and helping me sleep (Caroline), continually supporting me and spurring me on (Iain), reminding me that most importantly, I am your boring big sister (Shane) and above all for being my crazy siblings :)

I am lucky that I have a vast number of friends to thank who have put up with me at various points: I could full another ten pages if I said everything I wanted to. However, I'll try and keep it brief! Firstly, thanks to the PhD students in ApSci that I have met along the way for distracting, procrastinating with and more importantly motivating me at every point over the last few years. Some specific mentions need to be made however: Laura (for making your home my home and being there when you could have so easily have gone the other way), Kerry (my BFF!!! Love you) and Colyn, Sam, Kayt and Matt (for last minute 30STM jaunts and bouncy messages), April (for being utterly inspirational in every way), Sal

(my procrastination partner in crime), Lydia (my angel across the globe), Tina and Iain Joseph (for lots of cuddles), Jenn and Paul, Jane, Gill, Debbie, Ruthie, Ang (my girls from way up North), everyone Sheffside for your support from up in the Peaks, and finally Lynda and Peter Marston and family for your support and encouragement for the last three years.

Chris – I can't put into words how grateful I am that you have been there from such an early stage, through the highs, lows, tantrums and excitement that come as part of doing a PhD. Nothing I can say will ever thank you enough for your overwhelming and unconditional love and support for the whole of the Team M&M rollercoaster;) Two cups! Xxx

Finally, I would like to thank the family members who, sadly, are no longer around to see me get this far. In particular, I would like to thank my wonderful Granny whose love and support cannot be put into words. I hope you would be proud, and dedicate this to you.

Authors Declaration

I confirm that this thesis is my own work with the exception of the manuscripts below. As primary author on both papers, I was responsible for all aspects including idea development, data collection and analysis, interpretation and preparation of the manuscripts.

McKinley, E., and Fletcher, S. (2010). Individual responsibility for the oceans? An evaluation of marine citizenship by UK marine practitioners. Ocean & Coastal Management 53(7): 379-384.

McKinley, E., and Fletcher, S. (2010) Public Involvement in Marine Management? An Evaluation by UK Marine Practitioners. Littoral 2010 Conference Proceedings, London, 21st-23rd September. (in press)

CHAPTER STRUCTURE

Abstract	3
Acknowledgements	4
Detailed chapter structure	6
List of Tables	15
List of Figures	19
Abbreviations	20

PART ONE: Context of the Research

Chapter One: Introduction

1.1	Introduction	21
1.2	Rationale	21
1.3	Research aim and objectives	25
1.4	Structure of thesis	26
1.4.1	Part One: Context of the research	26
1.4.2	Part Two: Data collection and analysis	26
1.4.3	Part Three: Interpretation, synthesis and conclusions	27
	REFERENCES	28

CHAPTER TWO: THE TURN TO MARINE CITIZENSHIP

2.1	Introduction	31
2.2	Governance of the marine environment	31
2.2.1	Society and the marine environment	31
2.2.2	Governance and management of the global marine environment	33
2.2.3	Contemporary marine management	33
2.2.4	Marine management and the UK	35
2.2.5	Challenges to current UK marine management	37

2.3.	Turn to citizenship	38
2.3.1.	Citizenship	39
2.4	Environmental citizenship	41
2.4.1	Identification of models of environmental citizenship	42
2.5	Evolution of marine citizenship	46
2.6	Factors potentially influencing marine citizenship	48
2.6.1.	Information and its role in marine management	49
2.6.1.1.	Education	49
2.6.1.2.	Knowledge	51
2.6.1.3	Literacy	51
2.6.1.4.	Information	52
2.6.2.	Awareness, concern and marine citizenship behaviour	53
2.6.2.1.	Awareness	53
2.6.2.2	Concern	54
2.6.2.3	Behaviour and responsibility	55
2.6.4.	Personality and socio-demographics	57
2.6.5.	Socio-economic influences	59
2.7	Identification of factors specific to marine citizenship	59
2.7.1.	Public perception	60
2.7.2.	Public participation	61
2.7.3.	Proximity to the coast and sense of place	63
2.8.	Conceptual model of marine citizenship	65
2.9	Concluding comments	67
	REFERENCES	73

PART TWO: Data collection and analysis

CHAPTER THREE: GENERAL METHODOLOGY AND TELEPHONE INTERVIEW SCHEDULE

3.1.	Introduction	88
3.2	General methodological approach	88
3.2.1	Overall methodology	88

3.2.1.1.	Telephone interview schedule	93
3.2.1.2.	Thematic case study schedule	94
3.2.3	Quality assurance of data collection phases	94
3.2.4.	Overall morals and ethical considerations	95
3.2.5.	Scope and limitations of the general methodology	96
3.3.	Phase one data collection: Marine practitioner survey	98
3.3.1	Qualitative data collection	98
3.4.	Telephone interview schedule	99
3.4.1	Selection of interview method	99
3.4.2.	Use of semi-structured interviews	104
3.4.3.	Determination of case study option	105
3.4.4.	Method of telephone interviewee selection	110
3.4.5.	Telephone interview design	111
3.5	Telephone interview study	114
3.5.1.	Pilot study	114
3.5.2.	Actual study	115
3.6	Analysis of interviews	117
3.6.1.	Method of analysis	117
3.6.2.	Content analysis and coding	119
3.6.3.	Effect of interviewer	120
3.7	Limitations of phase one methodology	120
3.8	Summary	121
	REFERENCES	122

CHAPTER FOUR: RESULTS AND DISCUSSION OF PHASE ONE DATA COLLECTION

4.1	Introduction	126
4.2	Results of practitioner telephone interviews	126
4.2.1	Practitioner perception of current marine management	131
4.2.2.	Role of the individual in management of the marine	132
	environment	
4.2.3	Understanding of citizenship	134
4.2.4	Perception of citizenship for the environment	134

4.2.5.	Citizenship and the marine management	136
4.2.6.	Citizenship and management of the marine environment	137
4.2.6.1	Legislative implications of marine citizenship	138
4.2.7.	Potential factors influencing marine citizenship	139
4.2.8.	Responsibility for management of the marine environment	141
4.2.9.	Role of stakeholders and communities in marine management	142
4.2.9.1.	Implications of increased public involvement	144
4.2.10	Relationship between the marine environment and society	145
4.2.11.	Public awareness and concern for the marine environment	147
4.2.12	Methods of increasing public awareness of marine issues	148
4.2.13	Promotion of responsible behaviour towards the marine	151
	environment	
4.2.14	Summary of key points	153
4.3	Discussion of phase one results	154
4.3.1.	Practitioner perception of education and marine citizenship	154
4.3.2.	Influence of location on marine citizenship	156
4.3.3.	Livelihood and marine citizenship	157
4.3.4.	Participation, perception and culture relating to marine	158
	citizenship	
4.3.5.	Responsibility for management of the marine environment	159
4.4	Refined model of marine citizenship	161
4.5	Implications for phase two of research	163
4.5.1.	Identification of key themes for Phase Two	163
4.5.2.	Case study selection	164
4.6.	Summary	165
	REFERENCES	170

CHAPTER FIVE: METHODOLOGY OF PHASE TWO

CASE STUDIES

5.1	Introduction	174
5.2	Mixed methods approach	174
5.3	Case studies in research	175
5.3.1.	Types of case study research	176

5.3.2.	Use of multiple-case design case studies	177
5.3.3.	Limitations of overall case study research methodology	179
5.4.	Identification of key themes	180
5.4.1	Education	181
5.4.2.	Personal attachment	181
5.5	Selection of case study research strategy	181
5.5.1	Case study option one: urban and rural coastal communities	181
5.5.2	Case study option two: presence or absence of a marine	182
	campaign	
5.5.3.	Case study option three: sites based on marine environment	182
	type	
5.5.4.	Case study option four: combination	183
5.6	Identification of case study sites	183
5.6.1.	Poole bay	187
5.6.2.	Isle of Arran	188
5.6.3.	Birmingham	188
5.6.4.	Helsby, Cheshire	189
5.6.5.	North Antrim coast (Giant's Causeway)	189
5.6.6.	Milford Haven	190
5.7.	Data collection	190
5.7.1	Education	190
5.7.1.1.	School based questionnaires	190
5.7.1.2	Use of self-administered questionnaires	192
5.7.1.3.	Generation of questions	196
5.7.1.4.	Limitations of the method	198
5.7.1.5.	Selection of schools	199
5.7.1.6.	Pilot study for school based questionnaires	199
5.7.1.7.	Actual study	200
5.7.1.8.	Scope for teachers capacity survey	200
5.7.2.	Personal attachment	201
5.7.2.1	Use of structured interviews	202
5.7.2.2.	Generation of structured interviewed questions	204
5.7.2.3.	Interviewee selection	205

5.7.2.4.	Pilot study	205
5.7.2.5.	Actual study	206
5.7.2.6.	Limitations of the methodology	207
5.8	Data analysis and interpretation	208
5.8.1	Education	208
5.8.2.	Personal attachment	209
5.9	Summary	210
	REFERENCES	211

CHAPTER SIX: RESULTS AND INTERPRETATION OF PHASE TWO: UK CASE STUDIES

6.1	Introduction	214
6.2	Results of education thematic case study	214
6.2.1.	School survey	215
6.2.1.1.	Student knowledge of marine related issues	215
6.2.1.2.	Self assessed statements	225
6.2.1.2.1	Marine education available in school	227
6.2.1.2.2.	Capacity for decision making	227
6.2.1.2.3.	Student knowledge of the marine environment	228
6.2.1.2.4.	Links between society and the marine environment	229
6.2.1.2.5	Student awareness of marine issues	229
6.2.1.2.6.	Concern for the marine environment	230
6.2.1.3.	Evaluation of association between factors	231
6.2.1.4.	Additional student comments	233
6.2.3.	Teachers capacity survey	233
6.2.4	Education thematic case study concluding comments	234
6.3	Results of personal attachment thematic case study	235
6.3.1.	Analysis of personal attachment case study	235
6.3.2.	Public perception of marine management and governance	239
6.3.3.	Public awareness of the marine environment	240
6.3.4.	Public concern for the marine environment	242
6.3.5.	Lifestyle choices, public behaviour and the marine environment	243
6.3.5.1.	Awareness of everyday behavioural choices	244

6.3.5.2.	Food based consumer behaviour	245
6.3.5.3.	General consumer behaviour	246
6.3.5.4.	Public willing to change	247
6.3.6.	Public perception of responsibility for the marine environment	248
6.3.7.	Personal attachment and marine citizenship	251
6.3.8.	Associations between personal attachment factors	252
6.3.9.	Personal attachment thematic case study concluding comments	258
6.4	Discussion of thematic case study results	259
6.4.1.	Role of education for the marine management	260
6.4.1.1.	Potential education strategies	261
6.4.2.	Responsibility and management of the marine environment	263
6.4.2.1.	Behaviour and the marine environment	263
6.4.2.2.	Collaborative management between government and the public	264
6.4.3.	Socio-economics and marine citizenship	265
6.4.4.	Influence of location on factors of marine citizenship	266
6.4.5.	Socio-demographic factors	266
6.4.6	Personal connection to the marine environment	267
6.5	Summary	268
	REFERENCES	271

PART THREE: Interpretation, synthesis and conclusions CHAPTER SEVEN: SYNTHESIS AND DISCUSSION

7.1	Introduction	275
7.2	Marine citizenship: A synthesis of practitioner and public	276
	perceptions	
7.2.1	Role of the public in marine management	276
7.2.2.	Role of education in marine citizenship	277
7.2.3.	Influence of location	280
7.2.4.	Personal connection to the marine environment	281
7.2.5.	Behaviour	282
7.2.6.	Public participation in marine management	284

7.2.7.	Socio-economic factors	285
7.2.8.	Socio-demographic factors	286
7.2.9.	Summary of influences on marine citizenship	288
7.3	Marine citizenship: Development of conceptual model and	291
	definition	
7.3.1.	The need for a conceptual model	291
7.3.2.	What is marine citizenship?	292
7.3.3.	Conceptualisation of marine citizenship	294
7.3.3.1	Scenario 1	296
7.3.3.2.	Scenario 2a	297
7.3.3.3.	Scenario 2b	297
7.3.3.4	Scenario 3	298
7.3.4.	A conceptual model for marine citizenship	298
7.3.4.1.	Unsuccessful marine citizenship	304
7.3.4.2.	Frustrated marine citizenship	304
7.3.4.3.	Limited marine citizenship	305
7.3.4.4.	Successful marine citizenship	305
7.3.4.5.	Comments	306
7.3.5.	Progression of marine citizenship	306
7.3.5.1.	Route 1	307
7.3.5.2.	Route 2	308
7.3.5.3.	Route 3	308
7.4	Application to contemporary management strategies	308
7.4.1.	Implications of marine citizenship for current marine	308
	management and policy	
7.4.2.	Limitations of the conceptual model	310
7.4.3.	Strengths and weaknesses of marine citizenship	311
7.5	Summary	313
	REFERENCES	315

CHAPTER EIGHT: CONCLUSION

8.1	Introduction	320
8.2	Concluding comments	320
8.2.1	Role of marine citizenship	320
8.2.2.	Composition of marine citizenship	321
8.2.3.	A conceptual model of marine citizenship	323
8.2.4.	Key conclusions	324
8.3	Recommendations	325
8.3.1.	Recommendations concerning enabling factors	325
8.3.2.	Recommendations concerning personal components of marine	327
	citizenship	
8.4	Contribution of the research	329
8.5	Areas for future research	330
8.6	Future of marine citizenship in marine management	333
8.7	Final remarks	334
	REFERENCES	335

APPENDICES

Appendix 1	Email sent to marine practitioners prior to interviews	337
Appendix 2	Marine practitioner introduction and rationale letters used in Phase one.	338
Appendix 3	Marine practitioner pilot interview	341
Appendix 4	Actual marine practitioner interview	344
Appendix 5	Sample marine practitioner interview transcript (including highlighted segments used to determine themes and patterns during content analysis).	348
Appendix 6	Education thematic case study rationale letter	353
Appendix 7	Rationale for school involvement sent to participating schools.	354
Appendix 8	Instructions for student questionnaire	356
Appendix 9	Student questionnaire	357
Appendix 10	Instructions for Teachers' capacity survey	362
Appendix 11	Teachers' capacity questionnaire (Chapter Five)	363
Appendix 12	Personal attachment thematic case study community survey	366

Appendix 13Grid analysis conducted on the data collected throughout
the personal attachment thematic case study371

List of Tables

No.	Title of Table	Page
2.1	Common issues facing the marine environment and their association with society (Adapted from Atunes and Santos)	32
2.2	A selection of the political initiatives requiring consideration in UK marine management	36
2.3	Range of personality variables thought to have an influence on environmental citizenship	58
2.4	Strengths and weaknesses associated with public participation in environmental management and decision-making.	63
2.5	Comparison of the factors present in environmental citizenship and marine citizenship.	68
2.6	Factors influencing Marine Citizenship deduced through the literature review.	69
3.1	Differing approaches of inductive and deductive research in human geography (adapted from Kitchin and Tate, 2000)	89
3.2	Main characteristics of qualitative, mixed methods and quantitative approaches to research (Adapted from Creswell, 2009)	90
3.3	The three main mixed methods strategies that could have been applied to the research (adapted from Creswell, 2009)	91
3.4	Summary of research progression	92
3.5	Methods of assuring the quality of the data collected during	95
	both phases of the study.	
3.6	Benefits and problems associated with each of data collection technique considered (Adapted from De Leeuw, 1992).	101
3.7	Various dimensions of interviews (Adapted from Gillham, 2000).	102

15

3.8	Description of the five options considered for direction of	106
	research	
3.9	Comparison of the advantages and disadvantages associated	107
	with UK and International case studies and UK only case	
	studies.	
3.10	The most common qualitative research sampling techniques	111
	(Adapted from Marshall, 1996)	
3.11	The links between the knowledge gaps identified in the	112
	literature review, the data requirements of this project and the	
	appropriate questions generated for the telephone interviews	
3.12	Open Questions Vs Closed Questions (adapted from	114
	Denscombe, 2003).	
3.13	Table of the organisations interviewees were associated with.	116
4.1	Content Grid Analysis for Marine Practitioners Telephone	126
	Schedule	
4.2	Components of the two primary themes proposed for further	164
	investigation in the second phase of data collection.	
5.1	Illustration of the various types of research case study types	177
	(Adapted from Yin, 2003a; 2003b)	
5.2	Key themes identified through the marine practitioner	180
	interviews.	
5.3	Case study criteria as determined by Curtis et al. (2000) and the	184
	requirements of the research	
5.4	Criteria for assessing case study site suitability for the research	185
	(Adapted from Curtis et al., 2000)	
5.5	Description of thematic case study sites.	186
5.6	Current school system within the four UK countries (Adapted	191
	from the British Council, Online)	
5.7	Benefits and Weaknesses of the most common methods of self	195
	completion questionnaire administration (Adapted from	
	Bourque and Fielder, 1995).	
5.8	The potential advantages and disadvantages of using	196
	questionnaires as a data collection method (Adapted from	

Gillham, 2000)

5.9	Various approaches to research interviews (adapted from	203
	Denscombe, 2003).	
5.10	Commonly used methods of selecting sample participants	205
	(Adapted from Gillham, 2007).	
6.1	Average number of terms in each category in the three case	216
	study schools	
6.2	Indication of subjective knowledge based on students' term	217
	familiarity.	
6.3	Overall levels of student familiarity with marine terms (%	218
	response)	
6.4	Percentage of correct answers provided in the student quiz	219
6.5	Average number of correct answers (out of 5 questions) given	220
	by students in the marine quiz section of the student survey	
6.6	Table 6.6: Average number of marine related organisations	221
	known by students	
6.7	Student knowledge of organisations commonly associated with	221
	the marine environment (represented through percentage of	
	students selecting each option)	
6.8	Average number of marine designations known by students	222
6.9	Student knowledge of marine environmental designations	222
	(represented through percentage of students selecting each	
	option)	
6.10	Most common method through which marine information is	223
	sourced (represented through percentage of students selecting	
	each option)	
6.11	Television programmes related to the marine environment	224
	viewed by students (represented through percentage of students	
	selecting each option)	
6.12	Percentage responsibility attributed to management groups	224
	(represented through percentage of students selecting each	
	option)	
6.13	Percentage of students rating the self assessed questions as low,	226

moderate and high

6.14	Spearman correlation matrix of the relationships observed	232
	through the student education survey (N = 121 respondents)	
	(statistically significant p values given below each correlation)	
6.15	Results of the self-assessed personal attachment interviews.	237
6.16	Spearman's correlation matrix indicating the relationships	254
	between the factors investigated through the personal	
	attachment thematic case study. Q1-13 corresponds to the	
	interview questions as presented in Table 6.16.	
6.17	Association between community responses for awareness and	257
	concern which further explains the relationships identified by	
	PCA (Fig 6.1)	
6.18	Principal component analysis conducted on the data variables	258
	(a)-(m)	
7.1	Influencing factors of marine citizenship derived from synthesis	289
	of the results.	
7.2	Personal factors of marine citizenship and the optimum level of	293
	these factors at which a sense of marine citizenship would be	
	encouraged	
7.3	Conceptualisation of marine citizenship based on the presence	296
	of the key themes of education and personal attachment.	
7.4	Optimum conditions for successful expression of marine	301
	citizenship.	
7.5	SWOT analysis of marine citizenship and its potential role in	312
	marine management in the UK (adapted from Petts and Leach	

(2000) SWOT analysis on public participation)

List of Figures

2.1	Model of environmental citizenship (Taken from Hawthorne and	44
	Alabaster, 1999).	
2.2	Model of environmental citizenship (Taken from Berkowitz et al.,	45
	2005).	
2.3	A conceptual model of marine citizenship based on the observations	66
	of the literature review.	
3.1	Mixed methods design approach applied to the research.	93
4.1	Refined model of marine citizenship (Adapted from McKinley and	162
	Fletcher, 2010).	
5.1	Process of conducting a multiple-case case study (Adapted from	178
	Yin, 1994).	
6.1	PCA plot of relationships between factors investigated in the	256
	personal attachment case study.	
7.1	Progression of marine citizenship in the presence of key elements	299
	and appropriate enabling factors	
7.2	Model of marine citizenship comprising the identified components	302
	and recommended management strategies to engender successful	
	application of the concept	
7.3	Routes towards successful expression of marine citizenship	307

List of Abbreviations

ABPmer	ABP Marine Environmental Research
AONB	Area of Outstanding Natural Beauty
CCW	Countryside Council for Wales
Cefas	Centre for Fisheries and Aquaculture Science
Defra	Department for environment, food and rural
	affairs
EF	Enabling factors
EMS	Environmental Management Systems
EU	European Union
GES	Good Environmental Status
GESAMP	Joint Group of Experts on the Scientific Aspects
	of Marine Environmental Protection
HLMO	High Level Marine Objectives
ICZM	Integrated coastal zone management
JNCC	Joint Nature Conservation Commission
LOC	Locus of control
KS	Key stage
MCAA	Marine and Coastal Access Act
MCS	Marine Conservation Society
MMO	Marine Management Organisation
MP	Marine practitioner
MSFD	Marine Strategy Framework Directive
MSP	Marine Spatial Planning
NAW	National Assembly for Wales
NEP	New Environment Paradigm
NOAA	National
OCB	Organisational citizenship behaviour
OSPAR	Administers of the Oslo and Paris Commissions
PA	Personal attachment
QDAS	Qualitative data analysis software
RAMSAR	RAMSAR Convention on Wetlands Sites
SEPA	Scottish Environmental Protection Agency
SPA	Special Protected Area
SSSI	Site of Special Scientific Interest
SSMEI	Scottish Sustainable Marine Environment
	Initiative
UNCED	United Nations Conference on Environment and
	Development
UNCLOS	United Nations Convention on the Law of the
	Sea
UNECE	United Nations Economic Commission for
	Europe
UNEP	United Nations Environment Programme
UNESCO	United Nations Educational, Scientific and
	Cultural Organization

CHAPTER ONE

INTRODUCTION

1.1 INTRODUCTION

This chapter provides an overview of the research project, beginning with an introduction to the main academic and practical applications supporting the rationale for the research. The chapter outlines the aim and objectives of the research project followed by an overview of the structure of the thesis.

1.2 RATIONALE

With approximately 50% of the industrialised world residing within 50 kilometres of the coast, the marine² environment and its associated resources are of significant importance to the global community, providing a variety of ecosystem services and processes integral to everyday life (Boersma and Parrish, 1999; GESAMP, 2001; Rees et al., 2010; Fletcher et al, 2011). For example, within the UK alone, ecosystem services provided by the marine environment include regulation of pollution, provision of food, fuel and pharmaceuticals, physical and psychological wellbeing (Fletcher et al, 2011). Despite this, and the range of international agreements established to conserve the marine environment, frameworks regarding the regulation of human activity in the marine environment are limited (Rogers et al., 2007). Additionally, regardless of the evidence of social dependency on the marine environment, research suggests that there is a lack of public understanding of the strategic value of the marine environment (Costanza, 1999; Costanza et al., 1999; UNEP, 2006). This lack of understanding of the relationship between society and the marine environment has resulted in a sense of societal disconnection from the marine environment. Finkl and Kruempel (2005) view this lack of public connection as

² Throughout this chapter and the remainder of the thesis, 'marine' refers to coastal, intertidal and all undersea marine environments.

the primary barrier to sustainable marine management and an area that urgently needs to be addressed.

Further to this, traditional governance of the marine environment has been primarily top down, driven by the state but is undergoing an evolution into a more participatory, community based system (Edwards et al., 1997; McFadden, 2008; McKinley and Fletcher, 2010). As with other environmental policies, it is increasingly recognised that only partial responsibility for the marine environment lies with statutory government bodies (Hawthorne and Alabaster, 1999; Mamouni Limnios et al., 2009). Additionally, although the legal framework for inculcating marine citizenship is currently lacking, efforts are being made to encourage this participatory approach to marine management which has been recommended by a number of international conventions. Most relevant to this research are the recommendations set out by Agenda 21 at the Earth Summit in 1992 (Hawthorne and Alabaster, 1999; Kuijper, 2003; French, 2004; Matti, 2006), the Aarhus Convention (UNECE, 1998) and the Tbilisi Intergovernmental Conference on Education (UNESCO, 1978). Specifically, Agenda 21 assessed the importance of the part each individual plays in the promotion and development of environmental sustainability, whilst still implying that governments must take an active role in encouraging and motivating the global public to participate (Kuijper, 2003; Matti, 2006). Internationally, there have been movements towards improving public involvement and responsibility for the marine environment, with successes most notably in the U.S.A. (Steel et al., 2005; NOAA), Canada and Australia (Foster et al., 2005). Within the UK there has been increasing emphasis on community involvement and responsibility is further perpetuated by the Marine and Coastal Access Act (2009), which promotes a more participatory approach to marine management.

Whilst evidence from literature suggests that public participation and education are vital to improving marine management, relatively little is known of how public behaviour could be engaged effectively (Ducrotoy *et al.*, 2000; Kuijper, 2003; Jedrzejczak, 2004; Osborn and Datta, 2006). As changes in overall environmental governance occur, it has become increasingly apparent that the long term stability of the marine environment would benefit from a new form of

citizenship being developed: one that highlights the need for a greater sense of personal responsibility within society towards the environment as a whole. Rapid development of global coastlines has put the marine environment under unprecedented pressure, resulting in degradation of resources that can be partially attributed to collective lifestyle and behavioural choices made by individuals (McKinley and Fletcher, 2010). Traditionally, citizenship theory has been broadly defined as involving the rights, duties and public involvement associated with membership of a political community (Correia, 2002; Purcell, 2003; Seyfang, 2005). As citizenship theory evolved, the realisation that sustainable environmental management requires cooperation between citizens and their governments prompted the evolution of environmental citizenship (Dobson and Valencia Saiz, 2005). Environmental citizenship is now recognised as a vital part of environmental management by governmental agencies and management authorities (Strong, environmental 1998). Environmental citizenship is based on the ideal that citizens should be more environmentally responsible with an awareness of human impacts on the environment (MacRory, 1996; Hawthorne and Alabaster, 1999; Fletcher and Potts, 2007). An environmental citizen may, therefore, be defined as one who recognizes environmental issues when they arise, and considers them before taking an action that may influence the environment in a meaningful way. Theoretically, environmental citizens put long term benefits to the environment before short term gain, attempting to prevent environmental issues prior to their occurrence, and are generally continually interested in the environment, its sustainable development, and issues related to it (Roth, 1992). Given the benefits associated with environmental citizenship, it can be inferred that similar advantages could be attributed to the inculcation of marine citizenship.

Although the inclusion of the public has been described as integral to successful marine management (Ducrotoy *et al*, 2000; Kuijper, 2003; Jedrzejczak, 2004; Osborn and Datta, 2004), there has been relatively little consideration given to the role and facilitation of public participation in contemporary marine management. Therefore, the underlying rationale for this research is the proposition that a form of marine governance that engages individuals as actors in marine management through altered behavioural and lifestyle choices would

benefit marine management practices in the UK. It is proposed that this would be inherently more sustainable than traditional state driven marine policy as it would recognise the public as having a key role in the development and implementation of marine policy. Examples of the benefits for marine management expected to be associated with marine citizenship include enhanced public participation, inclusion of local knowledge in management practices and easier implementation of management practices. In addition, this research proposes that with a greater level of public involvement and responsibility, marine management could be developed at a more sustainable, long-term level. Theoretically this would aid the development of sustainable management of valuable marine ecosystems and resources, whilst facilitating economic and social development and stability. Societal dependency and the increasing concern regarding the impacts of the overall degradation of the marine environment provide a strong rationale for research into the development of marine citizenship.

The research is undertaken using an inductive approach (explained in Chapter Three, Table 3.1) which traditionally would not work from an initial hypothesis (Kell and Oliver, 2003). However, given the potential for interdisciplinary application of the research a tentative hypothesis for the research is suggested: Can a conceptual model for marine citizenship be developed and applied to contemporary marine management in the UK? With this provisional hypothesis in mind, the research examines the potential role of marine citizenship in UK marine management, considering the factors required to facilitate its application to contemporary marine management in the UK. This investigation into the role of marine citizenship provides a unique contribution to the ongoing debate regarding public involvement in marine management and planning.

1.3 RESEARCH AIM AND OBJECTIVES

The aim of the project is to critically evaluate the potential role of marine citizenship in the sustainable management of the marine environment in the UK. This will be achieved through the following objectives:

- 1. Development of a conceptual model of marine citizenship based on a systematic literature review.
- 2. The establishment of practitioner opinion on the applicability of marine citizenship to management of the marine environment in the UK.
- 3. Synthesis of a conceptual model for marine citizenship investigated through UK community thematic case studies.
- 4. Evaluation of the critical factors identified as having an impact on marine citizenship in the UK.
- 5. Establish a UK applicable definition and model of marine citizenship.
- 6. Generation of recommendations for the application of marine citizenship to contemporary sustainable marine management in the UK.

1.4 STRUCTURE OF THE THESIS

This thesis has three main parts, organised into eight chapters. A summary of these chapters and the resulting structure of the thesis are presented in Sections 1.4.1 - 1.4.3.

1.4.1. Part One: Context of the Research

Part One of the thesis provides an introduction to the research rationale and general methodological approach, discussing both its theoretical and practical applications. Chapter One outlines the underlying rationale behind this research and presents an overview of the structure of the thesis. Chapter Two provides an evaluation of current marine and coastal management, focusing in particular on the role of the public in management and decision-making processes. Relevant theories of citizenship in terms of the environment in general are then reviewed examining their relevance and application to the marine environment. The initial theoretical model for the concept of marine citizenship generated from observations made throughout the literature review is then discussed meeting the requirements of the first objective. Finally, the potential implications of marine citizenship for management of the marine environment are discussed and used to determine the main research areas forming the focus of Part Two of the thesis.

1.4.2. Part Two: Data Collection and Analysis

Part Two of the thesis begins by outlining the general methodology taken throughout the research, which follows a mixed methods approach. Chapters Three and Four focus on the data collection, analysis and brief interpretations of the first phase of data collection through a marine practitioner telephone interview schedule. Chapter Four concludes with a discussion of the key findings of the telephone interview schedule, in particular outlining how these findings would be used to guide the succeeding case study phase of data collection. Chapters Five and Six focus on the case study phase of data collection and analysis through thematic case studies. This phase of the research aimed to further examine the key relationships identified in the practitioner survey, allowing a comprehensive evaluation of influential factors to be carried out. The case studies provide valuable insights into community perceptions regarding marine management, as well as indications of general public awareness and concern for the marine environment. Chapter Six follows on to further interpret the key relationships identified throughout the research, and to assess the implications of these on the future promotion of the concept of marine citizenship. Part Two of the thesis meets the requirements of Objectives Two, Three and Four. Along with the initial conceptual model produced and discussed in Chapter Two, the data collected and analysed in Part Two of the research is the foundation for the refined model and recommendations of marine citizenship presented in Part Three of the thesis.

1.4.3. Part Three: Interpretation, Synthesis and Conclusions

Part Three of this thesis forms the final element of the research project and relates to Objectives Five and Six. Chapter Seven presents a synthesis of the observations made in both the telephone interview schedule and the case study research. Specifically, Chapter Seven begins by examining the influences found to affect an individual's sense of marine citizenship. These findings are then used to refine the preliminary models produced throughout the research and to identify a number of enabling factors required for successful expression of marine citizenship. From the model and overall synthesis of the observations, the implications of marine citizenship are determined and evaluated. Chapter Seven also outlines recommendations for how the promotion of marine citizenship could be applied for the benefit of developing sustainable management plans for the marine environment. Chapter Eight defined marine citizenship and conceptualises the central conclusions of this research and further evaluates the recommendations for marine citizenship being applied to the management of the marine environment in the UK. Finally, Chapter Eight identifies the original contribution of this research to its field and proposes The thesis concludes with the several areas for further development. consideration of future applications of marine citizenship to global marine management.

REFERENCES

Boersma, P., D. and Parrish. J., K. (1999). "Limiting abuse: marine protected areas, a limited solution." Ecological Economics 31: 287-304.

Correia, A. M. R. (2002). Information literacy for an active and effective citizenship: White paper prepapred for UNESCO, the U.S. National Commission on Libraries and Information Science, and The National Forum on Information Science. Information Literacy Meeting of Experts. Prague, The Czech Republic.

Dobson, A., and Valencia-Saiz, A. (2005). "Introduction." Environmental Politics 14(2): 157-162.

Ducrotoy, J., Shastri, S., and Williams, P. (2000). "Coastal sciences and management: the need for networking in higher education." Ocean & Coastal Management 43: 427-444.

Finkl, C., W., and Kruempel, C. (2005). Threats, obstacles and barriers to coastal environmental conservation: Societal perceptions and managerial positionalities that defer sustainable development. Proceedings of the First International Conference on Coastal Conservation and Management in the Atlantic and Meditteranean, Algarve, Portugal.

Fletcher, S., and J. Potts (2007). "Ocean Citizenship: An Emergent Geographical Concept." Coastal Management 35: 1-14.

Fletcher, S., Saunders, J. Herbert, R. and Roberts, C. (2011) Description of the ecosystem services provided by broad-scale habitats and features of conservation importance that are likely to be protected by Marine Protected Areas in the Marine Conservation Zone Project area. Natural England.

Foster, E., Haward, M. and Coffen-Smout, S (2005). "Implementing integrated oceans management: Australia's south east regional marine policy (SERMP) and

Canada's eastern Scotian Shelf integrated management (ESSIM) initiative." Marine Policy 29: 391-405.

French, P., W. (2004). "The changing nature of, and approaches to, UK coastal management at the turn of the 21st Century." The Geographical Journal 170(2): 116-125.

Kuijper, M., W., M. (2003). "Marine and coastal environmental awareness building within the context of UNESCO's activities in Asia and the Pacific." Marine Pollution Bulletin 47: 265-272.

Matti, S. (2006). The imagined Environmental Citizenship: Exploring the state -Individual Relationship in Swedish environmental policy. Department of Business Administration and Social Sciences. Lulea, Lulea University of Technology. PhD: 200.

MacRory, R. (1996). "Environmental Citizenship and the law: repairing the European Road." Journal of Environmental Law 8(2): 219-235.

Mamouni Limnios, E. A., Ghadouani, A., Schilizzi, S., G., M., and Mazzarol, T. (2009). Giving the consumer the choice: A methodology for Product Ecological Footprint calculation. Ecological Economics 68(10): 2525-2534.

McKinley, E., and Fletcher, S. (2010) Individual responsibility for the oceans? An evaluation of marine citizenship by UK marine practitioners. Ocean and Coastal Management

Purcell, M. (2003). "Citizenship and the right to the global city: Reimagining the Capitalist world order." International Journal of Urban and Regional Research 27(3): 546-590.

Osborn, D., and Datta, A. (2006). "Institutional and policy cocktails for protecting coastal and marine environments from land-based sources of pollution." Ocean and Coastal Management 49: 579-596.

Roth, C., E. (1992) Environmental literacy: Its roots, evolution and directions in the 1990s. Columbus, OH. ERIC Clearinghouse for Science, Mathematics and Environmental Education.

Seyfang, G. (2005). "Shopping for Sustainability: Can Sustainable Consumption Promote Ecological Citizenship?" Environmental Politics 14(2): 290-306.

Strong, C. (1998). "The impact of environmental education on children's knowledge and awareness of environmental issues." Marketing Intelligence and Planning 16(6): 349-355.

UNECE (1998) United Nations Economic Commission for Europe. Convention on Access to Information, Public participation in Decision-making and Access to Justice in Environmental Matters. Aarhus, Denmark. June 25th 1998.

UNESCO (1978). Final Report. Tbilisi, Intergovernmental Conference on Environmental Education.

UNEP (2006). Marine and Coastal Ecosystems and human well being. A synthesis report based on the findings of the Millennium Ecosystem Assessment. UNEP: 76pp.

CHAPTER TWO

THE TURN TO MARINE CITIZENSHIP

2.1 INTRODUCTION

This chapter provides a review of the existing research literature related to marine citizenship. Given that marine citizenship is an original concept, the volume of directly related literature was found to be limited. However, the transition from traditional state driven governance to one that champions the application of citizenship to environmental management was reviewed, with particular emphasis on the marine environment. Therefore the chapter begins by identifying the social value of the marine environment, outlining traditional and current management strategies. The movement toward concepts of environmental citizenship in relation to the marine environment is then investigated. Finally, the literature review is used to generate a theoretical model of marine citizenship which is presented at the end of the literature review. This model will be used to guide the remainder of the research, initially determining the focus of the first phase of primary data collection.

2.2 GOVERNANCE OF THE MARINE ENVIRONMENT

2.2.1. Society and the marine environment

The marine environment offers a wide variety of coastal and deepwater habitats from which society derives a number of goods and services, including coastal tourism, fishing and aquaculture, petrochemical industries and food provision (GESAMP, 2001; Rees *et al.*, 2010). Despite high rates of social dependency on the marine environment, many studies note that the value of the marine environment is often significantly under appreciated and misunderstood by the wider public (Costanza, 1999; Costanza *et al.*, 1999; UNEP, 2006). More recently, however, there has been a shift in public perception and there appears to be an increasing recognition of the intrinsic value of marine environment derived goods and services to society. In spite of this, a lack of public awareness of the societal impacts on the marine environment means there are a still a diverse range of issues that are not being addressed successfully. Table 2.1 illustrates that while many marine issues have natural drivers, the majority are strongly correlated with the demands placed on the marine environment by society.

Table	2.1:	Common	issues	facing	the	marine	environment	and	their
associa	ntion v	with society	y (Adap	ted from	n An	tunes and	d Santos, 1999)).	

Marine Issue	Driving Forces	Impact
Over Fishing	Human population growth Food requirements	Decreased catch effort Catching 'down the food web'
Land derived contamination	Urban and industrial development Agriculture and aquaculture Resource requirements	Health effects in marine species Effects to Human health Damage to coastal ecosystems Economic losses for tourism and recreation
Pollution and Dumping	Urban and industrial development Port and shipping activities Accidental oil spills, ballast water derived pollution.	Health effects in marine species Effects to Human health Damage to coastal ecosystems Economic losses for tourism and recreation
Destruction of coastal ecosystems	Urban and industrial development Population Growth Resource Needs Coastal Defences Recreation and Tourism	Biodiversity losses Changes in productivity Increased vulnerability of systems Industrial losses e.g. fishing Coastal erosion Changes to sediment flows
Coastal dynamics	Urban and industrial development	Property losses due to erosion Increased vulnerability of coastal areas Flooding of low lying coastal areas
Climate change	Population Growth Urban and industrial development Transportation and Resource requirements	Changes in biological productivity Heightened populations at risk due to flooding Property risk

2.2.2. Governance and management of the global marine environment

Traditional 'top-down' marine management approaches to policy development and implementation have been increasingly complimented by more 'bottom-up', community directed processes (Chaniotis and Stead, 2007). 'A Sea of Troubles' reported a number of weaknesses in the management of the marine environment including: poor governance of international seas coupled, a lack of stakeholder and community engagement, a traditionally fragmented approach to management causing poor coordination between sectors, ineffective communication between science, policy and the public, and low levels of public awareness (GESAMP, 2001).

Traditional management of the marine environment has been deemed to be unsuccessful following a tendency of governance bodies to manage on a shortterm basis through a fragmented regime where isolated departments deal with individual issues (Stojanovic et al., 2009). Increasingly, however, attempts to integrate social, economic and environmental responsibilities are being made by international governing bodies (Beierle, 1998; Appelstrand, 2002; Humphrey et al., 2000). As a result, the role of state bodies has evolved from direct 'governing' to a more collaborative style of 'governance' with the aim of including public, private and voluntary organisations across all scales of management (Peters and Pierre, 1998; Newman et al., 2004). Application of this change in direction to environmental management is promoted by Lawrence (2005), who suggests that environmental governance should be based on democratic and efficient management aimed at public involvement and providing reliable environmental information. In addition, management of marine resources has evolved to recognise a variety of socio-economic factors, as well as environmental issues that must be addressed (Clark, 1997; Okey, 2003).

2.2.3. Contemporary marine management

The UN Convention on the Law of the Sea (UNCLOS) 1982, Agenda 21 at the Earth Summit in 1992 (Foster *et al.*, 2005) and the Global Programme of Action for the Protection of the Marine Environment from Land-Based Sources in 1995

(Johnston and Vanderzwaag, 2000) all imply a degree of social responsibility to manage the marine environment more holistically (GESAMP, 2001). The recommendations proposed by these political initiatives prompted an evolution of marine management.

Natural resource management of any kind involves a diverse range of stakeholders and interests and therefore requires an integrated, interdisciplinary approach including an evaluation of economic, social, cultural and ecological issues (Clark, 1997; Hegarty, 1997; Cooper *et al.*, 2007). In the context of the marine environment, management is experiencing an ongoing shift in strategy moving away from sectoral management to a more integrated regime. Evidence of this can be seen in the promotion of Integrated Coastal Zone Management (ICZM) which was most recently championed as a key mechanism to tackle observed management failures (Juda, 1999; 2002/413/EC; Chaniotis and Stead, 2007). ICZM can be defined as a continuous and coordinated approach to sustainable development and protection of marine resources that applied equal considerations to the environmental, socio-cultural and economic requirements of an area (Clark, 1997; Cicin-Sain and Belfiore, 2005; Skourtas *et al.*, 2005; Chaniotis and Stead, 2007).

Although ICZM had previously been promoted as the key to successful marine management, to date there has been little evidence of complete integration among stakeholders (Cheong, 2008). Other proposed approaches to marine management have included co-management (Juda, 1999), ecosystem-based management (Cheong, 2008) and adaptive management (Clark, 1997). The key similarity between these approaches is the call for a higher level of public involvement in marine management coupled with improved integration between traditional stakeholders. These forms of more collaborative management are expected to result in a situation in which responsibility for marine management is shared between governments and stakeholders through integrated management (Juda, 1999). Collaborative approaches to management have been found to encourage a sharing of management responsibilities between authorities and local communities (Berkes and Turner, 2006) and are increasingly considered a

possibility for future marine management (Mikalsen and Jentoft, 2001), with comanaged schemes found to be successful in Oceania, Alaska and New Zealand (Berkes and Turner, 2006). Previous studies in Thailand (Nickerson-Tietze, 2000) and Mexico (Chuenpagdee *et al.*, 2002) have shown that these bottom-up approaches prompt enhanced levels of local participation, that take into account local requirements and encourage inclusion of local 'lay' knowledge to ensure efficient, easily implemented and well regulated management schemes.

Given that no one source or activity can be blamed for the deterioration of the marine environment, rather there is a collective societal responsibility to ensure sustainable marine management (GESAMP, 2001) the role of the public in integrated marine management is of increasing interest. This integrated approach is further supported through recommendations that a collaborative management approach to any environmental resource would serve as a mechanism to deliver efficient governance, meeting the multiple requirements of the users of these resources (Juda, 1999; Newman *et al.*, 2004; Cicin-Sain and Belfiore, 2005).

2.2.4. Marine Management and the UK

The UK, as an island, has the second longest coastline in Europe and as a result has a long-standing intimate relationship with the marine environment (Ducrotoy *et al.*, 2000). The UK coastline represents a diverse and varied environment in terms of physical features, natural resources and processes, and human settlement and usage (Ducrotoy *et al.*, 2000; French, 2004). This has given way to a complex system of management strategies with a variety of issues related to a myriad of ecosystem services and processes, ranging from heavy industry to fishing, tourism and recreation to shipping, and conservation (French, 2004). Historically, there has been no strategic framework guiding collective management of the UK marine environment; instead marine management has been dominated by a sectoral approach distributed between various departments within the central government (Defra, 2006). The UK marine environment is also subject to an extensive and diverse range of international, national and often regional legislation and designations. Table 2.2 presents a sample of these

further highlighting the complexities currently associated with UK marine management.

Table 2.2: A selection of the political initiatives requiring consideration inUK marine management

Major Political Initiatives impacting UK marine environments			
International Initiatives	European Initiatives	UK Initiatives	
 Interpol Convention on Oil Pollution Preparedness, Response and Co-operation (1990) The RAMSAR Convention on Wetlands (1971) UNESCO Convention Concerning the Protection of the World Cultural and 	 Habitats Directive (92/43/EEC) Water Framework Directive (2000/60/EC) Birds Directive (79/409/EEC) Bathing Waters Directive (79/409/EEC) Dangerous Substances Directive (76/464/EEC) 	 Marine and Coastal Access Act (2009) Shoreline Management Plans Marine spatial planning Marine Policy statements 	
 International Convention for the Prevention of Pollution from Ships (MARPOL) 1973/8 London Convention on the Prevention of Marine Pollution by Dumping Water and Other Matter (1972 and 1996 protocols) Earth Summit (UNCED) adoption of Agenda 21 in 1992 (Chapter 17) Basel Convention on the Control of Transboundary Movements of Hazardous Waste and their Disposal (1993) Convention on Biological Diversity (1993) UN Framework Convention on Climate Change (1993) Implementation of United Nations Convention of the Law of the Sea (1994) The Global Programme of Action for the Protection of the Marine Environment from Land Based Activities (1995) Code of Conduct for Responsible Fisheries (1995) 	 Assessment Directives (85/337/EEC and 91/11/EEC) Integrated Pollution Prevention and Control Directives (96/61/EEC) Nitrates Directive (91/676/EEC) Urban Waste Water Directive (91/271/EEC) Strategic Environmental Assessment Directive (2001/42/EC) Marine Strategy Framework Directive (2008/56/EC) Integrated Coastal Zone Management COM/00/547 in 2000 OSPAR Convention for the Protection of the Marine Environment of the Northeast Atlantic 		
--	--	--	
--	--	--	

Although there has been increasing awareness of marine issues since the 1970s, marine management has remained traditional, only undergoing significant alterations following the publication of Agenda 21 (Selman and Parker, 1997). Chapter 17 of Agenda 21 (UNCED, 1992) called for more integrated management of the marine environment with improved conservation of resources, reduction of pollution and increased overall understanding of marine ecosystems, the services they provide and the managerial challenges these present (Boesch, 1999; Foster *et al.*, 2005). Since Agenda 21, the focus of UK marine management schemes allowing for ecological conservation as well as sustainable social and economic development in coastal areas (French, 2004). Calls for more holistic management in the UK have ultimately led to the ratification of the Marine and Coastal Access Act (MCAA) (2009) aiming to

eliminate some of the issues that have developed from the historically fragmented management approach to marine management in the UK (French, 2004; Fletcher, 2007) through the development of a unifying management body in the form of the Marine Management Organisation (MMO).

2.2.5. Challenges to public involvement in UK marine management

The need for public involvement in marine management in the UK is well established (Edwards *et al.*, 1997). However, research conducted by Defra (2006) identified a number of challenges as contributing factors to the failure to deliver effective, participatory management of the coast, including:

- Lack of long-term vision for the management of the coast coupled with a limited understanding of coastal processes.
- Inefficient inclusion of stakeholders and end-users in decision-making processes relating to the marine and coastal environment.
- Uncoordinated sectoral legislation and policy.
- Lack of political and financial support for local initiatives to develop sustainable coastal management.

In addition to poorly coordinated management strategies identified by Edwards *et al.*, (1997), Stojanovic *et al.* (2009) highlighted the challenges caused by perceived cultural rifts between scientists and policy makers as a result of traditional conflict between the long-term horizons of science and the shorter term planning of public decision makers. Further to this, there has been a historical perception that the marine environment is an alien entity to the wider public (Jones, 1999) which has resulted in a lack of public concern and empathy towards marine management and conservation and poor relationships between governance bodies and the wider public (Ananda, 2007). Efforts have been made to improve coordination between marine stakeholders and a broad range of stakeholders, although there is still limited guidance available regarding the inclusion of the wider public. Movements towards achieving widespread public engagement are increasingly evident in a UK context, for example, the European Recommendation on ICZM (2002/413/EC) focuses on the necessity to involve

all interested parties in order to achieve sustainable coastal management. Additionally, the use of Coastal Partnerships³ as a management approach is increasingly widespread in the UK and recognises the need for integrated management of the coast (Stojanovic and Barker, 2008). The following sections discuss the role of citizenship in promoting the role of the individual and collective society in the context of the marine environment, and how this can be applied to current marine management strategies.

2.3 TURN TO CITIZENSHIP

This section of the literature review presents an evaluation of the original concept of citizenship, its evolution and the movement towards more modern definitions of citizenship. More recently, concepts of citizenship have been expanded to include environmental behaviour and attitudes expected from members of society, primarily as a mechanism to encourage the adoption of citizenship to deliver environmental benefit presents a new governmental approach to environmental management. This is a phenomenon that has been described by Valencia Saiz (2005) as the 'turn' to citizenship and is viewed as a potentially effective mechanism of policy delivery. The following sections highlight the basic principles of citizenship theories that can be included in the conceptual model of marine citizenship.

2.3.1. Citizenship

Throughout its evolution, the overarching concept of citizenship has become an integral part of everyday life, founded on the theory that society should contribute to the achievement of collective social, economic and environmental goals (Fletcher and Potts, 2007). Citizenship is broadly defined as involving the democratic rights and responsibilities held by members of a community (Ton and Dietrich, 1998; Hilton, 2001; Ferreira, 2002; Purcell, 2003; Mason, 2004; Seyfang, 2005). The extent of this community involvement was commented on

³ Coastal partnerships can be defined as a forum through which interested bodies and sectors are brought together in a bid to achieve sustainable management of the marine environment.

in research by Chamberlain (1997, Online) who stated that "citizenship requires strong enough identification with broader communities to lead people to live their lives in ways that are socially, economically, politically and environmentally responsible".

The definition of a citizen can depend on context, with the generic explanation referring to a citizen as a member of political community (Dfes, 2004). In this sense, it brings with it the responsibilities and rights that come with being a citizen and is often referred to as nationality. Alternatively, citizenship can be a reference to a person's involvement in public affairs i.e. the behaviour of a citizen (Dfes, 2004) referring to a wide range of activities, from taking part in elections, to having a general interest in public affairs (Mason, 2004; Diner, 2003; Smith, 1995). As an evolving concept, citizenship has aimed to encourage individuals to consider themselves global citizens rather than solely as citizens of one nation (Corrie, 2002), with particular importance placed on achieving a 'common good' (Kearns, 1995).

The concept of citizenship has evolved to encompass many facets of modern life including social, political and civil aspects. More recently it has expanded to include economic, environmental and cultural aspects (Corrie, 2002). Traditionally examined in the public sphere (Seyfang, 2005), the concept of citizenship can be defined as participation in public life and involvement in public affairs and decision-making (Corrie, 2002). This participation can be of varying degrees resulting in what is most commonly described as either active or passive citizenship (Corrie, 2002). Passive citizenship has been primarily associated with public rights (Selman and Parker, 1997) while active citizenship is considered to encourage greater empowerment of the public, enhancing individuals' awareness of their role in society, invoking a responsible and participatory citizenry (Diner, 2003). The concept of citizenship is being actively promoted by the current UK coalition government's⁴ plans for a 'Big Society' which seeks to empower citizens, encouraging their involvement in

⁴UK Coalition government elected in 2010.

communities and enhance sense of responsibility (The Conservative Party, Online).

While the concept of citizenship has grown in popularity, a growing number of different forms of citizenship have been proposed including social citizenship (Valdivielso, 2005), scientific citizenship (Irwin, 2001), ecological citizenship (Dobson, 2003; Carter and Huby, 2005; Seyfang, 2005) and environmental citizenship (Berkowitz *et al*, 2005; Hawthorne and Alabaster, 1999). Each of these has its foundation in the basic principles of the modern definition of citizenship, namely participation, capacity for active involvement through education and an awareness of individual rights and responsibilities with regards to their participation in society.

2.4 ENVIRONMENTAL CITIZENSHIP

According to Hawthorne and Alabaster (1999, p.25) 'Environmental citizenship is a...internationally stated objective". For example, Environment Canada, one of the first organisations to actively encourage the general public to embrace the concept, define environmental citizenship as "a personal commitment to learning more about the environment and to taking more responsible environmental action" (cited in Fletcher and Potts, 2007). The emergence of environmental citizenship reflects the recognition that the traditional governance of the environment by statutory organisations has been unsuccessful in sustaining and managing the environment (Dobson and Valencia Saiz, 2005).

Coupled with this shift in traditional environmental management techniques is a documented change in public opinion and compassion towards the environment (Williams, 2008). This transition indicates an improvement in public perception of the environment. Earlier research by Agyeman and Evans (2004) suggested that environmental issues were conventionally considered to be 'someone else's problem' resulting in a low sense of societal responsibility. More recent studies (Diner, 2003; Matti, 2006; Cruz, 2008) imply that the perception of environmental issues has evolved and it is now widely accepted that many of the issues facing the environment can be at least partially attributed to societal

behavioural choices, implicating an acknowledged level of individual and societal environmental responsibility. Given this change in perception, it has been asserted that a move to a more environmentally aware citizenry infers a willingness among the public to employ lifestyle changes and long-term alterations to societal behaviour (Smith, 2005; Matti, 2006). This is supported by research by Defra in 2007 and 2009, which highlighted an improvement in individual willingness to make behavioural adaptations for the benefit of the environment.

As mentioned in Section 2.3, as modern citizenship has evolved, so too has its application, giving rise to a growing number of type specific forms of citizenship. In each of these forms of citizenship, the focus is on a shift in the relationship between society and the state: that in return for the rights provided to individuals by the state, such as a right to security and medical care (Purcell, 2003; Chamberlin, 1997), the state expects certain behaviours and values from individuals that contribute to a 'common good' (Fletcher and Potts, 2007). Historically, this relationship was not specifically applied to the role of individuals in environmental management and impacts of societal behaviour on the environment. However, a movement to a more citizen led, bottom up form of management, termed the 'turn to citizenship' (Valencia Saiz, 2005) with regard to environmental management, is underway.

2.4.1. Identification of models of Environmental Citizenship

Environmental citizenship has been one of the more widely researched facets of the applications of modern day citizenship. A number of studies have generated detailed models of the influence and relationships between social, economic, cultural and other factors included in environmental citizenship (Hawthorne and Alabaster, 1999; Barnett *et al.*, 2005; Berkowitz *et al.*, 2005). Figures 2.1 and 2.2 depict two current models of environmental citizenship, illustrating the

complex web of relationships involved in an individuals' sense of citizenship towards the environment.



Figure 2.1: Model of Environmental Citizenship, (Hawthorne and Alabaster, 1999).



Figure 2.2: Model of environmental citizenship (Berkowitz et al., 2005)

For the purpose of this study, the model of environmental citizenship proposed by Hawthorne and Alabaster (1999) is used as a basis for the identification of the theoretical components of marine citizenship as it was considered to be the more comprehensive model of environmental citizenship. The model identifies a wide range of factors involved in environmental citizenship. These interconnected components result in environmental citizenship being defined as having awareness and understanding of environmental issues, and how they relate to personal, social and environmental values, whilst having the motivation, and capacity to act accordingly, having adequate knowledge of choices and consequences (Hawthorne and Alabaster, 1999; Berkowitz *et al.*, 2005).

Given the benefits observed following promotion of environmental citizenship and heightened community inclusion in terrestrial environmental management, for example, more successful implementation of management strategies and conflict resolution (Appelstrand, 2002; Ananda, 2007; Cooper *et al.*, 2007), it seems pertinent that a similar management approach could be applied to the marine environment. Taking these models and current thinking on environmental citizenship as the starting point, a number of elements were identified as potential components of marine citizenship. These elements and their interactions will now be discussed in terms of a proposed societal transition to a sense of marine citizenship and the generation of a conceptual model of marine citizenship.

2.5 EVOLUTION OF MARINE CITIZENSHIP

Like other specific forms of citizenship, environmental citizenship remains a relatively recent concept (Smith, 2003; Bell, 2004; Smith, 2005). Despite its youth, environmental citizenship is increasingly championed as a mechanism for overcoming the issues resulting from failed traditional, state driven environmental management models by enhancing public engagement in the process (Chaniotis and Stead, 2007; Young et al., 2007). Given the youth of environmental citizenship the majority of research to date conducted in terrestrial based environments (Mrazek, 1996; Ananda, 2007). The movement to inclusive, citizen driven management in terrestrial ecosystem management (e.g. forestry management) has proven to be successful, primarily through developing more sustainable and efficient management plans than previously in place (Ananda, 2007). Indeed, the very definition of environmental citizenship emphasises the importance of individual relationships with the environment, requiring a heightened sense of concern in conjunction with personal and collective responsibility for the environment. It is anticipated that inclusive and integrated systems of marine governance will generate similar benefits. Encouragingly, there is evidence to suggest that governance organisations are already actively working to enhance individual and collective engagement in marine management, at all levels, aiming to prevent further unnecessary deterioration of the marine environment (UNEP, 2006; Chaniotis and Stead, 2007). Examples of this can be seen in the UK's newly ratified Marine and Coastal Access Act (2009) and the longer standing initiatives of Canada's Oceans Act in 1997 (Berkes et al., 2001; Foster et al., 2005; Kearney et al., 2007) and Australia's Oceans Policy in 1998 (Foster et al., 2005), each of which promotes inclusive marine management.

By ensuring individuals have access to accurate and well-disseminated information, a move towards active participation and increased responsibility for the both individual and collective impacts on the marine environment could potentially be facilitated (Hawthorne and Alabaster, 1999; Sharp, 2002; Barr, 2003; Chaniotis and Stead, 2007; Fletcher and Potts, 2007; Rodriguez and Cruz, It has not yet been determined how marine citizenship can be 2007). encouraged, but research conducted by the Countryside Council for Wales (CCW) (Williams, 2008) suggested there is scope for its development. Their study indicated a general sense of admiration for the marine environment among communities, although individuals rarely consider the implications of their actions on the marine environment or how they might contribute to management procedures (Williams, 2008). Observations in earlier studies have partially attributed this to a lack of information available to individuals, coupled with the perception that their input has no influence (Steel et al., 2005; Williams, 2008). Stojanovic and Barker (2008) note that the relationship between communities and coastal managers appears to be showing evidence of positive change, with governing bodies now actively promoting stewardship of the marine environment. Research by Williams (2008) and Defra (2009) supports this, indicating that communities would like to take a more active role but do not currently feel they are provided with sufficient information to participate in marine management and decision-making processes.

Despite the lack of literature on marine citizenship, there are some basic concepts regarding general citizenship, which may be applied when exploring the concept. These include public awareness of rights and responsibilities, informed concern for the marine environment and the ability to articulate opinions and arguments for and against issues relating to the management of the marine environment (Fletcher and Potts, 2007). This emerging concept should be based on an individual's responsibility towards others, as well as for the protection and management of the global marine environment (Matti, 2006). Ideally, these should be influential in the development of marine management and citizens should be active in the preservation, management and development of the marine environment. Finally and most fundamentally, marine citizens should be

responsible in their actions and aware of the impact their actions and activities may have on the marine environment and its related resources (Dfes, 2004).

Building upon these ideas related to environmental citizenship (Berkowitz *et al.*, 2005; Hawthorne and Alabaster, 1999) and their potential role in a marine context, marine citizenship can therefore be tentatively defined as:

An awareness of individual rights and responsibilities related to the marine environment, coupled with the capacity to actively engage in marine decision making processes.

Taking this definition and current thinking on environmental citizenship as the starting point, there are a number of elements that require investigation in order to fully understand the potential elements that potentially constitute marine citizenship. These are discussed in the following sections.

2.6 FACTORS POTENTIALLY INFLUENCING MARINE CITIZENSHIP

In order to construct a working conceptual model of marine citizenship, it is necessary to clearly define the role of each of these factors in context of the marine environment. Sections 2.6.1 - 2.7 present an assessment of the identified factors, particularly focusing, where possible, on their relationship with the marine environment and the influence of this on the development and promotion of marine citizenship. The most closely related factors have been grouped into categories in order to highlight their potential connectivity when evaluating their role in marine citizenship. The results of the literature review are used in the composition of the conceptual model of marine citizenship presented in Section 2.8.

2.6.1 Information and its role in marine management

2.6.1.1 Education

Environmental education was first described in the 1970s by Stapp et al. (1970) who stated that in order for citizens to participate meaningfully in environmental management "it is vital that the citizenry be knowledgeable concerning their environment and associated problems" (p.14). As citizenship theory has evolved and become increasingly accepted as a key mechanism through which to promote societal responsibility, growing emphasis has been placed on the role of education (Correia, 2002). This is increasingly evident, not least by the inclusion of citizenship education as a mandatory component of the UK national curriculum (House of Commons, 2009). Given this, education is recognised as having a fundamental role in individual decision-making, providing humans with sufficient information (both environmental and otherwise) on which to base their decisions (Kuijper, 2003; Jenkin, 2003). Its role in enhancing environmental awareness and altering social perception is well recognised (UNESCO, 1977; Kuijper, 2003) and its importance as a component of an efficient and well functioning management system is significant (Beierle, 1998; Hay and Foley, 1998; Haklay, 2002).

Ferreira (2002) stated that 'informed and responsible citizens can participate actively and give adequate reasons to the environmental problems and issues'. Participation of an informed public has been identified as a key element in facilitating improved public engagement (Stapp *et al.*, 1970; Sears and Hughes, 1996; Aarhus Convention, 1998; Hawthorne and Alabaster, 1999; Ducrotoy, 2001; Berkowitz *et al.*, 2005) and it can be inferred that education would play a similar role in the inculcation of marine citizenship. Further evidence in support of the relationship with marine management and education is provided by Fletcher (2008) who cites the EU recommendations for ICZM as explicitly supporting the requirement for enhanced education in order to increase capacity and public knowledge. This is supported by the view that enhanced public awareness and knowledge through better education would result in increased public support to tackle issues facing the marine environment (Hawthorne and

Alabaster, 1999; Kuijper, 2003; Steel *et al.*, 2005). Furthermore, Agenda 21 emphasised the need for improved marine and coastal education (Ducrotoy, 2001). In order to meet these recommendations, current levels of marine specific education need to be improved, providing accurate, science-based information regarding both the natural and human elements of the marine environment (Correia, 2002; Kuijper, 2003; Berkowitz *et al.*, 2005).

Equally important is a move away from the traditional classroom based education towards increased provision of marine community education outreach programmes, in collaboration with enhanced marine education in schools (Ducrotoy *et al.*, 2000; Potts, 2000; Fletcher and Potts, 2007). Examples of small community-based marine education programmes have found an increased understanding of community impacts on the marine environment (Uneputty *et al.*, 1998) and a more marine aware community (Edwards *et al.*, 1997; Uneputty *et al.*, 1998; Storrier and McGlashan, 2006). Success stories of this nature support the common assumption that higher levels of public awareness and understanding linked to enhancement of knowledge are fundamental to better marine environmental protection and management (Ducrotoy *et al.*, 2000; Steel *et al.*, 2005).

Internationally, there has been a move towards an acceptance that current levels of marine education included in the formal teaching need to be improved. For example, in 2007 in the Unites States, NOAA (National Oceanographic and Atmosphere Administration) generated an Education Strategic Plan 2009-2029, providing further evidence of the importance of education in marine management and conservation. In the context of marine education in the UK, a number of studies have emphasised the need for improvements to be made that will enhance current levels of marine education in a bid to increase public awareness and understanding of the marine environment (Fletcher *et al.*, 2009; Castle, *et al.*, 2010).

2.6.1.2. Knowledge

Environmental knowledge has been identified as a key predictor of individuals' behaviour towards the environment (Hawthorne and Alabaster, 1999). The importance of public knowledge and understanding of environmental concepts is further supported by Tytler *et al.* (2001) who stated that public understanding of environmental issues directly influences perception of management. Finkl and Kruempel (2005) view this societal lack of understanding as a psychological barrier to the implementation of mechanisms required for conservation and management of the marine environment. It is possible that this is compounded by damaging personal behavioural choices arising from a lack of awareness of the marine environmental managers have adopted strategies to encourage higher levels of environmental knowledge among the wider public in a bid to engender a change in social behaviour (Cottrell and Graefe, 1997; Berkowitz *et al.*, 2005; Defra, 2007).

In the context of the marine environment, a number of researchers have related high levels of public knowledge about the environment to successful management (Steel *et al.*, 2005; Storrier and McGlashan, 2006). For example, Steel *et al.* (2005, p. 98) stated that "knowledge is vital in developing an individual perception of the oceans and the resources they provide" and that it is the "key to accomplishing effective environmental policies." Steel *et al.* (2005) further suggest that knowledge and concern are positively related in that people with relatively high levels of knowledge tend to have higher levels of environmental concern, and a greater desire to protect the marine environment.

2.6.1.3 *Literacy*

In the model of environmental citizenship generated by Hawthorne and Alabaster (1999) environmental literacy is explained as being a combination of environmental awareness and basic environmental education. The concept is further defined as having the knowledge and capacity to take appropriate action to maintain environmental systems (Roth, 1992), providing "a basic functional

education for all people, which provides them with the necessary knowledge, skills and motives to cope with environmental needs and contribute to sustainable development" (UNESCO, 1989). Studies conducted by Tytler *et al.* (2001) support the suggestion that an element of scientific literacy in citizens coupled with local knowledge can benefit decision making processes, often proving to be a good basis for citizen-based questioning of an environmental issue.

Diner (2003) suggests that environmental literacy focuses on environmental sensitivity, knowledge, skills, attitudes, personal investment/ responsibility and involvement. With reference to marine environmental literacy, research has indicated that levels of literacy and awareness among the public are very low (Fletcher and Potts, 2007; Steel *et al.*, 2005). Despite this, NOAA recognises a marine literate citizenry as being vital to achieving international marine management goals (NOAA, 2007). It can be inferred from research by Haklay (2001) that individuals with higher levels of literacy would be more likely to behave in a favourable manner towards the environment, exhibiting a higher level of awareness of issues and the impacts of their behaviour on the marine environment. In a UK context, there have been recent attempts at establishing coastal literacy programmes in particular by CoastNET whose coastal literacy project aims to educate and inform local people in a bid to enhance public understanding and engagement in their coastal environments (CoastNET Online).

2.6.1.4. Information

As identified in the models of environmental citizenship produced by Hawthorne and Alabaster (1999) and Berkowitz *et al.* (2005), the availability of accurate environmental information is known to be a crucial factor in environmental citizenship. Currently environmental information exists in vast and numerous forms, which need to be utilised to benefit particular situations (Hawthorne and Alabaster, 1999). Availability of information is an integral component of environmental citizenship and can be directly linked to the other factors of environmental citizenship. Agenda 21 states "each individual shall have appropriate access to information concerning the environment", researchers suggest it is imperative that information regarding the environment be accurate and accessible to the public (Haklay, 2002). The importance of accessible and accurate information has been emphasised through the development of the European Directive "Freedom of access to information on the environment" (90/313/EEC) and Aarhus Convention (UNECE 1998). Indeed, Lee and Abbott (2003) go as far as to state that access to information is the primary objective of the Aarhus Convention and establish it as the key starting point for initiating any improvements to public participation in decision making.

The literature review suggested a lack of research conducted on the links between access to information, marine conservation and management. However, given the strong links between information availability, knowledge and education, it can be assumed that public access to marine information could playa prominent role in marine management in the UK. These assumptions are supported in part by Haklay (2002) who identified direct links between the levels of information available, public interest and public participation in environmental matters. As such, it is imperative that practitioners, governance organisations and managers work to enhance public capacity to engage with marine management through increased availability of clear and accurate information (Dobson and Valencia-Saiz, 2005). Given this, it can be hypothesised that information availability and accessibility to the wider public would be a necessary component of marine citizenship.

2.6.2 Awareness, concern and marine citizenship behaviour

2.6.2.1. Awareness

Environmental awareness is a prime example of the interconnected nature of each of the components included in Hawthorne and Alabaster (1999)'s model of environmental citizenship. Acceptance of the relationship between environmental awareness and education is long-standing, initially linked at the Tbilisi Conference (UNESCO, 1978) stating that it should "foster clear awareness of, and concern about...ecological interdependence". This recommendation clearly highlights the connectivity between the factors of

awareness and concern (Hawthorne and Alabaster, 1999). In addition, the relationship between awareness and social behaviour towards the environment has been further cemented by Drevensek (2005, p.227) who stated that,

"it is only when people...know what is going on in the environment at the local, regional, national or even global level that they can play an active, responsible role in shaping policy-making in line with their own...needs".

A growing sense of environmental awareness among governance bodies regarding the requirements of environmental management has led to an increase in participatory methods to manage environmental resources (Ananda, 2007). As with many of the components of environmental citizenship, awareness, literacy and concern are inextricably linked (Hawthorne and Alabaster, 1999; Steel et al., 2005). This is supported by observations that people with a higher level of environmental knowledge exhibit a greater desire to learn about the environment and the potential threats it faces (Ananda, 2007). Current levels of public awareness of marine environmental issues in the UK evaluated by Fletcher et al. (2009) indicated that although there is an interest in the marine environment, marine awareness in the UK is considerably under developed. In addition to observed poor levels of public awareness in the UK, methods of raising public awareness of the marine environment are considered to be in need of some improvement as to date, they have proven largely ineffective (Fletcher and Potts, 2007; Fletcher et al., 2009;). In spite of current observations regarding marine awareness, research into environmental citizenship firmly places public awareness as an integral component, thereby providing sufficient evidence for its inclusion in an evaluation of a marine specific concept.

2.6.2.2. Concern

As a component of environmental citizenship, environmental concern has proven a difficult concept to define and therefore has been loosely described as "an individual's degree of emotional reaction to…reported damage to the environment" (Hawthorne and Alabaster, 1999 p.27). Like education, concern for the environment has been identified as a predictor for individual behaviour towards the environment (Bamberg, 2003). Various factors have been acknowledged as contributing to environmental concern, including; an individual's personality, locus of control (the belief that external events can be influenced by personal activities), sense of social responsibility and future orientation (a desire to help others/ the environment without any direct personal gain), sense of personal responsibility for environmental issues, economic status, emotionality, religious affiliations, and whether or not environmentally favourable behaviour is seen as the social norm (Hawthorne and Alabaster, 1999; Bamberg, 2003). These factors are strongly linked to personality and have been presented in more detail in Table 2.3. Poortinga *et al.* (2004) suggest that levels of environmental concern are directly correlated with an individual's values i.e. important life goals or standards that people aspire to, which provide the basis for individual behaviour These social influences on environmental concern are supported by Bamberg (2003), with levels of concern also attributed to perception, environmental knowledge and access to education.

2.6.2.3. Behaviour and responsibility

As discussed in Sections 2.6.1.1 and 2.6.1.3, strong links between environmental knowledge, education and responsible behaviour have been identified (Hawthorne and Alabaster, 1999; Haklay, 2002; Bamberg, 2003; Steel *et al.*, 2005; Defra, 2007). It is assumed that if an individual has a high awareness of environmental issues, their behaviour should, theoretically, reflect the relevant knowledge.

Hawthorne and Alabaster (1999) indicated strong correlations between environmental behaviour, both direct and indirect (Poortinga *et al.*, 2004), and environmental education. As well as influencing direct environmental behaviour, it has been shown that individual values can play an important role in indirect environmental behaviour such as consumerism, acceptance of policy and development of legislation (Poortinga *et al.*, 2004). In addition to the influence of education on behaviour towards the environment, Stern (2000) suggested that environmentally significant behaviour can be influenced by an individual's personal values and sense of responsibility. This is founded on early research on environmental behaviour, producing the New Environmental Paradigm (NEP) model (Dunlap and Van Liere, 1978) which measured individual views on human-environment interaction, thereby determining the influence of individual values on pro-environmental behaviour.

In order to attempt to harness particular behaviours towards the environment, it is necessary to understand the motivational forces behind individual behavioural choices. Research has investigated attitudes toward the environment and the resulting behavioural choices, with negative attitudes towards poor environmental behaviour expected to engender pro-environment behavioural choices in the same individual (Cottrell and Graefe, 1997; Hartig *et al.*, 2001). In addition Hartig *et al.* (2001) propose that an individual's perception and interest in the environment can have an impact on an individual's behavioural choices. The model produced by Hawthorne and Alabaster (1999) does not specifically consider the potential influence of perception on environmental citizenship. However, given the level of importance attributed to it in other studies (Kaivola *et al.*, 2003; Finkl and Kruempel, 2005; Tran, 2006), perception will be discussed as potentially influential factor for marine citizenship in Section 2.7.1.

Links between knowledge, education and responsible behaviour have been highlighted by recommendations made by the UK strand of UNED (United Nations Environment and Development) recognised the importance of education and its connection to marine environmental behaviour. Cottrell and Graefe (1997) and Thapa *et al.* (2005) suggest that in order for an individual to act responsibly towards the marine environment, they require knowledge about the situation. CCW (Williams, 2008) identified a lack of public understanding and awareness observed among coastal communities within the UK. However, contrastingly, a willingness to expand marine specific awareness and related levels of concern was still evident amongst the public, subject to enhanced guidance (Williams, 2008) as indicated by studies carried out by Defra (2006; 2009). This juxtaposition suggests that progress is required providing the public with accurate information, highlighting the relationships between the marine environment and everyday life, to increase their understanding of the impacts their behavioural choices have on the marine environment. If the demand for

more information identified by CCW (Williams, 2008) is met, prompting a change in social behaviour towards the marine environment, it stands to reason that more successful marine management can be developed. However, it has been recognised that individual behavioural choices are made based on a number of factors, each of which is subject to complex processes prior to a decision being reached (Barr, 2003; Urama *et al.*, 2006; Young *et al.*, 2007).

2.6.3. Personality and socio-demographics

Hawthorne and Alabaster's (1999) research into environmental citizenship suggests that a range of personal and social variables could influence marine citizenship. These personality related variables are presented in Table 2.3 and it can be assumed that they would have a similar influence on marine citizenship as has been observed in studies on environmental citizenship.

Table 2.3: Range of personality variables thought to have an influence on environmental citizenship

Personality Variables				
Attitude to the New Environmental Paradigm (Dunlap and Van Liere, 1978)	• Views humans as part of an integrated ecosystem with nature			
Locus of Control	 Locus of control (LOC) is strongly linked to the public and individual perception of environmental issues and ability to influence an event (Hawthorne and Alabaster, 1999). An individual with an internal LOC considers the results of certain situations to be directly linked to their own actions External LOC attributes events of this kind to higher power e.g. governmental bodies (Hawthorne and Alabaster, 1999). 			
Sense of social and personal responsibility	 A sense of social responsibility suggests an individual's desire to help others even in the absence of any personal gain (Hawthorne and Alabaster, 1999). Frequently used as indicators of environmental concern (Hawthorne and Alabaster, 1999). Assumes partial personal responsibility for environmental problems, and should, as such, be responsible for developing the solutions (Hawthorne and Alabaster, 1999). Related to environmental behaviour (Hawthorne and Alabaster, 1999). 			
Adherence to Social norms	 In cases when environmentally responsible behaviour is the perceived social norm, there may be an increased social pressure for individuals to behave in an environmentally favourable manner. This social pressure may cause people to behave with environmental issues taken into consideration, regardless of whether they were a personal concern or not (Hawthorne and Alabaster, 1999). 			
Emotionality	 Relationship between how an individual feels towards the environment and behaviour Individuals are more likely to exhibit a higher level of concern when they feel the issue is of personal relevance and may affect their lives (Hawthorne and Alabaster, 1999). 			
Religious and cultural affiliations	• Slight relationships established between an individual's level of concern and a low level of religious beliefs and culture (Hawthorne and Alabaster, 1999).			
Economic Orientation	 Hawthorne and Alabaster (1999) suggest economic orientation to be of high importance Impact environmental benefits before personal economic gain, favouring long term environmental protection over a less permanent state of economic gain. 			

In addition to personality attributes, a number of factors are included in sociodemographics, including age, gender, income and family background (Hawthorne and Alabaster, 1999; Barr, 2003; Dolnicar, 2006). These have been identified as having the potential to influence an individual's environmental decision-making process and should therefore be considered in the context of marine citizenship. In particular these factors have been related to education and concern for the environment (Hawthorne and Alabaster, 1999).

2.6.4. Socio-economic influences

Socio-economic factors directly link to numerous aspects of Hawthorne and Alabaster's model (1999), with Haklay (2002), for example, equating high levels of environmental education with 'middle class' society. Evidence of this is provided by Santos et al. (2005), who compared the socio-economic status of the visitors to two Brazilian beaches, examining the influence of this on the visitors' sense of environmental responsibility. It was found that litter generation on the two tourist beaches was found to be higher at the beach visited by individuals with a lower average annual income. This inferred a lower level of awareness than at the beach frequented by individuals of higher socio-economic status (Santos et al., 2005). In addition, research has suggested that access to information and education can often be limited to those of higher socio-economic standing (Haklay, 2002; Santos et al., 2005; Steel et al., 2005). This is supported further by Steel et al. (2005) who suggested that there is a "knowledge gap", separating people of higher and lowers socio-economic status, where people of a lower socio-economic standing tend to have a lower level of marine knowledge. These observations imply that efforts to address this social inequality are required if there is to be a successful expression of marine citizenship within UK society.

2.7 IDENTIFICATION OF FACTORS SPECIFIC TO MARINE CITIZENSHIP

Through the literature review, it is clear that existing models of environmental citizenship cannot be directly applied to the marine environment. A number of

other factors not included in Hawthorne and Alabaster's model (1999) were identified as potential additional factors requiring investigation with regard to the marine environment. These are discussed in Sections 2.7.1 -2.7.3.

2.7.1 Public Perception

Research has implicitly linked the efficacy of increased education and knowledge to public perception and understanding of available information (Hawthorne and Alabaster, 1999; Hartig *et al.*, 2001; Tytler *et al.*, 2001; Jenkins, 2003). The importance of this relationship is further emphasised by Barr (2003) who suggested that the efficacy of education projects could be dependent on public understanding of environmental issues. The potentially complex influence of public perception is explored by Jentoft (2007) who suggests that reality is a factor of what people perceive to be true, and is then confirmed as reality when individuals base their behaviour on these perceptions. This supports the necessity for widely accessible and accurate information regarding the marine environment.

It is becoming increasingly recognised that the most effective way of monitoring and protecting the marine environment is to modify the behaviour of the people that use it, to encourage users to adopt more environmentally friendly behaviour (Juda, 1999). The implementation of policy is dependent on citizen participation, and generally assumes that providing environmental education will encourage alterations in people's behaviour. However, it is acknowledged that an individual's perception of the importance of an issue and related solutions that will have the greatest influence on citizen behaviour (Eden, 1996). Behaviour depends on an individual's perception of the world; as understanding of the environment and its threats increases, behavioural changes within society can be expected (Juda, 1999).

Perception of issues can vary depending on socio-demographic factors as discussed in Section 2.6.4. Appelstrand (2002) shows evidence of this in the context of forestry management. For example, research has found rural residents to be more in favour of traditional uses of forest for production, whilst, in

contrast, urban residents were more supportive of new, innovative activities, recreational activities and preservation for future generations (Appelstrand, 2002). It is possible that similar would be true when comparing coastal and inland communities with coastal communities most concerned with issues relating to productivity and traditional livelihoods, while inland community concerns may focus on conservation issues.

It should be noted that in an environmental context, public perception could extend to justice issues often associated with environmental resources and management. The importance of public perception of a concept and associated justice issues has been found to influence the level of public participation (Rowe and Frewer, 2000). In the past, observations have shown public understanding and trust in scientific knowledge to be low. For example, in the early 1990s, 36% of the public were of the opinion that scientists were not informed as they professed, with 47% of the British public also stating that they did not trust the government to deal with environmental issues (Worcester, 1993). Although not included in Hawthorne and Alabaster's model (1999), environmental citizenship has been linked to the concept of environmental justice in later work (Agyeman and Evans, 2004). In the UK, attempts to encourage more inclusive marine management, and therefore engender a perception that the management has been developed in a just manner, have been successful on local scales through organisations such as Finding Sanctuary (Online).

2.7.2 Public Participation

The importance of public participation with the marine environment was first recognised in the 1970s, becoming increasingly important in modern coastal management (McNeil *et al.*, 2006; Kawabe, 2004; Appelstrand, 2002; Edwards *et al*, 1997). This was reiterated in 1992 at the Earth Summit, where the development of Agenda 21 stated, "one of the fundamental prerequisites for the achievement of sustainable development is broad public participation in decision making" (cited in Fletcher, 2003). Numerous authors have identified connections between public participation and successful environmental management (Eden, 1996; Carnes *et al.*, 1998; Chopyak and Levesque, 2002).

Participation was initially related to environmental issues in the recommendations proposed by Agenda 21, followed by the Aarhus Convention, the Convention on Access to Information, Public Participation in Decision-making and Access to Justice in Environmental Matters (UNECE, 1998). In spite of this, Hawthorne and Alabaster's model (1999) does not specifically mention participation as a factor of any significance in environmental citizenship. Given the potential influence of this factor, it is proposed that participation should be included in marine citizenship. In order to do so, there must be a thorough understanding of the role of public participation and how it can be applied to marine management.

Based on the suggestion that if a person has *the right* to participate in societal decision making, they also have a responsibility to be a 'good' citizen, public participation has been defined as the most effective method to enable the public in environmental decision-making (Smith, 2003). In the context of environmental policy development and management implementation, public participation has been identified as way of offering stakeholders the opportunity to share and exchange knowledge (Edwards et al., 1997; Chess and Purcell, 1999; Kearney et al., 2007). For example, Appelstrand (2002) observed increased participation in forestry management to result in the adoption of more efficient management regime. Failure to involve the public at the early stages of the decision-making process has been found to lead to a lack of trust, suspicion and dissent towards the management regime, reducing the possibility of future participation (Edwards et al., 1997; Appelstrand, 2002). Therefore in order to encourage the highest degree of acceptance, inclusion in the process is paramount as individuals are more likely to accept plans if they believe their opinions and input are valued (Appelstrand, 2002).

The suggestion that public participation could aid marine and coastal management was first accepted and promoted by the U. K. government in 1992. The U.K responded to this by accepting the fact that coastal and marine management schemes required integration of all stakeholders and users, in order to develop efficient management plans (Edwards *et al.*, 1997). In the past, citizen involvement has meant that key interested parties have been consulted

and the information provided by them has been discussed and publicised. However, this method has been highly criticised by people who felt that they should have been consulted earlier in the decision making process, so as to have some kind of meaningful impact (Edwards, *et al.*, 1997). It has been suggested that failure to involve the local communities in coastal management planning at an early stage could undermine UK coastal environment initiatives (Edwards *et al*, 1997). However, although research has shown that public participation can be beneficial to environmental management, it also necessary to consider the potential challenges associated with enhanced public involvement. Table 2.4 presents a variety of strengths and weaknesses currently associated with public participation in overall environmental management.

 Table 2.4: Strengths and weaknesses associated with public participation in

 environmental management and decision-making.

Strengths	Weaknesses
 Widening the number of interests represented in the decision-making process (Appelstrand, 2002; Fletcher, 2003). Early anticipation of conflicts (Appelstrand, 2002), Improvement of local sense of ownership of resources (Appelstrand, 2002; Fletcher, 2003). Inclusion of lay knowledge in planning and policy-making; (Tytler <i>et. al.</i>, 2001) More successful implementation of management and planning strategies (Dreversek, 2005). Comprehensive assessment of risks (Dreversek, 2005). 	 Poor participation as a result of low capacity for involvement (Fletcher, 2003; Lee and Abbott, 2003; Rowe and Frewer, 2000). Traditional over dependence on scientific and technological expertise (Rowe and Frewer, 2000). Potential for the process to be costly in both time and money (Beierle, 1998). Potential bias during management and decision-making as a result of over representation of certain stakeholders (Appelstrand, 2002; Rowe and Frewer, 2000).

2.7.3 Proximity to the coast and sense of place

An individual's sense of place and consequential sense of attachment to an area can be integral to their perception, understanding of issues and ultimately sense of responsibility towards the environment (Cantrill, 1998; Jorgensen and Stedman, 2001; Stedman, 2002; Cox *et al.*, 2008; Kusuma, 2001). Place

attachment is most commonly described as a positive relationship between individuals or communities and their environment, and has been related to knowledge, awareness, concern, emotional connections and sense of responsibility (Jorgensen and Stedman, 2001). Of particular importance to environmental management is an understanding of how sense of place can impact on public perception of natural ecosystems. For example, Cantrill (1998) presents evidence to suggest that different experiences will engender different reactions to different environments and any propositions of managing a particular resource.

In spite of its apparent complexity and the marine specific research conducted, sense of place is clearly an important component of any model including environmental behaviour, awareness and citizenship. Although not directly included in the model created by Hawthorne and Alabaster (1999), sense of familiarity and ownership invoked through a positive sense of place or attachment to a particular environment is often reflected in a desire to manage and protect environmental resources (Berkowitz et al., 2005; Jorgensen and Stedman, 2001). Therefore, in the context of environmental citizenship, attachment to place can already be strongly linked to developing an individual's overall sense of responsibility towards the environment (Kearns, 1995). With regards to its role in marine citizenship, the connection between attachment to the marine environment and levels of awareness was identified by Steel et al. (2005) with visits to the coast highlighted as one of two important situational variables influencing adult ocean literacy. Steel et al. (2005, p. 111) found that experience of the coast can influence an individual's level of concern for the marine environment and "brings people closer to the problems", encouraging them to behave in a more appropriate manner. However, supporting research has found that a sense of connection to the marine environment cannot be based solely on exposure and proximity to the coast. Williams (2008) identified a number of socio-demographic factors that superseded an individual's proximity to the marine environment, including age and life stage.

The next section of this chapter outlines the generation of a conceptual model relating on the observations made throughout the literature review.

2.8: CONCEPTUAL MODEL OF MARINE CITIZENSHIP

Having discussed the potential components of marine citizenship and the interand intra- relationships that could influence an individuals' sense of marine citizenship, a conceptual model of marine citizenship derived from the observations of the literature review can be generated as displayed in Figure 2.3. It is important to highlight that unlike the preceding models produced for environmental citizenship (see Figures 2.1 and 2.2); no hierarchy or preconceived relationships are indicated in this diagrammatic model. Although some of the information could have been inferred from the Hawthorne and Alabaster (1999) or Berkowitz *et al.* (2005) model, it was decided that the level of influence these factors and their connectivity might vary significantly in a marine specific model. The conceptual model of marine citizenship illustrated in Figure 2.3 reflects this complexity.



2.9 CONCLUDING COMMENTS

A synthesis of the findings presented in Sections 2.2-2.8 is now presented through discussion of a number of key conclusions taken from the literature review.

The first finding is that there is a significant lack of research into the application of environmental citizenship in a marine context. This acted as a founding rationale for further research into a concept of marine citizenship. Evidence was found of other type specific forms of environmental citizenship but these were dominated by terrestrial ecosystems. One source proposed a concept of 'ocean citizenship' and provided an evaluation of the benefits that could potentially be associated with a concept of this nature.

Secondly, the literature review identified a number of additional factors that had not been included in the primary models of environmental citizenship. Following an assessment of the role of these factors in environmental citizenship, the exclusion of these factors in existing models renders current environmental citizenship models effectively inapplicable in a marine context. This provides a strong rationale for furthering the research into the generation of a concept of marine citizenship and how it could be applied for the benefit of marine management. Table 2.5 presents the factors common to both environmental and marine citizenship, highlighting a number of additional factors with the potential to influence a marine specific concept of citizenship.

(Hawthorne and Alabaster symbol.	r, 1999) and marine cit	izenship indicated by the	e
Potentially Affecting	Environmental	Marine	
Factors	Citizenship	Citizenship	

Table 2.5:	Comparison of	the factors	s present in	ı environn	nental ci	tizenship
(Hawthorn	e and Alabaster,	1999) and	marine cit	izenship iı	ndicated	by the ✓
symbol.						

Factors	Citizenship	Citizenship
Education	✓	1
Responsibility	✓	1
Capacity	✓	1
Socio-economics	✓	1
Awareness	✓	1
Personality traits	✓	1
Desire to Act	✓	1
Literacy	✓	1
Attitude	✓	1
Concern	✓	1
Perception		1
Participation		1
Livelihood dependency		1
Proximity to the resource		1

As Table 2.5 illustrates, a concept of environmental citizenship could not be directly transferred to the marine environment. There are significant gaps with regard to the influence of livelihood dependency and geographical location on an individual's sense of citizenship towards the marine environment. These gaps serve only to further support the need to investigate and promote marine citizenship.

In addition to this, there is increasing recognition that management objectives are in need of adaptation so that long-term environmental benefits are considered. It is the proposition of this research that marine citizenship would theoretically engender a change in societal attitude and behaviour towards the marine environment.

The third conclusion of the literature review was the acknowledgement that successful expression of marine citizenship requires the presence of the factors identified. When contemplating the concept of marine citizenship, it is necessary to consider all potential factors that may have an effect on the presence of marine

citizenship in today's society presented in Table 2.6 with each of these parameters and the level hypothetically required to promote marine citizenship.

Table 2.6: Factors potentially influencing Marine Citizenship deducedthrough the literature review.

Factors	Level likely to encourage Marine Citizenship	Level likely to discourage Marine Citizenship
Socio Demographics		
Position in Society	Developed Region, Stable Income	Developing Region, Unstable Income
Background (parenting)	Pro-environmental background with high level of exposure to	Background with low level of exposure to environmental issues
Heritage/ Culture	High – Medium Dependency on the Marine Environment	Low Dependency on the Marine Environment
Proximity to Coast	Strong association with the Marine Environment	Weak association with the Marine Environment
Income	Medium to high	Low
Personality Variables		
Sense of social responsibility	High Sense of Social Responsibility	Low Sense of Social Responsibility
Sense of personal	High Sense of Personal	Low Sense of Social Responsibility
Delitical Affiliation	A filiation with and any incommental	Afficient with molitical martine with
Pontical Annation	political parties (usually liberal)	low emphasis on environmental issues (usually conservative)
Emotionality	High Connection/ Attachment to the marine/coastal environment – can be dependent on locality.	Low Connection/ Attachment to the marine/ coastal environment
Canacity		
Access to Information	High – Medium Access to	Low Access to Appropriate
Access to information	Appropriate Information	Information
Desire to Learn	High	Low
Suitability of other circumstances	Cultural, Economic, Social and Political Circumstances Suitable for Development	Cultural, Economic, Social and Political Circumstances Unsuitable for Development
Education	Well Educated in Environmental Issues	Poorly Educated in Environmental Issues
Education		
Access to education	High Access to Appropriate Education	Low Access to Appropriate Education
Literacy	High – Medium environmental literacy levels	Low environmental Literacy levels
Knowledge	Good level of knowledge, ability to act appropriately.	Low level of knowledge
Knowledge		
Local, National, Global	Good level of knowledge of marine environmental issues at all scales	Low level of knowledge of marine environmental issues

Factors	Level likely to encourage Marine	Level likely to discourage Marine
	Citizenship	Citizenship
Awareness		
I Indonaton din a officerson	High Madium Hudanstanding	I and I advector dia a
A manage of	High Medium Understanding	Low Understanding
Awareness of	of consequences of Management	Low Levels of Awareness of Consequences of Management Plans
Consequences	plans and policies	and Policies
Responsibility		
Locus of Control (LOC)	High (Internal) LOC	Low (External) LOC
Desire to Act	High – Medium desire to improve situation	Low desire to improve situation
Attitude to NEP	Agreement with the NEP	Disagreement with NEP, more accepting of the DSP
Concern		
Personality	Associated with high levels of pro- environmental personality variables	Associated with low levels of pro- environmental personality variables
NEP	Acceptance	
LOC	Internal	External
Behaviour		
Desire to Act	High Desire to Act in a Pro-	Low desire to act
	environmental manner	
Concern	High – Medium Level of Concern	Low Level of Concern about the
	about the Condition of the marine environment	Condition of the Marine Environment
Public Participation		
Responsibility	High – Medium Sense of	Low Level of Responsibility
	Responsibility	
Awareness	High – Medium Level of Awareness	Low Level of Awareness
Public Perception		
Attitude to NEP	Accept the NEP	Refute the NEP
Political Affiliation	Liberal	Conservative
Proximity to resource		
	Reside in a coastal region.	Reside in an inland region.
Livelihood Dependency		
	High levels of dependency on marine resources	Low levels of dependency on marine resources

The final conclusion to come out of the literature review is the recognition that as with the generation and examination of any new concept, it is necessary to consider the benefits and burdens it creates. Currently general public awareness and associated concern for the marine environment are relatively low, which has potential implications for the promotion of marine citizenship. It has become increasingly apparent that the condition of the marine environment is inherently linked to issues of social justice (Agyeman and Evans, 2002), financial dependency (Costanza, 1999) and cultural heritage (Costanza, 1999). With such diverse and complex relationships requiring attention, promotion of a marine citizenship could be a potentially arduous task. The development of a marine

citizenship model must consider the lessons learned through environmental citizenship (Hawthorne and Alabaster, 1999).

Following completion of the literature review three areas were identified for further investigation in this project:

- The lack of definition or framework for a concept of marine citizenship;
- An identification of the factors influencing both individual and societal conceptions of marine citizenship;
- Evaluation of the application of marine citizenship for the benefit of sustainable management of the marine environment.

Given these areas of investigation the specific data requirements for the remainder of the research were therefore identified as:

- The perception of marine practitioners regarding marine citizenship and *its role in management of the marine environment*. This will provide an evaluation of current management strategies, the role available to individuals in marine management, identification of the factors thought to be directly related to marine citizenship, and the potential role for marine citizenship in future management.
- Evaluation of the key factors identified by marine practitioners through thematic case studies. This will allow identification of the gaps between management and community perspectives regarding the application of marine citizenship and the factors that are of most importance when promoting this emerging concept. In addition to this, collection of this data will allow identification of where effort would be required in order to further the concept of marine citizenship and how it could be applied to future sustainable marine management.

This research will seek to fill the gaps identified through the literature review and will contribute to the debate concerning the role of citizenship in management of the marine environment and associated resources. The general methodological

approach adopted through which this data will be collected and analysed is presented in Chapter Three.
REFERENCES

Agyeman, J., and Evans, B. (2004). Just Sustainability': the emerging discourse of environmental justice in Britain? The Geographical Journal 170(2): 155-164.

Ananda, J. (2007). Implementing Participatory Decision Making in Forest Planning. Environmental Management 39: 534-544.

Antunes, P., and Santos, R. (1999). Integrated Environmental Management of the Oceans. Ecological Economics 31: 215-226.

Appelstrand, M. (2002). Participation and societal values: the challenge for lawmakers and policy practitioners. Forest Policy and Economics 4: 281-290.

Aoyagi-Usui, M., Vinken, H., and Kuribayashi, A. (2003). Pro-environmental Attitudes and Behaviours: An Internal Comparison. Research in Human Ecology 10(1): 23-31.

Barr, S. (2003). "Strategies for sustainability: citizens and responsible behaviour." Area 35(3): 227-240.

Bamberg, S. (2003). "How does environmental concern influence specific environmentally related behaviours? A new answer to an old question." Journal of Environmental Psychology 23: 21-32.

Beierle, T. C. (1998). "Public Participation in Environmental Decisions: An Evaluation Framework Using Social Goals."

Berkes, F., Mathias, J., Kislalioglu, M., and Fast, H. (2001). "The Canadian Arctic and the Oceans Act: the development of participatory environmental research and management." Ocean and Coastal Management 44: 451-469.

Berkes, F., and Turner N., J. (2006). "Knowledge, Learning and the Evolution of Conservation Practice for Social-Ecological System Resilience." Human Ecology 34(4): 479-494.

Bell, D., R. (2004). Liberal Environmental Citizenship. Citizenship and the Environment workshop, ECPR Research Sessions. Uppsala

Berkowitz, A., R., Ford, M., E., and Brewer, C., A. (2005). A framework for integrating ecological literacy, civics literacy, and environmental citizenship in environmental education. in Environmental Education and Advocacy: Perspectives of Ecology and Education. E. Johnson, A., and Mappin, M., J. Editors, Cambridge University Press. Cambridge. 227-266.

Boesch, D., F. (1999). "The role of science in ocean governance." Ecological Economics 31: 189-198.

Cantrill, J., G. (1998). "The Environmental Self and a Sense of Place: Communication foundations for regional ecosystem management." Journal of Applied Communication Research 26: 301-318

Carter, N., and Huby, M. (2005). "Ecological Citizenship and Ethical Investment." Environmental Politics 14(2): 255-272.

Castle, Z., Fletcher, S., and McKinley, E. (2010). Coastal and marine education in schools: Constraints and opportunities created by the curriculum, schools and teachers in England. Ocean Yearbook. 24: 425-444.

Chamberlin, C. and Trumpeter: 14 (1997). "The Practice of Citizenship as Support for Deep Ecology."

Available Online at http://www.icaap.org/iuicode?6.14.2.11

Chaniotis, C., Stead, S. (2007). "Interviewing people about the coast on the coast: Appraising the wider adoption of ICZM in North East England." Marine Policy 31: 517-526

Cheong, S. (2008). "A new direction in coastal management." Marine Policy 32: 1090-1093.

Chess, C., and Purcell, K. (1999). "Public Participation and the Environment: Do we know what works?" Environmental Science and Technology 33(16): 2685-2692.

Chopyak, J., and Levesque, P. (2002). "Public participation in science and technology decision making: trends for the future." Technology in Society 24: 155-166.

Chuenpagdee, R., Fraga, J., and Euan-Avila, J. I. (2002). "Community Perspectives Toward a Marine Reserve: A Case Study of San Felipe, Yucatan, Mexico." Coastal Management 30: 183-191.

Cicin-Sain, B., and Belfiore, S. (2005). "Linking marine protected areas to integrated coastal and ocean management: A review of theory and practice." Ocean & Coastal Management 48: 847-868.

Clark, J., R. (1997). "Coastal zone management for the new century." Ocean & Coastal Management 37(2): 191-216.

CoastNET (2010) Coastal change.

Available at http://www.coastnet.ord.uk/coastalchange. Accessed on 10/11/2010

Commission of the European Communities (CEC). Recommendation of the European Parliament and of the Council 30 May 2002 concerning the implementation of integrated coastal zone management in Europe. 2002/413/EC Official Journal of the European Union 2002; 45: 24-7

The Conservative Party. (Online) Available at http://www.conservatives.com/News/News_stories/2010/03/Plans_announced_to http://www.conservatives.com/News/News_stories/2010/03/Plans_announced_to <a href="http://www.conservatives.com/News/News_sto

Cooper, C., B., Dickinson, J., Phillips, T., and Bonney, R. (2007). "Citizen Science as a Tool for Conservation in Residential Ecosystems." Ecology and Society 12(2): 11-22.

Correia, A. M. R. (2002). Information literacy for an active and effective citizenship: White paper prepared for UNESCO, the U.S. National Commission on Libraries and Information Science, and The National Forum on Information Science. Information Literacy Meeting of Experts. Prague, The Czech Republic.

Costanza, R. (1999). "The ecological, economic and social importance of the oceans." Ecological Economics 31: 199-213.

Costanza, R., Andrade, F., Antunes, P., and M. van den Belt, Boesch, D., Boersma, D., Catarino, F., Hanna, S., Limburg, K., Low, B., Molitor, M., Pereira, J., G., Rayner, S., Santos, R., Wilson, J., Young, M. (1999). "Ecological economics and sustainable governance of the oceans." Ecological Economics 31: 171-187.

Cottrell, S., P., and Graefe, A., R. (1997). "Testing a conceptual framework of responsible environmental behaviour." Journal of Environmental Education 29(1): 17-28.

Cox, M., Johnstone, R., and Robinson, J. (2008) "A conceptual model of impacts of environmental change on human well-being." Available Online. Accessed at 17/11/2010.

Cruz, J. M. (2008). "Dynamics of supply chain networks with corporate social responsibility through integrated environmental decision making" European Journal of Operational Research 184: 1005-1031.

DEFRA (2006). Promoting an integrated approach to the management of the coastal zone (ICZM) in England.

DEFRA (2007). A framework for pro-environmental behaviours.

DEFRA (2009) Our Seas - a shared resource: High Level Marine Objectives.

Dfes (2004). Making sense of citizenship: a CPD handbook.

Dobson, A., and Valencia-Saiz, A. (2005). "Introduction." Environmental Politics 14(2): 157-162.

Dolnicar, S. (2006). "Nature conserving tourists: The need for a broader perspective." An International Journal of Tourism and Hospitality Research 17(2): 235-255.

Donald, B. J. (1997). "Fostering volunteerism in an environmental stewardship group: A report on the task force to bring back the Don, Toronto, Canada." Journal of Environmental Planning and Management 40(4): 483-505.

Diner, K. (2003). Aspects of European Citizenship and Environmental Education Through Information and Communications Technology. First International conference on Environmental Research and Assessment, Bucharest, Romania.

Drevensek, M. (2005). "Negotiation as the Driving Force of Environmental Citizenship." Environmental Politics 14(2): 226-238.

Ducrotoy, J. (2001). "Higher Education in Coastal Sciences and Management: Establishing a Partnership Between Education and " Innovations in Education and Teaching International 38(3): 239-250.

Ducrotoy, J., and Pullen, S. (1999). "Integrated coastal zone management: commitments and developments from an international, European and United Kingdom perspective." Ocean and Coastal Management 42: 1-18.

Ducrotoy, J., Shastri, S., and Williams, P. (2000). "Coastal sciences and management: the need for networking in higher education." Ocean & Coastal Management 43: 427-444.

Dunlap, R., E., and, Van Liere, K., D. (1978). The New Environmental Paradigm."The Journal Of Environmental Education 9(4).

Eden, S. (1996). Public participation in environmental policy: considering scientific, counter-scientific and non-scientific contributions. Public Understanding of Science 5: 183 - 204.

Edwards, S. D., Jones, P. J. S. and Nowell, D. A. (1997). Participation in coastal zone management initiatives: a review and analysis of examples from the UK. Ocean and Coastal Management 36(1-3): 143-165.

Ferreira, M. (2002). Environment and Citizenship: from the local to the global. Active Citizenship, sustainable development and Cultural Diversity.

Finding Sanctuary. Available at <u>http://www.finding-sanctuary.org</u>. Accessed on 10/11/2010/

Finkl, C., W., and Kruempel, C. (2005). Threats, obstacles and barriers to coastal environmental conservation: Societal perceptions and managerial positionalities that defer sustainable development. Proceedings of the First International Conference on Coastal Conservation and Management in the Atlantic and Mediterranean, Algarve, Portugal.

Fletcher, S. (2003). Stakeholder representation and the democratic basis of coastal partnerships in the UK. Marine Policy 27: 229-240.

Fletcher, S. (2007). Converting Science to policy through stakeholder involvement: An analysis of the European Marine Strategy Directive. Marine Pollution Bulletin 54: 1881-1886.

Fletcher, S., and J. Potts (2007). Ocean Citizenship: An Emergent Geographical Concept. Coastal Management 35: 1-14.

Fletcher, S., Potts, J., Heeps, C., and Pike, K. (2009). Public awareness of marine environmental issues in the UK. Marine Policy 33: 370-375.

Foster, E., Haward, M. and Coffen-Smout, S (2005). Implementing integrated Oceans management: Australia's south east regional marine policy (SERMP) and Canada's eastern Scotian Shelf integrated management (ESSIM) initiative. Marine Policy 29: 391-405.

French, P., W. (2004). The changing nature of, and approaches to, UK coastal management at the turn of the 21st Century. The Geographical Journal 170(2): 116-125.

GESAMP (2001). A Sea of Troubles. Rep. Stud., GESMAP (IMO/FAO/UNESCO-IOC/WMO/WHO/IAEA/UN/UNEP Joint Groups of Experts on the Scientific Aspects of Marine Environmental Protection) and Advisory Committee on Protection of the Sea: 35.

Haklay, M. (2002). Public Environmental Information: understanding requirements and patterns of likely public use. Area 34(1): 17-28.

Hartig, T., Kaiser, F., G., and Bowler, P., A. (2001). Psychological Restoration in Nature as a Positive Motivation for Ecological Behaviour Environment and Behaviour 33(4): 590-607.

Hawthorne, M., and Alabaster, T. (1999). Citizen 2000: Development of a model of environmental citizenship. Global Environmental Change 9: 25-43.

Hay, I., and Foley, P. (1998). Ethics, Geography and Responsible Citizenship."Journal of Geography in Higher Education 22(2): 169-183.

Hegarty, A. (1997). Start with what the people know: a community approach to integrated coastal zone management. Ocean & Coastal Management 36(1-3): 167-203.

Hilson, C. (2001). Greening Citizenship: Boundaries of Membership and the Environment. Journal of Environmental Law 13(3).

House of Commons (2009). National Curriculum: Fourth Report of Session 2008-2009, Volume 1. Children, School and Families Committee. London.

Hubert, D., Frank, A. and Igo, C. (2000). Environmental And Agricultural Literacy Education. Water, Air and Soil Pollution 123: 525-532.

Humphrey, S., Burbridge, P., and Blatch, C. (2000). US lessons for coastal management in the European Union. Marine Policy 24(4): 275-286.

Irwin, A. (2001). Constructing the scientific citizen: science and democracy in the biosciences. Public Understanding of Science 10: 1-18.

Jentoft, S. (2007). Limits of governability: Institutional implications for fisheries and coastal governances. Marine Policy 31: 360-370.

Jenkins, E., W. (2003). Environmental Education and the Public Understanding of Science. Frontiers in Ecology and the Environment 1(8): 437-443.

Johnston, D., M., and Vanderzwaag, D., L. (2000). The ocean and international environmental law: swimming, sinking and treading water at the millennium. Ocean and Coastal Management 43: 141-161.

Jones, P., J., S. (1999). Economic and sociocultural priorities for marine conservation. Effects of fishing on non-target species and habitats: biological, conservation and socio-economic issues. M. Kaiser, J., and de Groot, S., J. Oxford, Blackwell Science: 354-365.

Jorgensen, B., S., and Stedman, R., C. (2001). Sense of place as an attitude: Lakeshore owners' attitudes toward their properties. Journal of Environmental Psychology 21: 233-248. Juda, L. (1999). Considerations in developing a functional approach to the governance of large marine ecosystems. Ocean Development and International Law 30: 89-125.

Kaivola, T., Ferreira. M. M. ,Alexander, F., Valdivielso, M.C., Houtsonen, L.,Miranda, B., Slater, F., and Trobat, M. (2003). How should perceptions and conceptions of sustainability and citizenship influence teaching-learning processes for sustainable development education and intercultural sensitivity? UNESCO Conference on Intercultural Education, Jyvaskyla, Finland.

Kawabe, M. (2004). The Evolution of Citizen Coastal Conservation Activities in Japan. Coastal Management 32: 389-404.

Kearney, J., Berkes, F., Charles, A., Pinkerton, E., and Wiber, M. (2007). The role of participatory governance and community-based management in integrated coastal and ocean management in Canada. Coastal Management 35: 79-104

Kearns, A. (1995). Active Citizenship and local governance: political and geographical dimensions. Political Geography 14(2): 155-175.

Kuijper, M., W., M. (2003). Marine and coastal environmental awareness building within the context of UNESCO's activities in Asia and the Pacific. Marine Pollution Bulletin 47: 265-272.

Kurtz, H. (2002) Negotiating the Meaning of Citizenship: Alternative Visions for Citizen Participation in Environmental Justice Disputes", Annual Meeting of the Association of American Geographers, Los Angeles, California. April 2002

Kusuma, H., E. (2008). Attachment to Urban Settlement and Environmental Responsibility. International Symposium on Climate Change and Human Settlements, RCHS Indonesia & NILIM Japan, Denpasar. March 18-20.

Lawrence, A. (2005). Reluctant citizens? The disjuncture between participatory biological monitoring and environmental governance. International Sociology

Association Conference ' Environment, knowledge and democracy'. Luminy, Marseilles, France.

Lee, M., and Abbot, C. (2003). The usual suspects: Public participation under the Aarhus Convention. The Modern Law Review Limited. Published by Blackwell Publishing Ltd.

MacRory, R. (1996). Environmental Citizenship and the law: repairing the European Road. Journal of Environmental Law 8(2): 219-235.

Mason, M. (2004). Citizenship beyond national borders? Identifying mechanisms of public access and redress in international environmental regimes. Research Papers in Environmental and Spatial Analysis.

Matti, S. (2006). The imagined Environmental Citizenship: Exploring the state -Individual Relationship in Swedish environmental policy. Department of Business Administration and Social Sciences. Lulea, Lulea University of Technology. PhD: 200.

McNeil, T. C., Rousseau, F. R. and Hildebrand, L. P. (2006). Community based environmental management in Atlantic Canada: The Impacts and Spheres of influence of the Atlantic Canada Action Program. Environmental Monitoring and Assessment 113: 367-383.

Mikalsen, K., H., and Jentoft, S. (2001). From user-groups to stakeholders? The public interest in fisheries management. Marine Policy 25: 281-292.

Mrazeck, R. (1996). "Seeing the Trees through the Forest: Navigating Toward Environmental Citizenship in Canada." Canadian Journal of Environmental Education 1: 121-125.

Newman, J., Barnes, M., Sullivan, H., and Knops, A. (2004). Public participation and Collaborative Governance. Journal of Sociology and Politics 33(2): 203-223.

Nickerson-Tietze, D. J. (2000). Community-based Management for sustainable fisheries resources in Phang-nga Bay, Thailand. Coastal Management 28: 65-74.

Okey, T. A. (2003). Membership of the eight regional Fishery Management Councils in the United States: are special interests over represented? Marine Policy 27(193-206).

Osborn, D., and Datta, A. (2006). Institutional and policy cocktails for protecting coastal and marine environments from land-based sources of pollution. Ocean and Coastal Management 49: 579-596.

Painter, J. (1995). Spaces of citizenship: an introduction. Political Geography 14(2): 107-120.

Peters, B., G., and Pierre, J. (1998). Governance without government? Rethinking Public Administration. Journal of Public Administration Research and theory 8(2): 223-243.

Poortinga, W., Steg, L. and Vlek, C. (2004). "Values, Environmental Concern, and Environmental Behaviour." Environmental and Behaviour 36(1): 70-93.

Potts, J. (2000). Recent developments at the National Maritime Museum, UK, and their potential for capacity building. Marine Policy 24: 79-81.

Purcell, M. (2003). Citizenship and the right to the global city: Reimagining the Capitalist world order. International Journal of Urban and Regional Research 27(3): 546-590.

Rees, S., E., Antrill, M., J., Austen, M., C., Mangi, S., C., and Richards, J., P. (2010) Is there a win-win scenario for marine nature conservation? A case study of Lyme Bay, England. Ocean and Coastal Management 53: 135-145

Rodrigues, J., Domingos, T., Giljum, S. and Schneider, F. (2006). Designing an indicator of environmental responsibility. Ecological Economics 59: 256-266.

Roth, C., E. (1992) Environmental literacy: Its roots, evolution and directions in the 1990s. Columbus, OH. ERIC Clearinghouse for Science, Mathematics and Environmental Education.

Rowe, G., and Frewer, L. J. (2000). Public Participation Methods: A Framework for Evaluation. Science, Technology and Human Values 25(1): 3-29.

Santos, I. R., Friedrich, A., C., Wallner-Kersanach, M., and Fillmann, G. (2005). Influence of socio-economic characteristics of beach users on litter generation. Ocean & Coastal Management 48: 742 -752.

Sears, A., M., and Hughes, A., S. (1996). Citizenship Education and Current Educational Reform. Canadian Journal of Education 21(2): 123-142.

Selman, P., and Parker, J. (1997). Citizenship, Civicness and Social Capital in Local Agenda 21. Local Environment 2(2): 171-184.

Seyfang, G. (2005). Shopping for Sustainability: Can Sustainable Consumption Promote Ecological Citizenship? Environmental Politics 14(2): 290-306.

Sharp, L. (2002). Public Participation and Policy: unpacking connections in one UK Local Agenda 21. Local Environment 7(1): 7-22.

Skourtas, M. S., Kontogianni, A.D., Georgiou, S. and Turner, R. K. (2005).Valuing Coastal systems in Managing European Coasts: Past, Present and Future.Edited by Vermaat J., E., et al. Berlin, Springer-Verlag: 119-136.

Smith, S., J. (1995). Citizenship: All or nothing? Political Geography 14(2): 190-193.

Smith, G. (2005). Green citizenship and the social economy. Environmental Politics 14(2): 273-289.

Smith, P., D. and McDonough, M. H. (2001). Beyond public participation: Fairness in Natural Resource Decision Making. Society and Natural Resources 14: 239-249.

Stapp, W. B., Bennett, W. Jr., Fulton, J., MacGregor, J., Nowak, P., Swan, H., Wall. and Havlick, S. (1970). The concept of environmental education. American Biology Teacher 32(1): 14-15.

Stedman, R., C. (2002). Toward a social psychology of place: predicting behaviour from place-based cognitions, attitude and identity. Environment and Behaviour 34: 561.

Steel, B., S., Smith, C., Opsommer, L., Curiel, S., and Warner-Steel, R. (2005). ublic Ocean Literacy in the United States. Ocean and Coastal Management 48: 97-114.

Stern, P., C. (2000). Toward a Coherent Theory of Environmentally Significant Behaviour. Journal of Social Issues 56(3): 407-424.

Stojanovic, T., and Barker, N. (2008). Improving governance through local Coastal Partnerships in the UK. Geographical Journal 174(4): 344-360.

Storrier, K., L., and McGlashan, D., J. (2006). Development and management of a coastal litter campaign: The voluntary coastal partnership approach. Marine Policy 30: 189-196.

Sundberg, J. (2003). Conservation and democratization: constituting citizenship in the Maya Biosphere Reserve, Guatemala. Political Geography 22: 715-740.

Thapa, B., Graefe, A., R., and Meyer, L., A. (2005). Moderator and mediator effects of scuba diving specialization on marine-based environmental knowledge-behaviour contingency. The Journal of Environmental Education 37 (1): 53-67

Tonn, B., E., and Pietrich, C. (1998). Everyday Life's Constraints on citizenship in the United States. Futures 30(8): 783-813.

Tytler, R., Duggan, S., and Gott, R. (2001). Public participation in an Environmental dispute: implications for science education. Public Understanding of Science 10: 343-364.

Tran, K., C. (2006). Public perception of development issues: Public awareness can contribute to sustainable development of a small island. Ocean & Coastal Management 49: 367-383.

UNCED (1992) United Nations Conference on Environment and Development. The Earth Summit: The Rio Declaration. Rio de Janeiro, Brazil. 2nd-14th June 1992.

UNESCO (1978). Final Report. Tbilisi, Intergovernmental Conference on Environmental Education.

UNEP (2006). Marine and Coastal Ecosystems and human well-being. A synthesis report based on the findings of the Millennium Ecosystem Assessment. UNEP: 76pp.

UNEP. (2007). Global Marine Assessments: A survey of global and regional assessments and related activities of the marine environment.

Retrieved 12/05/2010, from

www.unep.org/dewa/assessments/EcoSystems/water/MarineAssessment/index.as p.

Uneputty, P., Evans. S. M., and Suyoso, E. (1998). The effectiveness of a community education programme in reducing litter pollution on shores of Ambon Bay (Eastern Indonesia). Journal of Biological Education 32(2): 143-147.

Urama, B., C., Kenyon, W., Burton, R., and Potts, J. (2006). Integrated value mapping for sustainable river basin management: Economics, ethics and social psychology. A. T. P. S. Network, African Technology Policy Studies Networks.

Valdivielso, J. (2005). Social Citizenship and the Environment. Environmental Politics 14(2): 239-254.

Valencia Saiz, A. (2005). Globalisation, Cosmopolitanism and Ecological Citizenship. Environmental Politics 14(2): 163-178.

Williams, L. B. R. (2008). Public attitudes towards the maritime environment around the Menai Strait and Conwy Bay. CCW Policy Research Report. CCW: 55.

Worcester, R., M. (1993). Public and elite attitudes to environmental issues. International Journal of Public Opinion Research 5(4): 315-334.

Young, O., R., Osherenko, G., Ekstrom, J., Crowder, L., B., Wilson, J., A., Day, J., C., Douvere, F., Ehler, C., N., McLeod, K., L., Halpern, B., S., and Peach, R. (2007). "Solving the Crisis in Ocean Governance: Place based management of Marine Ecosystems." Environment 49(4): 20-32.

CHAPTER THREE

<u>GENERAL METHODOLOGY AND TELEPHONE INTERVIEW</u> <u>SCHEDULE</u>

3.1. INTRODUCTION

This chapter presents the two-stage methodological approach utilised to accomplish the aim and objectives of the research project. The project is firmly rooted in the discipline of social geography, with potential applications to the delivery of marine management and policy, and as such follows an appropriate research method in both phases of data collection. Any potential ethical and moral issues related to the research project will be considered, as will measures of quality assurance. The second part of this Chapter will discuss the specific methodology applied to the first phase of data collection; the practitioner interview survey, which will be discussed in more detail in Section 3.5-3.8.

3.2. GENERAL METHODOLOGICAL APPROACH

3.2.1. Overall Methodology

Research projects can utilize either an inductive or deductive progression depending on the requirements of the study (Kitchin and Tate, 2000). The differences between these two techniques are outlined in Table 3.1. For the purposes of this research project, the research will progress inductively with each phase guided by the preceding data collection. The premise of inductive research is that as the data collection and analysis progress, any conclusions or observations made regarding the raw data become narrower, eventually focusing on specific components of the research (Kitchin and Tate, 2000; Thomas, 2003).

Inductive Research	Deductive Research
• Research is conducted before	• Theory is established prior to
theory is established (Kitchin	the research (Kitchin and Tate,
and Tate, 2000).	2000).
• Theory is established through	• Research is conducted to
the identification of themes and	examine theories, providing
patterns in the data (Kitchin and	validation of pre-existing
Tate, 2000).	concepts (Kitchin and Tate,
• Research is not based on testing	2000).
a hypothesis but on synthesis of	• Most commonly driven by
the theory (Kell and Oliver,	hypothesis testing (Kell and
2003).	Oliver, 2003).

Table 3.1: Differing approaches of inductive and deductive research inhuman geography.

The concept of marine citizenship is complex and ultimately concerned with human society and behaviour towards the marine environment. Given this complexity, a mixed methods approach was identified as the most appropriate research framework to ensure a comprehensive investigation. The characteristics of qualitative, quantitative and mixed methods research are outlined in Table 3.2. Mixed methods research can be a combination of qualitative and quantitative methods, or a mix of just qualitative methods or a mix of purely quantitative methods (Brannen, 2005). Historically, these traditions have been kept distinct from each other but in recent years, there has been a movement towards using the strengths of both to benefit research (Johnson and Onwuegbuzie, 2004).

Mixed methods approaches are often used in social investigation where research is complex and qualitative or quantitative research methods are deemed inadequate on their own (Creswell, 2009). Mixed methods can formally be defined as when the researcher combines quantitative and qualitative research approaches and methods in a single study (Johnson and Onwuegbuzie, 2004). It is seen as a complementary technique using the strengths and limiting the weaknesses of the more traditional approaches to research (Pope and Mays, 1995; Creswell, 2009). In order to allow the project a high degree of flexibility and adaptation, an inductive mixed methods approach was adopted to ensure comprehensive understanding of applicability of marine citizenship to the management of the marine environment.

Table:	3.2.	Main	characteristics	of	qualitative,	mixed	methods	and
quantit	ative	approa	ches to research	(Ad	apted from C	reswell,	2009)	

Qualitative Research	Mixed Methods	Quantitative Research
Takes a constructivist	Takes a pragmatic	Takes a post-positivist
participatory knowledge	approach to the data.	Uses close ended questions
claims.	Can use both open and	and tends to use numeric
	closed questions, emerging	data
Uses open ended	and predetermined	Tests theories and verifies
approaches, generally uses	approaches associated with	explanations.
text or image as data.	both qualitative and	
	quantitative data and	Needs identification of
Researchers have to	analysis.	variables for investigation.
consider their own position	Callects both quantitative	Uses validation standards
In the context of the data.	and qualitative data.	Uses validation standards.
Focus on a single	-	Observes and measures
phenomenon.	Integrates data at different stages	numeric information.
Studies the participant's		Employs statistical
settings and aims to obtain	Employs both qualitative	procedures.
insights into particular	and quantitative research	
phenomena.	approaches.	Relates the data and
		questions to hypotheses.
Considers aspects of	Tends to present visual	
quality assurance of the	pictures of methods in the	Uses surveys and
data.	study.	experiments.
Data tends to develop an agenda for reformation of a theory	Uses sequential, concurrent and transformative methods	
theory.	memous.	
Researchers bring personal		
values to the study.		
Uses strategies of		
phenomenology, grounded		
theory, ethnography, case		
studies and narratives.		

As with other research approaches, there are various directions mixed methods research could take (Creswell, 2009). The research took a sequential mixed methods design with an initial qualitative phase, followed by a concurrent mixed methods phase including both qualitative and quantitative data collection (presented in more detail in Table 3.3 and Figure 3.1). This allowed identification of key elements and views required for further investigation in the second phase of the research, as is common in mixed methodologies (Creswell, 2009).

Transformative	Sequential	Concurrent
Can be both sequential	Usually quantitative data	Most familiar and well
and concurrent in nature.	collection and analysis,	validated method
	followed by a qualitative	
A theoretical aspect	collection and analysis	Collection of both
underpins the basis of the	phase.	qualitative and
research providing a		quantitative data
directional question for	Can be weighted the	simultaneously making
the research.	other way with a	comparisons between the
	qualitative phase	two.
Can follow a sequential	conducted initially,	
pattern of two distinct	followed by a	Used so that the research
phases.	quantitative phase of	benefits from the
	data collection.	strengths of both
		qualitative and
	Mixing of the data	quantitative research.
	occurs prior to the	Mining of the data accura
	is informed by the initial	through transformation
	Is informed by the initial	ar comparison of the
	quantitative phase of	dete
	research.	uala.
	Typically used to	Can be difficult to
	investigate the	compare data in different
	quantitative phase	formats
	further	
	Can be either	Can be either
	explanatory or	triangulation (1 phase of
	exploratory in nature	data collection) or an
		embedded (2 phases of
		data collection)
		concurrent strategy

 Table 3.3: The three main mixed methods strategies that could have been

 applied to the research (Adapted from Creswell, 2009)

Table 3.4 summarises both phases of the research project the research project consisted of two phases of data collection; the marine practitioner telephone interview schedule and thematic case studies which are explained further in Sections 3.2.2.1 and 3.2.2.2.

Phase	Data Collection	Method
Pre-da	ta collection Literature l	Review
Phase One	Practitioner perception of	In-depth semi structured
Chapter 3 and 4	role of marine citizenship	telephone interviews
Phase Two	Case Study Examination	Personal attachment;
Chapter 5 and 6	of key elements identified in Practitioner survey (Phase 1)	examined through community based semi structured interviews
		Education; School based questionnaires

When conducting mixed method-based research, it is important to have a clear idea of how the research is going to progress and how the qualitative and quantitative components of data collection relate to each other. Figure 3.1 illustrates the overall progression of the research indicating an overarching sequential approach to the research including a concurrent mixed methods data collection phase.



Figure 3.1 Mixed method design applied to the research

3.2.1.1 Telephone Interview Schedule

In order to ensure that the research established a representative view of UK marine practitioners, data was sought from a diverse range of marine practitioners across a wide geographic and professional scale. A total of 42 telephone interviews were conducted using a semi-structured interview technique. Interviews were hand scribed, immediately typed and returned to interviewees for clarification and confirmation. In conjunction with providing valuable insights into practitioner perceptions of the role of, and elements influencing, communities in management of the marine environment, the telephone interviews also allowed identification of the areas requiring further investigation. The specific methodology and justification of conducting the telephone interview schedule is presented in Chapter Three, Section 3.3.

3.2.1.2 Thematic Case Study Schedule

Two key themes, education and personal attachment, were identified for further investigation following the practitioner telephone interview schedule, forming the focus of the second phase of the data collection. Various approaches were used to investigate these themes. A school survey was used to investigate the theme of education within three schools at selected case study sites, while a community based semi-structured interview was administered to examine the theme of personal attachment at five case study sites. A variety of locations were selected to ensure a comprehensive and representative assessment of the factors was carried out. A full description and justification for the methodological approach and analysis in the case study phase of research is described in Chapter Five. Synthesis of the telephone interview schedule and the case study research provided the research with data from which a model of marine citizenship could be derived (as presented in Chapter Seven), and recommendations for the application of marine citizenship to the sustainable management of the marine environment (presented in Chapter Eight).

3.2.3 Quality Assurance of Data Collection Phases

Throughout the research project, the data collected was quality assured in a number of ways to ensure reliability of the data collection and during the data interpretation and analysis as presented in Table 3.5.

Table 3.5: Methods of assuring the quality of the data collected during both phases of the study.

Quality Assurance Method	Phase of Research
Pilot Interviews	Phases 1 and 2
Hand scribing of interviews	Phases 1 and 2
Immediate typing and data collection of school based questionnaires	Phase 2
Hand written notes to accompany hand scribed interviews	Phases 1 and 2
Immediate reflection following interview	Phases 1 and 2
Continual back up of data	Phases 1 and 2
Generation of typed transcript of interview	Phases 1 and 2
Transcript returned to candidate for clarification to allow them to make alterations and check meaning where needed	Phase 1
Retyping of interviews, application of any changes noted following clarification by interviewee. This was followed by an immediate review of the transcript to establish important alterations.	Phase 1
Checking of coding system	Phase 1 and 2

3.2.4. Overall Moral and Ethical Considerations

The research was conducted in accordance with Bournemouth University's Research Ethics Policy and Procedures. As a mixed methods approach was used for this study, moral and ethical issues pertaining to both qualitative and

quantitative approaches should be considered (Creswell, 2009). The need for informed consent from participants involved in academic research has been well documented (Mason, 2002; Denscombe, 2003; Creswell, 2009). In the case of this study, informed consent was obtained either verbally through telephone calls or written consent through emails in phase one, and through written or verbal consent from participating schools and interviewees at case study locations in phase two. Creswell (2009) identifies a number of ethical elements that were included when obtaining consent from participating interviewees:

- The right to anonymity of participants.
- The right to a mutually beneficial process for both the researcher and the participant.
- The right to participate freely in the research while retaining the right to withdraw from the process at any time.
- The right to ask questions throughout the interview process.
- The right to be fully informed about the research and the role of the participants in the study.

Participants were provided with a short project rationale and briefing to ensure full understanding of the requirements of the surveys and interviews in both phase one and two. During the case studies, a short introduction was given upon approaching potential participants and it was ensured that participants fully understood and had given full verbal consent to continue with the interview

3.2.5. Scope and Limitations of the general methodology

The scope of this research was restricted to marine practitioner and community perceptions of the marine environment during the timeframe of this research. It should be noted that during the time period of the research project, statutory management of the marine environment in the UK underwent significant change with the ratification of the Marine and Coastal Access Act (2009), in addition to the EU Marine Strategy Framework Directive (2008/56/EC). This did not prove to be a significant issue for the selected research method, but it is a factor, which should be given consideration when discussing the general observations.

Although the general mixed methods approach allowed a comprehensive investigation into the emerging theme of marine citizenship, the length of time required to carry out sequential research of a project of this type could be perceived as a potential issue. However, it was determined that a distinct initial phase of data collection was integral to the success of the research as it allowed formation of a base model of marine citizenship and did not prove to be problematic for the progression of the research.

The concurrent second phase of mixed methods based data collection allowed investigation of the key elements identified in the initial phase of data collection. Creswell (2009) suggests that carrying out concurrent data collection may lead to difficulties as a result of requiring interpretation of data in both qualitative and quantitative formats. However, in the case of this study, it was decided that the qualitative element of the second phase of data collection provided necessary and detailed information for complete examination of the research question, supporting the use of concurrent mixed methods.

Creswell, 2009 suggests that any model derived from an inductive methodological approach will only be relevant to the context in which it was developed. In order to ensure wider applicability of the research, the telephone interviews carried out in Phase One of the data collection ensured that practitioners were selected from a wide geographical distribution. The research further acknowledges that the wider applicability, i.e. application outside the UK, of any model and recommendations concluded from this study will require further investigation. In spite of this, any observations made from this research are reported as an important contribution to the general debate surrounding the application of citizenship to the management of the marine environment.

Sections 3.3–3.7 outline the specific methodology applied to the first phase of data collection, the marine practitioner telephone interview schedule.

3.3 PHASE ONE DATA COLLECTION: MARINE PRACTITIONER SURVEY

Sections 3.3.1- 3.4 present the development and selection of methods used to collect data for the first phase of this project. This involved the generation of an interview for surveying individuals associated with organisations involved with the management, development and conservation of the marine and coastal environments.

3.3.1. Qualitative data collection

Gillham (2000) identifies a number of benefits that can be associated with a qualitative approach to data collection, which can be applied to the first phase of the research. These benefits include:

- An ability to conduct investigations that other methods may not allow due to practicality and ethics;
- Allow examination of under studied theories and concepts providing information for later research;
- The encouragement to explore complexities in greater detail in a manner that may not be possible through more controlled methods.

Qualitative research approaches are known to provide a more in depth, detailed level of content, which can then undergo various interpretations dependent on the requirements of the research project (Graneheim and Lundman, 2004; Kaplon and Duchan, 1988). This level of detail and interpretive flexibility lends itself well to investigation of emerging theories, as is the case in this research (Kaplon and Duchan, 1988). The flexible nature of this approach also allows adaptation of the data requirements as the project progressed, ensuring that the data formed the theory, rather than attempting to relate the data to existing theories.

Historically, qualitative research has moved away from numerical justification characteristic of quantitative research methods, and as a result, there are a number of issues that need to be addressed with regard to the validity of qualitative research (Morse *et al.*, 2002). The potential limitations of qualitative research have been identified as including the potential for the following: biased interpretation, incorrect transcription, unsubstantiated generalisation and unintentional focus on positive cases (Gibbs, 2002). The measures taken to overcome these challenges to ensure validity and accuracy of the data throughout the collection phases of this research are presented in Table 3.2.

3.4 TELEPHONE INTERVIEW SCHEDULE

3.4.1. Selection of interview method

The primary data collection for this part of the study involved conducting a survey, with the aim of gaining insight into the opinions held by individuals involved in the management of the UK marine environment on the role and influences of citizenship with respect to the marine environment. Various qualitative data collection methods were considered for this phase of data collection, including focus groups, interviews and questionnaires (Creswell, 1994; 2009). Based on criteria, including expense and logistical feasibility, it was decided that face-to-face interviews, telephone interviews and mail based surveys were the most appropriate potential data collection options for this study (Denscombe, 2003). These three methods are defined below with further explanation of related advantages and disadvantages outlined in Table 3.6.

- Face-to-face interviews are defined as the administration of a structured or partially structured questionnaire in the presence of a respondent (De Leeuw, 1992; Denscombe, 2003). It has been suggested that for research involving members of the public, face-to-face interviews are the most effective method of data collection (Denscombe, 2003).
- A telephone interview is based on the same premise as the face-to-face interviews and involves conducting interviews though a telephone, often

through a centralised centre with a team of interviewers (De Leeuw, 1992).

• In contrast, postal or email based, 'self completion' surveys (Denscombe, 2003) are often used in a case study environment, in which respondents receive a structured questionnaire and respond in their own time by returning the survey and following the instructions in the accompanying letter (De Leeuw, 1992). Postal surveys characteristically have low return rates with only an estimated 20% of potential participants typically returning surveys (Denscombe, 2003).

1992). Data Callantian Mathad	A discontances	Dicadvantano
Data Collection Method	Advantage	DISadvantage
Face to Face Interviews	Flexible and open method of interviewing,	• Can be time consuming as face to face
	allowing the researcher maximum control.	interviews are generally conducted over a
	• Interviewer is available to answer	longer time period than telephone
	questions and clarify any issues	interviews
	(Denscombe, 2003)	• Face-to-Face interviews can often be
	Allows utilisation of non-verbal responses	expensive, as they generally involve either
	and may produce more complete answers	the researcher or the respondent travelling
	Researcher has a high level of control	in order to facilitate the interview. With
	over respondents selected	regard to this project's international
	The researcher gets non-verbal signs as	
	well as verbal answers from the	• Can be intrusive on a respondents
	interviewee with regard to how the	personal time and space, making mgn
	interview should proceed	rerusal levels a possibility
	Verification of interviewees answers is	
	easy due to the interviewee's presence	
Telephone Interviews	Interviewer is available to answer	• Have been shown to have higher refusal
	questions and clarify any issues	rates than face to face interviews
	• Less intrusive on a respondent's personal	• Previous comparisons of methods have
	time and space	shown that telephone interviews can result
	• Increased sense of anonymity for the	in acquiescence or extremity in answers,
	participant (Sturges and Hanrahan, 2004).	which can mean the results may not
	Researcher has a high level of control	accurately represent public opinion on an
	over respondents selected	ISSUE
	Will allow an international selection of	
	respondents to be approached for	
	interview easily and with a lower financial	
	impact	
	Verification of respondents answers can	
	be carried out	
	• Less costly than interviewing in a	
	respondent's normal environment (Sturges and Hanrahan 2004)	
	uity 110111011011 400 1).	

Table 3.6: Benefits and problems associated with each of data collection technique considered (Adapted from De Leeuw,

с Г				_	-	
Data Col	lection Method		Advantage		Disadvan	tage
Mail Based Surv	eys/ Questionnaire	S	ondent has full control	ol on the time	Researcher has min	nimal control over the
		take	n to return the survey	(respondent complet survev (Denscombe	ing and returning the 2003)
			unal intrusion on maltime and snace	respondents	• Can have low ret	, ±000) turn rates as cannot
			onde and opace		guarantee potential r survey	respondents will return
				_	• There can be no ve to respondent's ansv	srification with regard vers
					Respondent has littl have any queries respondent is absent	e or no opportunity to dealt with as the
Table 3.7: Vario	us dimensions of in	iterviews (Adapte	d from Gillham, 2	000).		
Unstructured Structured			Semi-str	ructured		
Listening to and	Using natural	'Open ended'	Semi-structured	Recording	Semi-structured	Structured
observing participants conversation	conversation to investigate research	interviews directed by a few key	interviews using both open and closed	schedules i.e. verbally administered	questionnaires: multiple choice and open ended	questionnaires consisting of simple, direct
	questions	interview	questions	questionnaires	questions	and closed
		questions to				questions
		guide the interviewee				

Based on an assessment of the analysis in Table 3.6, telephone interviews were identified as the most efficient method of data collection in this phase of the research. Although, there is evidence to suggest that face-to-face interviews commonly result in higher response rates and can produce more complete answers as a result of their personal nature (De Leeuw, 1992), this method was deemed unsuitable due to time constraints and the greater expense associated with face-to-face interviews (Denscombe, 2003). Telephone interviews on the other hand provide many of the advantages exhibited by face-to-face interviews (Table 3.6), such as the interviewer being available for clarification whilst having the added advantage of being less intrusive than face-to-face interviews can be, as well as presenting other more logistical benefits, such as speed of administration and a lower financial burden (Denscombe, 2003; Bourque and Fielder, 2003). Previous research has indicated that interviewe participants often prefer telephone interviews as they preserve interviewe anonymity and are considered to be more convenient (Sturges and Hanrahan, 2004).

For the purpose of this research, telephone interviews also ensured that a wide geographical distribution of interviewees could be selected with minimal time and financial cost (Fenig *et al.*, 1993; Bourque and Fielder, 2003; Sturges and Hanrahan, 2004). When compared with self-completion postal or web-based questionnaires, telephone interviews allow greater opportunity for collection of more in depth information. This is made possible in telephone interviews through the inclusion of more questions than in a self-completion questionnaire (Bourque and Fielder, 2003). Many of the potential challenges of telephone interviews, such as high refusal rates (De Leeuw, 1992; Bonnel and Le Nir, 1998), dissatisfaction with interview length and poor cooperation from participants (Holbrook *et al.*, 2003), were mitigated for in this project from the outset by sending an introductory letter and short project rationale to each potential interviewee at least two to three weeks prior to the interview date, introducing the project and explaining for the proposed research. Examples of both documents are provided in Appendices 1 and 2.

3.4.2 Use of Semi-structured Interviews

Interviews are commonly used in social research as they can provide detailed information and allow deeper examination of information (Mason, 2002; Denscombe, 2003). Table 3.7 presents an illustration of the continuum of possible interview techniques available for use in data collection. Beneficial levels of flexibility and adaptability, as well as potential for clarification not possible in other methods of data collection, are characteristic of interviews (Denscombe, 2003; Bell, 2005). For the purposes of the practitioner survey, semi-structured interviews were deemed the most appropriate method of data collection (See Chapter Five, Table 5.8 for a comprehensive outline of the most common forms of interview used in research). Semi-structured interviews typically consist of a number of open-ended questions addressing a clear list of issues while encouraging researcher flexibility with regard to the order of the questions and potentially addressing new issues (Denscombe, 2003). This allows the participant to elaborate on areas of particular interest, providing additional detail for the research. The most common features of semi-structured interviews as identified by Mason (2002) are as follows:

- The dialogue of the interview should be an interactive exchange between interviewer and interviewee.
- The interviews should have an informal, conversational tone.
- Generally interviews are unlikely to consist of a complete, standard set of questions.
- Interviews should have a flexible structure to allow for movement away from the preconceived questions.
- Interviewee can elaborate on points most relevant to them and are given the freedom to guide the interview.
- Any information obtained should be considered in context of the interviewee.

These features were applied to the telephone interview schedule by generating a semi-structured interview template with the flexibility to cover topics thought to be important by the interviewee. The use of a conservational style of

interviewing ensured that the information obtained was detailed and could be used as an accurate representation of the topic being discussed by the interviewee.

3.4.3 Determination of Case Study Option

Before identifying organisations from which suitable candidates could be interviewed, it was necessary to determine the direction in which the research would progress in order to ensure that appropriate interviewees were selected. The aim of this phase of research was to generate a theoretical model of marine citizenship based on the opinions on marine practitioners, following which examination of the key elements of the model was carried out through case study research (Chapter Five). It was clear that there are numerous paths that the project could be taken, and each needed to be considered carefully as the choice of interview candidates for telephone interviews could determine potential case studies. These options were considered and are rationalised in Table 3.8. The challenges associated with conducting international case studies included the possibility of a less detailed, superficial investigation being conducted as a result of increased travelling times, language barriers and the need to alter questionnaires and interviews to suit the marine management context of each case study site. Given these concerns, the advantages and disadvantages of using both international and UK only case studies, outlined in Table 3.9, were considered in order to aid selection of the most appropriate case study option.

Options	Description of Options
Option One – UK based model with UK case studies	Model developed from interviews from UK based marine and coastal organisations. Model tested in UK case studies.
Option Two – UK based model tested internationally	Model developed as in option one, but would be tested on an international basis.
Option Three – International Model and Case studies	Model developed through the same method as options one and two, but on an international scale. Model tested through international case studies.
Option Four – combination of UK and International model and case studies.	Model developed by interviewing UK case studies and individuals from similar organisations in the case study countries. Model tested in both the UK and countries where international interviewees were based.
Option Five – UK based model tested in both UK and International case studies	Model developed as in options one and two, but tested in both UK and international case studies.

Table 3.8: Description of the five options considered for direction of research

Table 3.9: Comparison of the advantages and disadvantages associated with UK and International case studies and UK only case studies.

UK and In	ternational	uo NU	ly
Advantages	Disadvantages	Advantages	Disadvantages
Assesses transferability of the concept.	Financial constraints.	Minimal language barriers expected.	No international transferability of the concept.
Relatively good access to finances due to research fund available.	Language barriers.	Minimal costs with regard to time and finances.	Government devolution could prove challenging to the research.
Marine environment is an international resource.	Potentially over-stretching the concept.	Variety of case study locations available.	
Variety of places that could be selected as case studies.	Could it really be considered international with only a sample of countries tested?	Similar political systems across the four home countries.	
Cultural variances.	Significant time demands through travel to case study sites and potential need for	Devolution could lead to potentially interesting issues.	
	translation of data.	Allows detailed assessment of Marine Citizenship with depth and breadth of potential case studies.	
		Cultural variances in the UK.	

It was important that all five potential options available for this research were carefully considered so the most effective method for the generation of an accurate and efficient model could be chosen.

Option 1 suggested using a purely UK based model, tested only in UK case study areas. A benefit to this method was that there would be no question of the generated model's relevance to the case study areas. Although this method could potentially be considered restrictive due to its small geographical scope, the UK is a multicultural nation and given the 19,491 miles of coastline (British Cartographic Society, 2009 Online), its marine environment supports a diverse range of uses, stakeholders and communities. Studies have shown that culture and background can have a significant influence on the development of environmental citizenship, and it can be assumed that the same would be true of marine citizenship (Hawthorne and Alabaster, 1999). It was suggested that cultural differences could be considered within a UK only based case study, examining the differences that may be present as a result of devolution by having a case study site in each of the home countries (i.e. England, Scotland, Wales and Northern Ireland).

Option 2 proposed using a conceptual model generated solely using the results of UK based telephone interviews, and testing the model in an international arena. Although there have been calls for a globalisation of environmental management, it is imperative that the individuality of specific cases be considered in order for management regimes and development plans to succeed. Therefore, it was considered potentially beneficial to create a comprehensive working model in one region (UK) with the potential to evaluate its transferability in other areas. However, it was determined that this option would have been limited by an inability to carry out a completely international case study phase of research.

Given the time and financial constraints of this project, option 3 would not be feasible. It would have been impractical to attempt a completely international model in the time available. In order for it be accurate, representatives from the majority of, if not all, global nations would be required for interview, as well as requiring a high number of case studies with sufficient geographical spread to
ensure it could be considered an "international model". Without this, it would be difficult to establish what exactly would constitute an international model and therefore it could not be treated as a comprehensive, working international model.

Option 4 proposed that case study areas would be selected prior to the interviews taking place. It is then suggested that some of the candidates selected for interview be from these international case study sites, in order to ensure that the model generated from the results of the interview can be related to these particular sites. Although this method would give the research an international element, it could be suggested that the model would be as geographically bound as option 1 would be, and that the research would not be applicable to anywhere but the nations in which the telephone interviewees were based. It is also important to consider difficulties that may be present as a result of language barriers; this could be particularly problematic whilst attempting to conduct telephone interviews.

Option 5 appeared to be the most effective method of introducing an international element into the research within the time and financial budget available. Option 5 proposed the generation of a model for marine citizenship based on interviews conducted within the UK, including UK offices of international organisations. As described, the model would then be tested in two phases; UK based case studies, as well as international cases. This would ensure that the research can be applied and tested globally, and would simultaneously examine the model's transferability to varying locations.

Although Option 5 was identified as the best method of ultimately generating an international working definition and model of marine citizenship, it was decided that a comprehensive, detailed evaluation of a model for marine citizenship in the UK would be of more benefit. It was a concern that attempts to produce a model with a broader geographical distribution would be less detailed and would not generate the standard of model desired. As a result, it was decided that Option One (UK based model and case study phase) would take this research in the appropriate direction. It was acknowledged that it is important that consideration

be given to the influence of location dependent variations in social, economic and cultural factors when selecting the case study sites. As previously mentioned, it has been found that cultural heritage and background can have an influence on the development of the sense of environmental citizenship within a community (Hawthorne and Alabaster, 1999). Thus it can be inferred that similar factors will be influential components in generating a model for marine citizenship. Given the multicultural diversity of the UK and the vast range of users and uses of the marine environment, it was decided that the UK could provide the detail required to produce an all-inclusive, wide-ranging template for marine citizenship.

3.4.4. Method of telephone interviewee selection

One of the challenges facing social research is the inability to collect data from every individual within a research category and, as a result, researchers are reliant on information obtained from a representative portion of the sample population (Denscombe, 2003). Once the direction of the research was decided, a sample group for the interviews in Phase One had to be determined. In order to determine which organisations to approach for potential interview candidates, a number of stakeholders, their involvement and interest in sustainable management and thus, the development of the concept of marine citizenship, had to be considered. It was decided that organisations that could potentially benefit from marine citizenship and those involved in marine management would be the most appropriate interviewees. This included management authorities as well as policy makers, as successful implementation of management regimes and legislation will be influenced by acceptance and input from the public. Throughout the interview process there was an element of "snowballing"⁵ with many interviewees suggesting other potential interviewees. Unlike in quantitative based research, random sampling is often not the best method of investigating the complexities relating to human behaviour (Marshall, 1996; Denscombe, 2003). For this study, a combination of two commonly used sampling strategies was used; namely, judgemental and theoretical sampling (Table 3.10). By using a combination of these methods, it was possible to select

⁵ Snowballing applies to the identification of potential interviewees during interviews.

interviewees who could contribute meaningfully to the research (Denscombe, 2003) as well as allowing for 'snowballing' (Marshall, 1996; Fink, 2003a). Reaching theoretical saturation⁶ was used as a guideline as to how many interviewees were required for a comprehensive investigation.

Qu	alitative Sampling Technic	lues
Convenience	Judgement	Theoretical
Selection of most	Most common technique	Samples are done
accessible candidate		dependent on the
	Active selection of the	requirements of the
Has time, cost and effort	most productive sample	research
benefits		
	Can lead to snowball	Samples are selected
	sampling	based on emerging
		theories to encourage
	Encourages selection of	further elaboration
	confirming and	
	disconfirming subjects	

 Table 3.10: The most common qualitative research sampling techniques

 (Adapted from Marshall, 1996)

3.4.5. Telephone Interview Design

The preceding literature review was used to identify gaps in the current knowledge regarding citizenship and its application to the marine environment. In accordance with an inductive approach, the information from the literature review was used to develop the interview template for the second phase of the data collection, the telephone interviews. Table 3.11 illustrates the link between the literature gaps, data requirements for this project and the interview questions. The interview was designed in order to achieve the following objectives:

- To establish the understanding of citizenship, both general and in an environmental context, held by organisations involved in marine and coastal management, and involved in dealing with the general public;
- To determine the level of concern, awareness and sense of responsibility thought to be held by members of the public with regard to the marine environment;

⁶ Theoretical saturation is the point at which no new information is derived from the data collection.

• To identify the perceived requirements for increased public involvement in the marine environment and its management, as well as the benefits this would be considered as having for its long-term sustainability.

Table 3.11: The links between the knowledge gaps identified in the literature review, the data requirements of this project and the appropriate questions generated for the telephone interviews.

Gaps in the Literature Review	Data Requirements for Research	Telephone Interview Questions
Poor understanding of citizenship in management, development and conservation of the marine environment – necessary for successful development of marine citizenship	General Understanding of Citizenship	In your opinion what is the role of the individual in the management and decision-making process with regard the marine and coastal environment? What is your understanding of Citizenship? And what do you understand it to mean in an environmental context?
No real understanding of the benefits industry would have from a greater sense of marine citizenship Currently there is no connection between the current meaning of citizenship applies to the marine environment.	Benefit of developing marine citizenship	How do you think the concept of citizenship relates to the sustainable management of the marine environment and its associated resources? In recent times, there has been a global promotion of the concept of citizenship in areas of political development. Should also apply to the marine and coastal environment? If so, to what extent?
There is a need to examine how effective increasing public involvement in marine environmental issues and management would be. There is a lack of knowledge regarding promotional methods for increasing environmental responsibility.	Benefit of increasing public involvement with marine management	 What effect do you think increasing public involvement in the management of marine resources would have? How do you think public involvement in marine and coastal management would influence future policy implementation? How do you think the sense of environmental responsibility within today's society should or could be altered and promoted?
There is a lack understanding of the factors influencing citizenship in an environmental context, and little research has been conducted in a marine specific context	Factors that could be considered as having an influence on the development of marine citizenship	What factors do you think influence a sense of citizenship amongst the public towards the marine environment and its resources?

Gaps in the Literature Review	Data Requirements for Research	Telephone Interview Questions
There is no record of practitioner opinion on current management of the marine environment in the UK the role of the public and how it would benefit management of the marine environment.	Opinion of current marine and coastal management	How do you think the current management of the marine environment affects its long-term sustainability? Who do you consider responsible
As above		for the management of the marine environment and its resources? – for each answer, attempt to explain at what level and why?
As above		How active a role do you think communities and stakeholders should take in the management and decision making processes with regards to the marine environment?
There is no real understanding of the level of awareness and understanding of issues facing the marine environment. Practitioner perception of this could aid in	Public knowledge of marine and coastal issues and legislation	How concerned do you think the general public are regarding the condition of the marine environment?
promotional techniques and establishing the factors considered most important		How do you think public awareness can be encouraged?
considered most important		Who should be responsible for providing the resources for this?
		With regard to the previous question, what scale do you think people's awareness of the marine environment and the issues facing it are? (I.e. local, regional, national, global?)
No suggestions of how to improve from those currently involved in marine management. This would aid in indicating	Methods of encouraging a sense of stewardship within the general public as well as a sense of stakeholder responsibility	How do you think a sense of responsibility towards the marine environment can be promoted?
potentially influential factors of marine citizenship.		Whose responsibility do you think it is to do this?
There are perceived gaps in public understanding of the interaction there is between the marine environment, social and economic systems.	Examine the level of awareness of the link between environmental, economic and social systems	To what degree do you think environmental issues, in this case mainly focusing on marine issues, relate to every day life in human society?
		Does the state of the environment and its management have an impact on social and economic issues? Explain answers

Interviews and questionnaires lend themselves to different forms of questioning, namely open and closed questions (Denscombe, 2003). The advantages and disadvantages of both of these formats are explained in Table 3.12. For the purposes of the practitioner survey of the data collection, given the desired semi-structured nature of the interview, it was decided that relatively open questions would be most appropriate.

Table 3.12: Open Questions Vs Closed Questions (adapted from Denscombe,2003).

Open Q	uestions	Closed (Questions
Advantages	Disadvantages	Advantages	Disadvantages
Respondent can choose wording and length of answer depending on what they are comfortable with.	Require more effort from respondents which can reduce willingness to be involved.	Structured answers in categories pre- established by the researcher.	Less scope for answers that reflect the feelings of the respondents.
Data gathered is likely to rich and complex, adding detail to the data being collected.	The data collected is often raw and requires intense analysis to reach conclusions.	Responses can be simple or complex, depending on the requirements of the data. Pre-coded data collected is easily quantified and analysed.	Respondents can get frustrated at the lack of opportunity to express their views without restriction.

3.5 TELEPHONE INTERVIEW STUDY

3.5.1. Pilot Study

Prior to the main study being conducted, a pilot interview survey was carried out in order to test the methodology. A pilot study refers to a small scale version of an intended full scale study, enhancing the likelihood of a successful main study (van Teijlingen and Hundley, 2002). Conducting a comprehensive pilot study also allows evaluation of participant understanding of the questions to limit any potential difficulties in the main study (Lancaster *et al.*, 2004). The pilot study interviews were carried out during August 2008 and consisted of interviewing five individuals who worked for organisations related to the marine management. The transcripts of these interview, interviewees were asked to evaluate the interview questions and style of interview. Analysis of the pilot study prompted some minor changes to the interview questions, with some alterations to phrasing made along with the addition of two questions. Both the original and altered versions of the interview are included in Appendices 3 and 4.

Although interviews are commonly audio-recorded, studies have shown that if detailed notes are taken during the interview and written up within 24 hours of conducting the interview, recording is not necessary (Darke *et al.*, 1998). Given this, it was decided to record the interview by hand and then type them to produce an electronic transcript. Following the pilot study, it was determined that in order to ensure recording had been accurate, in the actual study typed interview transcripts should be returned to the interviewee via email requesting confirmation that the transcript was an accurate record of what had been discussed. This opportunity to allow the interviewee to reflect on a researcher's interpretation of their responses has been recommended by other studies as a method of ensuring the collection of rich and accurate data (Travers, 2001).

3.5.2. Actual Study

The actual practitioner survey study took place between September 2008 and January 2009 with a total of 42 marine practitioners interviewed, representing a response rate of 49% of the individuals contacted⁷. A wide range of individuals were contacted with individuals from various sectors, organisations, both governmental and NGOs, as well as academics were interviewed to achieve as broad an evaluation as possible (presented in Table 3.13). The average length of the telephone interviews was approximately 30 minutes, with the shortest being 25 minutes and the maximum time taken for interview being 80 minutes. Each candidate was initially contacted by email and was provided with the information and rationale as produced for the pilot study (Appendix 1 and 2). As determined by the pilot study, interviews were recorded by hand, later word-processed and returned to the interviewee for confirmation of their accuracy. This proved to be a relatively successful method with 24% of interviewees returning their transcripts with changes. These were mainly of a clarification nature where the

⁷ None of the individuals contacted refused to participate in the study; however, during the interview period marine practitioners in the UK were producing the now Marine and Coastal Access Act, and it is most likely that potential interviewees did not have the time to participate meaningfully in the study.

interviewee wished to enhance the information given rather than correct misinterpretation or mistakes. In accordance with inductive research, the total number of interviews conducted was not pre-determined, instead it was determined that interviews should be conducted until theoretical saturation (Guest *et al.*, 2006), was reached.

Categories	Examples of Organisations
Local Governance Organisations	Bournemouth Beach Forum
	Scottish Sustainable Marine Environment
	Initiative (SSMEI) – Sound of Mull
	SSMEI – Berwickshire
	SSMEI – Shetland
	SSMEI – Firth of the Clyde
	Scottish Environmental Protection Agency
	(SEPA)
Regional Governance Organisations	Dorset Coastal Forum
(Government and NGOs)	Dorset County Council
	Forth Estuary Forum
	Solent Forum
	English Heritage
	Department of Environment N. Ireland
	Sea Fisheries Cornwall
	Ulster Wildlife Trust
	Devon Maritime Federation
	Sefton Coastal Partnership
	Severn Estuary partnership
	River Hamble Authority
	Durham County Council
	Association of British Ports
National Governance Organisations	Countryside Council for Wales
(Government and NGOs)	Natural England
	World Wildlife Foundation
	Defra (Department for Environment, Food
	and Rural Affairs)
	Cefas (Centre for Environment, Fisheries and
	Aquaculture Science)
	The Crown Estate
	Scottish Environmental Protection Agency
	Marine Conservation Society (MCS)
	Environment Agency
	JNCC (Joint Nature Conservation
	Committee)
	The National Trust
	The Environment Agency – Cymru
	The Wildlife Trust
	Earthwatch
	Marine Directorate Scottish Government
	National Assembly Wales (NAW)

Table 3.13:	Table of the	organisations	interviewees	were associ	ated with.
-------------	--------------	---------------	--------------	-------------	------------

Categories	Examples of Organisations
International Governance Organisations	UNEP (United Nations Environment
	Programme)
	OSPAR (Administrators of the Oslo and
	Paris Commissions)
Academics and Other Groups	University of Hull
	University of Glamorgan
	Other groups interviewed included:
	ABPMer
	Dialogue Matters
	-

3.6 ANALYSIS OF INTERVIEWS

This section begins by outlining the advantages and disadvantages of conducting analysis on the data through both computer-based qualitative data analysis software (QDAS), as well as manual analysis. Having explained the choice made, it then goes onto to detail the process of analysis and the effect of the interviewer on both the collection of the data and its analysis and interpretation.

3.6.1. Method of Analysis

Initially there was uncertainty as to whether electronic or manual analysis would be applied to the data collected. Analysis of the data collected through the interviews in the first phase of data collection occurred concurrently with data collection in accordance with qualitative data analysis guidelines (Basit, 2003). It is important that this is not a separate phase of a research project and that observations can be made and applied to the remainder of the data collection if necessary. Analysis may not always be in a formal manner but the researcher should be considering codes and categories that can be used to describe the phenomenon observed (Basit, 2003). These rules are applicable to both manual and computer-based analysis and were therefore adhered to throughout analysis of the practitioner interview phase data.

Technological advances have altered the traditional format of qualitative research in both data collection and analysis (Gibbs, 2002). Initially, Qualitative Data Analysis Software (QDAS) was considered an additional complication in an already complex research field (Kaczynski, 2004). Demand for user friendly programs has ensured development of more simplistic QDAS to mirror the almost guaranteed use of data analysis tools for quantitative data (Kaczynski, 2004). Researcher interaction with the data is of utmost importance when it comes to efficient analysis of qualitative data (Walsh, 2002; Basit, 2003). Given this, concerns have been expressed about the potential for use of computer aided data analysis to distance the researcher from the work, invoking a more quantitative, rather than qualitative, analysis of the data (Webb, 1999; Walsh, 2002; Gibbs, 2002). Although use of QDAS is undoubtedly a quick and simple way of reliably analysing large data sets (Walsh, 2002), the risk of losing information through use of software programs was considered to be too high at this point in the research. It was also decided that given the applicability of manual analysis to the smaller data sets characteristic of doctoral research (Webb, 1999), the time taken to comprehensively understand a QDAS and be able to use it competently would have proved redundant.

Although there are a number of benefits associated with the use of QDAS, there is the possibility that use of software in this way would result in a debilitating rigidity with regard to analysis of the data (Kaczynski, 2004). There was also the concern that the researcher would not be fully comfortable with the software being used, running the risk of losing some of the detail of the data through the analysis procedures. The primary benefit of computer analysis is of a clerical nature, with software providing a more efficient and less time consuming way of managing data and coding theories (Morrison and Moir, 1997). Previous studies have indicated that those with limited experience in qualitative data analysis would benefit most from completing manual data analysis as they would gain much needed insight into their data which are essential for any method of data analysis (Webb, 1999; Basit, 2003). Given this, it was decided that analysis of the data from Phase One would be carried out without software for the following reasons:

• To ensure in-depth familiarity and connection with the data,

- To prevent the data from becoming distanced and to maintain control and understanding of the data,
- As a result of the recognition that software can create distance between the data and the researcher.

3.6.2 Content Analysis and Coding

Although software programs have been shown to be an efficient method of analysing hundreds of pages of transcriptions, it was decided that working through the data in a comparative manner was most appropriate (Denscombe, 2003). Unlike statistical surveys, where the data is precoded, qualitative, openended based interviews are coded post interview completion (Fink, 2003b). Information collected in qualitative data research projects can take numerous forms, including transcripts of interviews or focus groups, field notes following personal observations, as well as transcripts of written, spoken or filmed material observed by the researcher (Fink, 2003b). For the purposes of this research, as outlined in Section 3.5.1 telephone interviews were selected as the most efficient method of data collection. As a result, the data produced were qualitative interview transcripts that required coding and interpretation following completion of the telephone schedule.

As recommended by Gillham (2000), transcription of the interviews was carried out as soon possible following the completion of each interview. Previous work has show verbatim transcription of interviewees to allow identification of repetitious and redundant information within the data collected (Gillham, 2000). Given the qualitative nature of the data, content analysis through coding was used to examine and interpret the data collected throughout the telephone interviews. Content analysis allows the identification of substantive statements and concepts within the data (Gillham, 2000).

The data analysis was based on a process of data reduction and interpretation as discussed in Creswell (1994) and Gillham (2000) so that patterns within the data could be identified. In the case of this research, the data from the first phase was

analysed through identification of substantive statements, creation of textual segments and coding them to allow comprehensive content analysis to be conducted on the data. The initial simple codes identified then underwent recontextualising to develop meaning from the broader categories as described in numerous studies (Darke *et al.*, 1998; Jones, 2007; Creswell, 2009). For the purposes of this study, the data analysis involved interpretive coding being assigned to the data in order to identify themes and categories (an example of a processed interview is provided in Appendix 5). Analysis of the interpretive coding was used to identify a number of relationships relating to the application of citizenship to marine management, with two broad themes encompassing a number of more specific components selected for further research.

3.6.3. Effect of the Interviewer

It has been suggested that participant responses vary according to their perception of the interviewer with gender, ethnicity and age commonly altering how a participant will answer questions (Denscombe, 2003). It is important to realise that most of these personal attributes cannot be changed although efforts can be made to put participants at ease by being punctual, polite and neutral (Denscombe, 2003; Bell, 2005). As the interviews in Phase 1 were carried out over the telephone most of these issues were avoided.

3.7 LIMITATIONS OF PHASE ONE METHODOLOGY

As with any qualitative research project, it is important to acknowledge temporal, spatial and situational influences when interviewing (Marshall, 1996). The limitations specific to telephone interviews were considered when determining the most appropriate method for conducting the initial data collection phase (See Table 3.6), including a low response rate associated with telephone interviews and a possibility that interviewees will provide the answers that they feel are required to end the telephone call. Both these limitations are associated with the fact that telephone interviews, particularly those involving home numbers, can be perceived as being intrusive and an invasion of the interviewees' privacy. In the

case of this research, this was mitigated for by contacting interviewees prior to the interview to arrange a suitable time in order to minimise the impact on the interviewee's personal and working time.

A second limitation of telephone interviews that has occurred with increasingly widespread use of mobile telephones as noted by Denscombe (2003) is that individuals may not always be available to talk despite answering the call, and that the network signal may experience interruption during the interview. In the case of this study, the contact numbers used throughout the telephone interview schedule were preferably work based telephone numbers or when necessary home telephone numbers provided by the interviewe.

A further potential limitation of the telephone interview methodology used in this research is the absence of audio recording equipment throughout the interviews. As mentioned, it has been shown that recording is not necessary for successful interviews to be conducted. In order to ensure that the interviews were recorded correctly and accurately, typed transcripts were returned to the interviewees who were asked to verify that the transcript was an accurate record of their interview.

3.8 SUMMARY

This chapter has outlined the general methodological approach of mixed methods with consideration given to the limitations and scope of the overall methodology as well as of that specific to the first phase of data collection. As well as the overarching research approach, this chapter detailed the selection of the methods for the initial phase of data collection (See Table 3.6). The final part of the chapter discussed the processes of data analysis and interpretation following the initial data collection phase.

REFERENCES

Basit, T., N. (2003). Manual or electronic? The role of coding in qualitative data analysis. Educational Research 45(2): 143-154.

Bell, J. (2005). Doing your research project: A guide for first time researchers in education, health and social science. 4th Edition, Open University Press.

Brannen, J. (2005) Mixed Methods Research: A Discussion Paper. ESRC National Centre for Research Methods. NCRM/005

Bourque, B., L., and Fielder, E., P. (2003). How to conduct telephone surveys, 2nd Edition, SAGE Publications Ltd.

Bonnel, P., and Le Nir, M. (1998). The quality of survey data: Telephone versus face-to-face interviews. Transportation 25: 147-167.

Creswell, J. W. (1994). Research Design; Qualitative and Quantitative Approaches, Sage Publications, Inc.

Creswell, J., W. (1998). Qualitative Inquiry and Research Design, Sage Publications, Thousand Oaks, USA.

Creswell, J. W. (2009). Research design: Qualitative, quantitative and mixed methods approaches. 3rd Edition, SAGE Publications, Inc.

Darke, P., Shanks, G., and Broadbent, M. (1998). Successfully completing case study research: combining rigour, relevance and pragmatism. Information systems Journal 8: 273-289.

De Leeuw, E., D. (1992). Data quality in mail, telephone and face-to-face surveys, Netherlands Organization for Scientific Research: 177.

Denscombe, M. (2003). The Good Research Guide for small Scale research projects. Open University Press.

Fink, A. (2003). How to Manage, Analyze and interpret survey data. 2nd Edition, SAGE Publications Ltd.

Fink, A. (2003) How to sample in a survey. 2nd Edition, SAGE Publications.

Fenig, S., Levav, I., Kohn, R., and Yelin, N. (1993). Telephone vs. Face-to-Face interviewing in a community psychiatric survey. American Journal of Public Health 83: 896-898.

Gibbs, G., R., Friese, S., and Mangabeira, W., C. (2002). The use of new technology in qualitative research: Introduction to Issue 3(2) of FQS. Forum: Qualitative Social Research 3(2).

Gillham, B. (2000). Case study Research Methods, Continuum

Graneheim, U., H., and Lundman, B. (2004). Qualitative content analysis in nursing research: concepts, procedures and measures to achieve trustworthiness. Nurse Education Today 24: 105-112.

Guest, G., Bunce, A., and Johnson, L. (2006). How many Interviews are enough? An experiment with data saturation and variability. Field Methods 18(1): 59-82.

Hart, C. (2001). Doing a Literature Search: A comprehensive guide for the Social Sciences, SAGE Publications Ltd.

Hawthorne, M., and Alabaster, T. (1999). "Citizen 2000: Development of a model of environmental citizenship." Global Environmental Change 9: 25-43.

Holbrook, A., L., Green, M., C., and Krosnick, J., A. (2003). Telephone versus Face-to-face interviewing of national probability samples with long questionnaires: Comparisons of respondent satisficing and social desirability response bias. Public Opinion Quarterly 67: 79-125.

Johnson, R., B., and Onwuegbuzie, A., J. (2004). Mixed methods research: A research paradigm whose time has come. Educational Researcher 33(7): 14-26.

Jones, M., L. (2007). Using Software to Analyse Qualitative Data. Malaysian Journal of Qualitative Research 1(1): 64-76.

Kaczynski, D. (2004). Examining the Impact of Qualitative Data Analysis Software upon the Analysis Process. Australian Association for Research in Education. Melbourne, Australia.

Kaplon, B., and Duchon, D. (1988). Qualitative and Quantitative Methods in Information Systems Research: A Case Study. Management Information Systems Quarterly 12(4): 571-586.

Kell, D., B., and Oliver, S., G. (2003). "Here is the evidence, now what is the hypothesis? The complementary roles of inductive and hypothesis-driven science in the post-genomic era." BioEssays 26: 99-105.

Kitchin, R., and Tate, N., J. (2000). Conducting Research into Human Geography: Theory, Methodology and Practice, Prentice Education Limited.

Lancaster, G., A., Dodd, S., and Williamson, P., R. (2004). Design and analysis of pilot studies: recommendations for good practice. Journal of Evaluation in Clinical Practice 10(2): 307-312.

Marshall, M., N. (1996). Sampling for Qualitative Research. Family Practice 13(6): 522-525.

Mason, J. (2002). Qualitative Researching. 2nd Edition, SAGE Publications

Morison, M., and Moir, J. (1998). "The role of computer software in the analysis of qualitative data: efficient clerk, research assistant or Trojan horse?" Journal of Advanced Nursing 28(1): 106-116.

Morse, J., M., Barrett, M., Mayan, M., Olson, K., and Spiers, J. (2002) "Verification strategies for establishing reliability and validity in qualitative research." International Journal of Knowledge, Culture and Change Management 1(2).

Pope, C., and Mays, N. (1995). "Reaching the parts other methods cannot reach: an introduction to qualitative methods in health and health services research." British Medical Journal 311: 42-45.

Sturges, J., E., and Hanrahan, K., J. (2004). "Comparing Telephone And Face to face Qualitative Interviewing: a Research note." Qualitative Research 4: 107-118.

Thomas, D., R. (2003). A general inductive approach for qualitative data analysis, School of Population Health, University of Auckland.

Travers, M. (2001). Qualitative research through case studies, SAGE Publications Ltd.

Walsh, M. (2003). "Teaching Qualitative Analysis Using QSR NVivo." The Qualitative Report 8(2): 251-256. Retrieved 19/02/10 from <u>http://www.nova.edu/ssss/QR/qr8-2/walsh.pdf</u>

Webb, C. (1999). "Analysing qualitative data: computerized and other approaches." Journal of Advanced Nursing 29(2): 323-330.

Van Teijlingen, E., and Hundley, V. (2002). "The importance of pilot studies." Nursing Standard 16(40): 33-36.

CHAPTER FOUR

<u>RESULTS AND DISCUSSION OF MARINE PRACTITIONER</u> <u>TELEPHONE INTERVIEWS</u>

4.1 INTRODUCTION

In this chapter the analysis and key observations of the practitioner telephone interviews are described. The chapter begins with general observations made following analysis of the data and examines how this data was used to guide the successive phase of data collection and analysis. All quotes⁸ are unattributed to the participants in order to maintain anonymity and assure confidentiality in the interview. The views expressed by the interview participants do not necessarily reflect the official view of the organisation they are associated with.

4.2 RESULTS OF PRACTITIONER TELEPHONE INTERVIEWS

This section will outline the results of the analysis of the telephone interviews. Although the headings are derived from the interview questions, the results have been grouped together for discussion when relevant. Table 4.1 presents a grid analysis of the data collected through the marine practitioner telephone interviews. This table illustrates how frequently interviewees mentioned the key elements identified through content analysis conducted on the data. In addition, the grid analysis was used to support the generation of the key themes investigated in the remainder of the research (discussed in Section 4.3). The main results of the marine practitioner phase will be briefly discussed in Section 4.3 with further interpretation in Chapter Seven. Sections 4.2.1 - 4.2.13 presents the key elements highlighted through the telephone interviews.

⁸ Throughout this chapter, quotes taken from the practitioner telephone interviews are included in italics in support of the observations made through comparative analysis of the data. The quotes also act as an illustration of the typical comments made during interviews.

Table 4.1: Content Grid Analysis for Marine Practitioners Telephone Schedule

Note: The interviewee codes are based on the date on which the interview was conducted and is prefixed with MP, an abbreviation for Marine Practitioner. The categorical coding was carried out using the methodology of content analysis and coding as outlined in

hapter 3.	The 🗸 symbo	ols indicate t	he number of	times an it	nterviewee	mentioned	the theme i	n their inte	rview.			
gory ng	Disconnection with the marine environment	Need for more Sustainable Management and Governance	Personal & social responsibility	Socio- economics	Devolution	Social Background	Public participation & engagement	Ownership	Livelihood/ Dependency	Proximity and Location	Environmental justice	Personal connection and Belonging
erviewee e												
01	>	//					>			>		>
0 2	>	>					//					
3	>	>	~ ~	>			>	//				>
4	>	>					>		//			
5	>	>			<i>、、</i>		>		<		>	
9 (>	\ \ \	~ ~	>					111	>		>
L .	>	>	>		>		>	~ /		//		
8	>	>	>		>		///			>		>
6	>	~ / /	<i>``</i>	>			>					
• 10		>					>		<	//		
11			>	>			//		<	>		>
12	>		>				>					
13		\ \ \	>				>		<	>		
14		>	>				>	///	//	///		
15			>				>	>	<			//
16			>				>	>	~	>		>
17	< <	~ / /		>		>	>					
18	~		<i>∕</i>		//		//					
19	>	>	~				>		~	~		>

127

Category coding	Disconnection with the marine environment	Need for more Sustainable Management and Governance	Personal & social responsibility	Socio- economics	Devolution	Social Background	Public participation & engagement	Ownership	Livelihood/ Dependency	Proximity and Location	Environmental justice	Personal connection and Belonging
MP 20	^	>	///				~ / /					
MP 21		>						~	>	>		~
MP 22			///	>			>			>		
MP 23										>		<
MP 24			///	>		>	>	< < < <		~ /		>
MP 25	//		//				>			//		~
MP 26	///			>			1111	<				<
MP 27	//		>							>		<
MP 28	>		>			>	>	~		>		<
MP 29	///		>	>			>					
MP 30	~	>	>			>	>					~
MP 31	~	~					~ /			11		1
MP 32			~					1				
MP 33	^ /		>	~			< <		~	<i>▶</i>		~
MP 34	~		////				>		~	/		11
MP 35	>	`	\ \ \	>			`	<	^	//		
MP 36		>	>			>	>		~	111		
										///		
MP 37		~	///							11		
MP 38	~						>			/		1
MP 39	~	< <				>	< <		~	/		
MP 40	//		//			~	~ /	111				1
MP 41	~	~	ノノノ			>	`			1		
MP 42	~		>						~			>

Rights and responsibilities		>		>													</</th <th></th> <th></th> <th>></th> <th></th> <th>`</th> <th></th> <th></th> <th>></th>			>		`			>
Public perception									//					//	>	>		>							
Value of marine resources														>		11									
Role of the individual and communities							>	>			<u>^</u>	>	<u>^</u>	>						>	~ ~	>	ノノノ	11	アア
Understanding of links between marine environment and society			>			~	//			~	///					//		//		////	~	~			~
Role of Awareness		//	>	//			>	>	~ /	//	~ ~	//	~ ~	//		//		111	~ ~	~ ~		>	~	/	~
Education and Information availability		1111		1111		~ ~	>	///	>		~			>	//	1111		//	~ /	~ /	>		~ ~	111	111
Role of the media		>		111	<i>ر</i>				>			>			>	>	>		///		>				~
Age/ Life stage		>						>				>		>		>					>		~	>	/ //
Capacity to engage		>										//						>		>					
Behaviour							>						^		>	>			>	>					~
Culture, Heritage and Ethnicity				>				>	>	>								>		>					
Category coding	Interviewee Code	MP 1	MP 2	MP 3		MP 4	MP 5	MP 6	MP 7	MP 8	MP 9	MP 10	MP11	MP12	MP13	MP14	MP15	MP16	MP 17	MP 18	MP 19	MP 20	MP 21	MP 22	MP 23

Rights and responsibilities				< <		>		>	>		>								
Public perception									>									>	
Value of marine resources				>							>								
Role of the individual and communities				>		>		//		/ 					>	>			1
Understanding of links between marine environment and society		>			· ·	>	~ ~	, 	· ·				////		>	>	>	//	<
Role of Awareness	ノノ	>	>	`			>	>			>			`	< </td <td></td> <td>></td> <td>~ /</td> <td></td>		>	~ /	
Education and Information availability		///		>	>	>	< <	>		///	1111	>	>	>				~ ~ ~	//
Role of the media	~ ^	>	>		>		//	>	>	///	//		//		>	>	>	>>	<
Age/ Life stage							~ /	>	>	>			>				//	\ \ \	~
Capacity to engage												>			~				
Behaviour	ノノ	/	//	/	/	<			/	/		11			/			~ /	
Culture, Heritage and Ethnicity	>			>		>			>					>		>		>	
Category coding	MP 24	MP 25	MP 26	MP 27	MP 28	MP 29	MP 30	MP 31	MP 32	MP 33	MP 34	MP 35	MP 36	MP 37	MP 38	MP 39	MP 40	MP 41	MP 42

4.2.1. Practitioner perception of current marine management

The overall perception of current marine management by interviewees was that it is 'fragmented', 'ineffective', 'short-sighted' and 'disconnected' with one interviewee feeling that it is "probably not promoting...a sustainable future for the marine environment" with another emphasising the point further by saying that current management is having a "disastrous effect on the long term sustainability". One interviewee linked sustainable, long-term management with an integrated system suggesting that "current management is not capable of It will require a healthy dictatorship by some sustainable management. groups...guiding activity". Current management strategy was thought by the majority of interviewees to "lack a holistic overview" and generally interviewees felt that successful marine management required a more "long term approach" and should aim to "develop win-win [situations] rather than compromises [between user groups]". The short-term nature of the current management of the marine environment was mentioned by another interviewee who suggested it could cause other social problems as "[management] is all short term orientated rather than long term. There is a legacy...when it comes to the marine environment, along with...cultural issues that we don't understand". There was also a suggestion from interviewees that promotion of other approaches to management would be of benefit to the marine environment, for example "the promotion of the ecosystem approach to management should bring benefits to the environment and [to] those who use it".

Practitioner consensus was that although current marine management was deemed ineffective, there was a sense that improvements are being made with "sustainability becoming [an important] driver for policy and management" although 'things are improving but not fast enough'. Improvements were not considered sufficient with one interviewee saying marine management is "changing but even then it's not sustainable until all sectors are engaged". Interviewees did highlight the promotion of increased efforts to achieve sustainability in international frameworks citing Agenda 21, the "Jakarta Mandate, Marine Strategy Framework and the efforts of OSPAR" as examples. Evidence provided by interviewees indicated that improvements are being made

on a more local scale in the UK, although it was acknowledged that the implementation of these improvements is subject to a number of constraints, such as lack of resources and time restraints. The importance of local involvement and the role of the individual in marine management were linked to citizenship with one interviewee stating that '*citizenship is likely to play an important role in resource management and development as if people are more conscious of their role they will be more empowered to make decisions*'.

4.2.2. Role of the individual in management of the marine environment

Interviewees were of the opinion that ideally the public would have a very active role in the development of marine management plans and decision-making with only six interviewees failing to mention public participation in their interview (Table 4.1). Currently, however, it was suggested by interviewees that 'there is virtually no role for the individual' in marine management as there is 'no adequate mechanism [for public involvement]'. General consensus suggested that early consultation and engagement would be the key to encouraging active public involvement in marine management, and ensuring that 'public requirements are considered and met'. A moral and ethical obligation on the part of authoritative bodies to encourage public involvement as "part of the democratic process" was also recognised by interviewees. This was further supported by an interviewee who stated that the 'UK government is democratic and therefore everybody has a responsibility [to the marine environment]'. This opinion was complemented by the suggestion that successful marine management would require effort from statutory management bodies to utilise the "obligation [for individuals to participate], if they want to have an input into the future".

The role of the individual in the management processes appears to be seen as valuable but currently very limited. The majority of interviewees were of the opinion that although there is a public role, it needs improvement and is currently restricted with the need for a "mechanism to enable the public to have an active role" and "allow them to contribute comfortably". Although it was generally accepted that the public should be involved in marine management at some level,

their involvement would require guidance from a more statutory level as "everyone should take care of the environment but it is for people higher up to deliver policies, to feed them down [to the public] and enforce them". One interviewee differentiated between stakeholders and members of the general public and implied that, as a result of these differences, a variety of mechanisms would be required in order to facilitate effective public participation, stating that "the general public have a different role and different levels of inclusion...need to make opportunities for the public to have involvement but think it would be at a superficial level...Stakeholder involvement would mean as much involvement as possible...whereas citizens do not need to be". Contrastingly, a minority of interviewees did not consider the public to have a useful role in management of the marine and coastal environment, with one stating that they were "not entirely convinced bv the argument that the general public are stakeholders...[the] contribution they make is often arbitrary". There appeared to be clear examples of local attempts at encouraging high levels of public involvement, thereby facilitating a public role in marine management with one local council interviewee stating that "Bournemouth beach forum currently go for Blue Flag awards. One of the requirements for this it to engage with stakeholders...this has involved setting up forums to get stakeholder engagement across a section of people".

Finally, it was recognised that the role an individual plays can be dependent on a number of factors, for example, the "*nature of the individual*", "*locality*", both globally and "*within the UK*", "*individual's knowledge*" and the "*opportunities to be involved*". One interviewee suggested that they imagined '*most people wouldn't know where to begin*' with regards to marine management. In spite of the general approval of enhanced public involvement, there was some disagreement as to the level of involvement that would be required, or indeed would be possible for the public, with one interviewee agreeing that everyone has a role in the management process while simultaneously questioning the extent of the role of the general public stating that "everyone has their part to play, but how much?".

4.2.3. Understanding of citizenship

The primary theme underpinning interviewee understanding of citizenship was of "belonging to a political group, such as a country", and about "[where] you come from and where you live". In addition to this, a significant relationship between citizenship and "having a sense of responsibility within the community" and "where people should think about their role in society rather than their own selfish requirements" was apparent. One interviewee stated that citizenship is an awareness of "personal responsibility and responsibility of one's actions and *impacts on other people*", while another suggested that it is a "sense of belonging" to a particular nation or state to which an individual has various rights and responsibilities". The importance of this sense of responsibility was supported by a further link identified by one interviewee who stated that they "believe that citizenship entails responsibility" and the 'interactions between the individual and society', suggesting that citizenship is about a 'contract between the citizen and the state'. Interviewees suggested that an enhanced sense of public "ownership" of and connection to the marine environment would lead to more successful implementation of sustainable management. Interviewees were of the opinion that if individuals exhibited a sense of attachment to the marine environment it would encourage them to work for 'the common good'. Another factor thought to have the potential to influence citizenship was that of social and cultural differences within communities resulting in a need to be "culturally aware in a human context".

4.2.4. Perception of citizenship for the environment

Environmental citizenship was not a concept that the majority of interviewees were familiar with, with one suggesting it was "an emerging concept". The focus of environmental citizenship was not thought to be purely environmental with interviewees linking the concept to 'societal equality and economic well being'. A number of the interviewees were unsure as to the relevance of citizenship with regard to the environment, with one interviewee directly questioning the level of influence attempts to develop environmental citizenship would actually have. Interviewees were aware that the public have other

concerns often more directly related to their every day life style choices and may not be receptive to efforts to generate societal environmental citizenship, let alone marine citizenship. This concern was addressed by another interviewee who suggested that given that "there are many complexities to decision making and environmental costs are often not considered [a] priority. For example, [people] understand the environmental costs of flying but...still choose to fly However strong environmental citizenship is, there are still other drivers to influence decisions". Interviewees recognised that there are potential limitations to the development of marine, or indeed, environmental citizenship with one interviewee stating that they "personally find it difficult to apply in an environmental context". In spite of some initial uncertainty regarding environmental citizenship, one interviewee defined an environmental citizen as someone who 'lives as sustainably as possible and does what they can in their However, interviewees felt that 'citizenship of the own personal lives'. environment has to be encouraged by a sense of awareness and then caring – everyone lives in the environment but not everyone is an environmental citizen' suggesting that there are a number of challenges facing the promotion of environmental citizenship.

A number of factors were identified through the marine practitioner interviews as having a potential to influence the concept of environmental citizenship. These factors included 'awareness', 'concern', 'individual behaviour', 'responsibility for both current and future generations' and 'knowledge'. As with general citizenship interviewees commonly linked individual and public sense of responsibility with the concept of environmental citizenship. One interviewee linked citizenship and responsibility directly by suggesting that environmental citizenship is connected to "taking our environmental responsibilities seriously", and another stating it was related to having "a personal responsibility for the environment". Environmental citizenship and associated responsibility were further linked to the sustainability of the environment; as one interviewee commented that environmental citizenship "extends to being responsible for resources... [in] both current and future generations". Another interviewee suggested that the onus for enhanced responsibility is on both authoritative bodies and individuals alike, and stated that "citizenship means that people

responsible for the environment have a responsibility to those who live in it and as a citizen, there is a responsibility to protect the environment". Others suggested contributing factors included environmental awareness, "being aware of rights and needs across a wide range of issues and acting with responsibility", and environmental concern, given that the "quality of the environment is something that concerns everyone". Active involvement in environmental decision-making and management was also identified as a key component of environmental citizenship with one interviewee believing in "active citizenship and participative democracy...through collaborative processes".

4.2.5. Citizenship and the marine environment

The general consensus reached by interviewees was that there is scope for a marine specific concept of citizenship, suggesting that citizenship has a potentially important role to play in the management of the marine environment. The majority of interviewees highlighted a need for a more integrated system for efficient management, conservation and development of all environments; for example, one interviewee stated that we "should promote all concepts of citizenship in environmental issues". Others felt that although they agreed with an overarching concept of environmental citizenship "marine issues may need extra attention" and that there was a need to "link land and sea in terms of management". However, it was the opinion of other interviewees that "the coast is often ignored in comparison to the terrestrial environment" and that given that "the marine environment is a fluid and ever changing environment and [that] it is imperative that everyone works together to protect it", there was a very real need for a separate concept. This was further supported by acknowledgement of the 'different problems and lack of enforcement ability' in the marine environment in comparison to its terrestrial counterparts. One interviewee championed the benefits of marine citizenship stating that it would "increase people's understanding of environments and the connections between uses [of the marine environment]" implying it would have some influence on individual behavioural choices.

In addition to supporting the need for a marine specific concept of citizenship, potential challenges were highlighted by a number of interviewees. Issues with regard to ownership and property rights related to the marine environment were identified as a potential problem as "the marine environment cannot be owned [and] so connections and feelings of ownership can be very different [therefore]...the principles of terrestrial environmental citizenship will be difficult to translate [to the marine environment]". A common concern illustrated by Table 4.1 was that of public disconnection from the marine environment resulting in a lack of awareness of marine issues and societal impacts on the marine citizenship was the challenges posed by a perceived sense of public disconnection from the marine environment with "a sense of coastal citizenship...easier to address, while subtidal issues would be rather more difficult [to engage the public with]".

4.2.6. Citizenship and management of the marine environment

In general, interviewees strongly associated the concept of citizenship to sustainable management of the marine environment and its resources, with most interviewees directly linking citizenship and marine management saying that "they go hand in hand" and that sustainable management "needs to be based on citizen involvement". Interviewees also suggested that citizenship is required to ensure management of the marine environment is sustainable, successful and effective. A statement made by one interviewee directly associated citizenship with the marine environment stating that, "[marine] citizenship is a wider understanding, appreciation and acceptance that the sea has an environmental, social and economic function". This holistic approach was further supported by another interviewee who suggested that, "citizenship...needs to be holistic and recognise legitimate uses of the marine environment". These statements and others made by interviewees inferred that "a feeling of ownership and responsibility for the maritime environment is fundamentally important in ensuring we can achieve sustainable management". Although interviewees felt that, ideally, sustainable management should be linked to citizenship, many were of the opinion that currently "it doesn't link well" and that "improving

engagement of stakeholders in management would make management easier and would increase the desire to be engaged".

The concept of marine citizenship was brought into question by one interviewee who believed that there might be "difficulties when it comes to large open [bodies of] water". Another mentioned the possibility of conflicting priorities of development and conservation for sustainable management suggesting that "in the near future there will be a drive for wave and tidal energy which will conflict with conservation; which has the higher environmental good?" This potential conflict was also referred to by another interviewee, who said that "citizenship refers to the individual role and sustainable management needs a consensus from government level so individual perceptions do not affect management" suggesting that the public role in management of the marine environment is, and should remain, minimal.

4.2.6.1 Legislative implications of marine citizenship

There was a wide range of opinions on this subject, with some interviewees being of the opinion that legislation would be an important component of promoting marine citizenship. One interviewee suggested that marine citizenship and its potential influences on individual behaviour "should be more statutory than voluntary" as "anything that translates policy and legislation into something that is more real for people and makes it clear to them what they should do, then it is a good thing". One interviewee felt that considering citizenship when developing marine specific legislation would 'place a duty on all of us to try and achieve sustainable management of the marine environment'. Contrastingly others were of the opinion that there was no mechanism through which marine citizenship could be legislated for and that including citizenship in legislation seemed against the very nature of citizenship stating, "coercion doesn't go hand in hand with citizenship" and another negating the need for legislation as people can't be forced "to feel stewardship, it's a way of life". Most frequently interviewees felt that considerations of citizenship should be made within the legislative process but it depended on a number of factors. One interviewee suggested that it depended on "[the] definition of citizenship and the

level of legislation being considered". In spite of this, it must be noted that another interviewee stated that they didn't "*know how it would be encapsulated*". Another interviewee felt that a potential issue would be "[*the*] *very real danger of social engineering through legislation which would put people off, it could potentially have a very negative impact*".

Acknowledgement of the Aarhus Convention implied some interviewees were of the opinion that citizenship was already being legislated for. For example, one interviewee suggested that "there should be [best] practice legislation; which is already present in a way in the Aarhus convention" although they were of the opinion that it wasn't clear how it was being implemented. It was also suggested that citizenship was considered during the development of the Marine and Coastal Access Act (2009). One interviewee clearly supported this sentiment saying that "the Marine Bill⁹ promises to involve coastal communities in future decision making in the marine environment. This should instil a sense of stewardship, so in a sense it does exist in policy or as a goal of policy". It was also suggested by one interviewee that "governments have a responsibility to develop policy that citizens want" while another felt that "policy and legislation should outline the citizen's role". Results suggested that there was a discrepancy in practitioner perception as to how policy and legislation could include citizenship and what the appropriate mechanisms for encouraging marine citizenship would be.

4.2.7. Potential Factors Influencing Marine Citizenship

A variety of demographic factors, including personal connections to the coast, socio-economics, age and occupation, were identified as impacting the development of a sense of citizenship amongst the general public towards the marine environment. Interviewees identified an individual's sense of personal connection to the marine environment as a key factor in generating a sense of marine citizenship. Importantly, a variety of forms of personal connection were

⁹ At the time of the telephone interview schedule, the Marine and Coastal Access Act was being promoted as the upcoming 'Marine Bill'. References to the 'Marine Bill' in this study therefore correspond to the newly ratified Marine and Coastal Access Act, December 2009.

identified ranging from "historical connections", "personal attachments such as memories and family connections", and "holidays and recreation". Location and proximity to the coast were identified by interviewees as primary components of marine citizenship, as it was "expected that coastal communities are more connected to the marine and coastal environments and would have a higher sense of marine citizenship".

Dependency on the marine environment, in terms of livelihood and income, was also commonly identified as a potential influencing factor with one interviewee directly linking dependency on the marine environment with marine citizenship suggesting for example, that "fishermen...should theoretically be the main promoters of marine citizenship since their economic stability is based on the condition of the sea. They reap the benefits of the marine environment and therefore ought to be prime candidates". Personal wealth was also considered to be a potential factor in generation of marine citizenship as 'people involved in marine leisure activities [associated with personal wealth] are generally very aware with a genuine and inherent interest...about the marine environment'. One interviewee thought that marine citizenship could be partially income related but it would 'depend on the demographic of people living in the area'.

Interviewees highlighted culture as having an impact on an individual's level of marine citizenship. This was considered on different levels, with one interviewee suggesting that there are clear "social differences in perception" towards the marine environment. This was supported by others who questioned the perception of minority groups within the UK, suggesting it was "culturally not as engrained in their culture as ours". This statement in itself suggested that there is a cultural affiliation with the marine environment in the UK due to the "historical implications of the UK being an island nation". Similarly another interviewee considered that the same would be true of other island nations, stating that "island nations who have a long term, historical connection with the marine environment, such as the UK, would be expected to have a greater understanding than those from a land locked nation." A conflicting sentiment was expressed by another interviewee who suggested that, "British people take the marine environment for granted". Cultural issues within the UK were also

questioned with one suggesting that marine citizenship is a "middle class" concept and another stating that "people being well educated with a high level of environmental awareness often with a middle class background would be those with a higher sense of citizenship towards the marine environment".

Other factors identified by interviewees as having the potential to influence an individual's sense of marine citizenship included "aspects of school curricula could also play a factor [as] a high level of environmental education will encourage a greater sense of citizenship". It was, however, acknowledged that this approach would have significant limitations in that formal structured education is currently very much directed towards younger generations, and although the majority of interviewees did not connect age with a sense of citizenship, it is possible that it would have an impact. There were however conflicting views on this subject; for example one interviewee stated that "age probably is a factor as children are being taught more about the marine environment and climate change at school", while contrastingly another suggested that "as people get older, they understand more and potentially care *more*". Age was also implicated as being a factor when considering individuals access to marine and coastal environments with one interviewee commenting that many coastal communities comprise of retired communities who may have problems accessing these environments.

4.2.8. Responsibility for Management of the Marine Environment

The general consensus was that the responsibility for marine management and its associated resources currently falls to the government, albeit fragmented and at a number of different levels; "Government makes the decisions, policy and drivers...down to the regulators, industries and everybody who has an interest in marine environment that it is managed sustainably". It was indicated that although the "ultimate responsibility lies with the government", the "current fragmented system means that there is a lack of transparency". A number of interviewees indicated what the ideal situation would be with regard to management responsibilities and suggesting a "tiered system of development and implementation".

The numerous levels of responsibility were mentioned by several interviewees, for example, stating that there are 'a whole range of people responsible' as "there are many levels of governance; global, regional, national; there has to be [public] engagement at each of these levels". Interviewees acknowledged that responsibilities fall at different levels with "local levels having more engagement with the community...[and] national level is not really directly involved with management, more with setting out a strategic direction that should reflect stakeholder needs, putting legislation and frameworks in place to deliver policy". One interviewee suggested that 'partnerships' would be an efficient strategy of managing the marine environment, implying the need for a more collaborative approach to marine management. In support of this, one interviewee stressed the opinion that we "all have a responsibility to work together and deliver management effectively, educating other people and getting input from other users'.

Although marine management was considered to be currently predominantly government driven, interviewees felt there was a "need to develop a finer grained bottom up approach and develop legislation". This perception was contested by another interviewee who argued that this would not be the most successful method and that we "need a top-down framework from government to implement policy, with NGOs and interest organisations acting as advisors and lobbyists". Clearly there is a lack of consensus even among marine practitioners in the UK with regard to who is responsible for management of the marine environment: this observation is highlighted by one interviewee who suggested that "perhaps the question should be who isn't responsible [for the marine environment]?" Interviewees also suggested that the significance of individuals in marine management has increased, implying links to citizenship for the marine environment and that 'people need to manage their own individual activities and understand the correct way to behave [towards the marine environment]'.

4.2.9. Role of stakeholders and communities in marine management

There was a general consensus amongst interviewees that "*ideally there should* be an active role" for stakeholders and the wider community in marine

management, with one interviewee suggesting it is "core to the decision making" process". It was also noted that it is important to reach a balance between government-directed management and the general public with "everyone having a right to participate should they wish to" and that "without them it would be an arbitrary management system". Although public involvement was accepted by most interviewees as being an important component of marine management it was suggested that the role of the public necessitates guidance as "there is a requirement for strong governance to lead [public involvement] to ensure a balance is reached". Only one interviewee was of the opinion that there was 'no role' for the public in the management of the marine environment suggesting a general consensus that enhanced public involvement in marine management would be advantageous. The majority of interviewees suggested that there should be an active public role, particularly in coastal areas, where local authorities are often viewed as being driven by terrestrial issues 'often missing that...constituents of coastal local authorities are coastal workers and the impacts of the marine environment are relevant in these areas'.

Overall, general consensus was that the public should have input but that an enhanced level of public involvement could potentially make managing the marine environment more complex. A number of difficulties with regard to the level of involvement possible for the general public were also remarked upon. These included a risk of "*stakeholder fatigue*" and "*consultation fatigue*", in addition to restrictions on time and resources meaning it "*may be difficult to facilitate local [marine interest] groups continuously*". However, in spite of challenges, the general consensus was that "*if you ignore stakeholder engagement, you ignore it at your peril*". Interviewees recognised the difficulties of making involvement statutory suggesting that forced engagement would create a negative perception of marine management. The potential for increased complications in management as a result of public involvement was also recognised as a possible challenge. However, success stories regarding community driven management, such as that of Lamlash Bay¹⁰ in Scotland, which was considered a 'good example of a community led initiative' implied

¹⁰ Lamlash Bay is a Community Marine Conservation Zone encompassing the first Scottish No-Take Zone designated in September 2008 (C.O.A.S.T. Online).

that overall interviewees were in favour of community involvement and were of the opinion that '*citizens need to have a sense of ownership in order to increase the desire to engage*' in management of the marine environment.

4.2.9.1. Implications of Increased Public Involvement

Both advantages and disadvantages of increasing public involvement in management of marine resources were identified throughout the telephone interviews. Firstly it was observed that many of the interviewees expected increased involvement to result in an elevated sense of ownership towards the marine environment within society. Sense of ownership was linked to an increase in knowledge and awareness, with one interviewee stating that "generally speaking the greater the number of people involved, the greater the knowledge, [better] evaluation of decisions made, [greater] sense of ownership and the greater chance of sound marine management plans being formulated, delivered and implemented". In order for public involvement to be effective, interviewees identified that the public need to be educated to ensure they have sufficient knowledge and capacity to engage meaningfully in marine management processes.

The majority of interviewees predicted that increased awareness and understanding within general society would encourage a greater demand for increased political support for marine and coastal issues, suggesting that marine issues would have a "higher profile politically meaning it would be taken up by more of the electorate". According to another interviewee, this "higher profile" and "higher publicity" could theoretically be linked to an "increased concern and awareness". Practitioners hypothesised that increased ownership and awareness would "Lead to more sustainable decision making [process]...that can actually be delivered and achieved", encouraging a better result with regard to the development of management and policy. It was indicated by interviewees that enhanced public engagement would "highlight other issues that had not been considered", encourage a reduction in conflict between interest groups and "objections to plans" and would encourage a 'greater appreciation of challenges facing the marine environment'. A further potential advantage identified was
that of an increased knowledge base resulting in the use of both scientific and local knowledge. Interviewees also identified a number of disadvantages associated with an increase in public involvement suggesting it could make the management process "difficult", "harder to manage and take everyone's interests on board" whilst also having the potential to make the process more time consuming and expensive. There was also acknowledgement that public opinion can be affected by what is perceived as being fashionable and that *"increased public involvement will mean decisions made are affected by changes"* in current trends...[there] needs to be a fair method that is resilient to changes in public moods". However in spite of the disadvantages identified by interviewees, it was the general consensus that "the advantages [of increased public involvement] far outweigh the disadvantages if the process is facilitated well". Further to this, it was suggested that there is a need to "increase public capacity to be involved" in order to make public engagement in marine management and decision making effective in the best possible manner.

4.2.10. Relationship between the marine environment and society

Awareness of marine issues was perceived by the interviewees, which they attributed to an overriding sense of disconnection between the public and the marine environment. The idea that "most people don't see the impacts [of their behaviour] on their daily lives" was a common theme, with numerous interviewees suggesting that "links were not seen" as the marine environment is "difficult to access" and "an abstract concept" for the general public. Although there was a recognition by the interviewees that there are "impacts of marine environmental issues on daily lives; [for example] health issues, socio-economics and climate control". They also thought that these impacts are not recognised by the public who "generally do not think about the relationship between their lifestyles and the environment". One interviewee gave further evidence to this point stating that 'people do not connect waste water with what ends up on the beach... [or] where goods have come from or the impacts of shipping...on the *marine environment*'. Interviewees compared society's relationship with the marine environment and its terrestrial counterpart suggesting that as 'terrestrial environments are very visual and issues facing the marine environment are not very visible, it has led to a sense of disconnection [from the marine environment]'. This lack of connection was linked to a low level of awareness with one interviewee saying that they don't "think that a lot of people are aware of the impacts they have" and another suggesting that "beyond a basic awareness, there is very little understanding of the issue". One interviewee suggested that the recent lack of connection between society and the marine environment could be changing in response to increased media publication of these relationships. Levels of awareness and connection were, however, considered to be variable and dependent on other factors, for example, proximity to the coast, as suggested by one interviewee "there is limited awareness, people who live by the coast may have awareness but not those who live further away". Another interviewee highlighted the possibility that "personal connection or interest in the coast" and those who are directly dependent on marine resources may have an influence on how aware an individual is.

Although public awareness was thought to be low, the suggestion was made by a small number of interviewees that the general public "know about big issues like climate change and sea level rise" and that there is "awareness of what is perceived as being important". It was also suggested by some that there is a "varied level of concern", supported clearly by one interviewee who stated that "it varies: some people are very aware, certain sectors and individuals have strong ideas about issues and understand, while there are others that have no connection, perhaps due to lack of personal engagement and experience". There was a sense that the public have other concerns that need to be taken into consideration when examining their sense of citizenship and awareness towards the marine environment. The underlying complexity of decision making was identified as interviewees as a potential challenge to marine citizenship, which was clearly supported by one interviewee who explained the practitioner perception that there is a "hierarchy of public concerns; Health and safety, economy and environment, in that order". The hierarchical perception of public concern suggests that environmental issues in general are in need of promotion in order to elevate them to a position of public priority.

4.2.11. Public awareness and concern for the marine environment

The general level of awareness of marine issues among the general public was considered to be very low by interviewees. It was suggested that this could be related to a number of factors, with one interviewee stating that environmental awareness could be "dependent on where you live, profession, background, education", and another suggesting it related to "locality, schooling and numerous other factors". It was also implied that awareness can be related to personal circumstances and is "dependent on what each individual has to worry about in their personal lives". The results hypothesised that where people do have awareness it is of highly publicised issues such as an "awareness of key issues like sea level rise and climate change, but not aware of sedimentary processes and marine pollution issues such as increasing ocean acidification, coral reef depletion and depleted fish stocks for example". Interviewees identified a lack of public connection to the marine environment as a limiting factor with regard to public awareness. One marine practitioner interviewed suspected that "people are not aware unless they are actively involved or affected by it".

The interviews indicated that practitioner perception of public concern towards the marine environment ranged from very concerned to a generally low level of concern, the latter being the theme of the majority of the responses. This low level of concern was attributed to the fact that 'the public aren't engaged enough with or informed enough about the marine environment and its condition, and their ability to appreciate it is very limited'. It was suggested that "the sea is too remote from most people's experiences" and that the public are "as a whole, less concerned than they should be". Those who suggested a degree of public concern thought that there is a "relationship between awareness and concern" and that concern can be "issue driven...by charismatic species like cetaceans and other marine mammals". Interviewees were of the opinion "that people aren't aware of their own effects but are concerned about the marine environment". Practitioners' interviewed suggested that "there is a general anxiety about [the marine environment] but there is a difference between general concern that the environment is in trouble, and individual choices and ability to make difficult *decisions*". Consensus was that "*if information level was increased then the level of concern would grow substantially*" supporting the need to improve the availability of information to the public in order to mobilise an enhanced level of public concern.

Importantly, as with levels of awareness, some interviewees linked public concern for the marine environment with personal attributes and circumstances, with one suggesting that "*it depends on how much you worry about it and the type of person that you are*" while another supported this statement with "*they're not concerned… people are concerned with their own lives and personal circumstances*". Differentiation between the terrestrial and marine environment was identified in this question with one interviewee stating that "*there is not the same level of concern for marine habitat loss as there is when compared to loss of rainforests; there is no equivalent for the marine environment as a whole*". Although general concern was considered to be low, interviewees were of the opinion that if there were an increase in information about marine issues, concern for the marine environment would be enhanced.

4.2.12. Methods of increasing public awareness of marine issues

The majority of interviewees indicated that in their opinion a significant amount of work is still required with regard to raising public awareness of marine issues, particularly on a local level, with one interviewee stating that there is a "need for increased awareness of species in local waters" and another affirming the need to "empower local communities" with regard to their role in the management of the marine environment. Interviewees were of the opinion that although it may be "relatively easy to set up opportunities ...it will really only interest people who already have a concern" stressing the need for new initiatives regarding marine education.

Raising public awareness of marine issues was identified as being an "*uphill struggle*" with interviewees identifying a number of potential methods to increase public awareness. Media coverage of marine events was most commonly identified as the most effective method of increasing general public

awareness. The use of different forms of media was identified by one interviewee as being "the best way to target a wide variety of people" while another labelled it as "the number one method, particularly television", and a number of interviewees championing the use of TV programmes such as Coast and Blue Planet. One interviewee commented on the importance of basing programmes of this type in the UK to ensure people have knowledge of their own environment with the aim of enhancing local interest and awareness. Interviewees emphasised the importance of using "good science [in order] to make programmes exciting and factual". Although various forms of media were championed as a mechanism for increasing awareness amongst the public, interviewees highlighted the need for responsible journalism and the "need for accurate portrayal [of marine issues] without sensationalisation".

Overall, interviewees identified education as a key mechanism of increasing public awareness of marine issues with one interviewee stressing that "the importance of the country's dependency on the marine environment needs to be central to education". Inclusion of marine issues in formal teaching would target younger generations, which was suggested to be one of the more important avenues for increasing awareness among the general public. It was recognised that "[marine] education needs to start from a young age" as "kids teach their parents and make them act" and that "schools are [the] ideal way [to increase awareness] as kids are the next generation". Interviewees felt that if marine issues were to be included in formal education, it would be important that marine education 'be taught in tandem and to the same depth of the terrestrial environment'. Potential limitations of this method were recognised however by one interviewee who stated that "it would be good to have an element in the national curriculum as it is important to teach younger generations, although there will be a time lapse before it filters into older society".

Marine practitioners interviewed felt that there is a sense of disconnection towards the marine environment among the public which be challenging given that "the difficulty with the marine environment is that people do not encounter it in their normal everyday lives". An increase in information availability was recommended by a number of interviewees in a variety of different ways "public events", "focusing education on local level to engage communities", "public conferences", as well as "general awareness raising - summer events, aquariums; having events to raise awareness with speakers and educational displays". The analysis also suggested that linking people's everyday lives is an important factor, with the promotion of green consumerism suggested as a method to do this, in order to allow the public to make "links between marine produce and their everyday lives, for example, through the Marine Conservation Society's Good Fish Guide Campaign" and other "supermarket promotions". The current economic downturn was identified as a further challenge to raising public awareness with interviews of the opinion that it may have resulted in environmental issues 'slipping down the agenda'.

Although interviewees acknowledged that responsibility for marine management falls at various levels including "coastal partnerships", "non governmental organisations", "government departments", "local authorities", "commercial enterprises" and "educational bodies", interviewees also suggested that 'there is a responsibility for those who manage the marine environment to ensure there is an...awareness of impacts of society on the marine environment'. Overall, a number of different bodies and organisations were identified by interviewees as being responsible for providing the resources for increasing public awareness. Interviewee perception varied with regard to the level of responsibility held by the government towards the marine environment with some interviewees feeling that the responsibility fell solely on the government, while others noted that "it is easy to say government should fund it, but it is not that straight forward". The issue of devolution in the UK was mentioned by some of the interviewees, with one interviewee suggesting that resource provision should be allocated "state by state...nation specific with UK level funding". Other interviewees felt that financial support had to come from the government although it was not purely their responsibility with a "government based overarching campaign that local authorities and other organisations can link into" is required. Although it was accepted that "everyone has their part to play...where extra resources are needed, this should ideally come from [all marine organisations] with [guidance] from the government" indicating the need for a collaborative relationship between the government, other organisations and individuals. There was a suggestion that industry could also play a role with one interviewee stating that it may be "*interesting if marine industries took more of an active interest in this are, as it may take away some of the bias present in the information*". Responsibility for public awareness was considered fragmented by one interviewee who thought this further reiterated the need for the Marine Management Organisation (MMO) proposed in the Marine and Coastal Access Act (2009). Finally, interviewees implied an inequality between terrestrial and marine environments with regard to public education and awareness, emphasising "the a need to ensure that the marine environment is treated equally to the terrestrial environment when considering funding...the marine environment is often considered inaccessible and therefore less deserving of funding".

4.2.13. Promotion of responsible behaviour towards the marine environment

The general opinion of the interviewees was that there is "always scope to do more" but that there is the very real "danger that we will turn people off by going on about the environment". As with many of the other questions, a comparison was made between terrestrial and marine environments by a few interviewees suggesting that although "people are beginning to appreciate it more...the terrestrial environment has traditionally had more attention than the marine environment". The general consensus was that there is currently not enough promotion of marine issues to generate a sense of marine citizenship among the public, although there has "been an increase in the volume of material but it is still low". It was suggested that the level of promotion varied with location and it was suggested that promotion of marine issues was "not at a national level but certainly among some coastal communities".

Interviewees commonly linked an enhanced sense of responsibility to a number of other potential influencing factors, for example, increases in education and participation in marine issues. It was also recognised that links between the marine environment and lifestyle must be made more apparent to the general public. The importance of relevancy was promoted by one interviewee who felt that there was a need to make marine issues *"real for day to day life and [raise*

awareness of the] small actions that people can take", with another suggesting that "links to educate and empower people" would heighten a sense of responsibility. A number of solutions were expressed by different interviewees; including methods that act to "[bridge] the gap between science and the public", "increased realisation of the productivity of the sea" and encouraging "consumer choice" and "getting people to value the resource" although there was acknowledgement that it is "difficult to change people's behaviour". A personal sense of attachment was also strongly linked to development of a sense of responsibility as "the marine environment needs people to feel attachment and familiarity before they will take action to protect it". The power of having a personal influence was commented on by one interviewee who felt that citizens "need to know that modifying their behaviour will have positive effects [on the *marine environment*]". Potential for personal gain as a result of environmentally responsible behaviour towards the marine environment was directly linked to an enhanced sense of responsibility with one interviewee suggesting that "if there was money in it, [the public] would act more responsibly". In direct contrast to the benefits associated with an enhanced sense of public responsibility, one interviewee questioned what effect raising a sense of responsibility would have, saying that "even if you increase a sense of responsibility, are you changing behaviour or are you just increasing people's sense of guilt?...It may make people feel quite defeated."

It was acknowledged that although interviewees did not consider there to be an optimum level of promotion of marine issues, it has *"improved over recent years"* and that in the UK particularly there was a *"hope that the Marine Bill would improve it"*. One interviewee was optimistic, saying that *"there has been a gradual increase in appreciation for the marine environment"* supporting the observation that promotion has increased public awareness and appreciation for the marine environment. Recognition of the importance of accurate information in promotional materials was highlighted by one interviewee who believed we *"have to work out where people get their information from as there can be conflicting messages. For example, if people can see a fish and chip shop in their village, they are less likely to consider problems of over-fishing or see the issue as being real". The influence of media portrayal on perception of marine*

issues should also be considered with a minority of interviewees feeling that *"media attention has been negative"* potentially having a detrimental effect on levels of public awareness and concern towards the marine environment.

4.2.14 Summary of key points observed in marine practitioner interviews

The analysis of the marine practitioner telephone interviews (Sections 4.2.1-4.2.13) highlighted a number of key points summarised below and discussed in more detail in Section 4.3:

- Interviewees acknowledged that traditional marine management has failed and that a move towards a more participatory form of management would be widely beneficial,
- Results of the interviews suggest the need for more collaborative marine management through enhanced public involvement in the process,
- Interviewees identified the role of education in enabling an inculcation of a societal sense of marine citizenship,
- The importance of personal connection with the marine environment was identified in relation to a number of components of marine citizenship,
- Interviewees highlighted a potential relationship between location and an individual's expression of marine citizenship,
- The influence of dependency on an individual's awareness, concern and overall behaviour towards the marine environment.

4.3 DISCUSSION OF PHASE ONE RESULTS

This section provides a short discussion the main observations obtained from the first phase of the data collection. The implications of these results for the second phase of data collection are outlined in Section 4.4 - 4.5. Further discussion and synthesis with the second phase of data collection is presented in Chapter Seven.

4.3.1. Practitioner perception of education and marine citizenship

Observations from the literature review indicated levels of education, knowledge and awareness to be strongly associated with a sense of environmental citizenship (Sears and Hughes, 1996; Hawthorne and Alabaster, 1999; Berkowitz et al., 2005). As outlined in Section 4.2.1 - 4.2.13, this sentiment was echoed in the analysis of the telephone interviews, with many interviewees implying that an increase in education and information availability about the marine environment would encourage higher levels of awareness, concern, responsibility and ultimately aid the generation of a societal sense of marine citizenship. Availability of accurate information coupled with an efficient education process was strongly linked by interviewees to the promotion of awareness and concern towards the marine environment by the interviewees. The relationship between education, awareness, concern and behaviour has been well researched (Haklay, 2002; Ananda, 2007) with general consensus being that individuals with higher levels of knowledge exhibit a higher sense of responsibility and concern towards the environment. With a specific focus on the marine environment, research has highlighted the importance of public education (Ducrotoy et al., 2000; Correia, 2002; Steel et al., 2005) in enhancing public understanding of the marine environment and its management (Ducrotoy, 2001). Links between education and delivery of effective marine management have been highlighted in previous research (Kuijper, 2003; Osborn and Datta, 2006; Williams, 2008; Castle et al., 2010).

Previous studies support the proposal by UK marine practitioners that by enhancing education, long-term solutions to challenges facing the marine environment may become apparent, with the potential to alter community attitude and behaviour towards the marine environment (Uneputty et al., 1998; Hartig et al., 2001; Tytler et al., 2001). Steel et al. (2005) suggested that having a more in depth understanding of public knowledge of the marine environment could allow the development of more effective education methods. The importance of providing communities with accurate information was highlighted by interviewees, who raised concerns that irresponsible education could engender a "blame culture" as communities become overwhelmed by the magnitude of some of the challenges facing the marine environment and may feel unable to participate meaningfully. Interviewees championed the provision of accurate information and efficient education programmes as a mechanism of combating the perception that the marine environment is 'out of sight, out of mind' and 'someone else's problem', working to promote the potential role of each individual and community can take. The results are supported by previous work conducted by Tytler et al. (2001) which proposed that addressing the perceived current deficit in public knowledge and awareness of the marine environment would lead to a better community understanding of research initiatives, legislation and management procedures. Interviewees were of the opinion that increasing the availability of various education programmes would encourage an individual's sense of personal responsibility for their behaviour towards the marine environment. Earlier research by Cottrell and Graefe (1997) supports this proposal stating that the stronger an individuals sense of responsibility as a result of heightened awareness and knowledge, the stronger their commitment to behaving in a particular manner. Given the observations made following the telephone interviews, and the available supporting research, it was inferred that in order to encourage a wider level of public involvement in marine management, efforts need to be made to increase public understanding of the marine environment and societal dependency on the numerous resources its provides.

Analysis of the telephone interviews suggested that increasing the level of marine education in schools would be the most effective method of promoting a societal move towards marine citizenship. Younger generations were considered the key audience for targeted marine education suggesting that age may be an indirect factor in the efficacy of education, an opinion observed in earlier research conducted by CCW (Williams, 2008). The implications of sociodemographics for the future of marine citizenship are further investigated through the second phase of data collection.

A number of potential mechanisms through which marine awareness could be promoted were suggested by interviewees, ranging from traditional classroom based education, awareness days at marine attractions and through increased publicity through the media. Previous research supports these suggestions with the importance of the media in marine-education discussed by Steel *et al.* (2005). However, the potential for access to education to be limited by socio-economic circumstances and background has been suggested (Steel *et al.*, 2005). Given this, it can be implied that an individual's socio-economic status could influence facilitation and expression of marine citizenship.

4.3.2 Influence of location on marine citizenship

Throughout the practitioner interviews frequent references were made to the potential influence of location and proximity to the coast on an individual's sense of marine citizenship. Further to this, interviewees suggested that perception of the marine environments' condition could vary greatly across the UK, potentially influencing promotion of marine citizenship on a UK wide basis, with particular differences expected between inland and coastal communities. Inhabitants of coastal communities were expected to have a '*wider affinity [with the marine environment]*' than those in inland areas suggesting a perception that coastal communities should exhibit a comparatively superior level of personal connection to the marine environment.

Other factors, such as marine awareness and concern were associated with residency in close proximity to the marine environment, and the potential impact on marine citizenship was determined to need further investigation. It was expected that individuals living in close proximity would be likely to have a higher sense of citizenship towards the marine environment, although there has been research that contests this hypothesis. Steel *et al.* (2005) suggested that individuals living in close proximity to the coast would express a high level of

knowledge, and thus have an enhanced sense of concern and responsibility for the marine environment than inland communities. This hypothesis was used by CCW (Williams, 2008), whose work indicated that individuals in inland regions of Anglesey, North Wales had a stronger connection than individuals in coastal areas due to a higher sense of appreciation. Given these discrepancies within preceding research regarding the relationship between proximity to the marine environment and the components of marine citizenship (Figure 2.3), it was proposed that this relationship should be an area of further inquiry in the second phase of the research.

Interviewees were of the opinion that management of the marine environment was frequently given less consideration than its terrestrial counterpart, suggesting an inequality between the two. Rose *et al.* (2008) showed that there have been fewer studies investigating the public perception of the marine environment in comparison to the terrestrial environment, supporting the interviewee's perception that there has been a traditional lack of public and governmental interest in the marine environment. Sense of place and personal connection has been identified as precursors to individual behavioural choices with regards to the environment (Cantrill, 1998; Stedman, 2002). Given this, and the perceived lack of connection between the British public and the marine environment, interviewees suggested that an individual's connection to the marine environment would influence their expression of marine citizenship.

4.3.3 Livelihood and Marine Citizenship

It was suggested by interviewees that an enhanced sense of marine citizenship would be expected from individuals who are more dependent on the marine environment i.e. coastal communities whose livelihoods are intrinsically linked to the condition of the marine and coastal environment. Evidence to support this was provided by one interviewee who identified fishermen as the "*prime candidates for marine citizenship*" given their dependence on marine resources. However, although it was expressed that the fishing community should have an almost integral sense of marine citizenship, as a group they were generally considered to be irresponsible by interviewees with regard to the sustainable use

of the marine environment. The model of environmental citizenship proposed by Hawthorne and Alabaster (1999) did not consider livelihood to have a significant impact on the expression of environmental citizenship. In contrast, this relationship was perceived by interviewees as being vital to the development of marine citizenship with interviewees positively linking livelihood to the successful development of the concept of marine citizenship. This is further supported by previous research which indicated that dependence on an environment can be crucial to an individual's sense of connection (Cox *et al.*, 2008; Jorgensen and Stedman, 2001).

4.3.4 Participation, perception and culture relating to the Marine Citizenship

Overall, interviewees felt that the lack of public connection with the marine environment has been a challenge to meaningful public engagement. Although interviewees acknowledged that participation in isolation does not constitute citizenship, it was stressed that it is a crucial component as observed in previous research (Barnett *et al.*, 2005). Given the perceived importance of active participation, interviewees suggested that a balance between statutory and community governance, could aid deliverance of successful long-term marine management. This sentiment is directly connected to recommendations proposed in Agenda 21 which promotes citizen involvement in environmental management (Agyeman and Evans, 2004).

Interviewees also highlighted the possibility that culture could have an influence on a community sense of marine citizenship. It was suggested by one interviewee, for example, that many British communities have become disillusioned as the marine environment is "*out of sight, out of mind*" resulting in a diminished sense of personal and collective responsibility. In addition, interviewees implied that social standing may impact the inculcation of marine citizenship within the UK, suggesting that the marine environment is a resource predominantly used by *'white, middle class communities'*, with few ethnic minority communities utilizing it to the same degree. Given this, concerns were expressed by interviewees that, in some cases, promotion of marine citizenship could viewed by the public as being exclusive, and restricting participation from communities that are "socially deprived". Interviewees identified personal perception of the marine environment as having the potential to significantly impact individual behavioural choice, a relationship that had been previously been examined and identified as a precursor for individual engagement with the marine environment in a study conducted on the Menai Strait, North Wales (Williams, 2008). Between this and the observations during the practitioner interviews, it was inferred that the level of connection felt by individuals towards the marine environment could have a potentially important role in the development of marine citizenship. Interviewees expressed a concern that the perceived lack of public connection to the marine environment could be detrimental to the development of marine citizenship. In order to determine how best to promote marine citizenship, it may be necessary to determine what the public perceive as marine citizenship, and what forms of promotion they would best respond to.

The issue of long-term sustainability of the marine environment and how the public perceives its resources could also be inferred following analysis of the interview responses. As stated by Suarez de Vivero (2007) there is currently no governance regime that has ascertained how to facilitate all of the perceived uses of the marine environment. Both Suarez de Vivero's research (2007) and the observations of this study suggest a system that would maintain maritime identity and culture for coastal communities while reducing efforts at traditional activities, such as fishing, and encouraging ongoing but sustainable development of coastal regions to ensure continued social and economic stability is required. Interviewees proposed that this collaborative approach cannot be achieved without a greater sense of public responsibility and desire for sustainable solutions to marine issues, which should, according to this research, accompany a higher level of marine citizenship.

4.3.5 Responsibility for Management of the Marine Environment

The general consensus from the telephone interviews was that increased knowledge from effective marine education induces an increased awareness of marine issues, coupled with an enhanced sense of responsibility. However, a consensus was not reached with regards to who practitioners felt responsibility for the marine environment fell to. Interviewees suggested that a cooperative approach between government and society would benefit sustainable management of the marine environment. Potential limitations to this were highlighted in the analysis of interview responses. Recognition of the lack of property rights for most of the marine environment, coupled with perceived access difficulties highlighted limitations for a collaborative approach between the government and society to the management of the marine environment. It is also a question of whether the public feel they are sufficiently capable of contributing meaningfully to management and decision-making. In order to aid an advance in a sense of societal responsibility, it is imperative to understand whom society currently deems responsible for marine management. In order to evaluate this, it was decided that elements of responsibility for marine management required investigation in the case study phase of data collection.

When considering the relationship between responsible participation and marine citizenship, interviewees expressed concern as to how the inclusion of public opinion and requirements could be facilitated under current marine management. Interviewees were concerned that it would be challenging to encourage an enhanced level of public involvement, and thereby inculcate marine citizenship, without a transparent mechanism for including public opinion in the development of management and policy.

In addition to public involvement in policy development and management, practitioners addressed the potential for public involvement in policy and management implementation. Marine practitioners were of the opinion that there would be a clear value in enhancing public involvement in marine management with the public acting as a mechanism for the development of successful decision-making (McKinley and Fletcher, 2010). It was evident that practitioners were optimistic that through improved public education and awareness programmes, a societal willingness to change behaviour for the benefit of the marine environment would be encouraged.

4.4 REFINED MODEL OF MARINE CITIZENSHIP

The results observed through the practitioner survey have been used to refine the initial conceptual model of marine citizenship generated following the literature review. Figure 4.1 shows the refined model, which combines the findings of the literature review and the marine practitioner survey. Content analysis of the marine practitioner interviews highlighted a number of themes and factors potentially influencing the development of a marine specific concept of citizenship.

Figure 4.1 also illustrates the marine practitioner perception of marine citizenship and how it could be used to benefit the long-term management of the marine environment. Elements of marine citizenship identified by marine practitioners as having the most potential to have a positive impact on management of the marine environment are illustrated on the left column of the refined model (See Figure 4.1.). The refined model highlights the interconnected nature of a number of the factors influencing marine citizenship, with each impact on management affected by a number of components. This refined model illustrates the perception of marine practitioners that the generation of marine citizenship would not be a linear relationship; consideration would be required for the connections and cumulative impacts of each of these individual factors. The proposed final outcome of promoting marine citizenship was that of more efficient, more sustainable long-term management of the marine environment. The refined model required further investigation at a community level to examine how it could be applied to marine management. It was proposed that the model could be further examined through case study work discussed in further detail in Chapter Five.



Figure 4.1: Refined model of marine citizenship (Adapted from McKinley and Fletcher, 2010)

4.5. IMPLICATIONS FOR PHASE TWO OF RESEARCH

Given the inductive approach used for this research, the observations and refined model generated following the telephone interviews were used to guide the second phase of data collection through case studies. The implications of the observations presented and discussed in Sections 4.1- 4.4 of this chapter are outlined below.

4.5.1. Identification of key themes for Phase Two

The telephone interview schedule allowed the identification of a number of potentially important relationships between the factors of marine citizenship. From these specific relationships, two broad key themes were identified as being the factors, which would have the most significant impact on the development of marine citizenship. Table 4.1 presented a grid analysis of the results following manual content analysis on the data collected through the telephone interviews. The grid analysis was used to guide the development of the surveys used throughout the case study phase of research (presented in Chapter Five).

Following the comprehensive content analysis of the telephone interviews displayed in Table 4.1, the broad themes of education and personal attachment were selected as the key areas for investigation in the subsequent phase of the research project. These broad themes encompass a number of specific components identified through both the literature review and the telephone interview schedule. By selecting broad themes, a number of more specific factors within the key themes could be evaluated through the case study phase of data collection (as displayed in Table 4.2). The broad themes were directly related to the main components of marine citizenship identified by the interviewees (Figure 4.1). The selection of these themes and their influence on the methodological approach of the second phase of data collection will be further explored and discussed in Chapter Five, Section 5.4.3.

Broad Theme	Component of refined	Factors included in the	
	model of marine	broad themes investigated	
	citizenship		
Education	- High awareness of	- Access to Information	
	marine issues	- Awareness	
	- Concern for the	- Knowledge	
	marine environment	- Literacy	
	- Sense of personal and	- Related Concern	
	social responsibility	- Capacity to engage	
	- Marine education	- Perception	
Personal Attachment	- Social and personal	- Participation	
	responsibility	- Recreational Involvement	
	- Participation in	- Job Dependency and	
	marine management	Livelihood	
	- High awareness of	- Cultural links and	
	marine issues	connections	
	- Concern for the	- Residence in close	
	marine environment	proximity to the coast	
	- Pro-environment	- Holiday destination	
	behaviour and	- Family connections	
	perception of the		
	marine environment		

Table 4.2: Components of the two primary themes proposed for further investigation in the second phase of data collection.

4.5.2 Case study selection

Table 4.1 indicates the recognition of UK devolution as a potential challenge to marine citizenship, given that differences in the management of marine resources across the UK may be apparent¹¹. Interviewees were of the opinion that there was a possibility that the devolution of governments could vary in their valuation of the marine environment which in turn, could be potentially be mirrored in the level of marine citizenship exhibited by the citizens of the four home countries (England, Scotland, Wales and Northern Ireland). It was proposed that the

¹¹ Recently, evidence of these differences has begun emerging with the development of separate marine legislation, namely through the Marine (Scotland) Act of April 2010 and the Marine and Coastal Access Act for England and Wales, given Royal Assent in December 2009.

variation in statutory governance and its relative impact on community sense of personal and social responsibility towards the marine environment could be investigated by selecting case studies across a wide UK geographical distribution. In spite of reference to devolution in the interview responses, there is little literature available on the potential impacts for marine management. This in itself identified the issue of devolution and its potential political influence on the promotion of marine citizenship within the UK as an area that requires further investigation. The geographic spread of the case studies ensured that the research could be considered representative of the UK as whole. In addition to this, interviewees had expressed clear differences between inland and coastal communities and their respective sense of awareness, concern and responsibility for the marine environment. Therefore, selection of the case study sites included both inland and coastal regions in order to allow the relationship between proximity to the coast and an individual's sense of marine citizenship to be fully investigated. The process of selecting case study locations and details of the case study phase are discussed in further detail in Chapter Five.

4.6 SUMMARY

This primary phase of data collection assessed practitioner understanding and perception of the potential role of marine citizenship on the management of the marine environment. The general areas for investigation identified in Chapter Three (See Table 3.8) were: the establishment of the level of general understanding of environmental citizenship and how it could be applied specifically to the marine environment; factors that could be considered to influence an individual's sense of marine citizenship and how these would impact on the efficacy of the concept; potential mechanisms for promotion of enhanced awareness and concern regarding the marine environment and finally, the perception of public awareness and knowledge of marine issues and the relationship between the marine environment and society. This summary provides an overview and synthesis of the key themes identified through the practitioner survey, establishing areas requiring further research.

Section 4.2.1 investigated marine practitioner perception of current management of the marine environment in the UK. Overall, practitioners considered marine management in the UK to be in need of improvement particularly with regard to the inclusion of the public in management and decision making. The perceived role of the individual in management of the marine environment was described in Section 4.2.2. General consensus among marine practitioners was that ideally there should be a higher level of public involvement in managing the marine and coastal environment.

Sections 4.2.3 - 4.2.7 provided an assessment of the level of practitioner understanding of the term 'citizenship' in an environmental context, with specific reference to the marine environment and how citizenship could be applied to marine management. The key relationship identified in this context was that of a strong association between general citizenship and individual and social rights and responsibilities. These sections also showed practitioner recognition of the complex nature of individual decision-making processes. There was an acceptance that environmental citizenship in general has a critical role to play in management of the environment. In spite of this, there was acknowledgement by practitioners that the diverse nature of the marine environment requires more specific attention. Potential challenges to marine citizenship were also highlighted in these questions with respondents alluding to issues caused by a lack of societal lack of connection and education regarding the marine Although potential difficulties were discussed, the overall environment. consensus was that marine citizenship, if directed at increasing public understanding and involvement in the decision-making and management processes, would prove beneficial to managers.

Section 4.2.7 outlined practitioner opinion regarding factors with the potential to influence promotion of a societal sense of marine citizenship in the UK. As expected a wide variety of factors were suggested by practitioners, supporting the observations made by Hawthorne and Alabaster (1999) investigating the component parts of their environmental citizenship model (displayed in Chapter 2, Figure 2.1). Personal connections of various natures were identified as having a particularly strong level of influence on marine citizenship, and how an

individual reacts to and treats the marine environment. Notably, practitioners felt that the traditional island history of the UK should theoretically result in a more marine aware citizenry.

Following this, Section 4.2.8 evaluated practitioner perceptions regarding the level of responsibility towards the marine environment and the role of stakeholders, including the general public, in management of the marine environment. Practitioners highlighted their concerns regarding the fragmentation of management of the marine environment at the time of the interview schedule. Respondents were of the opinion that responsibility needed to extend further than traditional statutory management organisations, incorporating more involvement from stakeholders and the general public. The possible implications of increasing the level of responsibility and involvement held by communities was discussed in Section 4.3.10 with the overall opinion that increased involvement would serve to benefit the long-term management of the marine environment.

Sections 4.2.9 to 4.2.13 evaluated practitioner perceptions of the relationship between the public and the marine environment, including elements of public concern and awareness of issues facing the marine environment. Questions of sufficient public promotion and availability of information regarding issues facing the marine environment were also considered in these sections. The key observation of this section was that practitioners felt that there is a debilitating lack of knowledge and awareness, coupled with a resulting lack of concern and sense of responsibility towards the marine environment. A number of suggestions regarding how this problem could be dealt with so as to improve overall public capacity for involvement were made. The impact of personal connection to the marine environment was once again emphasised, reiterating the importance of various elements of personal attachment to the marine environment to effective promotion of marine citizenship. Overall practitioners were of the opinion that publication of issues facing the marine environment requires improvement, although it was acknowledged that it been progressing positively over recent years. The final finding throughout these sections was the

commonly expressed suggestion that levels of knowledge, awareness and concern would differ between coastal and inland areas.

Section 4.4 presented a brief discussion of the main relationships identified through analysis of the practitioner telephone interview schedule. Of particular importance are the two key themes, displayed in Table 4.2, selected to undergo further investigation in the case study phase of the research. An improved and refined version of the initial conceptual model generated following completion of the systematic literature review was also produced and discussed. This refined model and the observations made regarding practitioner opinions of marine citizenship and its potential role in management of the marine environment were then used to guide the methodological approach and determination of the research questions for the second phase of research. The implications of this phase of research on the remainder of the research project are explored further in Chapters Five and Six.

Finally, Section 4.5 described the implications of the results obtained through the first phase of data collection on the case study phase, presented in further detail in Chapter Five. Key themes identified through comparative content analysis of the data collected through the telephone interviews were also presented.

The results of the telephone interview schedule highlighted the complexities associated with environmental decision-making. Interviewees particularly mentioned socio-economics as taking priority over environmental issues, with some interviewees focusing on the current global economic downturn. It was proposed by one interviewee that there is a '*hierarchy of concern*' within society, with environmental issues placed at the bottom of the rank. Taking this into consideration, it was proposed that in order for a comprehensive investigation into the emergent concept of marine citizenship, it would be necessary to ascertain the individual and collective impact of a variety of potentially influential factors.

Chapter Five outlines the methodological approach selected to further examine and interpret the relationships observed in the telephone interview survey and their potential influence on marine citizenship in the UK. Chapters Six and Seven further discuss the specific observations of the case study phase of research and synthesise the observations made in both data collection phases respectively.

The next phase of data collection aims to examine the key themes and patterns identified through the practitioner surveys, investigating the level of influence each of these factors would have on the concept of marine citizenship. More specifically, the key points requiring further investigation arising from the next phase are:

- The relationship between education and individual sense of marine citizenship;
- The relationship between forms of personal attachment and inculcation of a concept of marine citizenship;
- The identification of the conditions required to enable expression of marine citizenship.

REFERENCES

Agyeman, J., and Evans, B. (2004). "Just Sustainability': the emerging discourse of environmental justice in Britain?" The Geographical Journal 170(2): 155-164.

Ananda, J. (2007). "Implementing Participatory Decision Making in Forest Planning." Environmental Management 39: 534-544.

Barnett, J., Doherty, B., Burningham, K., Carr, A., Johnstone, G., and Rootes, C. (2005). Using Science to create a better place. Environmental Citizenship: A Literature Review, Environment Agency: 36.

Berkowitz, A., R., Ford, M., E., and Brewer, C., A. (2005). A framework for integrating ecological literacy, civics literacy, and environmental citizenship in environmental education.

Cantrill, J., G. (1998). "The Environmental Self and a Sense of Place: Communication foundations for regional ecosystem management." Journal of Applied Communication Research 26: 301-318.

Castle, Z., Fletcher, S., and McKinley, E. (2010). Coastal and marine education in schools: Constraints and opportunities created by the curriculum, schools and teachers in England. Ocean Yearbook. 24: 425-444.

Community of Arran Seabed Trust (online) Available at <u>http://www.arrancoast.com/</u>. Accessed on 08/11/2010.

Correia, A. M. R. (2002). Information literacy for an active and effective citizenship: White paper prepared for UNESCO, the U.S. National Commission on Libraries and Information Science, and The National Forum on Information Science. Information Literacy Meeting of Experts. Prague, The Czech Republic.

Cottrell, S., P., and Graefe, A., R. (1997). "Testing a conceptual framework of responsible environmental behaviour." Journal of Environmental Education 29(1): 17-28.

Cox, M., Johnstone, R., and Robinson, J. "A conceptual model of impacts of environmental change on human well-being."

Ducrotoy, J., Shastri, S., and Williams, P. (2000). "Coastal sciences and management: the need for networking in higher education." Ocean & Coastal Management 43: 427-444.

Ducrotoy, J. (2001). "Higher Education in Coastal Sciences and Management: Establishing a Partnership Between Education and " Innovations in Education and Teaching International 38(3): 239-250.

Haklay, M. (2002). "Public Environmental Information: understanding requirements and patterns of likely public use." Area 34(1): 17-28.

Hartig, T., Kaiser, F., G., and Bowler, P., A. (2001). "Psychological Restoration in Nature as a Positive Motivation for Ecological Behaviour." Environment and Behaviour 33(4): 590-607.

Hawthorne, M., and Alabaster, T. (1999). "Citizen 2000: Development of a model of environmental citizenship." Global Environmental Change 9: 25-43.

Jorgensen, B., S., and Stedman, R., C. (2001). "Sense of place as an attitude: Lakeshore owner's attitudes toward their properties." Journal of Environmental Psychology 21: 233-248.

Kuijper, M., W., M. (2003). "Marine and coastal environmental awareness building within the context of UNESCO's activities in Asia and the Pacific." Marine Pollution Bulletin 47: 265-272.

McKinley, E., and Fletcher, S. (2010). "Individual responsibility for the oceans? An evaluation of marine citizenship by UK marine practitioners." Ocean & Coastal Management. (in press)

Osborn, D., and Datta, A (2006). "Institutional and policy cocktails for protecting coastal and marine environments from land-based sources of pollution." Ocean and Coastal Management 49: 579-596.

Rose, C., Dade, P., and Scott, J. (2008). Qualitative and quantitative research into public engagement with the undersea landscape in England, Natural England Research Reports, NERR019.

Sears, A., M., and Hughes, A., S. (1996). "Citizenship Education and Current Educational Reform." Canadian Journal of Education 21(2): 123-142.

Stedman, R., C. (2002). "Toward a social psychology of place: predicting behaviour from place-based cognitions, attitude and identity." Environment and Behaviour 34: 561.

Steel, B., S., Smith, C., Opsommer, L., Curiel, S., and Warner-Steel, R. (2005)."Public Ocean Literacy in the United States." Ocean and Coastal Management 48: 97-114.

Suarez de Vivero, J., L. (2007). "The European Vision for oceans and seas -Social and political dimensions of the Green Paper on Maritime Policy for the EU." Marine Policy 31: 409-414.

Tytler, R., Duggan, S., and Gott, R. (2001). "Public participation in an Environmental dispute: implications for science education." Public Understanding of Science 10: 343-364.

Uneputty, P., Evans. S. M., and Suyoso, E. (1998). "The effectiveness of a Community education programme in reducing litter pollution on shores of Ambon Bay (Eastern Indonesia)." Journal of Biological Education 32(2): 143Urama, B., C., Kenyon, W., Burton, R., and Potts, J. (2006). Integrated value mapping for sustainable river basin management: Economics, ethics and social psychology. A. T. P. S. Network, African Technology Policy Studies Networks.

Yuliani, E., L., Adnan, H., and Indriatmoko, Y. (2008). The use of appreciative inquiry as a tool for enhancing adaptive capacity in natural resources management. IASC Conference. Cheltenham, UK. 14-18 July 2008

Williams, L. B. R. (2008). Public attitudes towards the maritime environment around the Menai Strait and Conwy Bay. CCW Policy Research Report. CCW: 55.

CHAPTER FIVE

METHODOLOGY OF PHASE TWO: CASE STUDIES

5.1 INTRODUCTION

This chapter outlines the methodological approach employed in the second phase of data collection of this study. In accordance with the inductive mixed methods approach applied to the project, the methodology of the second phase of data collection was guided by observations made in the initial data collection phase (discussed in Chapters Three and Four). The chapter begins with a description of how the methodological approach to the research has evolved throughout the preceding phase of data collection. The chapter then discusses the use of case study based data collection and its suitability as a method for addressing the aim and objectives of the study. Finally the chapter presents a brief discussion of the key observations made following analysis of the data collected throughout the case study phase of research with further interpretation presented in Chapter Seven.

5. 2 MIXED METHODS APPROACH

By applying an overall mixed methods approach to the research, described in Chapter Three, the strengths of both qualitative and quantitative data collection and analysis could be utilised (Gable, 1994; Johnson and Onwuegbuzie, 2004; Creswell, 2009). As outlined in Chapter Three, a sequential mixed methods approach¹² was applied to the research with the second phase of data collection directly guided by observations made in the practitioner telephone interview schedule (Chapters Three and Four).

¹² See Table 3.3 in Chapter Three for an explanation of commonly used mixed methods strategies.

As discussed by Gable (1994) the use of case studies in social research was traditionally based on a qualitative approach. However the use of qualitative case studies in conjunction with more quantitative data collection methods such questionnaires and structured interviews is accepted as a common as methodology (Gable, 1994). A mixed methods approach was applied to the case study phase of data collection. Quantitative data collection and analysis was employed to investigate the themes of education and personal attachment as identified through the marine practitioner interviews (Chapter Four). This approach allowed the researcher to establish the most significant relationships relating to the application of marine citizenship to marine management in a community setting. Simultaneous collection of qualitative data was used to further investigate these relationships, which allowed detailed and collaborative interpretation of both data types (Kaplan and Duchon, 1988; Gable, 1994; Gillham, 2007). For the purposes of this research, the use of quantitative and qualitative methods was found to be complimentary, with the interpretative qualitative data enhancing the observations through analysis of the quantitative data.

5.3 CASE STUDIES IN RESEARCH

Yin (2003a) defines case study research as "an empirical enquiry that investigates a phenomenon within its real-life context, especially when the boundaries between phenomenon and context are not clearly evident" (p.13). Gerring (2007) further explains the case study method as an intensive study of single or multiple cases in order to obtain wider understanding of similar events. Case study research is commonly used to investigate specific research questions and can comprise of an individual, a group or a community of any number of variants depending on the requirements of the research (Gillham, 2000). This approach is often used by social researchers due to the ability to investigate an event in natural settings (Stake, 1978 in Gomm *et al.*, 2000). Gillham (2007) identified a number of guidelines that should be observed for successful case study research including critically reading relevant literature and early determination of the research questions and aims of the case studies. Given that the use of case studies has been identified as an effective method of developing emerging theories (Darke *et al.*, 1998; Curtis *et al.*, 2000), case study based research was considered an appropriate mechanism through which to investigate the potential role for marine citizenship in the management of the marine environment. The primary aim of the case study phase was to further investigate the key elements identified in the analysis of the practitioner interviews. The mechanisms for doing this are discussed further in Sections 5.6 to 5.9.

5.3.1. Types of Case Study Research

There are three main types of case study research commonly used including exploratory, descriptive and explanatory cases (Table 5.2). Importantly, each of these approaches to case studies can be conducted through the investigation of single or multiple cases depending on the requirements of the research project. Yin (2003b) recommends that the selection of the appropriate case study choice should be dependent on the phenomena being explored through the research. For the purposes of this research, a descriptive case study approach was applied to the investigation of marine citizenship factors. This allowed a thorough investigation of the two thematic factors identified in Chapter Four. The data was collected with the aim of describing the current situation regarding education and personal attachment to the marine environment in the context of marine citizenship.

 Table 5.1: Illustration of the various types of research case study types

 (Adapted from Yin, 2003a; 2003b)

Case Study Types				
	Exploratory	Descriptive	Explanatory	
Single and multiple methods.	Defines questions or areas of research for a subsequent study of any nature.	Presents a complete description of event/ phenomena within its context.	Examines how or why events happen and explains these phenomena.	
	Data collection takes place prior to final determination of questions.		Often used to trace events over time.	
	Often used as a prelude to social research.			
	Research may take an intuitive direction, generating theory from the data.			

5.3.2 Use of multiple-case design case studies

As illustrated in Table 5.2 case study research can be comprised of single and multiple cases. Although, single case design studies are the most common format used in case study research; the use of multiple case studies has increased in frequency (Yin, 1994). Multiple cases are often used as comparative studies with the evidence produced from this type of study often considered more compelling than single case designs (Yin, 1994). The use of multiple cases can be complex, and as such Yin (1994) recommends the use of theoretical replication i.e. ensuring that each case has a distinctive purpose thereby producing predictably contrasting results. Using multiple cases in this way allows comparisons to be made promoting a more comprehensive and diverse investigation into the phenomenon being examined (Darke *et al.*, 1998). Figure 5.1 illustrates how multiple cases for each case study followed by a cross-case synthesis of the results in order to further develop the theory of marine citizenship.



Figure 5.1: Process of conducting a multiple-case case study (Adapted from Yin, 1994).

As discussed in Section 4.5, the themes of education and personal attachment were identified as key themes in the context of marine citizenship through analysis of the practitioner interview survey. It was determined that the most appropriate method for assessing these themes would be through thematic case study research. Given the common assumption that public opinion and perception may vary depending on location, it was determined that case study sites should be selected across a wide geographic distribution. This also ensured that the study could be considered a viable representation of the UK. The case study sites proposed were initially based solely on geographical location in an attempt to locate sites that could be used to examine both key themes identified in Chapter Four. However, this had the potential to geographically limit the data, resulting in a poorly constructed evaluation of marine citizenship. Thematic case studies¹³ were therefore determined to be the most efficient and effective method of thoroughly assessing the themes identified in the practitioner interviews. Using thematic case studies allowed investigation of each theme in a number of places, fundamentally acting as multiple cases of one study and allowing comparisons to be made.

5.3.3 Limitations of overall case study research methodology

Johnson and Onwuegbuzie (2004) suggest that qualitative researchers can often neglect to rationalise interpretation of the data they collect. A common challenge is the assumption that data collected through different means can be added together to generate a more rounded perception of the data (Brannen, 2005). In order to avoid this, the qualitative and quantitative data were analysed independently and were found to provide mutually supportive interpretations. Yin (1994) has identified a number of concerns traditionally associated with case study research:

- There is often a perceived lack of rigour in data collection when conducting case study research.
- There can be the potential for researcher bias to influence the direction of the findings or observations derived from the case studies.
- Single case studies can provide little basis for generalization of theories.
- Case studies can often result in lengthy data collection periods which can be a limiting factor for research.

Both the education and personal attachment themes were investigated using structured interviews with a clear set of predetermined questions minimising the potential for researcher bias during data collection. Thematic case studies using multiple locations allowed the results to be used as a representation of public consensus in the UK. As suggested by Yin (1994) case study research can often be time consuming and generate a high volume of data. For the purposes of this research, the time required was accounted for and the high volume of data was

¹³ Thematic case studies involved the investigation of one theme across multiple locations i.e. the theme is the case study not the location in which the data collection takes place

considered a benefit as it ensured a thorough investigation of the key themes. Each of these issues has been considered and is discussed in Sections 5.5 with specific reference to the relevant thematic case study.

5.4. GENERATION OF KEY THEMES

Analysis of the practitioner interviews highlighted a number of potentially significant themes within the data outlined in Sections 4.5. These provided the foundation for the community based investigation into the application of marine citizenship, establishing areas for further investigation. The factors proposed for investigation included the influence of education on an individual's sense of marine citizenship, the influence of socio-economics, including livelihood and dependency, location, exposure to the marine environment and also culture, on the inculcation of marine citizenship within society, as well as examining societal perceptions of management responsibilities towards the marine environment. These individual factors were further categorised into two broad primary theme categories (Chapter Four) for use in the case study investigation phase of the research. The specific methodologies used to investigate these two broad themes, displayed in Table 5.2 will now be discussed.

Primary Theme	Includes the following factors:
Education	Awareness
	Concern
	Access to Efficient Information
	Knowladge of Marina Issues
	Knowledge of Marine Issues,
	Capacity for involvement,
	Location
	Responsibility.
Personal Attachment	Responsibility,
	Concern,
	Awareness,
	Dependency i.e. Livelihood,
	Culture i.e. traditions, family heritage,
	Socio-economics,
	Politics,
	Location,
	Proximity to the marine environment.

 Table 5.2: Key themes identified through the marine practitioner interviews.
5.4.1. Education

Education was recognized as having a vital role in the evaluation and promotion of marine citizenship, with strong links made between marine environmental education in schools and society's sense of concern and awareness towards the marine environment. It was decided that the factors of concern, awareness, knowledge and information availability, although considered separate parameters in the Hawthorne and Alabaster model (1999), they are intrinsically linked to education and therefore can be considered collectively as a primary theme.

5.4.2. Personal attachment

The theme of personal attachment was determined as encompassing the more specific themes of location influencing an individual's sense of place, the power of cultural perception of the marine environment and how it links with an individual's everyday life, the impact of personal connections to a particular place and finally the importance of a person's dependency on the marine and coastal environment. In addition, this theme evaluated the influence of connection to the marine environment on their sense of responsibility, concern and awareness.

5.5. SELECTION OF CASE STUDY RESEARCH STRATEGY

A number of potential combinations for the progression of the case study phase of research were considered, discussed in Sections 5.5.1-5.5.4.

5.5.1. Case Study Option One: urban and rural coastal communities

It was proposed that an investigation into the potential differences between rural coastal communities and urban coastal communities would provide information for many of the research questions prompted by the practitioner interviews. This option would have allowed an examination of the cultural influences, socio-economic factors and education identified by practitioner survey as having a potential influence on the inculcation of a societal sense of marine citizenship. It

was also thought that this would allow an investigation into the influence of community dependency on the marine environment on the concept of marine citizenship. However, given the importance attributed to proximity to the coast by the majority of the marine practitioners, it was decided that this option was inappropriate. By focusing solely on coastal locations, an investigation based on these criteria would not allow a comprehensive investigation into the relative influence of proximity on an individual's perception of the marine environment, and thus their sense of marine citizenship.

5.5.2. Case Study Option Two: Presence or absence of a marine campaign

Option two proposed the selection of case studies based on the presence and absence of a marine campaign¹⁴. The rationale behind this option was that it would be expected that an area where there had been a successful marine campaign would exhibit a higher level of marine citizenship within its community. As in Option One, this option would allow the factors of education, socio-economics and societal perception of management responsibilities to be examined. However, concerns were raised that case study sites selected on this basis may not give a comprehensive a view into the effects of culture. Again, as with option one, it was decided that this option would not necessarily provide the most appropriate case study sites for a thorough investigation into the themes to be conducted. In addition to this, similarly to option one, the implications of proximity to the coast could not be investigated.

5.5.3. Case Study Option Three: Sites based on marine environment type

The third option considered proposed the selection of case studies based on the presence of certain marine environments. This option would have allowed investigation into the influence of education and socio-economics in the areas selected, and would also encourage examination of public perception of various marine environments. Potential environments proposed included estuarine, sandy and beach environment and cliff based environments. As with the first

¹⁴ Marine campaigns were considered to include local and regional marine conservation or education initiatives.

two options, this option for selecting case studies would have prevented an examination of the influence of proximity to the coast on marine citizenship with site selection restricted to coastal areas.

5.5.4. Case Study Option Four: Combination

It was decided that a combination of Options One, Two and Three would permit the most comprehensive investigation into the proposed factors of marine citizenship, simultaneously examining the public perception of its role and application in the management of the marine environment. By combining the aforementioned options it was possible to evaluate the proximity to the coast, dependency on the marine environment and the influence of marine features while simultaneously investigating the themes of education and personal attachment. As discussed in Section 4.4.3, analysis of the practitioner interviews highlighted a common practitioner perception that there would be an assumed lack of attachment, responsibility and awareness of marine issues within communities' further inland in comparison to their coastal counterparts. Therefore, in order to fully examine the potential influences of proximity to the coast on marine citizenship, it was proposed that an inland community should be used as a case study site. By conducting the research in a variety of coastal locations supporting different industries and processes, the influence of these factors in the value placed on the marine environment by individuals could be assessed.

5.6. IDENTIFICATION OF CASE STUDY SITES

Given the vast coastline of the UK, and the diverse range of habitats and communities based around it, any number of locations could have been selected as sites to investigate the thematic case studies. In order to decide on the best option for this research, a number of logistical criteria as recommended by Curtis *et al.* (2000) were considered in conjunction with criteria identified through the research question. Table 5.3 illustrates the criteria used to determine the direction of progression for the case study phase of research. Table 5.4 illustrates the suitability of the proposed locations based on a number of selection criteria

recommended by Curtis *et al.* (2000) and the research requirements. Further explanation of how these criteria apply to the specific locations selected for case study research is presented in Section 5.6.

Table 5.3: Case study criteria as determined by Curtis *et al.* (2000) and the requirements of the research

Case study criteria				
Criteria based on research	Criteria based on research questions			
practicalities				
Relevant sampling strategy.	Presence of marine campaign.			
High generation of information.	Coastal area.			
Enhanced theoretical generalizability.	Access to schools.			
Provision of accurate information.	Access to a variation in socio-			
Ethical sampling strategy.	economic situations.			
Feasible sampling plan.				

Case study site						
Criteria	Poole Harbour	Lamlash Bay	Birmingham	North Antrim Coast	Milford Haven	Helsby, Cheshire
Relevant sampling strategy	1	1	1	1	1	✓
High generation of information	1	1	1	1	1	✓
Enhancement of theoretical generalizability	1	1	1	1	1	√
Provision of believable descriptions	1	1	1	1	1	✓
Ethical sampling strategy	1	1	1	1	1	√
Feasible sampling plan	1	1	1	1	1	v
Presence of marine campaign	1	1	Х	1	1	Х
Coastal area	1	1	Х	1	1	√
Access to schools/ education system	1	J	J	J	✓	1
Access to variation in socio- economics				<i>✓</i>		1

Table 5.4: Criteria for assessing case study site suitability for the research with presence indicated by the ✓ symbols (Adapted from Curtis *et al.*, 2000)

Tables 5.4 and 5.5 indicate that each of the proposed case study sites has the potential to provide detailed information about the themes associated with marine citizenship being investigated in this research phase. The flexible and adaptive methodology allowed for a potential increase in the number of geographical locations during the case study phase of research to ensure that the themes were exhaustively examined. It was also decided that a diverse range of coastal types should be examined throughout the thematic case studies; a description of the proposed case study sites is displayed in Table 5.5.

Case Study Site	Key Points
Poole Bay, Dorset, England.	Coastal region, Holder of a number of national and international designations. Main industries include tourism and recreation.
Isle of Arran, Scotland	Coastal region, Traditional fishing culture. Main industries are tourism and fishing.
Birmingham, West Midlands, England.	Inland region. Approximately 100 miles from the nearest coastal area. Main industries were traditionally manufacturing and engineering but are now dominated by the services and tourism sector.
Milford Haven, Pembrokeshire, Wales.	Coastal region. Pembrokeshire Coast. Main industries include heavy industry, petrochemicals and traditional fishing
Helsby, Cheshire, England.	Estuarine/ river basin environment, Historically dependent on agriculture now dominated by manufacturing and commuter professions.
North Antrim Coast, Antrim, Northern Ireland	Coastal region Site of the Giant's Causeway, UNESCO World Heritage Site of global cultural importance, Main industry is tourism.

Table 5.5: Description of thematic case study sites.

In order to ensure the study could be considered a UK wide project, it was important to identify sites that would be representative of the variety of coastlines and socio-economic relationships between UK society and the marine environment. It is expected that people living near/on the coast will express higher levels of environmental concern and knowledge, regarding both environmental issues and management policies (Cicin-Sain and Knecht in Steel *et al.*, 2005). It has been argued that people inhabiting the coastal zone are likely

to view the marine environment and its resources in more immediate and more tangible terms than inland communities. Given this, it was decided that it was important to assess the influence of the hypothesised factors on a sense of marine citizenship among citizens residing in both coastal and inland communities. An introduction to each of the sites selected and their suitability as a case study site in this research project is outlined in Sections 5.6.1- 5.6.7.

5.6.1. Poole Bay

Poole Bay comprises of a 16Km stretch of coastline on the Southern coast of England extending from the Sandbanks Peninsula to Hengistbury Head (Poole and Christchurch Bay Management Group, Online). Poole Bay supports a wide variety of activities, both commercial and recreation. It is of local, national and international conservation importance and has been the subject of many conservation designations. Poole Harbour has been designated a wetland of international importance under the RAMSAR convention since 1971. The European Marine Site (EMS) Special Protected Area (SPA) was designated in 1999 while the harbour within Poole Bay falls within an area of outstanding Natural Beauty (AONB). In addition, there are two designated Sites of Special Scientific Interest (SSSI) including Poole Bay Cliffs SSSI, Christchurch Harbour SSSI and three local nature reserves at Branksome to Southbourne Overcliff, Branksome Dene Chine and Hengistbury Head (Poole and Christchurch Coastal Group, 2007).

Due to its diverse range of uses and community requirements Poole Bay was determined to be an excellent candidate for case study based investigation for this project. Its multiple uses will ensure a detailed examination of the influence of socio-economics on the marine environment and the potential for the development of marine citizenship in the surrounding community. Poole Bay is also a popular tourist region meaning that there is scope for investigating the level of marine citizenship and how this concept could be applied among the tourist population, as well as among local residents. Numerous factors made Poole Bay an appropriate case study site; logistically it was a sensible option being that it in close proximity to Bournemouth University and therefore incurred minimal travel or accommodation expenses.

5.6.2. Isle of Arran

Arran is the largest island situated in the estuary of the Firth of the Clyde on the west coast of Scotland inhabited by a population of approximately four thousand people (McLellan, 2008). Over time, the population has become increasingly concentrated around Brodick, the main pier on the island, and the nearby towns of Lamlash and Whiting Bay. A number of marine practitioner interviewed in phase one highlighted the efforts of the community in this area following the development of a community led campaign for a MPA in Lamlash Bay, off Arran, West Scotland. Given this, Lamlash Bay was identified as an area where marine practitioners expected there to be an inbuilt sense of marine citizenship. It is proposed that, given this recognition by marine management practitioners, Arran should be considered for the second case study site examined in this project.

5.6.3. Birmingham

Birmingham is commonly considered to be the central point of the United Kingdom with the nearest coastal area being almost 90 miles away from the city. Traditionally Birmingham has been dominated by manufacturing and production with the area traditionally acting as one of the key focal points for trade throughout history. Given that challenges posed by differences between inland and coastal communities were highlighted by marine practitioners, it was determined that an inland case study would allow this potential relationship to be examined. Interviewees frequently named Birmingham as the place furthest from the coast in the UK when drawing comparisons between inland and coastal regions, prompting the selection of Birmingham as a suitable case study site.

5.6.4. Helsby

Helsby is a rural village in the North West of England in Cheshire, situated on the Mersey estuary. Traditionally, the area has a strong agricultural history although more recently the area has become dependent on major manufacturing industries in the area, as well as being a commuter town for Liverpool, Warrington, Chester and Manchester. Given its location, Helsby is neither fully inland nor coastal and was thought to be a potentially interesting case study as a result of its estuarine nature.

5.6.5. North Antrim coast (Giants Causeway)

The area surrounding the Giants Causeway on the North Antrim coastline has been identified as a case study site as a result of it's designation as a United Nations Education Scientific and Cultural Organisation (UNESCO) World Heritage Site (WHS) in 1986 (Watson, 2000). The area was recognised as a WHS site based on two main criteria: it is considered to be an important example of the earth's evolutionary history during the Tertiary period, whilst supporting rare natural phenomena, including rare bird species, the Chough. This was further followed up by its designation as a National Nature Reserve by the Department for Environment of Northern Ireland in 1987, with further designation as the Causeway Coast Area of Outstanding Natural Beauty in 1989 recognising the causeway and the surrounding area as being of national importance. These designations help develop the causeway in a manner that benefits both local communities and visitors to the area, whilst promoting education and research. It is inferred that an area with WHS status would face different issues than other coastal areas due its recognised international cultural and historical importance. The North Antrim coast also provided the research project with a Northern Ireland case site allowing investigation into the implications of the devolved Northern Irish Government on marine citizenship.

5.6.6. Milford Haven

Milford Haven is the largest town in Pembrokeshire, Wales, and expands the geographical spread of the study to encompass the whole United Kingdom. Milford Haven has experienced an industrial shift, traditionally dependent on a thriving fishing industry; its dependence on the coastal and marine environment is now based on heavier industry, with the construction of oil refineries in the area. The town itself is of particular interest to this project as the town is within the Pembrokeshire Coast National Park boundary. Milford Haven was identified as a suitable site for the thematic case study due its evolving industrial history, its coastal location and its role as the Welsh case study site. In addition, the presence of the Pembrokeshire Coast National Park suggests a certain level of awareness of marine issues in the area.

5.7 DATA COLLECTION

In a mixed method research project, case studies can typically involve a variety of data collection techniques (Darke *et al.*, 1998). The methods used in this phase of data collection are outlined in Sections 5.7.1-5.7.2.

5.7.1. Education

5.7.1.1. School based questionnaires

The first theme identified through the practitioner interviews was the role of education in the development of marine citizenship. This was strongly linked with the feeling that younger generations would be the most productive demographic at which to aim environmental education. The decision to focus on schools was directed by both the literature and observations taken from the practitioner interviews. Primarily, the practitioner interviews suggested that attempts to improve societal responsibility and awareness towards the marine environment should be directed at school children, as they are the decision makers of the future. Furthermore, both the academic literature (Hawthorne and Alabaster, 1999; Steel et al., 2005; McKinley and Fletcher, 2010) and the

interview results indicated that the level of awareness of environmental issues should theoretically increase with age. Therefore, it was proposed that a school questionnaire should be administered to identify the level of knowledge within school aged children regarding the marine environment and its management.

Devolution within the UK has resulted in the development of alternative education systems. Currently, the national curriculum in England, Wales and Northern Ireland is divided into four blocks called key-stages (KS) with KS3 and KS4 completed during secondary education. In Scotland, the national curriculum is divided into single year blocks with stages (S) 3 and 4 completed at secondary level. Table 5.6 presents an outline of the key stages and student ages in UK national.

Age	England and Wales	Northern Ireland	Scotland		
3	Nursery (non-compulsory)				
	During any Variate as 1	During any Vary stage 1	Numerow (non-commulation)		
4 5	Primary Key stage 1	Primary Key stage 1	Nursery (non compulsory)		
4-5 5 (Veer 1	Year 2	Drimowy (D)		
5-0	I cal I	real 2	\mathbf{P}		
67	Voor 2	Voor 2			
0-7	I cal 2	real 5	F Z		
	Kev stage 2	Key Stage 2			
7-8	Year 3	Year 4	P 3		
8-9	Year 4	Year 5	P 4		
9-10	Year 5	Year 6	P 5		
10-11	Year 6	Year 7	P 6		
	Key Stage 3	Key stage 3	P 7		
11-12	Year 7	Year 8			
12-13	Year 8	Year 9	Secondary (S)		
			S 1		
	Key Stage 4	Key Stage 4			
13-14	Year 9	Year 10	S 2		
14-15	Year 10	Year 11	S 3		
15-16	Year 11	Year 12	S 4		
END OF COMPULSORY EDUCATION					
16-17	Year 12 (lower sixth)	Year 13	S 5		
17-18	Year 13 (upper sixth)	Year 14	S 6		
	· · · · /				

 Table 5.6: Current school system within the four UK countries (Adapted from the British Council, Online)

At the end of KS4 and S4, students complete their compulsory schooling and are approaching fully active citizenship within society. In addition the national curriculum for England and Wales stipulates that citizenship education is compulsory for KS4 students and is included in Scottish education. Therefore students at this level of education (KS4 in England, Wales and Northern Ireland, and S4 in Scotland) were identified as the most appropriate age group through which to investigate the theme of education. As discussed by Castle *et al.* (2010), the national curriculum in the UK does not stipulate exactly what must be taught in schools meaning that although marine examples may be selected by teachers, this is not guaranteed. However, the national curriculum does provide teachers and schools with a structured framework from which to choose subject matter, identifying opportunities for marine education.

5.7.1.2. Use of Self administered Questionnaires

Questionnaires are a structured form of data collection and can be used in a variety of circumstances. Denscombe (2003) suggests that they are best suited to collecting data, if and when the following conditions are met:

- If a large number of respondents is required across a wide geographic area,
- If fairly straight forward information is required from the data collection,
- If standardized responses are necessary without the need for face-to-face interaction between the researcher and the respondent,
- If the research allows for time and cost issues related to the administration of the questionnaire, printing and data preparation,
- When the respondents are expected to be capable of reading and understanding the questionnaire without the need of clarification from the researcher.

Responses from self-completion questionnaires are most commonly based on fact or respondent opinion of the topic under investigation (Denscombe, 2003). Given the nature of the theme being investigated, and the age of the participants, it was decided that the majority of the questions would be worded in such a manner that both factual information and student perception of the marine environment could be examined. In addition to developing appropriately worded questionnaires, there are some routine elements that must be incorporated into the development of any questionnaire used in research. This additional information should include details about the sponsor of the project, the purpose of the research, assurances of interviewee confidentiality, return address and date (if required) and finally a note of thanks to the respondent for their participation (Denscombe, 2003). Each of these conditions was met in this study with the Bournemouth University logo included on all documentation and closing questionnaires by thanking the participants. In order to ensure the other information was read, the purpose of the research, assurances of confidentiality and other information included in the information supplied to the teacher. Copies of these documents can be found in Appendices 6 and 7.

Self-administered questionnaires are one of the more commonly used data collection techniques in research (Bourque and Fielder, 1995). There are a number of ways that questionnaires of this type can be delivered, as presented in Table 5.8 (Bourque and Fielder, 1995). For the purposes of this study, it was decided that group administration of the questionnaire would be most appropriate. Classroom based administration of questionnaires is a common method of collecting data of this type, coupling the benefit of targeting a larger group of people thereby obtaining a higher volume of data with the additional benefit of facilitating semi-supervision of the data collection process (Bourque and Fielder, 1995). This ensured that, within each location, all participants were in the same place and in addition the distributor (in this case, the teacher) could deal with questions and deliver instructions to the participating group. In addition, Bourque and Fielder (1995) suggest the use of close ended questions in self administered questionnaires as only the most self motivated of respondents will feel sufficiently confident and willing to respond to a questionnaire dominated by open ended questions. This recommendation was followed in the construction of the questionnaire used at this stage of the research. The advantages and disadvantages of using self-administered questionnaires are presented in Table 5.8.

Table 5.7: Benefits and Weaknesses of the most common methods of selfcompletion questionnaire administration (Adapted from Bourque andFielder, 1995).

Type of Questionnaire Administration	Strengths	Weaknesses
One-to-One	 Interviewer present to answer questions; Higher confidence levels with regard to results from face-to-face interviews; Allows assessment of the answerability of questionnaire. 	- Expensive in time and money.
Group	 Consistency with regard to deliverance of questionnaires; Some assessment of answerability of the questions (particularly useful in pilot stage); allows completion to be monitored; Administrator can provide instructions and answer queries. 	- Not appropriate when testing general populations.
Semi-supervised	 Administrator can provide instructions and answer queries; Efficient; Less costly financially; Useful for pilot studies and pre-testing; Some ability to monitor completion of questionnaires. 	 Frequently results in unrepresentative samples; Can result in inconsistency in instructions.
Unsupervised	 Consistent stimulus for all respondents; Potential generation of a more representative sample. 	 No control over who completes questionnaire; No feedback regarding answerability of the questions; Questionnaire must be developed so that it stands alone to minimise misunderstandings and queries.

Advantages	Disadvantages
 Low costs both in time and money; Targets a large audience easily and quickly; Analysis can be simple due to the common use of closed questions; Not as much pressure for immediate response; Allows generation of standardised questions and responses; Reduction in interviewer bias; Increased guarantee for respondent anonymity; Results provide evidence suggesting areas for further investigation. 	 Issues with data quality; Typically have a low response rate; Can be difficult to motivate respondents; Seeks information purely by asking questions; Lack of control over order of completion of the questionnaire; People often talk more easily than they write; Generally no opportunity for clarification; Wording of the questions must be carefully considered.

Table5.8: The potential advantages and disadvantages of usingquestionnaires as a data collection method (Adapted from Gillham, 2000).

As with generation of the interviews in the previous data collection phase, the format of the questionnaire was an important consideration (Table 3.9). In interviews and questionnaires, it is possible to use both open and closed questions. The benefits of closed questions lend themselves best to this portion of the data research generating a structured questionnaire with minimal potential for misunderstanding. Given the high volume of students being approached, the use of closed questions also meant a standardisation of the responses, making data analysis less complex (Denscombe, 2003).

5.7.1.3. Generation of Questions

As described in Section 5.1, the questions generated for the self-administered school questionnaire were guided by the observations made regarding education in Chapter Four. It was important to assess the level of knowledge held regarding the marine environment by the next generation of 'decision makers'. As such, the questionnaire aimed to collect primarily factual information with students required to answer a number of questions about common and well

publicised marine issues. These questions were partially directed by a questionnaire administered by Steel *et al.* (2005) to address public ocean literacy, in order to allow for some comparison between the studies. This will be highlighted and discussed further in Chapters Six and Seven.

In order to assess student knowledge and awareness of marine related issues, a number of indicators adapted from a study conducted by Steel et al. (2005) were used in a quantitative data based survey. As in Steel et al. (2005)'s study an indicator of both subjective knowledge (personal perception of knowledge) and objective knowledge (actual knowledge) was used. Students' subjective knowledge was investigated through survey participants' completion of a 'term familiarity' exercise, which required students to indicate their level of familiarity with a number of terms and phrases commonly encountered in relation to the marine environment. It was decided that not all terms used in Steel et al. (2005)'s study were applicable to the UK. Terms mentioned on a regular basis with regard to the issues facing the marine environment and the potential role of marine citizenship in the UK were used in place of those considered inappropriate to the study. Given that marine practitioners were commonly of the opinion that there is higher public awareness about global issues than local or regional issues, it was considered important to include terms that applied to both the national and international marine environments.

Levels of objective knowledge were investigated using a series of multiplechoice questions concerning well-documented issues facing the marine environment. All but one of the questions used in this section were taken from the work done by Steel *et al.* (2005) to allow for direct comparisons to be made between the two studies¹⁵. Students were then required to indicate their knowledge of marine designations, management organisations and indicate how they sourced marine information. Finally, students were asked to complete a series of questions based on self-assessment regarding their level of concern,

¹⁵ The 'marine quiz' used in this survey included the question: The transportation of sediment along the coast is known as: sediment drift; across coast drift or longshore drift. This question replaced a question on the El Nino phenomenon that was used in the survey in Steel *et al.*'s (2005) study. It was decided to include a question on coastal erosion and sedimentation given the severe impacts of sediment transport in numerous UK coastal areas e.g. Happisburgh.

knowledge and awareness of the marine environment and its inclusion in the national curriculum. An example of the student survey is presented in Appendix 9.

5.7.1.4. Limitations of the method

As with all methods, there are a number of limitations that must be considered and mitigated for when using questionnaires as a method of data collection. Some of the issues, such as low completion rates and lack of opportunity for clarification (Gillham, 2007) did not strictly apply to this research, as the questionnaires were administered by teachers who could provide clarification and would ensure that students completed the surveys to the best of their ability. Misunderstandings were considered to infer a lack of knowledge of the topic, which will be discussed further in Chapter 6. The brevity of the questions, which is often considered a disadvantage (Gillham, 2007) was helpful in this case as the target group was school students and it was important to ensure they did not lose interest before completing the entire questionnaire.

By using closed questions in the student questionnaire, it was a possible risk that the resulting data may be directed more towards the researcher's way of thinking (Denscombe, 2003). This was mitigated for by ensuring the wording of the questions could not lead the responses and by using the questions to obtain predominantly factual information. As described in Table 3.9, close-ended questions can be frustrating for respondents, as they tend to lack the opportunity to give a detailed response. However in this case, the questions were investigating level of knowledge and self assessed levels of awareness, calling for factual responses to the questions, rather than an investigation of opinion.

As with all self-completion surveys and questionnaires (Fink, 2003), the potential for missing data was high in the school questionnaire component of data collection. As missing information could have been a function of low awareness or knowledge regarding the subject matter, only questionnaires with more than three items of information missing were excluded. Two questionnaires were returned with only the first page completed and as a result, were excluded from the final analysis. This gave an overall response rate of approximately 98%.

Finally, there was also the issue of 'cleaning' the collected data. Data, once entered into a database can be considered 'dirty' as a result of miscoding, inaccurate data entry and missing data (Fink, 2003). In order to mitigate for any of these potential issues, all data was entered by the primary researcher so as to maintain continuity with regards to data coding, entry of the data into the online database and the method of dealing with missing data and incomplete questionnaires.

5.7.1.5. Selection of Schools

Schools were selected based on the following criteria; that the schools provided secondary level education at the appropriate level for this study and that they were co-educational institutions.

5.7.1.6. Pilot Study for School Based Questionnaires

Pilot studies are a valued component of case study research and allow the researcher to ensure their selected methodology is appropriate for the circumstances of the study (Bourque and Fielder, 1995; Lancaster, 2004). In the case of this study it also allowed the wording of the questions and the accompanying instructions to be evaluated in a working environment. A pilot study was conducted at a Bournemouth secondary level school in June 2009 with 22 students taking part. The questionnaire was emailed to the relevant teacher along with instructions regarding the administration of the questionnaire. Students were given the opportunity to comment on the structure of the questionnaire in order to identify any areas requiring alterations. No comments were made from the students, however, when it was suggested that abbreviated names and acronyms be accompanied with the full name of an organisations or marine designation, the teacher agreed that this would be a beneficial change. The teacher involved in the pilot study reported no issues with the administration

of the study or understanding of the instructions so no changes were made in either area for the actual study.

5.7.1.7. Actual Study

The study was carried out in three secondary level schools between October 2009 and March 2010. The schools selected and that agreed to participate in this study were from the Isle of Arran, the Poole Bay and Helsby case study areas. It was hypothesised that awareness and knowledge may vary according to geographical location and proximity to the coast so it was important that schools from different areas participated. Initially, it was proposed that a school from Birmingham would be included in the study to ensure inclusion from an inland area. However, although a number of schools were approached and asked to participate, no school could facilitate the study in the time period. It was felt however, that this would not detract from the data collected as the schools included are located in a range of geographical areas and could therefore provide sufficient information regarding the potential relationship between location and the theme of education.

The target group for the questionnaire were the Key Stage 4 (England and Wales)/ Year 3 (Scotland) as described in Section 5. Questionnaires were emailed or posted to the relevant teacher, accompanied by a brief project rationale and simple instructions explaining the purpose of the questions, how the students were required to answer them and a return address and date (Appendices 7, 8 and 9). The teachers administered the questionnaires during one of their teaching sessions ensuring a high response rate. The questionnaires were then returned by post and the transcripts were typed. In total 126 surveys were completed by students at the three selected schools. Data entered into both Microsoft Excel spreadsheets and onto the online survey tool Survey Monkey.

5.7.1.8. Scope for Teachers capacity survey

Given the emphasis placed on the role of formal education for younger generations by the telephone interviewees, it was considered important to assess the current capacity of the education system to deliver marine specific education. In order to investigate this, a short self-completion survey investigating teachers' capacity to deliver marine education effectively was provided to the schools participating in the student survey. It was requested that the survey was given to teachers of subjects most likely to involve marine education, namely biology and geography (as identified in research by Castle et al., 2010). The survey consisted of 15 open-ended questions examining teachers' perception of marine specific education currently available in the UK national curriculum and the potential influences of this on their students' future decision making and behavioural choices regarding the marine environment. A copy of the teachers' capacity survey is presented in Appendix 11. Although the teachers' capacity questionnaire was provided to a number of teachers, only three completed teachers' capacity surveys were returned. This was a lower number than expected and means that the observations made regarding this data cannot be considered representative of the overall UK teaching community. However, following analysis of the data collected, it was decided that it would be valuable to include the results, as they could be discussed in relation to the student education questionnaire.

5.7.2 Personal attachment

The second theme identified by the practitioner survey phase of data collection was that of personal attachment¹⁶. As suggested earlier, the theme of personal attachment can encompass a number of very different aspects of an individual's life when applied to the marine environment. This phase involved generation of a structured short interview conducted within five case study site communities. The survey investigated how sense of place has the power to influence awareness, and desire to behave in a pro-environmental manner towards the marine environment.

¹⁶ Personal attachment was defined as including a variety of factors including but not limited to livelihood dependency, childhood memories, recreational ties, and historical or cultural ties to an area.

It was proposed that conducting street interviews within the case study site communities would be the most efficient way of collecting a wide range of data from people of a variety of socio-demographics. Conducting a survey in this way provided a better representation of society than focusing on groups, which may already have involvement with the marine and coastal environment.

5.7.2.1. Use of structured interviews

Interviews are most commonly used to allow the researcher to go into more depth than is usually associated with data collection. Among other reasons, they are often used to obtain data based on the participant's emotions and opinions regarding the subject matter (Denscombe, 2003). There are a variety of approaches that can be taken when conducting interviews as illustrated in Table 5.11.

		Types of In	terviews used in Resea	ırch	
Structured	Semi-structured	Unstructured	Group	One to One	Focus Groups
Questions tightly	Clear questions to	Emphasis more	Allow generation of	Most common form of	Sessions tend to revolve
controlled by	be addressed;	firmly placed on	consensus views and	any of structured, semi	around a theme to focus the
researcher;		interviewee opinions;	richer responses than	structured and	discussion;
			one-to-one interviews;	unstructured	
				interviews;	
Predetermined	Flexibility to allow	Researcher	Number of challenges:	Relatively easy to	Interaction within the group is
questions with	respondent to move	introduces a topic	Conversation can be	arrange;	important to the outcome;
limited	interview in	and is flexible	dominated by stronger		
responses;	another direction;	regarding direction	characters;		
Use of pre-coded	Allows respondent	Lend themselves best	Views expressed may	Opinions only coming	Collective views are obtained;
answers;	to develop on their	to in depth	only be those viewed	from one source so	
	ideas;	investigations.	to be acceptable to the	easier to control	Difficult to record;
			group.	interview.	
Used to collect	Open ended				Participants may be reluctant
large volumes of	questions.				to disclose certain information.
data;					
Cimilar to					
onestionnaires					

Table 5.11: Various approaches to research interviews (adapted from Denscombe, 2003).

In contrast to the initial phase of data collection where semi-structured interviews were used to obtain practitioner opinions of the subject matter, structured interviews were found to be more appropriate. Structured interviews are often used in research where large volumes of data are required (Denscombe, 2003), as is the case in this study, and therefore lent themselves well to this data collection phase. In order to allow participants opportunity to express their opinions without complete restriction, a 'comments' area was made available after each question, allowing participants to expand on particular points. This qualitative data was collected to support the relationships identified through the quantitative data. A copy of the interview transcript can be found in Appendix 12.

5.7.2.2. Generation of Structured Interview Questions

The questions generated for investigation of the theme of personal attachment were guided by the practitioner interviews. In order to identify if there were discrepancies between practitioner and public perception of the role of the public in marine management and overall public awareness and concern, a number of the questions included in the personal attachment survey were taken directly from the practitioner survey. For example, interviewees commonly referred to consumerism as a potential mechanism through which to express marine citizenship and as a result two questions were included relating to individual's consumer behaviour.

The questions were structured to have a closed format to ensure that the interview could be conducted rapidly and accurately. This was an important consideration given the sampling technique used by the interviewer. Closed questions are generally only used when the answers required are factual (Gillham, 2007). However, in the case of this research the closed questions were used to identify general trends with an opportunity for the participants to make comments following each question should they want to elaborate on any point. Closed questions are rarely sufficient as a data collection on their own and often act as a component of mixed methodologies (Gillham, 2007) as was used in this research. The structured questions acted as the quantitative data collection,

whilst simultaneously providing an opportunity for interviewees to make comments about issues that were important to them.

5.7.2.3. Interviewee selection

Interviewee selection is commonly based on their capacity to provide information related to the research. There are a variety of sampling method commonly used during administration of interviews; random, convenience and quota sampling (Gillham, 2007; 2000), as illustrated in Table 5.9. Given the nature of the data collection, and the desire to talk to members of the general community rather than people who are already involved in management of the marine environment at some level, convenience sampling was deemed the most appropriate method of selecting participants.

Table 5.9: Commonly used methods of selecting sample participants(Adapted from Gillham, 2007).

	Sampling 7	Fechniques	
Random	Convenience	Quota	Systematic
When each person has an equal chance of being selected to participate through generating a random sequence of numbers. The smaller the sample, the less likely it is that it is representative.	Selection of the most convenient respondents e.g. selecting people at random on the street or at a particular location.	Selection based a representation of the characteristics of the whole sample.	Based on a systematic method of selection e.g. every fifth person.

5.7.2.4. Pilot study

A pilot study of the personal attachment thematic case study interviews were carried out, with the aid of two undergraduate students from Bournemouth University, in June 2008 on Bournemouth Seafront. The sixteen individuals interviewed were asked to make comments on the interview structure, format and delivery. However, no issues were raised regarding the structure of the interview

or the wording of the questions during the pilot study. It was noticed that people were often reluctant to participate, as it was not clear who was conducting the research. Individuals also appeared concerned that they would be required to make some form of financial donation or contribution despite interviewers carrying identification stating they were working on behalf of Bournemouth University. In order to mitigate for this, a Bournemouth University t-shirt was provided for the actual studies. This easily identified the interviewer as an academic researcher rather than working for a commercial organisation.

5.7.2.5. Actual Study

The actual study was carried out between July 2009 and February 2010. As a result of adverse weather conditions between November 2009 and January 2010, there was a considerable time lapse between the penultimate and final sites. The sites selected for investigation under the theme of personal attachment were Poole Bay, the North Antrim Coast (specifically the Giant's Causeway), Birmingham, the Isle of Arran and Milford Haven. During the Poole Bay and Isle of Arran case studies a team of trained interviewers aided the data collection.

In total, 275 interviews were conducted during this phase of data collection with the average time taken to complete each interview approximately 10 minutes. As in the practitioner survey, neither the number of interviews required from each location nor the number of sites chosen for interviews were predetermined. As suggested by Guest *et al.* (2006), the data requirements for this study were that data saturation¹⁷ regarding the theory was reached. Given the convenience sampling approach taken in the personal attachment case study, individuals were approached while they were conducting their business around the case study site. It is necessary to note that not all individuals were willing to take the time to complete the supplementary comments section of the interview but did complete the structured questions section. The potential limitations of this are considered

¹⁷ Saturation is the point at which no new information or themes are observed in the data. In order to ascertain when this has been reached, it was important to ensure that transcripts were read and analysed concurrently with the data collection.

in Section 5.8.2.6. The data collected through the interviews was entered into Microsoft Excel spreadsheets, and prepared for data analysis and interpretation.

Voice recorders were not used during these interviews as research has found that recording interviews can often make participants nervous and concerned about the anonymity of their responses (Darke *et al.*, 1998). The predominantly structured nature of the responses meant that this was not an issue as the interview could be recorded accurately by the interviewer at the time. Additional comments made by the interviewees were also noted by hand. Data collected from the interviews was recorded electronically at the end of each day to ensure responses remained clear and to minimise any confusion during transcription.

5.7.2.6. Limitations of the Methodology

As with other qualitative research methods, elements of personal identity, such as gender, ethnicity and age, can effect how an interviewee will respond to questions, depending on their perception of the researcher (Denscombe, 2003). In order to ensure these elements had a limited impact on the data collected it was important that the interviewer presented themselves in a neutral manner, in both courtesy and appearance and that they remained noncommittal throughout the interview.

The use of a convenience sampling strategy meant that there was no opportunity to pre-arrange interviews with potential interviewee candidates as had been the case in the telephone interview schedule. As a result, not all interviewees were prepared to take the time to complete both the quantitative and qualitative phases of the interview. Although the supplementary comments were useful, the quantitative data collected could be used to determine the most common behavioural patterns with regard to the public relationship with the marine environment. Therefore, interviews without the additional comments were considered to be very useful and to contain an extensive amount of information on which to base observations that could be supported by the qualitative comments. The convenience sampling approach also meant that individuals could not be approached later for clarification purposes. However, the quantitative nature of the score based questions prevented this from becoming a concern as the overall trends and behavioural patterns could still be established.

In the case of this study, it was a concern that approaching potential candidates at random at the case study site could be detrimental to the data collection process, with potential participants unwilling to be included. In order to put participants at ease, interviewers wore Bournemouth University t-shirts. Interviewers were also provided with an introductory script in order to introduce themselves as researchers for the university and provide a brief explanation of the purpose of their participation. Participants were assured that the interviews would remain entirely anonymous, as no personal details were required.

The implication of poor weather conditions on outdoor convenience sampling should also be noted. It was the initial aim that the five individual sites selected for investigating the theme of personal attachment would be visited consecutively over a period of five months from June to November 2009. This time schedule went according to plan for the first four sites but the final case study site was postponed until February 2010 as a result of poor weather conditions (including a number of severe weather warnings announced by the MET office).

5.8 DATA ANALYSIS AND INTEPRETATION

As outlined in Section 5.3, both qualitative and quantitative data were collected concurrently throughout the case study phase of the research. The data interpretation techniques used to analyse the data collected during the thematic case studies will now be discussed, with the results further outlined in Chapter Six.

5.8.1. Education

As explained in Section 5.7.1, the main component of the education thematic case study was an evaluation of student perception of marine education through formal education practices. The data collected was primarily quantitative and

underwent analysis using two programmes. First the data was manually uploaded onto the online survey tool, Survey Monkey¹⁸ which allowed basic descriptive analysis for each exercise in the student questionnaire to be conducted. Further analysis was conducted on relationships identified through the initial analysis using Minitab statistical package Version 15. As with all questionnaire data, there was the issue of missing data. For the purposes of this research, any incomplete surveys i.e. surveys with missing answers were not included in the data analysis so as to avoid potential issues when assessing the relationships present between identified factors. It should be noted that a small number of students provided additional comments at the end of their surveys which, although they were not officially requested as part of the data collection, gave valuable insight into student perception of the marine environment and are therefore included in the discussion where relevant.

The teachers' capacity survey data underwent content analysis adhering to the same guidelines as presented in Chapter Three, Section 3.8.2. Due to the low number of respondents, the use of content analysis allowed the identification of the most common themes regarding teachers' perception of marine specific education currently included in formal teaching in UK schools.

5.8.2. Personal Attachment

As discussed in Section 5.7.2.1, the personal attachment thematic case study called for the simultaneous collection of both qualitative and quantitative data. Given this, various forms of analyses were carried out on the data collected in order to fully interpret both the qualitative and quantitative components of the data.

As with the student education surveys, the quantitative data collected was manually uploaded onto the online survey tool Survey Monkey. Given the high volume of data collected through the personal attachment street interviews, the

¹⁸ Survey monkey is an internet based survey tool, which allows the researcher to create and manage their research questionnaires online. The researcher can then manually upload collected data responses to the Survey monkey server and can conduct basic analysis in order to identify key trends in the data.

use of Survey Monkey ensured the data could be easily managed, stored securely and could undergo basic analysis. Through the basic analysis tools provided by Survey monkey, potentially significant relationships were identified within the data with the significance of these relationships further assessed through statistical analysis. Statistical analysis was carried out using Minitab V15 which is discussed in more detail in Chapter Six.

With regards to the qualitative data collected through the personal attachment interviews, a similar analytical methodology as described in Chapter Three, section 3.8.2, was used with the data subjected to manual content analysis. In brief, textual segments were identified from within the data and categorised to generate a list of common themes. The most common themes were categorised in order to establish evidence of social trends within the data which could then be used in support of the observations from the quantitative analysis. Detailed results of this analysis are presented in Chapter Six and are discussed further in relation to the preceding phases of research.

5.9 SUMMARY

This chapter discussed the methodological approach chosen to further investigate the observations made in the practitioner survey phase of data collection. It explained the choice of thematic case studies as the most appropriate mechanism to investigate such broad themes, the rationale behind the selection of case study sites and the specific methods applied to each of the themes identified. Finally, the chapter focused on the data analysis and interpretation processes of both the qualitative and quantitative data collection that occurred in this phase of the research. The results of the case study research are outlined and discussed in Chapters Six and Seven.

REFERENCES

Bourque, B., L., and Fielder, E., P. (1995). How to Conduct Self-Administered and Mail Surveys, SAGE Publications Ltd.

British Council (online) Available at <u>http://www.britishcouncil.org/flasonline-uk</u> <u>education-system.pdf</u>. Accessed on 08/11/2010.

Castle, Z., Fletcher, S., and McKinley, E. (2010). Coastal and marine education in schools: Constraints and opportunities created by the curriculum, schools and teachers in England. Ocean Yearbook. 24: 425-444.

Creswell, J. W. (2009). Research design: Qualitative, quantitative and mixed methods approaches. , SAGE Publications, Inc.

Darke, P., Shanks, G., and Broadbent, M. (1998). "Successfully completing case study research: combining rigour, relevance and pragmatism." Information systems Journal 8: 273-289.

Denscombe, M. (2003). The Good Research Guide for small Scale research projects. 2nd Edition, Open University Press.

Fink, A. (2003). How to Manage, Analyze and Interpret survey data. 2nd Edition, SAGE Publications Ltd.

Gable, G., G. (1994). "Integrating case study and survey research methods: An example in information systems." European Journal of Information Systems 3(2): 112-126.

Gerring, J. (2007). Case Study Research: Principles and Practices, Cambridge University Press.

Gillham, B. (2000). Case study Research Methods. Continuum

Gillham, B. (2000). Developing a Questionnaire 1st Edition. Continuum

Gillham, B. (2007). Developing a Questionnaire 2nd Edition, Continuum.

Guest, G., Bunce, A., and Johnson, L. (2006). "How many Interviews are enough? An experiment with data saturation and variability." Field Methods 18(1): 59-82.

Hawthorne, M., and Alabaster, T. (1999). "Citizen 2000: Development of a model of environmental citizenship." Global Environmental Change 9: 25-43.

Lancaster, G., A., Dodd, S., and Williamson, P., R. (2004). "Design and analysis of pilot studies: recommendations for good practice." Journal of Evaluation in Clinical Practice 10(2): 307-312.

McLellan, R. (2008). Arran. Revised by Norman Newton, David & Charles Publications Inc.

Poole and Christchurch Shoreline Management Plan Group (2007). Available at http://www.twobays.net/poole_bay.htm . Accessed on 08/11/2010.

Steel, B., S., Smith, C., Opsommer, L., Curiel, S., and Warner-Steel, R. (2005). "Public Ocean Literacy in the United States." Ocean and Coastal Management 48: 97-114.

Watson, P. (2000). The Giant's Causeway and the North Antrim Coast, The O'Brien Press Ltd.

Yin, R., K. (1994). Case Study Research: Design and Methods. 2nd Edition, SAGE Publications Ltd.

Yin, R., K. (2003). Case Study Research: Design and Methods 3rd Edition, SAGE Publications.

Yin, R., K. (2003). Applications of Case Study Research 2nd Edition, SAGE Publications Ltd.

CHAPTER SIX

RESULTS AND INTERPRETATION OF THEMATIC CASE STUDIES

6.1 INTRODUCTION

This chapter presents the analysis and interpretation of the data collected through the community based thematic case study phase of research. The chapter begins with a presentation of the results observed through the Education thematic case study, as described in Chapter Five. This was a quantitative phase of data collection aimed at establishing current levels of student knowledge of issues facing the marine environment, and their perception of the facilities available to them for marine education. In addition to the results obtained through the student survey, the results of a short teachers' capacity survey are presented.

Following this, the observations made through analysis of the data collected during the Personal Attachment thematic case study are outlined. Given the mixed methods approach applied to the research project (Chapter Three), both qualitative and quantitative data were collected simultaneously in order to exhaustively examine the theme of personal attachment. The personal attachment thematic case study investigated public perception of the marine environment, sense of public awareness, concern and responsibility and the factors potentially influencing marine citizenship in a community. Quotes made by interviewees during the personal attachment case study are included in the text in italics.

6.2 RESULTS OF EDUCATION THEMATIC CASE STUDY

The education thematic case study assessed two aspects of the delivery of marine specific education in UK schools. The first component investigated student perception of marine education currently included in the UK national curriculum, the students' self assessed levels of awareness and concern, and the relationship

between these factors and indicators of the students' subjective and objective knowledge of the marine environment. The results of this survey are presented in Sections 6.2.1.

The second component of the education theme case study was the Teachers' Capacity survey, which examined teachers' opinion of their current level of capacity to deliver effective marine education. The survey allowed observations to be made regarding teacher perception on the potential impact of marine education on student behaviour and marine awareness. The results of this survey are outlined in Section 6.2.2.

6.2.1 SCHOOL SURVEY

A total of 121 surveys were returned fully completed by students in attendance at secondary level education facilities located in three of the case study sites (Isle of Arran, Poole Bay and Helsby¹⁹) as described in Section 5.8.2.5. Any incomplete surveys were excluded from the data analysis.

6.2.1.1. Student knowledge of marine related issues

The first section of the student survey assessed both the subjective and objective knowledge of the marine environment respectively through a term familiarity exercise and a short marine environment based quiz. Table 6.1-6.3 present the results of the Term Familiarity exercise. As shown in Table 6.1, students in Helsby were found to know and understand more terms on average (5.8) than students from schools in Poole Bay and Arran (4.3 and 4.6 respectively). The results presented in Table 6.1 did not indicate any form of significant correlation between students' knowledge of the marine environment and the location of the case study sites.

¹⁹ It should be noted that for the education thematic case study, Helsby was considered the least coastal of the case study sites and is therefore the more inland of the examples throughout Section 6.2.1.

	Arran	Poole Bay	Helsby
Know Term and Understand	4.6	4.3	5.8
Heard of but don't understand	2.9	3.2	2.5
Have never heard term	4.5	4.6	3.7

 Table 6.1: Average number of terms in each category of the term familiarity exercise.

Table 6.2 presents the average numbers of terms known and understood by students within the total sample and in each of the three case study sites, and shows that on average, students knew and understood 4.8 of the 12 terms. As indicated in Table 6.2, only 5 of the terms included in the term familiarity exercise were known and understood by over 50% of the participating students. The terms with which students were most familiar appeared to be climate change and sea level rise with 92.9% and 85.6%, respectively, of the total students' interviewed indicating that they 'knew and understood' these terms. The three terms students were least familiar with included two of the most commonly used terms currently related to marine and coastal management in the UK, namely Integrated Coastal Zone Management, Exclusive Economic Zone and the Marine Bill²⁰. These terms received a low level of familiarity with only 5.6%, 5.6% and 2.4% (respectively) of students indicating that they knew and understood these terms, with over 70% of students indicating that they had not heard of these terms as shown in Table 6.3.

In addition to marine related terms, students were required to indicate their familiarity with the term 'citizenship' given its inclusion at KS4/ S4 levels of education. A total of 51.3% of students indicated that they 'knew and understood' the term 'citizenship'. Considering this subject is a compulsory component of the national curriculum for this stage of the UK education programme, it was expected that the levels of student knowledge might have been higher in relation to this term. The potential implications of this observation on the development and further inculcation of marine citizenship will be explored further in Section 6.4.

²⁰ Now known as the Marine and Coastal Access Act, 2009
Variation between the school case study sites was evident in the results obtained through the education survey as shown in Table 6.2, further supporting the hypothesis that location and proximity to the marine environment influences levels of knowledge and other factors. For example, students from the Arran case site exhibited a higher level of familiarity with certain terms, including 'no-take zone', and 'ecosystem' with a 58.5% and 63.4% respectively.

Table 6.2: Indication of subjective knowledge based on students' termfamiliarity

Term Familiarity: Please identify which of these terms you are familiar with i.e. indicate by ticking the correct box whether they are terms that you know and understand, terms that you have heard of but do not know what they mean or if you have never heard of them.

	Percent (%) that know and understand term				
	Total	School 1 %	School 2	School 3%	
	%	Arran	%	Helsby	
			Poole Bay		
Ecosystem	46.8	63.4	35.7	42.1	
Biodiversity	24.0	9.8	2.4	68.4	
Coral bleaching	9.6	4.9	7.1	13.2	
Over fishing	55.6	63.4	38.1	73.7	
Climate change	92.9	92.7	90.5	100	
Sea level rise	85.6	82.9	83.3	94.7	
Coastal erosion	66.7	39	78.6	81.1	
Exclusive economic zone	5.6	7.3	7.1	2.6	
Integrated coastal zone					
management	5.6	2.4	7.1	10.5	
Marine Bill*	2.4	0	2.4	5.3	
Citizenship	51.6	39	54.8	73.7	
No-Take Zone	28.6	58.5	16.7	13.5	
Average number of	4.84				
terms known					
Number of students =					
121					
*Manine D:11 41	1	1.0 41		10 + 1	

*Marine Bill was the original name proposed for the UK Marine and Coastal Access Act (2009)

Table 6.3: Overall levels of student familiarity with marine terms (%response)

Term Familiarity: Please identify which of these terms you are familiar with i.e. indicate by ticking the correct box whether they are terms that you know and understand, terms that you have heard of but do not know what they mean or if you have never heard of them.

	Know and	Heard of but	Have never
	understand	do not	heard of Term
	term	understand	
		term	
Ecosystem	46.8	42.1	11.1
Biodiversity	24.0	32.8	43.2
Coral bleaching	9.6	39.2	52.8
Over fishing	55.6	16.7	27.8
Climate change	92.9	5.6	1.6
Sea level rise	85.6	12.8	1.6
Coastal erosion	66.7	15.1	18.3
Exclusive economic zone	5.6	22.2	72.2
Integrated coastal zone			
management	5.6	23.0	71.4
Marine Bill*	2.4	19.8	77.8
Citizenship	51.6	35.7	12.7
No-Take Zone	28.6	19.8	51.6
Average number of terms known Number of students = 121	4.84		
rumber of students 121			

*Marine Bill was the original name proposed for the UK Marine and Coastal Access Act (2009)

Table	e 6.4: Percentage of correct answers provided in the student quiz				
Marin	ne Quiz: Please underline the correct answer in each of the following sta	tements			
			Percent (%)	of correct ansv	vers
		Total %	Arran	Poole Bay	Helsby
а.	Ocean fisheries are affected by: climate change; red tides; over-	43.7	62.9	24.4	37.8
	fishing; all of the above				
b.	Most sea life: lives in the top 500ft of the oceans; lives on the sea	14.8	14.6	14.6	13.5
	floor; lives in the great ocean basins; is evenly dispersed through				
	the ocean depths				
C.	The movement of cold, nutrient rich water to the surface of the	62	70.7	24.5	69.4
	ocean is referred to as: upwelling; southern oscillation; trade winds;				
	reversal tide	46.6	19.5	41.5	70.3
d.	The transportation of sediment along the coast is know as: sediment				
	drift; across coast drift; longshore drift	28	39	12.2	21.6
e.	By-catch refers to: regular fish caught by nets; over-fishing; fish				
	that are harvested, but not sold or kept for personal use; a				
	climate phenomenon				
N = 1	21				
*corré	ect answers are marked in bold text				

As described in Chapter Five, the level of students' objective knowledge was assessed through the completion of a short marine quiz comprising of five statements pertaining to issues facing the marine environment. Table 6.3 presents the results of this quiz based on the percentage of students that provided the correct answer for each of the statements. Interpretation of results shows that less than half of the total student sample able to provide the correct answers to all five of the marine related statements. Only one question received over 50% of correct answers, with Question c in Table 6.3 answered correctly by the majority of the students (62%). The remaining four questions received 46% or less of participating students providing the correct answer, with only 14.8% able to correctly answer Question b.

A similar trend with regards to students' level of objective knowledge was observed in the individual case sites. Table 6.5 presents the average number of correct answers given by students overall in the marine quiz. The results gathered in this section of the survey indicate that, on average, the total student sample could provide a correct answer for 1.8 of the statements in Table 6.4. Participating students from the Arran case study school, on average, answered more questions correctly than those from the other two case study schools, with Arran students answering 2.2 of the questions. Examination of the results presented in Table 6.3 and 6.4 showed students' current level of objective knowledge of the marine environment to be low. As expected, differences were observed in the frequency of correct answers provided by students in each of the three schools, potentially adding weight to the hypothesis that location and proximity to the marine environment may have an influence on components of marine citizenship.

 Table 6.5: Average number of correct answers (out of 5 questions) given by students in the marine quiz section of the student survey

	All Schools	Arran	Poole Bay	Helsby
Average number of correct answers in marine quiz	1.8	2.2	1.1	2.1

In addition to assessing students' familiarity and knowledge of common terms and issues related to the marine environment, students' knowledge of groups and designations directly linked with the marine environment and its management in the UK was also established as an indicator of students' knowledge of the marine environment and its management. Table 6.5 shows that student knowledge of marine organisations was found to be quite low with participants in the survey, on average, indicating that they were familiar with less than half of the marine organisations listed in the survey. As shown in Table 6.6, students surveyed had, on average, heard of just 1.5 of the marine organisations listed in the school questionnaire (shown in Table 6.7). Examination of the differences in student knowledge of these organisations between the three case study schools, it was observed that students from Helsby averaged the highest level of knowledge with students knowing of 2.0 of the marine organisations named in the survey. Students based at the Poole Bay and Arran case sites exhibited a lower level of knowledge, indicating knowledge of just 1.2 and 1.4 of the named marine related organisations (respectively).

 Table 6.6: Average number of marine related organisations known by

 students

	All Schools	Arran	Poole Bay	Helsby
Average number of marine	1.5	1.4	1.2	2.0
related organisations known				

Table 6.7: Student knowledge of organisations commonly associated with the marine environment (represented through percentage of students selecting each organisation)

Marine Group	% Total	Arran	Poole	Helsby
	Students	%	Bay	
Marine Conservation Society (MCS)	41.3	48.5	44.8	58.3
Marine Stewardship Council (MSC)	5.8	6.1	6.9	8.3
Defra	6.6	9.1	3.4	11.1
Crown Estate	9.9	12.1	17.2	8.3
World Wildlife Fund (WWF)	65.3	84.8	62.1	91.7
UNESCO	17.4	15.2	31.0	19.4

Table 6.8 illustrates the level of students' knowledge of marine environmental designations common around the UK coastline and around the case study site locations. Given that at least one of these designations can be found in each of

the case study sites surveyed for this phase of the research, it was expected that participating students would have some knowledge of them. However, as shown in Table 6.9, on average students knew of just 1.6 designations out of a possible five options. Students attending the case study school on Arran exhibited the highest knowledge of these designations, with students indicating knowledge of 2 designations on average. In contrast, students from the Poole Bay based case site exhibited the lowest level of knowledge, knowing, on average, just 1.1 of the named designations.

 Table 6.8: Average number of marine designations known by students

	All Schools	Arran	Poole Bay	Helsby
Average number of marine environment designations known	1.6	2.0	1.1	1.6

Table 6.9:Student knowledge of marine environmental designations(represented through percentage of students selecting each designation)

Marine Environmental	% Total	%	%	%
Designations	students	Arran	Poole Bay	Helsby
Site of Special Scientific Interest	18.2	39.4	16.0	21.7
(SSSI)	28.9	39.4	44.0	47.8
Marine Protected Areas (MPA)	34.7	84.8	20.0	39.1
No Take Zone	22.3	30.3	20.0	52.2
Area of Natural Beauty (AONB)	3.3	9.1	4.0	0.0
RAMSAR	38.0	48.5	80.0	87.0
World Heritage Site				

In addition to assessing the level of student knowledge concerning the marine environment, students were also asked to specify how and where they obtained information about the marine environment and its management (Table 6.10). As expected, students indicated that television was the most common source of marine information with 71.4% of respondents selecting this option. A further 50.8% of students felt that they obtained marine related information through formal education at school. The discrepancies between this observation and the students' perception of whether they received sufficient marine information at school will be discussed further in Section 6.4. The third most common media (47.6%) source was the internet.

Most Common method of	%Total	Arran	Poole	Helsby
obtaining marine information	students		Bay	
School	50.8	36.6	45.2	73.0
Television Programmes	71.4	75.6	71.4	86.5
Internet	47.6	36.6	40.5	70.3
Peers	15.9	14.6	11.9	21.6
Radio	7.9	9.8	4.8	8.1
Newspapers or Magazines	31.7	41.5	21.4	35.1
Other	22.2	31.7	23.8	10.8
	1			

Table 6.	10: Mos	t common	method	through	which	marine	information	is
sourced	(represei	ited throug	gh percer	ntage of st	udents	selecting	g each source	e)

Following the marine practitioner interviews of phase one, it was expected that television would be identified as one of the more common source of marine information. Given this assumption, students were also requested to indicate which, if any, of the most frequently broadcast and popular marine related television programmes they had viewed. The selection of television programmes was guided by the marine practitioner telephone interview schedule and included "*Blue Planet*" and "*Planet Earth*". 25 of the students did not provide an answer for this question which was taken to suggest that these students had not viewed any of the television programmes named in the survey. Taking this into consideration, it was calculated that 81% of the total number of participating students had viewed at least one of the television programmes named in the exercise.

Television Programmes Viewed	Total	Arran	Poole	Helsby
	%		Bay	
Planet Earth	69	77.4	84.8	90.6
Blue Planet	43.7	54.8	51.5	59.4
Oceans	11.9	16.1	12.1	18.8
South Pacific*	7.94	6.5	24.2	15.6
Spring Watch	38.9	51.6	33.3	53.1
25 students did not provide an answer				
to this question				

Table 6.11: Television programmes related to the marine environment viewed by students (represented through percentage of students selecting each option)

*South Pacific is a marine environment documentary series based in the South Pacific Ocean that was broadcast while the initial student survey was conducted.

Student perception of where responsibility for the marine environment should lie was also assessed. Table 6.12 presents the results of this assessment with 67% of students selected the option of 'Everyone' while only 38.3% of students who took part in the survey indicated that the government should be responsible for the marine environment and its management. The category of 'Individuals' was selected by 19.2% of the total student respondents as being responsible for the marine environment. Both these observations suggest that a proportion of the students have an awareness of the individual and public role in maintaining the sustainability of the marine environment.

	% Total	Arran	Poole Bay	Helsby
	students			
Individuals	19.2	19.5	13.2	27.0
Non-government agencies	13.3	7.3	7.9	24.3
Government	38.3	31.7	23.7	59.5
Coastal groups	46.7	46.3	34.2	56.8
Everyone	67.5	63.4	78.9	62.2

Table 6.12: Percentage responsibility attributed to management groups (represented through percentage of students selecting each management group)

The level of students' direct involvement with the marine environment was examined through establishing the number of students who engaged in marine related hobbies. A majority of students (78%) did not take part in any hobby related to the marine environment. Of the 22% who confirmed their participation in hobbies linked to the marine environment, most frequently mentioned were fishing and swimming. Other activities mentioned by students included "*walking on the beach*", "*kayaking*", "*sailing*" and "*rowing*". Finally students were asked, based on their own experiences and in their opinion, if there was a threat to the marine environment. The majority (63.5%) of the total student sample indicated that, in their opinion, there was a threat to the marine environment. Twelve students refrained from answering this question, with one commenting that they did not "*know enough*" about the marine environment to comment on the matter.

6.2.2.2. Self Assessed Statements

The final section of the student education survey required students to complete a series of self-assessment based questions (presented in Table 6.13). Students were asked to rate their responses on a scale of 0 to 5 with 0 indicating the lowest level and 5 indicating the highest. Table 6.13 presents the results for each of these self assessed questions with the answers categorised as low (answers rated 0 and 1), moderate (answers rated 2 and 3) and high (answers rated 4 and 5). Given that the data collected in this component of the student survey was ordinal, certain guidelines had to be followed regarding student rating of their perceptions. For example, a student rating their awareness as 4 could not be said to have twice the awareness of a student who rated their awareness than the student giving their awareness a lower rating. These observations are now discussed in brief in relation to Table 6.13, with further interpretation presented in Section 6.4.

In addition, the potential relationship between location and gender and the components of marine citizenship being investigated through the education survey was examined. This was done through non-parametric analysis of variance using Kruskall-Wallis tests, the results of which are presented in Table 6.13.

to	Int	
dn	ifics	
ded	igni	
unc	ch S	
n re	wit	
bee	ools	
ave	sche	
6 h:	en	
۰) ۱	twe	
higl	be be	
nd	nces	
te a	erei	
era	diff	
pou	for	
W, I	les	
s lo	valı	
ns a	ity	
stio	abil	
ant	rob	
ed c	st pi	
sess	s tes	
fase	alli	
seli	≥-	
the	kal	
ing	Ins	
rat	SS K	
ents	note	d.
pnde	de	bol
of st	alue	l in
ge	P <	hteo
enta	e).	hlig
erce	olac	higl
3: P	al F	ips
6.13	cim	nsh
ble	e de	atio
Tal	ONE	rel:

relationships highlighted in bold.		1				I	
		Total	Arran	Poole Bay	Helsby	Location	Gender
		Students	(%)	(%)	(%) (%)	$\mathrm{DF} = 2$	DF = 1
		(%)N = 121	N=4	N=42	N=38		
To what extent is marine education covered in	LOW	56.2	70.7	65.9	31.6	H= 17.07	H = 8.78
school?	MODERATE	39.7	26.8	33.3	60.5	P < 0.001	P = 0.377
	HIGH	4.1	2.4	2.3	7.9		
		010	100	150	27.0	11 11 02	11 - 1 00
To what extent do you think you are provided	LUW	48.2	6.00	65.9 22.2	30.9 77.0	H= 14.95	H = 1.90
with enough information to make appropriate	MUDEKAIE	38.8 F	2.62	55.5 2 2	2.00	p = 0.001	P = 0.168
decisions about the marine environment?	HICH	C	4.4	C.7	6.1		
How informed do you think you are about the	LOW	52.9	53.7	66.7	36.9	H = 10.05	H = 4.44
marine environment?	MODERATE	42.1	39.0	31.0	57.9	P = 0.007	P = 0.035
	HIGH	5	7.3	5.5	5.2		
How much of an impact do you think your day-to-	TOW	33.9	31.7	42.9	26.3	H = 1.86	H = 0.41
day activities have on the marine environment?	MODERATE	51.2	48.8	47.6	57.9	P = 0.395	P = 0.521
	HIGH	14.9	19.5	9.5	15.8		
Haw would von rate vour awareness of the	LOW	355	29.2	54.8	21.0	H = 13 69	H = 3,29
marine environment?	MODERATE	54.5	63.4	28.6	60.5	P = 0.001	P = 0.070
	HIGH	9.9	7.3	16.7	18.4		
To what extent is the conservation of the marine	LOW	34.7	46.3	52.4	28.9	H = 7.56	H = 0.14
environment important to you?	MODERATE	35.5	34.1	30.1	42.1	P = 0.023	P = 0.708
2	HIGH	21.5	19.5	16.7	28.9		
	r 0111	Ţ	t c				
To what extent do you care about the marine environment?	LOW MODERATE	31.4 49.6	31.7 68 3	40.5 35 7	13.2 44 7	H = 5.90 P = 0.052	H = 0.11 P = 0.745
	HIGH	19.0	0	23.8	34.2	1	2

As shown by Table 6.13, gender was only found to have a statistically significant relationship with level of perceived informedness. Location was found to be significantly related to five out of the seven components investigated, further supporting the practitioner suggestion that location and proximity to the coast could influence expression of marine citizenship. The implications of these observations for the application of marine citizenship are discussed in Section 6.4 and Chapter Seven.

6.2.2.2.1. Marine education available in school

In order to ascertain student perception of marine education currently available through the national curriculum, student participants were requested to assess the level of marine specific education provided to them through school. Overall, slightly more than half (56.2%) of all the students interviewed rated the level of marine specific information available through formal education to be low. Given the importance attributed to location in marine citizenship, the relationship between student perception of school based marine education and location was examined and found to be statistically significant (p <0.001). Although it was expected that location would be a significant factor, closer examination of the results indicated that the relationship was not as expected. For example, students from Arran most commonly gave low ratings to marine education in school with over 70% of students scoring marine education through formal teaching as between 0 and 1. In contrast, students from Helsby were of the opinion that marine specific education was moderate with 60.5% of students scoring between 2 and 3 for this question.

6.2.2.2.2. Capacity for decision making

In addition to assessing the level of marine information included in formal classroom based teaching, students also rated their perception of the efficacy of this information in guiding their decisions about the marine environment. Students were asked to assess their perception of information availability in an attempt to evaluate student capacity for involvement in the marine environment. As shown in Table 6.13, 48.2% all students interviewed scored the availability of

marine specific information and its impact on appropriate decision making as being low (between 0 and 1). As with the section 6.2.2.2.1, location was found to have a significant relationship with capacity for decision making (H= 14.95, p= 0.001). Again, the results were not as expected with 65.9% of students from both of the coastal locations i.e. Arran and Poole Bay scoring this as low. In contrast, students from Helsby most commonly scored the availability of marine information as average with 55.2% rating it between 2 and 3.

6.2.2.2.3. Student knowledge of the marine environment

Students were asked to rate their own perception of their personal knowledge of the marine environment and the complexities of its management and conservation. Only 5% of the total number interviewed perceived their knowledge of the marine environment as being high (Table 6.13). Overall, 53% of students perceived their personal knowledge of the marine environment as being low scoring it between 0 and 1, but the percentage was lower for Helsby students (36.9% than for Arran and Poole Bay students (65.9%). A further 42% of all students interviewed rated their knowledge as moderate (scores between 2 and 3). Such poor levels of knowledge were mirrored in each of the three case sites with only 7.3%, 5.5% and 5.2% of students at Arran, Poole Bay and Helsby (respectively) scoring their personal knowledge as high (4 and 5).

Discrepancies between individuals' perception of their own behaviour and awareness, and their observed behaviour and awareness are often expected, particularly when collecting data through questionnaires and surveys (Gillham, 2000). Unexpectedly students did not over-estimate their own personal knowledge of the marine environment when comparing the self-assessment questions with the subjective and objective knowledge indicators. As shown in Table 6.13, students perceived their knowledge of the marine environment was low; an observation mirrored in the results obtained through the Term familiarity and marine quiz exercises included in the survey.

6.2.2.2.4. Links between society and the marine environment

Given the level of societal dependence on the marine environment, students perception of the impacts of their lifestyles and the choices they make (or those that are made on their behalf by parents or guardians) on the marine environment was evaluated. Table 6.13 shows that overall only 14.9% of students considered their impacts on the marine environment to be high while 51.2% of interviewed students rated the impact of everyday life as moderate (between 3 and 4). At each of the case study sites, less than 20% of students were found to consider the impact of their everyday lives on the marine environment to be high. Students from Helsby appeared to have most awareness of their potential impacts on the marine environment with only 26.3% of students rating their impact as low, in comparison to the low rating given by 31.7% and 42.9% of students at Arran and Poole Bay respectively. Unexpectedly, location was not found to have a statistically significant influence on students' awareness of their impacts on the marine environment (p = 0.395, Table 6.13). The implications of this lack of awareness of the links between the marine environment and society are further examined and discussed in Section 6.2.3.

6.2.2.2.5. Student awareness of marine issues

The majority of students interviewed perceived themselves as having a medium level of awareness of marine issues with 54% of interviewees rating their awareness between 3 and 4. However, collectively 35.5% of students rated their awareness of the marine environment as being low meaning that approximately 90% of students perceived their own awareness of the marine environment to be medium or lower. This trend was also observed in each of the three case sites with over 80% of students at each school indicating only a low to moderate level of awareness of the marine environment, but the percentage scoring low was lower at Helsby (21%) and highest at Poole Bay (54.8%), leading to significant differences between locations (p = 0.001). It was expected that students from Arran would exhibit higher levels of awareness given the proximity of the Lamlash marine conservation zone to their school, but the percentage of students

saying they had high awareness levels was lower at Arran (6.7%) than at Poole Bay (16.7%) and Helsby (18.4%).

6.2.2.2.6. Concern for the marine environment

Student concern and care for the marine environment was assessed through two questions; one focused on the perceived importance of marine conservation and the second addressed the level of personal concern for the marine environment by the students.

The marine practitioner telephone interviews conducted in the first phase of the research commonly referred to the younger generation of school goers as the key target audience at which to direct new initiatives aimed at improving societal behaviour towards the marine environment. However, this could only be achieved if the students themselves felt that the marine environment was worth being protected and conserved. Concern for the marine environment was assessed through students rating the importance of the conservation of the marine environment. Overall students rated the importance of the marine environment as being quite low with only 21.5% indicating that it was of 'high' importance to them. Conservation of the marine environment was least important to students from the Poole Bay case site with over 52% of students rating it between 0 and 1 compared to only 28.9% of student interviewees from the Helsby case site. The percentages of students rating their level of concern for the marine environment as high were lowest at Poole Bay (16.7%) and Arran (19.5%) and highest at Helsby (28.9%). Given the location of the study sites, the observed results are not as expected and indicate that the link between location and sense of concern towards the marine environment may not be as strong as previously implied by marine practitioners. The relationship between location and its potential influence on the factors of marine citizenship being investigated are explored further in Section 6.2.3.

Finally students were asked to personally assess their perception of their level of concern for the marine environment by asking them to rate how much they care about the marine environment. Overall students exhibited a moderate level of

concern for the marine environment with the majority of students rating their concern or level of care as either 2 or 3, although variations were evident between the different schools (Table 6.13).

6.2.2.3. Evaluation of the association between factors

Analysis of the data highlighted a number of potentially significant relationships. These included the potential influence of location on students' knowledge and awareness of the marine environment, the efficacy of the marine education currently included in the national curriculum and the relationship between students' actual knowledge and their perceived awareness and concern for the marine environment. Through statistical analysis, the significance of these potential relationships was investigated with regard to education and its potential role in the inculcation of marine citizenship. Table 6.14 presents a correlation matrix of the statistical analysis conducted on the student survey. Each of the correlations found to be statistically significant are highlighted in bold typeface. The relationships identified through this analysis are outlined below with the implications of these on the development of marine citizenship discussed further in Sections 6.4.

Each response factor was found to have at least one statistically significant correlation with another factor although some factors exhibited more frequent correlations than others. The frequency of statistically significant correlations between factors further emphasises the connectivity between the factors of marine citizenship highlighted throughout the research. For example, as shown in Table 6.14, awareness (Factor M) had statistically significant correlations with ten of the additional factors, location (Factor A) was found to have a significant relationship with eight of the other factors while students' knowledge and understanding of terms (Factor B) was found to have eleven significant correlations exhibiting p values of less than 0.05. The implications of Table 6.14 will be discussed in more detail in Section 6.4 and Chapter Seven.

res]	pondents) (statistic	ally sigr	iificant I	o values	given b	elow ead	- ch correl	lation hi	ghlight	ed in bol	(p			
V	Term Familiarity	V	в	С	D	E	Ъ	9	Н	I	ſ	K	Γ	W
В	Quiz results	$0.296 \\ 0.001^{*}$												
U	Groups known	0.526 < < 0.001	$0.109 \\ 0.236$											
D	Designations	0.472	0.269	0.471										
E	Perception of	0.229	-0.042	0.153	0.064									
	marine education in	0.012	0.652	0.097	0.490									
Ţ	schools Availability of	0.287	-0 059	0.220	0 099	0.724								
-	marine education	0.001	0.524	0.0016	0.282	<0.001								
J	Student perception	0.383	0.036	0.190	0.269	0.450	0.502							
	of marine	<0.001	0.698	0.038	0.003	<0.001	<0.001							
	information													
H	Perception of	0.036	0.048	-0.017	0.098	0.175	0.155	0.355						
	impacts	0.692	0.602	0.854	0.287	0.057	0.091	<0.001						
I	Consumer	0.175	0.019	0.182	0.175	0.260	0.364	0.393	0.303					
	behaviour	0.055	0.839	0.047	0.057	0.004	<0.001	<0.001	0.001					
ſ	Marine hobbies	-0.277	-0.148	-0.245	-0.472	0.122	-0.019	-0.203	-0.043	-0.115				
2	Darcantion of	0.350	/ 01.0	0.00/	100.02	01.102 0175	0110	0.020	0.040	0.209	0 126			
4	threats to the	<0.001	0.795	0.031	0.014	0.057	0.105	0.043	0.237	0.085	0.139			
	marine environment													
Γ	Marine awareness	0.381	0.085	0.210	0.312	0.306	0.304	0.549	0.251	0.268	-0.213	-0.379		
		<0.001	0.358	0.021	0.001	0.001	0.001	<0.001	0.006	0.003	0.019	<0.001		
Σ	Importance of	0.303	0.109	0.319	0.356	0.120	0.215	0.277	0.312	0.475	-0.291	-0.339	0.495	
	marine conservation	0.001	0.236	<0.001	<0.001	0.195	0.018	0.002	0.001	<0.001	0.001	<0.001	<0.001	
Z	Concern for the	0.308	0.172	0.241	0.362	0.147	0.231	0.297	0.306	0.420	-0.271	-0.313	0.569	0.803
	marine environment	100.0	0.001	0.00	100.0>	0.11U	110.0	100.0	100.0	100.0>	c00.0	100.0	100.0>	100.0>

Table 6.14: Spearman correlation matrix of the relationships observed through the student education survey (N = 121

232

6.2.2.4. Additional student comments

Following the completion of the survey, students were given the opportunity to make comments. Although the majority of students refrained from doing so, a small number wrote short statements at the end of the survey, which are now discussed in brief. The lack of student awareness of marine issues as observed through the survey (see Table 6.13) was clearly recognised by one student and was partially attributed to the fact that "marine problems are underreported, especially [considering] the pressing concern of global fish stock depletion" possibly implying a need for improved information availability. Comments like this suggested a desire among some students for an increase in marine specific education with one student stating that in their opinion "more education in this area would be beneficial and help to save the coasts" with another commenting that they "want to help but I don't know how".

6.2.3. TEACHERS CAPACITY SURVEY

Comments from participants in the teachers' capacity survey suggested that the current level of marine education in the national curriculum is considered to be 'very basic' with 'very little' included at KS4 and S4 level. One participant commented that the level of marine specific education improved at higher stages of education, giving examples of teaching students through visits to the coastal environment at A Level stage of education. In addition participants expressed a concern that 'there does not seem to be any focused work in the current curriculum' related to the marine environment and that 'more would be welcome'. As a result of this lack of inclusion of marine education in the national curriculum, teachers perceived student level of awareness to be 'low'. One teacher suggested that student awareness is 'about the same as other environmental issues' suggesting that overall environmental awareness among students is relatively low. A potential explanation for this was provided by the suggestion that marine education would not be perceived as being overly relevant to students unless '*[they] felt that it would have an impact on their wider life*'. Teachers perceived the currently 'very controlled curriculum' as being challenging to the successful delivery of marine education. Other examples of issues that the delivery of formal classroom based marine education included the concern that there is a '*lack of time and resources*' to allow teachers to include marine related topics in their teaching. Teachers suggested that enhancing the provision of marine specific education would be beneficial to students' future involvement with the marine environment, as it would improve '*their awareness and how they respond when presented with marine issues in the media*' although this was deemed to be dependent on if students '*felt it as relevant to their everyday lives*'.

Primarily, the teachers' capacity survey indicated that there is scope for further research in to the current capacity of teachers to deliver marine specific education within national curriculum guidelines at KS4/S4²¹. In order to improve student awareness of marine issues, participants recommended the expansion of the national curriculum to include marine based examples within the subjects taught currently. Participating teachers emphasised the need for more active involvement from students' parents coupled with better utilisation of materials and information provided through more informal education channels, for example, museums, television programmes and other media sources. The potential implications of the observations made through the teachers' capacity survey for the future delivery of marine specific education are addressed further in Sections 6.4 and in Chapter Seven.

6.2.4. Education thematic case study concluding comments

Throughout analysis of both phases of the education case study, a number of key areas were identified for discussion in the context of marine citizenship. The key areas are outlined below and will be discussed in Section 6.4 and in relation to the first phase of data collection in Chapter Seven.

- The observation that current levels of marine education included in the national curriculum are lacking.
- An observation that current levels of marine education in the UK are ineffective.

²¹ Explained in Chapter Five

- An identification of low levels of student knowledge and awareness of the marine environment and the issues facing it.
- An identification of informal education methods as the most common avenue through which students obtain their information about the marine environment.
- The identification of location as a significant factor in the context of marine education.

6.3 RESULTS OF PERSONAL ATTACHMENT THEMATIC CASE STUDY

The second thematic case study investigated the theme of personal attachment as identified through the marine practitioner interview (Chapter Four). As a broad category it was found to encompass numerous specific themes hypothesised to influence an individuals' sense of personal attachment to a location or environment. Through the theme of personal attachment, the influence of an individual's sense of place, cultural perception of the marine environment, links with everyday life, and finally the importance of a person's dependency on the marine environment on societal marine citizenship was examined.

As outlined in Chapter Five, an individual's sense of personal attachment was examined through community interviews at five locations in the UK. The interview was comprised of both quantitative and qualitative components, with the qualitative comments made by interviewees used to further examine and support the relationships identified through analysis of the quantitative data.

6.3.1. Analysis of Personal Attachment Case Study

As discussed in Section 5.3, given that both qualitative and quantitative data were collected through the personal attachment interviews, it was necessary to conduct different forms of analysis appropriate to the data in order to ensure complete analysis and interpretation.

Each of the interview questions included in this phase of data collection required the interviewee to assess their personal perception of a particular hypothesised component of marine citizenship, rating their answers on an ordinal scale of between 0 and 5, in the same manner as the final section of the education theme student survey. As described in relation to the student survey, during analysis of the data the ratings were grouped into low (0 and 1), medium (2 and 3) and high (4 and 5). Table 6.15 presents the collated data from both the total interviewee sample and the results of the individual case sites, represented through the percentage frequency of ratings given as an answer to each interview question. The initial observations provided by this data are outlined in Sections 6.3.2.-6.3.7 with further analysis of associations outlined in Section 6.3.8. Quotes obtained through the analysis of the qualitative data collected are included in italics throughout this section in support of observations made through the quantitative data analysis.

As discussed in Chapter Four, it was hypothesised that there would be potential relationship between location and the proposed factors of marine citizenship (Section 4.3.3). In order to establish the significance of these relationships, analysis of variance was conducted using a non-parametric Kruskall-Wallis test to examine the influence of case site location, socio-demographics and gender on the other factors assessed through the personal attachment interviews. The results of this analysis are presented alongside the data collected through the community interviews in Table 6.15 with significant relationships highlighted in bold type face.

Table 6.15: Results of the self-assessed personal attachment interviews. P value denotes Kruskall-Wallis test probability values for differences between schools with significant relationships highlighted in bold.

Personal Attachment Interview Questions		Total %	Poole Bay %	North Antrim Coast (Giants'	Birmingham (Inland) %	Arran (Lamlash Bay) %	Pembrokesh ire Coast (Milford Haven) %	KW * DF=4 Locatio n	KW DF= 6 Age group	KW DF= 1 Gender
				Causeway) %						
a) How would you rate your awareness	Low	17.5	19.4	28.2	14.9	13.4	6.9	H=12.16	H=28.28	H= 4.43,
of problems facing the marine	Moderate	57.5	63	46.2	59.6	57.7	38.2	p= 0.011	p <0.001	p =0.109
environment?	High	25	17.6	25.6	25.6	28.8	43.8			
b) How much do problems facing the	Low	9.5	15.7	12.9	2.1	5.7	0.0	H=8.36	H=10.72	H= 1.99
marine environment worry you?	Moderate	45.1	36.0	55.9	57.5	44.3	48.3	p=0.079	p =0.097	p =0.369
	High	45.5	48.1	33.3	40.4	50	51.7			
c) How important is the conservation of	Low	2.2	1.9	5.1	2.1	1.9	0.0	H=30.00	H=12.64	H = 2.97
the marine environment to you?	Moderate	31.6	47.2	35.9	21.2	17.3	10.3	P<0.001	p =0.049	P=0.227
	High	66.2	50.9	59	76.6	80.7	89.7			
d) To what extent do you think your	Low	37.1	54.7	12.9	19.2	30.7	44.8	H=36.96	H=4.56	H = 0.05
lifestyle impacts the marine	Moderate	53.8	41.7	74.3	61.7	59.6	48.2	P=0.001	p =0.602	p =0.977
environment?	High	9.09	3.7	12.8	19.4	9.6	6.9			
e) To what extent do you consider the	Low	36.4	56.5	30.8	27.7	13.5	24.1	H=50.04	H=9.93	H=4.84
marine environment when you buy	Moderate	41.8	34.3	43.6	57.5	48.1	31.0	P<0.001	P=0.128	P=0.089
food?	High	21.8	9.3	25.6	14.9	38.5	43.8			
f) To what extent do you consider the	Low	51.6	78.7	35.9	34	30.8	37.9	H=72.14	H=3.93	H=12.16
marine environment when you buy other	Moderate	30.9	14.8	51.3	47.9	38.5	20.6	P<0.001	P=0.686	P<0.002
products e.g. washing detergents etc?	High	17.5	6.5	12.9	17.0	30.8	41.3			

237

Personal Attachment Interview		Total %	Poole Bav	North Antrim	Birmingham (Inland) %	Arran (Lamlash Bav)	Pembrokeshi re Coast	KW * DF=4	KW DF=6	KW DF=1
Questions		5	%	Coast (Giants' Causeway)		%	(Milford Haven) %	Location	Age group	Gender
g) To what extent would you be	Low	17.5	33.3	⁄₀ 2.6	6.4	11.5	6.9	H=19.02	H=4.56	H=0.23
prepared to change your behaviour if it	Moderate	45.8	36.2	71.8	48.9	44.3	44.8	p< 0.001	P=0.602	P=0.890
would benefit the marine environment?	High	36.7	30.6	25.6	44.7	44.2	48.2			
h) To what extent would you say you	Low	0.4	0	2.6	0.0	0.0	0.0	H=18.24	H=8.79	H= 1.96
care about the marine environment?	Moderate	25.8	36.1	33.3	21.2	13.4	6.9	p< 0.001	p =0.186	p =0.376
	High	73.8	63.9	64.1	78.7	86.6	93.1			
i) How responsible do you feel for the	Low	19.6	38.9	25.7	12.7	11.5	0.0	H=18.24	H=5.48	H= 0.73
condition of the marine environment?	Moderate	45.8	45.4	43.6	57.4	46.2	31.0	p< 0.001	p =0.484	p =0.694
	High	30.9	15.8	30.7	29.8	42.3	0.69			
j) To what extent do you think the	Low	2.5	2.8	5.1	2.1	1.9	0.0	H = 5.12	H = 4.60	H= 1.17
government should be responsible for	Moderate	9.8	9.3	12.8	6.4	3.6	6.9	p= 0.275	p =0.595	p =0.558
the marine environment?	High	88.4	88	82	91.5	88.4	93.1			
k) To what extent do you think the	Low	1.1	1.8	0.0	2.1	0.0	0.0	H=14.28	H = 4.52	H= 3.85
public should be responsible for the	Moderate	28	42.6	25.7	21.2	17.3	6.9	p= 0.006	p =0.606	p =0.146
marine environment?	High	70.9	55.6	74.3	76.6	82.7	93.1			
I) How effective do you think current	Low	24	10.2	33.3	46.8	27.4	20.7	H=55.29	H=9.77	H =0.65
management of the marine environment	Moderate	53.1	47.3	61.5	48.9	54.9	6.89	p< 0.001	p = 0.135	p =0.724
is?	High	23.6	43.5	5.1	4.3	17.6	10.3			
m) To what extent would you say you	Low	11.6	13.9	18	8.5	9.6	3.4	H=32.47	H=4.68	H=3.22,
have a personal connection with the	Moderate	34.5	59.3	23.4	48.9	38.5	17.2	p< 0.001	P= 0.586	p= 0.200
marine environment?	High	42.9	26.9	48.7	42.6	57.9	79.3			

From Table 6.15, the following factors are shown to have a significant relationship with location: perceived awareness of the marine environment (p value = 0.011), perceived importance of marine conservation (p<0.001), perception of the impact of lifestyle on the marine environment (p<0.001), public willingness to change behaviour towards the marine environment (p=0.001), public sense of care towards the marine environment (p=0.001), interviewee sense of collective public responsibility (p=0.006), perception of the efficacy of current management strategies for the marine environment (p<0.001) and finally interviewees sense of personal attachment to the marine environment (p<0.001). The results further support the hypothesis proposed in the first phase of data collection that location has a significant influence on an individual's sense of marine environment. The implications of these relationships will be explored in more detail in Section 6.4 and in relation to overall research in Chapter Seven.

Given that age and gender are two of the major socio-demographic factors (Ruchter *et al.*, 2010; Ewert *et al.*, 2005), it was expected that there would be a relationship between these factors and the components of marine citizenship. As indicated by Table 6.15 gender was only seen to be significantly related to Question f, inferring that gender would not strongly influence marine citizenship. In contrast, age group was found to be significantly related to the factors of awareness (a) and concern (b and c). The implications of these observations in the facilitation of marine citizenship will be explored further in Section 6.4.

6.3.2. Public Perception of Marine Management and Governance

In order to establish how marine citizenship could be applied to marine management at a community level, it was necessary to evaluate public perception on the efficacy of current marine management strategies. As shown in Table 6.15, the majority of interviewees rated current management between 2 and 3, suggesting that management is satisfactory but is in need of improvement. As the thematic case study was investigated at different locations, there was the possibility that public perception of management would vary based on the

regional management approach at each site. The data presented in Table 6.15 indicates a potential relationship between public perception of management and location with 43.5% of Poole Bay case study interviewees rating the efficacy of marine management between 4 and 5. In comparison to this, the percentage of interviewees rating the efficacy of current marine management as high was less than 20% in each of the other case study sites. Content analysis of the qualitative data further implied the presence of these regional variations with one interviewee suggesting that managers are "working quite hard at it in some areas, for example, Cornwall". However, there was the perception that although there are attempts at "good management" of the marine environment "people don't understand certain impacts of their behaviour".

6.3.3. Public awareness of the marine environment

As seen in Table 6.15, only 17.5% of the total public interviewee sample rated their awareness of the marine environment as low (indicated through a rating of 0 or 1), with the majority (57.5%) of interviewees considering their level of awareness to be moderate, rating it as between 2 and 3. This trend was mirrored in each of the case study sites with most interviewees perceiving their own awareness as being approximately average (Poole Bay = 63%, North Ulster Coast = 46.2%, Birmingham = 59.6%, Arran = 57.7%, and Milford Haven = 38.2). Milford Haven was the only one of the sites where more interviewees rated their awareness as high with 43.8% scoring their personal awareness of marine issues as between 4 and 5.

Although the figures in Table 6.15 do not suggest a significant variation in levels of awareness based on location, qualitative analysis showed the potential influence of location and proximity to the marine environment to be frequently suggested by interviewees. A grid analysis of the qualitative data collected is presented in Appendix 13 and shows that 18% of interviewees commented on the issue of location in relation to individual and public awareness of the marine environment. As outlined in the literature review and through the marine practitioner interviews, location of the case studies was expected to result in variation in individuals' level of awareness of the marine environment. This

expected level of awareness appeared to be present among the interviewees, with one interviewee from Poole Bay stating that their level of awareness was "*not as high as it should be considering where I live*". The potential influence of proximity to the marine environment on individuals' sense of awareness towards the marine environment was furthered by a Birmingham interviewee who stated that "*there is a perception that what happens in Birmingham doesn't affect the marine environment because [the sea] is far away*".

A second perceived influence on individuals' awareness of the marine environment was that of an employment based or recreational dependency on the marine environment. For example, interviewees who indicated that their livelihood was dependent on the marine environment, for example individuals in the position of '*Marina Coordinate'*, '*fishermen*', or those involved in '*yachting*' were seen to rate their awareness of marine issues as high (between 4 and 5). One individual linked their involvement with the marine environment directly to their awareness stating that in their own opinion they had ''*fairly [high awareness] because [I] do salt water, fly fishing and surfing*''.

Although public knowledge and education were not directly assessed in the personal attachment case study, a potential link between education, knowledge and awareness was highlighted through the analysis of interviewee comments. Table 6.15 indicates that overall participants assessed their awareness of the marine environment as between moderate and high. Interviewees indicated that although they perceived themselves to be aware, they were "aware but relatively ignorant" and another saying that they "hear things on the news but they wouldn't be overly aware". The implications of links between marine education and knowledge on an individuals' level of awareness, and therefore their sense of marine citizenship, are explored further in Chapter 7 in relation to the results obtained in both Phase One (practitioner telephone interviews) and the education thematic case study.

6.3.4. Public Concern for the Marine Environment

Public concern for the marine environment was assessed through Questions b, c and f, each of which evaluated how worried interviewees were about marine issues they were aware of, how much importance they placed on the conservation of the marine environment and finally how much they personally cared about the marine environment and the issues facing it respectively.

As shown in Table 6.15 (Question b), over 90% of the total interviewees assessed their level of concern for the marine environment as either medium (45.1%) or high (45.5%). Less than 10% of interviewees perceived their levels of concern as low and although the qualitative data supported relatively high levels of concern, content analysis suggested that the actual level of concern may be considerably lower than the level perceived by the individuals themselves. As shown in the grid analysis (Appendix 13), varying levels of concern for the marine environment were expressed with interviewees stating that they "had never really thought about it" and that they "were not overly concerned". Others expressed varying levels of concern, acknowledging, "It's a big problem...it's not a little concern...everything can suffer" but as another pointed out society "[has] other things to think about". From the results presented in Table 6.15, there does not seem to be any significant variation in interviewees' level of concern for the marine environment.

The degree of importance interviewees placed on the conservation of the marine environment was also examined through Question c) with over 65% of the total interviewee sample indicating that marine conservation was of high importance to them with one interviewee stating that as "so much is [related] to the oceans, it seems to be that you ignore them at your peril". Despite most interviewees indicating that the conservation of the marine environment was highly important, analysis of the qualitative data suggested that to some people "[the marine environment] is something you don't think about". Interviewees' highlighted links between the marine and wider environment with one individual stating that conservation of the marine is "important as [the oceans] are the lungs of the earth". Although the majority of the overall sample rated marine

conservation as highly important, the percentage of interviewees giving it this rating was seen to vary between the case study sites. For example, 50.9% of interviewees from Poole Bay rated the conservation of the marine environment as highly important to them in comparison to 89.7% of the interviewees in Milford Haven.

Finally, Question f) asked interviewees to assess how much they personally cared about the marine environment and the issues facing it. As shown in Table 6.15 the majority of interviewees (73.8%) rated their level of care (or concern) for the marine environment to be high, between 4 and 5. Content analysis on the qualitative data showed that levels of public knowledge were referred to as an important factor when considering public concern for the marine environment. One interviewee stressed that although the marine environment "*[is] important but [I] don't know too much about it*". A lack of public capacity for involvement and a tendency to transfer responsibility for marine management elsewhere was commented upon by one interviewee who stated that "*[the marine environment is] important but other people can deal with it*".

6.3.5. Lifestyle choices, public behaviour and the marine environment

The marine practitioner telephone interviews conducted in Phase One highlighted the perception that the public are relatively unaware of the links between the marine environment and their everyday decision-making. Given this initial finding, the personal attachment interviews sought to establish general public perception of the impacts of their personal behavioural and lifestyle choices on the marine environment through Questions d, e and f. In addition to this, Question g required interviewees to evaluate how willing they would be to make changes to their lifestyle for the benefit of the marine environment. Overall, participants found it difficult to identify links between their lifestyles and the marine environment. Interviewees exhibited a lack of awareness of both the direct and indirect links between everyday living and the marine environment, with only 9% of interviewees suggesting that the impacts of their lifestyle on the marine environment would be high. Most frequently, interviewees considered themselves to have a moderate impact (rated between 2

and 3) on the marine environment. In contrast to the results presented in Table 6.15 regarding public awareness, concern and perception of current management strategies, the results of the self-assessment component of the interview do not show any clear variation based on location.

6.3.5.1 Awareness of everyday behavioural choices

The overall trend of interviewees rating their impact on the marine environment as relatively low was mirrored in each of the case study site locations. Although the data presented in Table 6.15 suggested interviewees were unaware of their impacts on the marine environment with between 30 and 54% of interviewees perceiving their impact as moderate, the fact that "society has a high impact on the environment. [For example], most of our sewage will end up in the sea" was acknowledged by one interviewee in Arran. It seemed that although a proportion of interviewees could see that they had an impact, they were not always clear on what these impacts were and how it was caused. This was particularly evident in a comment made by one interviewee who stated that their personal impact on the marine environment was low but as an afterthought added that "I eat seafood and have holidays there so there must be some impact". Interestingly some of the interviewees who indicated a perception that their impacts on the marine environment would be relatively low still acknowledged that everything has an impact "every time you flush the toilet, detergents, washing powder... I definitely have an impact but I am aware and conscious of it...it all has an impact".

Direct links between interviewees own behaviour and the condition of the marine environment seemed difficult to establish. Despite the expectation that coastal communities would exhibit a higher level of awareness or responsibility for their behaviour, interviewees in general were unaware of the impacts of their behaviour regardless of case study site location. Therefore, it is important to note that this inability to link everyday behaviour to the marine environment was not particular to Birmingham (the inland case study site) and was a trend that continued in the other case study sites. Some participants did realise that the impacts were "*higher than I think*" and that their lifestyle choices could potentially impact the marine environment "*without even knowing it*". It was common for people to immediately link the consumption of fish and seafood to this question and it seemed difficult for them to visualise the wider implications, although encouragingly, one who felt she had a low impact identified that it can be linked to other factors such as *"importation from ships"*. Qualitative analysis showed that a minority of interviewees expressed an awareness of more indirect impacts of society on the environment in general with one interviewee stating that they *"don't like food being flown in [as it causes] pollution"* implying a concern regarding the sourcing of goods being purchased.

6.3.5.2 Food based consumer behaviour

In order to assess actual behaviour in conjunction with perception of individual impact on the marine environment, interviewees' consumer behaviour was evaluated through Questions e and f. The first of the two questions addressed the potential implications for the marine environment related to the purchase of food. Based on the total results, only 21.8% of interviewees rated their consideration of these implications as high. When examining the individual case sites, there was a similar trend with the majority of interviewees claiming to give a moderate level of consideration to these issues when purchasing food (See Table 6.15 for further details). Milford Haven was the only case site in which a majority of interviewees (43.8%) rated their concern for this issue as being high (either 4 or 5). Content analysis conducted on the qualitative data collected for this question showed interviewees to most commonly relate consumer behaviour and the marine environment to the consumption of fish products as a food source. As a result, interviewees who did not eat fish were of the opinion that their lifestyle had a minimal, if any, impact on the marine environment, for example, stating that implications for the marine environment "generally wouldn't cross my mind".

In contrast, some interviewees indicated a level of behavioural awareness in the context of consumer choice with one interviewee commenting that they were "quite conscious of what I buy.... [I am] aware enough to check source, [for example] that Tuna is dolphin friendly" although there was a lack of trust in the information currently available as one interviewee commented "[you] don't

know how much is true on packaging and labels." Overall, participants felt that they would at least consider the implications, although this was considerably lower where people did not eat fish. The wider implications of food on the marine environment were only really considered by a minority of people with one clearly linking other factors stating that they "[didn't] eat fish but ships bringing food to the UK have an impact and I do eat imported food". The importance of ensuring the availability of information regarding these issues was commented upon by one interviewee who suggested that they "try [to] buy sustainable tuna but imagine still [I] buy fish that are at risk" implying a lack of knowledge regarding even something as widely reported as sustainable fisheries. Lack of knowledge was remarked upon by another interviewee who commented that "people don't consider it and should know more about it, [there] needs to be more awareness".

6.3.5.3. General consumer behaviour

The impact of lifestyle choices on the marine environment was investigated further with an assessment of the participating individuals' consumption of other goods, such as cleaning products. In general, the level of consideration interviewees gave to the consumption of other products was low with over 50% of interviewees giving scoring it between 0 and 1 on the self-assessment component of the interview. Interviewees who indicated a high concern indicated that the implications for the marine environment when purchasing household products for example were "something they thought of more often than the implications of food", while in contrast others admitted it was "not connected in my head...don't always think of the considerations..." The influence of education and awareness on individuals' behaviour was highlighted with a number of participants stating that the wider public would behave more responsibly "if they knew more". These comments implied a sense of public willingness to become more education about the marine environment and to behave in a marine environmentally friendly manner if they were felt sufficiently informed about the best choices to make.

Section 6.3.5.4 Public willingness to change

As well as perception of public behavioural choices, the personal attachment interview aimed to assess the level of public willingness to make behavioural changes to benefit the marine environment. As shown in Table 6.15, based on the self assessed component for this question, the majority of the total interviewee group rated their willingness to engage in behavioural change between 2 and 3. Only 17.5% indicated that their level of willingness to adapt for the benefit of marine conservation was low meaning that over 80% of the total sample was prepared to make changes to their current behaviour. This trend was further emphasised in the analysis of the qualitative data collected through which it was apparent that, overall, interviewees expressed a degree of willingness to make lifestyle changes for the benefit of the marine environment. Firstly, content analysis indicated a general consensus among interviewees across all of the locations was a desire to become more aware of the links between their behaviour and the impacts on marine environment. Interviewees' willingness to adapt their behaviour for the benefit of the marine environment was often dependent on a level of convenience, with one interviewee commenting that "if it greatly inconvenienced me then [I] wouldn't [change behaviour] but if it was reasonable then yes". A minority of interviewees indicated that they would not be willing to change their own behaviour, in spite of self proclaimed concern for the marine environment, with one interviewee stating that "it is down to the people who care and have nothing better to do with their lives, if that's how they choose to live their lives then that's their choice". Other interviewees, unsurprisingly, indicated a lack of willingness to make behavioural adaptations for the benefit of the marine environment. This was highlighted by one who commented that they "eat fish regularly and you can't change your diet because of the environment". Individuals interviewed were often of the opinion that changes to their own behaviour would be futile, with one interviewee stating that "what we do won't affect anything".

In addition to a lack of willingness to change, there was a sense that pro-marine environmental behaviour was perceived as being expensive and impractical if, for example, "*you're feeding a family of five in Glasgow on £200… It's not fair*

to expect someone in that situation to care as much". This interviewee expressed a valuable insight into the complexities associated with everyday life, particularly when socioeconomics are taken into consideration by commenting that "everyone has different pressures". The implications of socio-economics on marine citizenship are presented in Section 6.4 and Chapter Seven.

6.3.6. Public perception of responsibility for the Marine Environment

Questions i, j and k in Table 6.15 addressed public perception of responsibility as a factor of marine citizenship on different scales: individual responsibility, general public responsibility and governmental responsibility for the marine environment.

As presented in Table 6.15, over 75% of interviewees rated their sense of personal responsibility for the marine environment as being moderate or high. Overall, interviewees were of the opinion that they, as individuals, behaved in a highly responsible manner towards the marine environment and perceived themselves as having a level of responsibility for its sustainable management. However, given the comments made regarding other factors such as awareness and concern, it is possible that although individuals perceived themselves as being responsible, the results regarding public awareness and behaviour would suggest that currently the wider public do not possess the capacity to be actively and wholly responsible. This possibility is explored further in Section 6.4 and discussed in relation to the overall development and application of marine citizenship in Chapter Seven. In the individual case studies, similar observations were made with the exception of Milford Haven where 100% of the interviewees indicated a medium to high level of personal responsibility. This was in direct contrast with the results from the Poole Bay site where almost 39% of interviewees rated their personal responsibility for the marine environment as low.

The second responsibility themed question addressed the issue of governmental responsibility for the management and the conservation of the marine environment. The quantitative data allowed an insight into public perception of

the efficacy of current statutory governance of the marine environment and the role that the government should play in marine management. The data presented in Table 6.15 indicates that the majority of interviewees were of the opinion that government bodies should take on considerable responsibility for the marine environment, with over 80% of interviewees in both the total interviewee sample and each of the case sites rating government responsibility as between 4 and 5. In contrast to the other questions, as seen in Table 6.15, there was little if any variation in public perception at the locations of the case study sites. The data suggested an overall consensus that the government should play an active role in the management of the marine environment as "they are the ones that are supposed to be looking after the future environment", while also working to improve engagement of the wider public with marine management. The idea that the government "need to make people more aware so that they can take more responsibility [for the marine environment]" was strongly promoted by community interviewees. In addition to this, analysis of the qualitative data collected highlighted a public lack of trust in the government. The implications of this for marine citizenship and how it could be incorporated into current management strategies are addressed in Section 6.4.

With regards to the level of responsibility that should be attributed to the government for the marine environment, the interviewees were generally of the opinion that the government should be "more responsible than they are" and "should be more responsible for what they do [and the decisions they make]". This was based on the rationale that the government "are the custodians of the marine environment" and that they are "our representatives". Participants in the community interviews also expressed the opinion that the government has an obligation to raise public awareness through "more media awareness [and] education" as "the public need to be educated as to the consequences of actions and how the resource is used". This was further emphasised by an interviewee who suggested that there "is a need [for government] to educate people... they need to provide information...so that people can make the choice". Interviewees were of the opinion that the government should take more of an active role to inform the public about the marine environment and should "be able to say to people about... links to the marine environment".

The data collected through the supplementary comments made by interviewees suggested that the government is perceived with a lack of trust, with one interviewee directly stating that they "don't trust the government" which was supported by another's comment that they "don't think that the government respond to what peoples' thoughts are". This was manifested in the perception that although in public opinion government bodies should take more responsibility for marine management; politicians tend to work to "their own agenda", rather than for the good of the environment. One interviewee suggested that "[political] parties push what they think will get them into power and do not always base their information on accurate science". The lack of trust was further emphasised by an interviewee who stated that in their opinion, the government is "more interested in their profit... rather than conservation" and another stating that "instead of lining their own pockets they should be taking care of the environment". With respect to the efficacy of current marine management, interviewees felt that "[managers] have a far greater responsibility to educate the population with regard to our responsibility". One interviewee commented that "there is definitely a management issue" when it comes to the marine environment implying a sense of dissatisfaction with current management and governance.

As with the results for government responsibility, the results indicated that public responsibility should also be high with over 70% of the entire interview sample rating it between 4 and 5 (high). Variation in the results observed at different locations was observed with only 55% of Poole Bay interviewees rating public responsibility as high, while over 70% of interviewees at each of the other sites rated the level of responsibility the public should have as high. Over 90% of interviewees were of this opinion at the Milford Haven case site. Qualitative analysis of the data indicated that in general, the public were of the opinion that a collaborative relationship between government and the public would be beneficial to the management of the marine environment, as "government can only do so much". The grid analysis conducted for this question in Appendix 13 shows that the theme of cooperation between the public and government was mentioned 160 times throughout the personal attachment interviews. This

relationship and its potential role in the application of marine citizenship to marine management are discussed further in Section 6.4.

Content analysis conducted on the qualitative data indicated a relationship between education and responsibility for the marine environment. This was highlighted by interviewees, with one directly relating the two factors, commenting that *"if everyone had an education about [the] importance [of the marine environment] they would behave responsibly"*. Since education was not an assessed component of the personal attachment interviews, the relationship between interviewees' perception of their own awareness of the marine environment and the responsibility factors was further analysed. The results of this analysis are presented and explained in Section 6.4.

6.3.7. Personal attachment and marine citizenship

The final interview question aimed to directly evaluate individuals' sense of personal attachment and connection to the marine environment. As this is a relatively abstract concept, the qualitative data was used to establish how and why interviewees perceived themselves to be connected to the marine environment and why. In total, 42.9% of interviewees rated their personal attachment to the marine environment as high. Again more Milford Haven interviewees (79.3%) were found to perceive themselves as having a high level of personal attachment in comparison to the other case study sites (See Table 6.15).

As discussed in Chapter Five, the theme of personal attachment encompassed a wide variety of components including livelihood, dependency and cultural links to the marine environment. Qualitative analysis of the data collected through the personal attachment interviews provided further evidence for this, with a variety of reasons given by interviewees to explain their connection to the marine environment. In the case of Milford Haven, an area where a significant proportion of the community is dependent on the marine environment and its resources, interviewees were more aware of current management plans and legislation. For example, one interviewee directly commented on UK fisheries

policy stating that the "Common fisheries policy needs to be a lot stronger, government needs to be a lot less willing to compromise on marine policy".

These observations further suggested a relationship between the location of the case study sites and the level of connection to the marine environment perceived by the general public. Interviewees acknowledged that public knowledge and awareness of the marine environment has deteriorated with one interviewee commenting that the UK as a nation has *"lost the island culture"*. The responsibility of retaining the island traditions of the UK was highlighted by interviewees that statutory governance bodies *"are the custodians of the marine environment [and] it is their responsibility to manage our heritage"*.

6.3.8. Associations between personal attachment factors

Following the initial analysis conducted on the data, the data was further analysed to establish whether these observations were significant and which of the factors examined were likely to have the greatest impact on marine citizenship. The data collected were in ordered categories of responses, which could be coded as numerical levels (0-5).

Given the number of potential relationships expected within the data, correlation between the factors was analysed using Spearman's' correlations and the results of these tests are presented in Table 6.16. Table 6.16 presents a correlation matrix of the data with the statistically significant (p values less than 0.05) marked by bold typeface. Through the matrix in Table 6.16 a number of significant relationships are evident within the data. Questions 1-9 were found to exhibit statistically significant correlations. For the remainder of the questions, the level of correlations varied. For example, government responsibility (Q10 in Table 6.16) exhibits the least correlation with other factors, exhibiting significant relationships with only two factors, namely public awareness and sense of public responsibility. The factor of personal connection was found to exhibit significant correlations with nine of the other factors examined in the personal attachment interviews. The high proportion of statistically significant correlations identified throughout the data indicates the complex relationships between each of the
components of marine citizenship. The implications of this connectivity and interdependence will be discussed further in Chapter Seven.

Following the analysis presented in Sections 6.3.1-6.3.7, the correlations identified in Table 6.15 were summarised into key components through Principal Component Analysis (PCA). Although the data were only ordered categorical and not strictly quantitative variables, it was considered useful to conduct a PCA on the data to establish how the variables group into explanatory components (i.e. the hypothesised factors of marine citizenship), to most effectively define the inter-relationships between the variables and factors (Henderson and Seaby, 2008). Given that the variables were all of the same kind of data and of the same scales (i.e. 0-5) and potential variance, the PCA was undertaken using a covariance matrix. PCA is a method of summarising how variables are correlated and how they can be combined as one or more principal components. Variables given similar weightings in determining a particular component are positively correlated. In Figure 6.1, the PCA plots of two components, respondents with more similar responses will be plotted closer together. In addition, the relationship between responses and the two components illustrated by the plot presented in Figure 6.1 are shown by the length and direction of arrows of each response on the PCA plot. The placing of the response factor points in Figure 6.1 highlights the presence of four clear groups of response factors within the data which can be taken to represent (i) those aspects affecting individual behaviour, (ii) those regarding the general public and separately, the role of (iii) the government and (iv) managers. The identification of the clear groups seen in Figure 6.1 could potentially be applied to marine management, using the PCA to identify similar components that could be managed simultaneously through the development of holistic management strategies.



Figure 6.1: PCA plot of the relationships between factors investigated in the personal attachment thematic case study. The names of the spectral lines on the plot correspond to the questions asked in the Personal Attachment (PA) interview.

The closeness of the spectral lines in Figure 6.1 indicates a correlation between the factors in the components; for example, from Figure 6.1 scoring on marine awareness is strongly correlated with level of concern for the marine environment. Table 6.17 presents an example of the relationships presented by the PCA by examining the association between community responses regarding awareness and concern, indicating that individuals who expressed high levels of awareness also rated their level of concern as high.

Table 6.17: Association between community responses for awareness and concern which further explains the relationships identified by PCA (Fig 6.1)

			Aware	ness							
	Score	0	1	2	3	4	5				
		Number of individuals giving each score									
Concern	0	0	0	1	3	0	0				
	1	3	9	15	10	2	1				
	2	2	1	17	24	15	2				
	3	2	4	11	28	32	17				
	4	0	0	3	8	28	13				
	5	0	0	0	3	4	10				

The results of the PCA are presented in Table 6.18. This shows the interview responses to questions assessing individuals' personal sense of awareness, concern and impacts for the marine environment (Question variables (a)-(i) and (m) in Table 6.15) are correlated and formed the first and dominant principal component and accounted for 39.2% of the total variance observed in the data. The implications of this observation will be discussed further in Section 6.4 and in Chapter Seven.

Variable	PC1	PC2	PC3	PC4	PC5
(a) Awareness	0.247	0.324	-0.025	0.387	0.352
(b) Concern for marine	0.292	0.420	0.129	0.107	0.183
environment					
(c) Importance of marine	0.255	0.195	0.075	0.220	0.059
conservation					
(d) Lifestyle impacts	0.246	-0.099	0.256	-0.271	0.577
(e) Food implications	0.387	-0.306	-0.361	-0.104	0.117
(f)Product implications	0.418	-0.343	-0.546	0.091	0.045
(g) Lifestyle changes	0.295	-0.075	0.325	-0.463	-0.163
(h) Care for the marine	0.199	0.162	0.069	0.022	-0.117
environment					
(i) Personal responsibility	0.356	-0.071	0.230	-0.238	-0.371
(j) Government responsibility	-0.001	0.102	-0.163	0.190	-0.303
(k) Public responsibility	0.155	0.106	-0.131	0.164	-0.378
(I) Management efficacy	-0.103	0.573	-0.519	-0.602	0.055
(m) Personal attachment to	0.340	0.269	0.088	0.046	-0.275
marine environment					
Variance explained by PC	7.971	2.881	1.615	1.357	1.206
(Eigenvalue)					
% of total Variance	39.2	14.3	8	6.7	4.9
Cumulative % of total	39.2	53.4	61.4	68.1	74.1
variance					

Table 6.18: Weights given to each response variable following PCA conducted on the data variables (a)-(m) in Table 6.15.

6.3.9 Personal attachment thematic case study concluding comments

As sections 6.3.1-6.3.8 illustrate, a number of relationships have been identified through analysis of the personal attachment thematic case study interviews. These relationships have been grouped into a number of key categories and will be discussed further in Section 6.4.

• The potential influence of location on the expression of marine citizenship.

- The correlation between education, knowledge, awareness and concern and its implications for marine citizenship.
- The perception that there is a lack of public capacity for effective participation in a meaningful manner.
- An observed lack of public awareness of marine issues and the implications for marine citizenship.
- The need for collaborative management between the public and traditional marine management bodies expressed by interviewees.
- The role of socio-economic conditions as an enabler of marine citizenship.

6.4. DISCUSSION OF THEMATIC CASE STUDY RESULTS

Analysis of the data collected through the two thematic case studies highlighted a number of potentially important elements for discussion in the context of marine citizenship and the factors influencing its inculcation in UK society. The key elements of these observations are discussed briefly in Sections 6.4.1-6.4.6 and will be interpreted in relation to the first phase of data collection and the future application of marine citizenship in Chapter Seven.

The community interviews conducted through the personal attachment case study indicated a significant lack of confidence in current marine management strategies, particularly with regard to the ability of the government to generate effective management plans. There is a discrepancy between the perception of marine management held by marine practitioners (established in Phase One of data collection) and public perception (determined through the personal attachment and education case studies). The relationship between the two phases of research and the potential implications of this on the inculcation of marine citizenship and its role in marine management are discussed further in Chapter Seven.

6.4.1. Role of education in marine citizenship

Although only one of the thematic case studies directly investigated education as a theme, analysis of data collected on both themes further indicated education as a key element of marine citizenship. As discussed in earlier chapters, the relationship between education and an individuals' sense of awareness, concern, sensitivity and responsibility towards the environment has been long established (Barbas *et al.*, 2009; Steel *et al.*, 2005; Berkowitz *et al.*, 2005; Hawthorne and Alabaster, 1999; Strong, 1998; Kearns, 1995; Chawla, 1998; Tbilisi Convention, 1978).

Both the student and teacher components of the education thematic case study indicated that currently the UK national curriculum does not allow sufficient emphasis to be placed on these theories in relation to the marine environment and the challenges facing its conservation and management. Although earlier studies suggested that students generally exhibit a high level of environmental awareness (Goodwin et al., 2009) this was not the observation of the investigation in marine education. Jenkins (2003) suggests that the traditional formula of the natural world and its conservation as the basis for environmental education has undergone an evolution and has been expanded to encapsulate the emerging theories of sustainable development, environmental stewardship, environmental literacy and citizenship. The results obtained through the education thematic case study indicated a significant lack of knowledge about the marine environment and its resources among school leaving aged students. Although marine examples can be included in formal teaching, there is no mandatory inclusion of marine specific information in the current UK national curriculum (Castle et al., 2010). As a result, it was expected that, overall, participating students would be found to have a relatively low level of knowledge and awareness regarding the marine environment. With regards to marine environmental issues, students at KS4 level were observed to have a low level of both subjective and objective knowledge. In addition to this, in general, students did not perceive themselves to be sufficiently informed or aware of the marine environment and the issues facing it to make appropriate decisions regarding their behaviour towards the marine environment. The potential for younger

generations to impact societal behaviour has been well documented (Goodwin *et al.*, 2009; Strong, 1998) on the basis that increased environmental education will encourage higher rates of intergenerational discussion of environmental issues. However, the results from this survey, and earlier research, suggest that currently the younger generation in the UK are not currently equipped with the knowledge or capacity on which to base any behavioural adjustments (Castle *et al.*, 2010).

Although interviewees perceived themselves to be relatively aware of and concerned for the marine environment, and were of the opinion that overall their lifestyle and behavioural choices had minimal impact on the marine environment, observations made through content analysis of the qualitative data contradicted this. It was evident from the data that in general, the public are not fully aware of the societal impacts on the marine environment. Earlier research has suggested that a lack of awareness can often be due to a lack of connection and dependency on the marine environment, and a tendency for the public to neglect the lesser-known components of the marine environment (Novacek, 2008). The implications of this lack of awareness were seen to resonate through perceptions of public concern, responsibility and appropriate behaviour throughout the personal attachment thematic case study.

6.4.1.1. Potential education strategies

Various studies have recommended that enhancing public knowledge and awareness about the marine environment would increase public support for and involvement with the marine environment, its conservation and management (Castle *et al*, 2010; McKinley and Fletcher, 2010; Fletcher and Potts, 2007; Steel *et al*, 2005). Although research has shown children to rely heavily on formal school-based education, these are increasingly being enhanced through informal methods of educating and raising public awareness of environmental issues as a mechanism of supplementing formal education (Ruchter *et al.*, 2010; Fletcher *et al.*, 2009; Goodwin *et al.*, 2009; Haklay, 2002; Potts, 2000; Fortner, 1985). With regards to the marine environment, this has been championed as a key avenue through which to address public education and capacity issues related to the marine environment.

The role of various methods of informal education in order to provide marine specific information was emphasised in the observations of both of thematic case studies and is supported by earlier research (Ruchter et al., 2010; Barbas et al, 2009; Fortner, 1985). The results supported this; given observations that the majority of students did not feel their knowledge of the marine environment was primarily obtained through their formal education. As expected, the majority of students attributed their marine knowledge to alternative sources, most commonly citing television or the internet as the method in which they obtain information. This was indicated further throughout the student education survey in which the majority of students had watched at least one of the marine environment based television programmes listed in the survey. Although Steel et al. (2005) expressed concerns that inclusion of marine information in the wider media gave only an illusion of higher levels of public awareness, interviewees in the personal attachment case study stressed the importance of the media as a mechanism of delivering marine education, given that, as one interviewee suggesting, the public are "only aware of what's in the media [about the marine environment]".

Overall, the results of both thematic case studies supported and recommended further use of alternative methods, such as newspaper articles, internet based promotion of campaigns and broadcasting of marine based television programmes, to increase public knowledge and awareness of the marine environment. Previous research has also identified these methods as a key avenue through which to address the issues of public awareness raising (Novacek, 2008; Ruchter et al, 2010). However, interviewees expressed concerns, that in the past, the material produced through these avenues has been Although in general the results from both case studies poorly informed. championed the use of alternative approaches to providing education on a wide scale, some concerns were raised about the manner in which various media report on marine environmental issues. Interviewees in the personal attachment case study suggested that negative publicity about the marine environment appeared to be a "scare mongering [tactic] from the media and government" in a bid to engender certain changes to societal behaviour. Evidence from Novacek (2008) further supports the concerns expressed by the public regarding the media's treatment of marine citizenship, and the potential for conflicting messages may serve only to confuse the target audience. However, although the analysis of the data suggested a certain level of cynicism among the interviewees regarding current media coverage of the marine environment, it was predominantly seen as an effective method of delivering marine based information. Further to this, analysis of the interviews suggested that the use of television programmes, news bulletins, newspapers and the internet, have already aided in encouraging behavioural change among society. A number of interviewees commented that they ensure they only purchase *"things like dolphin friendly tuna, and don't buy cod"* suggesting that the media reporting of these particular issues has resonated with these individuals and caused them to make particular consumer choices.

Chapter Seven will further explore the potential role of both formal and informal education in the development of a more participatory form of marine governance through the promotion of marine citizenship. Common themes identified in both the first and second phase of data collection and the resulting conclusions regarding the place of enhanced public marine education will also be discussed in Chapter Seven.

6.4.2. Responsibility and management of the marine environment

Responsibility for the marine environment and its management was identified as an integral factor of marine citizenship through previous research on environmental citizenship (McKinley and Fletcher, 2010), with more specific evidence found through the marine practitioner interviews resulting in its inclusion in the personal attachment interviews.

6.4.2.1. Behaviour and the marine environment

One of the more common themes arising from the analysis of the personal attachment survey is that the general public are not aware of the impacts their decisions may have on the marine environment. Pro-environmental consumerism has been referred to by earlier research (Smith, 2005), as well as through the marine practitioner interviews as potentially the most realistic avenue through which the wider public can currently participate in the conservation of the marine environment. The results obtained through the personal attachment interviews suggested overall the public are increasingly conscious of the environmental implications of their purchases. However, the general consensus was that the public would be willing to make more marine environmentally favourable choices if they were more aware of the alternatives. The implications of this sense of willingness to alter societal behaviour in a manner that would benefit the marine environment have an important role to play in the inculcation and promotion of an effective concept of marine citizenship. This is explored in relation to the results from Phase One and discussed in further detail in Chapter Seven.

6.4.2.2. Collaborative management between government and the public

One of the more common themes arising from the analysis of the personal attachment interviews was that of an improved relationship between government bodies and the wider public with greater cooperation between the two. One interviewee commented that responsibility should lie with "both the government and the public" and that it "should be a compromise" although members of the public seemed to be of the opinion that "major policies need to come from the top". This collaborative relationship was commented on further with one interviewee commenting that the government "should pass legislation but it has to be made law for the public to listen" given that "they are the ones that tell us what to do". This emphasises a general observation that although the wider public appear to be willing to behave in a more marine environmentally favourable manner, its initiation would require a significant level of guidance through statutory governance (McKinley and Fletcher, 2010). The concept of collaborative management of the environment is becoming increasingly popular in environmental management with numerous studies advocating its success (Carnes et al., 1998; Juda, 1999; Kawabe, 2004). Research has shown that by improving the relationship between governments and wider society, public involvement in decision-making with regards to environmental resources and acceptance rates of management strategies and implementation is higher (Juda, 1999; Chopyak and Levesque, 2002; Irvin and Stansbury, 2004; Newman *et al.*, 2004).

In addition to the results obtained regarding public perception of the components of marine citizenship, the PCA plot presented in Figure 6.1 highlighted commonalities between the various factors. Interpretation of the similarities between the factors could potentially be used to aid development of appropriate mechanisms that management could undertake to promote marine citizenship. By identifying the similar components of marine citizenship, management strategies could be developed to address a particular group of factors simultaneously. For example, public perception of government responsibility and the efficacy of current management were found to be similar. It could be proposed that projects promoting public engagement in marine management would address the challenges posed by low public opinion of government and management efficacy in current marine management. The implications of how this could be utilised in marine management are explored further in Chapter Seven.

6.4.3. Socio-economics and marine citizenship

Although socio-economics were not directly examined through either of the case studies, data collected through the personal attachment case study prompted an evaluation of the role socio-economics in marine citizenship. Previous studies have found that socio-economic status can significantly impact an individual's level of environmental literacy (Santos *et al.*, 2005; Steel *et al.*, 2005; Ruchter *et al.*, 2010) and in addition is considered an integral component of the Hawthorne and Alabaster (1999) model. In addition to the financial concerns associated with enhanced environmental behaviour, community interviewees acknowledged that although they perceive the conservation of the marine environment as an important issue, people often have other issues they need to consider, such as raising a family. Through analysis of the evidence socio-economic status was found to have a potentially significant impact on the inculcation of marine citizenship, predominantly in relation to behavioural changes and the perceived

cost associated with these changes. The role of socio-economics in, and as a potential barrier to the facilitation of marine citizenship is outlined in more detail in Chapter Seven.

6.4.4. Influence of location on factors of marine citizenship

The relationship between location and related proximity to the marine environment and an individuals' sense of concern and ultimately, marine citizenship, was a recurring theme throughout the data analysis. One interviewee explained that movement away from the coast could result in a reduction in sense of care stating that they "*always used to be at the coast...but live inland here so [I have] no real connection*". There did not seem to be a definitive relationship between proximity to the coast and a sense of awareness and concern for the marine environment. Although literature suggested that an individual living in a coastal region would have a higher sense of awareness and concern, resulting in environmentally favourable behaviour (Steel *et. al.*, 2005), CCWs research into public perception of the marine environment contradicted this common assumption (Williams, 2008). The implications of this for the future of marine citizenship are discussed in Chapter Seven.

6.4.5. Socio-demographic factors

The current capacity of younger generations to participate in the future sustainable management of the marine environment was investigated through the education thematic case study. As discussed in Section 6.2.1, current levels of marine specific education are quite low with students exhibiting a low level of knowledge and awareness of marine terms and issues. However, in addition to these observations, analysis of the qualitative data collected through the personal attachment interviews emphasised the role of the younger generation in future marine management. One interviewee commented that "*the up and coming generation should be more passionate* [about the marine environment]". With regard to the public perception and concern for the wider environment, there is existing research to suggest that life-stage should be considered a contributing factor when assessing environmental behaviour (Williams, 2008). The results of

this research indicated that, statistically, age group is positively correlated with the three factors of awareness, concern and perception of individual impacts on the marine environment, as shown in Table 6.15. This relationship could have potential implication in the selection of target audiences for the promotion of marine citizenship and the direction approaches to enhance awareness should take. This will be explored further in the context of the wider research objectives in Chapter Seven.

In addition to life-stage, previous studies have suggested gender to have a significant influence on an individuals' perception of environmental issues in general (Hawthorne and Alabaster, 1999), inferring this factor could have an influence on future promotion of marine citizenship. This is further supported by Steel *et al.* (2005) who identify gender as being one of the most important variables throughout research on ocean literacy. These earlier studies provide sufficient rationale for examining the relationship between gender, levels of awareness and concern. A direct comparison can be made between two studies regarding whether males or females are considered to be more marine environmentally aware and concerned – whether they are more likely to have a sense of marine citizenship. However, in the case of this research, gender was not found to have a significant relationship with any of the other components evaluated through the personal attachment interviews.

6.4.6. Personal connection to the marine environment

As presented in Table 6.15, less than 50% of personal attachment interviewees indicated a high sense of personal connection with the marine environment. Earlier research has commonly linked enhanced public engagement with natural environments to a personal connection with a particular environment (Novacek, 2008). In addition, research has indicated that exposure to a natural environment should result in a higher level of knowledge and therefore, concern for the natural environment (Hawthorne and Alabaster, 1999; Ruchter *et al.*, 2010). Through content analysis, interviewees highlighted poor accessibility to the marine environment as a significant challenge for marine educators and managers.

The concept of cultural heritage influencing an individuals' sense of concern, awareness and responsibility towards the marine environment was a common theme throughout the data analysis. Firstly interviewees frequently commented on the importance of the UK's traditional heritage as "*an island*", implying that this inherited culture should theoretically breed an inherent sense of concern and responsibility for the marine environment. The theme of the UK as a community of island dwellers was a common reasoning for the need for enhanced public education and awareness-raising. This link to a traditional heritage could be viewed as a rationale on which to base the mandatory provision of marine information in formal education. The implications of the relationship between personal connection and the other components of marine citizenship are outlined in Chapter Seven.

In order to fully assess the potential role for marine citizenship in management of the marine environment, the observations made through the education and personal attachment thematic case studies are synthesised with the key observations of the telephone interview schedule. The implications of the synthesised results from both research phases are further interpreted and discussed in Chapter Seven.

6.5 SUMMARY

This second phase of data collection aimed to further investigate the two key themes identified through the telephone interview evaluation of marine practitioner perception of the potential role for marine citizenship. As discussed in Chapters Four and Five, these themes were identified as education and personal attachment. This chapter provided a detailed account of the results obtained through these thematic case studies and concluded with a brief discussion of the potential implications of these findings in relation to current literature.

Sections 6.2 outlined the results of the education based thematic case study which was aimed at assessing both student and teacher perception of current marine specific education included in formal teaching in the UK. Sections 6.2.1 and 6.2.2 presented the results of the school based student education survey conducted at three case study sites in different locations around the UK. Through a variety of exercises students' level of knowledge of the marine environment was assessed in a self-completion questionnaire. Overall, students were found to have a low level of knowledge and awareness of the marine environment, and it was observed that there was a heavy dependency on informal education methods, such as television and the internet, rather than formal education. In general, students were concerned for the marine environment, with the majority indicating that the marine environment is under threat. However, there was a clear lack of awareness and students were generally of the opinion that schoolbased marine education was not sufficiently informative. Section 6.2.3 presents the results of the teacher capacity survey which indicated that currently marine specific education is not effectively included in the delivery of the national curriculum. Participating teachers were of the opinion that there is a need for improved formal marine education in the UK and that an enhancement in education would serve to improve students' behaviour towards the marine environment, both while attending school and in the future.

Section 6.3 presented the results of the personal attachment thematic case study. This comprised of a survey at a variety of locations around the UK. Sections 6.3.3 - 6.3.7 outlines the observations made through the self-assessed component of the personal attachment community interviews. Overall, it was clear that there is a high degree of public concern for the marine environment, but interviewees did not perceive themselves to be significantly aware of issues facing the marine environment to know what the impact of their lifestyle might be, and how they could alter their behaviour to the benefit of the marine environment. Section 6.3.8 presented the results of statistical analysis of the data indicating a significant relationship between location and a number of the other factors being investigated through the personal attachment interviews. It was also found that life stage could potentially be an important component with regards to the mechanisms through which marine citizenship could be promoted. Following this, the overall observations made through the personal attachment thematic case study are then outlined in Section 6.3.9.

Section 6.4 presented a brief discussion of the main findings observed through the thematic case studies. Analysis of the data collected through the education and personal attachment case studies highlighted a number of key elements for discussion in the context of marine citizenship in communities.

Chapter Seven illustrates a synthesis of the results observed in both the marine practitioner and the thematic case study phases of research, discussing the key observations of both phases in relation to the inculcation of the concept of marine citizenship. The implications of the research findings for future marine management and the application of marine citizenship in this context are also explored in Chapter Seven.

REFERENCES

Barbas, T., A., Paraskevopoulos, S., and Stamou, A., G. (2009). "The effect of nature documentaries on students' environmental sensitivity: a case study." Learning, Media and Technology 34(1): 61-69.

Berkowitz, A., R., Ford, M., E., and Brewer, C., A. (2005). A framework for integrating ecological literacy, civics literacy, and environmental citizenship in environmental education.

Carnes, S. A., Schweitzer, M., Peelle, E. B., Wolfe, A.K., and Munro, J. F. (1998). "Measuring the success of public participation on environmental restoration and waste management activities in the U.S. Department of Energy." Technology in Society 20: 385-406.

Castle, Z., Fletcher, S., and McKinley, E. (2010). Coastal and marine education in schools: Constraints and opportunities created by the curriculum, schools and teachers in England. Ocean Yearbook. 24: 425-444.

Chawla, L. (1998). "Significant life experiences revisited: A Review of research on sources of environmental sensitivity." Journal of Environmental Education 29(3): 11.

Chopyak, J., and Levesque, P. (2002). "Public participation in science and technology decision making: trends for the future." Technology in Society 24: 155-166.

Ewert, A., Place, G., and Sibthorp, J. (2005) Early-life outdoor experiences and an individual's environmental attitudes. Leisure Sciences 27:225-239

Fletcher, S., and J. Potts (2007). "Ocean Citizenship: An Emergent Geographical Concept." Coastal Management 35: 1-14.

Fletcher, S., Potts, J., Heeps, C., and Pike, K. (2009). "Public awareness of marine environmental issues in the UK." Marine Policy 33: 370-375.

Fortner, R. W. (1985). "Relative effectiveness of classroom and documentary film presentations on marine mammals." Journal of Research in Science Teaching 22(2): 115-126.

Gillham, B. (2000). Developing a Questionnaire (1st edition), Continuum.

Haklay, M. (2002). "Public Environmental Information: understanding requirements and patterns of likely public use." Area 34(1): 17-28.

Hawthorne, M., and Alabaster, T. (1999). "Citizen 2000: Development of a model of environmental citizenship." Global Environmental Change 9: 25-43.

Henderson, P., and Seaby, R. (2008) A practical handbook for multivariate analysis. Pisces Conservation Ltd.

Irvin, R., A. and Stansbury, J. (2004). "Citizen Participation in Decision Making: Is it worth the effort?" Public Administration Review 64(1): 55-65.

Jenkins, E., W. (2003). "Environmental Education and the Public Understanding of Science." Frontiers in Ecology and the Environment 1(8): 437-443.

Kawabe, M. (2004). "The Evolution of Citizen Coastal Conservation Activities in Japan." Coastal Management 32: 389-404.

Kearns, A. (1995). "Active Citizenship and local governance: political and geographical dimensions." Political Geography 14(2): 155-175.

McKinley, E., and Fletcher, S. (2010). "Individual responsibility for the oceans? An evaluation of marine citizenship by UK marine practitioners." Ocean & Coastal Management. Newman, J., Barnes, M., Sullivan, H., and Knops, A. (2004). "Public participation and Collaborative Governance." Journal of Sociology and Politics 33(2): 203-223.

Novacek, M., J. (2008). "Engaging the public in biodiversity issues." Proceeding of National Academy of Sciences 105(1): 11571-11578.

Potts, J. (2000). "Recent developments at the National Maritime Museum, UK, and their potential for capacity building." Marine Policy 24: 79-81.

Ruchter, M., Klar, B., and Geiger, W. (2010). "Comparing the effects of mobile computers and traditional approaches in environmental education." Computers and Education 54: 1054-1067.

Santos, I. R., Friedrich, A., C., Wallner-Kersanach, M., and Fillmann, G. (2005). "Influence of socio-economic characteristics of beach users on litter generation." Ocean & Coastal Management 48: 742 -752.

Smith, G. (2005). "Green citizenship and the social economy." Environmental Politics 14(2): 273-289.

Steel, B., S., Smith, C., Opsommer, L., Curiel, S., and Warner-Steel, R. (2005). "Public Ocean Literacy in the United States." Ocean and Coastal Management 48: 97-114.

Strong, C. (1998). "The impact of environmental education on children's knowledge and awareness of environmental issues." Marketing Intelligence and Planning 16(6): 349-355.

UNESCO (1978). Final Report. Tbilisi, Intergovernmental Conference on Environmental Education.

Williams, L. B. R. (2008). Public attitudes towards the maritime environment around the Menai Strait and Conwy Bay. CCW Policy Research Report. CCW: 55.

CHAPTER SEVEN

SYNTHESIS AND DISCUSSION

7.1 INTRODUCTION

The review of the existing relevant literature, presented in Chapter Two, outlined the recent transition of marine management from a predominantly top-down, state driven process to the current attempts at bottom-up, holistic and participatory management approaches. Further to this, the literature established the increasingly integral role of citizenship in the context of environmental management. In addition, the literature review identified the components of environmental citizenship that could be applied to the marine environment, and highlighted gaps in additional factors expected to be included in a marine specific concept of citizenship.

The literature review identified a number of factors with the potential to influence the inculcation of a marine specific concept of citizenship and its application to the management of the marine environment. In order to ensure a comprehensive evaluation of these factors, two phases of data collection and analysis were undertaken to establish both practitioner and public perceptions of the concept of marine citizenship. Chapters Three and Four outlined the telephone interview investigation into marine practitioner perception of the role of the public in marine management and the potential application of marine citizenship in this context. The key observations of this first phase of data collection were further investigated through education and personal attachment thematic case studies, reported in Chapters Five and Six.

This chapter is structured into three main sections. First, it presents a synthesis of the results and observations made in Chapters Two, Four and Six, identifying the key influencing factors that affect marine citizenship in the UK (Section 7.2). The synthesis of these results identifies the key elements of marine citizenship

and examines its role in engaging the wider public in contemporary marine management.

The second part of the chapter (Section 7.3) builds on the synthesis of the results of the telephone interview and case study phases of research to guide the establishment of the first working definition and model of marine citizenship. Through the generation of this model, the gaps in current understanding regarding the role of the wider public and how a concept of marine citizenship could be applied to UK marine management are assessed.

Thirdly, it examines how the emergent concept of marine citizenship can be applied to the sustainable management of the marine environment in the UK (Section 7.4). The evaluation of the model's application to marine management will then allow a number of recommendations to be generated that will enable marine practitioners to identify the approach required to ensure an enhanced level of public awareness and concern for the marine environment, and a more involved citizenry. These recommendations are presented in Chapter Eight.

7.2 MARINE CITIZENSHIP: A SYNTHESIS OF PRACTITIONER AND PUBLIC PERCEPTIONS

As discussed in Chapters Two, Four and Six the research has identified a diverse range of components expected to have a varying level of influence on marine citizenship and its wider application in marine management. Sections 7.2.1-7.2.9 present a synthesis of the results identifying the key elements of marine citizenship. Although the results did not suggest any form of hierarchical order for the factors of marine citizenship, the order of the sections is set out so as to highlight the natural progression of the factors' inclusion in marine citizenship.

7.2.1. Role of the public in marine management

Central to the success of marine citizenship is the role of the public in marine management. Recent moves towards more participatory forms of environmental management (Section 2.2) suggest that the traditional concept of 'stakeholders' is

undergoing an evolution, encompassing not only end-users and direct stakeholders, but members of the public (UNECE, 1998; Rowe and Frewer, 2000; Berkes, 2004). In addition to this, earlier research indicates that the involvement of the public increases the success rate of marine management (Van Dyke, 1996; Dalton, 2005; Storrier and McGlashan, 2006). In this context, one of the key objectives of the research was to establish how marine citizenship could be applied to the role of the public in marine management. Although, currently there are no guidelines for the inclusion of the wider public in marine management (Salthouse *et al.*, 2010), practitioner perceptions suggested that an enhanced public presence is desirable.

In spite of the desire for enhanced public engagement, concerns were expressed that the wider public lack the capacity to be meaningfully involved in management processes (See Chapter Four, Section 4.2.2). Practitioner concerns regarding public capacity for involvement were justified through the case studies with public knowledge and awareness of the marine environment to be relatively low (Chapter Six). Despite community interviewees expressing a lack of capacity for involvement relating to low levels of knowledge and awareness, individuals expressed a desire for enhanced engagement and a willingness to adjust their behaviour if appropriate measures were taken to make it possible. These observations suggested that the promotion of marine citizenship would first need to address the currently low levels of public knowledge and awareness of the marine environment in order to affect the societal behavioural changes promoted by marine citizenship. The implications tied to lack of capacity and the symptomatically low levels of public awareness, knowledge, concern and sense of public and societal responsibility (Section 6.4) are discussed further in Section 7.3.

7.2.2. Role of education in marine citizenship

Correlations between the main components of environmental citizenship models, including education and public awareness, concern, sense of responsibility and behavioural choices have been well documented (Tbilisi convention, 1978; Sear and Hughes, 1996; Hawthorne and Alabaster, 1999; Haklay, 2002; Alessa *et al.*,

2003; Berkowitz *et al.*, 2005; Steel *et al.*, 2005; Ananda. 2007; Teisl *et al.*, 2008; Castle *et al.*, 2010; McKinley and Fletcher, 2010). Given that education was identified as an integral component of both environmental citizenship (Section 2.6.1) and marine citizenship (Chapters Four and Six), the multiple roles of education in marine citizenship were apparent.

Observations through both phases of data collection highlighted an overall consensus that improved marine education for the public is integral to the success of future marine management. This is supported by earlier research, which indicated that there has been a need for improved marine education in the UK for some time (Castle et al., 2010; McKinley and Fletcher, 2010). Marine practitioners identified younger generations as the key target audience at which efforts to encourage marine citizenship should be directed. The primary challenge to this approach, expressed by practitioners, was the concern that current marine-education included in UK formal teaching is not sufficient to engender an informed, responsible behaviour. This was further supported by the results of the education thematic case study (Chapter Six, Section 6.2.1) which illustrated a lack of knowledge and awareness among school-leaving age students at a variety of UK locations. Considering the relationship between environmental education and concern, awareness and behaviour are well documented, the low levels of these factors exhibited by students and teachers in the education thematic case study (Section 6.4) suggested that current marine education is not adequate. However, it was commented that no more could be facilitated due to a somewhat restricted curriculum and a lack of resources. This observation was supported by earlier research suggesting that levels of marine specific education in the UK are insufficient and in need of improvement in order to engender effective levels of awareness and responsibility among students (Castle et al., 2010).

As discussed in Section 4.4.1, and suggested by earlier research (Hay and Foley, 1998; Steel *et al.* 2005; McKinley and Fletcher, 2010) both formal and informal education techniques have an important role to play with regard to individuals' sense of responsibility for the marine environment. Given that school children were identified as the target audience for promoting a sense of marine

citizenship, gaps in current school based marine education need to be addressed. In order for formal marine education to have the desired effect of producing future generations with the capacity to be fully involved in the management of the marine environment, improvements need to be made to the level of mandatory marine specific information included in formal teaching. There are three core areas within the current national curriculum through which marine education could be taught at a higher level – geography, citizenship and science (Fletcher and Potts, 2007; Castle et al., 2010). Inclusion of more marine specific information within these three subject areas would allow all aspects of the marine environment to be covered including the ecological, social and economic issues. There have already been attempts to improve the level of marine specific education included in formal teaching; for example, the National Maritime Museum developed a classroom based pack which was aimed at developing students' levels of marine awareness (Fletcher et al., 2006). Delivery of this form of education would ensure students were provided with the information required to produce a marine aware society, capable of making appropriate decisions regarding the marine environment.

Although research has shown a heavy reliance on formal school-based education, these are increasingly being enhanced through informal methods of educating and raising public awareness of environmental issues as a mechanism of supplementing formal education (Fortner, 1985; Potts, 2000; Haklay, 2002; Fletcher *et al.*, 2009; Goodwin *et al.*, 2009; Ruchter *et al.*, 2010). In addition to formal techniques, the role of informal education techniques has been identified as an increasing presence in the delivery of environmental education to the wider public through environmental television programmes, museum displays, community awareness projects and the internet (Barbas *et al.*, 2009; Fletcher and Potts, 2007; Steel *et al.*, 2005). A number of marine environmental teaching method in either a home or classroom based setting, as suggested by Barbas *et al.* (2009).

Both practitioner and community interviewees indicated and supported theories that enhanced marine education would engender more favourable individual and collective behaviour towards the marine environment (as suggested by Tytler et al. 2001). Practitioners' perception that enhanced education would induce more favourable societal behaviour towards the marine environment (discussed in Section 4.4.1) was further supported by observations made through the personal attachment case study (Chapter Six). One clear discrepancy between practitioner and public perceptions related to the perception of information available to the wider public regarding the marine environment. Practitioners were generally of the opinion that the information required to guide appropriate decision-making was widely available and accessible; an opinion widely disputed by comments made by members of the public in the personal attachment interviews. Results of the personal attachment survey suggested that, in general, the public are not aware of how or where to access information pertaining to marine issues and feel incapable of making environmentally favourable decisions. This was further supported by observations, discussed in Section 6.3.3, public concern for the marine environment limited by a lack of public knowledge and awareness regarding issues and potential solutions.

7.2.3. Influence of location

Although location was often found to have a statistically significant influence on individuals' knowledge, awareness and concern for the marine environment, content analysis of the qualitative data showed the relationship was not as explicit as expected. This suggested that education and awareness-raising should not focus solely on coastal communities. Further to this, taking the traditional island culture of the UK into consideration, promotion of marine education programmes should not, therefore, be focused solely in coastal areas, but should be disseminated nationwide. One of the key observations of this research was a lack of public knowledge and awareness of social dependence on marine derived goods and services, and the importance of marine ecosystem services to both coastal and inland areas. This suggests that the potential for marine citizenship extends beyond coastal communities and therefore efforts should be made to engender nationwide citizenship towards the marine environment. The observations made in this study are supported by earlier research by Williams (2008) who indicated that inland communities frequently, somewhat unexpectedly, exhibit a higher sense of awareness and concern towards the marine environment than their coastal counterparts who often take it for granted.

7.2.4. Personal connection to the marine environment

Individual's sense of care and responsibility towards the marine environment was identified as a contributing factor in relation to individual behaviour towards the marine environment. Sense of connection to a particular environment has been well established as having a role in invoking responsible behaviour towards an environment (Burley et al., 2007). Lack of connection to the marine environment was observed through both phases of interviews in this research. In particular, practitioners expressed concerns that the UK appeared to have lost its island culture and that this has engendered a lack of societal respect and responsibility towards the marine environment. Practitioners identified an expectation that cultural links could engender a degree of individual and/ or collective social consideration for the marine environment. Given this, practitioners were of the opinion that members of coastal communities would exhibit a greater personal connection with the marine environment than those from inland regions, due to an expected higher level of dependency on marine derived goods and services. This connection between coastal communities and their environments has also been documented in earlier research (Williams, 2008; Davis and Wagner, 2006).

In spite of the traditional island culture of the UK commented upon by both practitioner and a small proportion of the case study interviewees, the majority of interviewees (58.1%) in the case study phase did not express high levels²² of personal attachment to the marine environment. A small number of individuals indicated a generational connection to their local marine environment due to a familial history in fishing, but this was not a common theme. A connection of this type was mentioned most frequently at the Milford Haven case study site, an area where marine derived industries are directly responsible for a high proportion of employment in the area. Previous research has indicated that

²² Levels of factors were identified through interviewees scoring each factor between 0 and 5. Low = 0 and 1, moderate = 2 and 3, and high = 4 and 5. (See Chapters Five and Six)

individuals lacking in attachment to an environment will perceive it in a different manner to those with a degree of connection (Twigger-Ross and Uzzell, 1996; Stedman, 2002; Williams, 2008; Ruchter *et al.*, 2010). This sentiment was echoed in the personal attachment case study, where sense of connection was most commonly linked to exposure to the marine environment through residence in coastal areas, holidays and/or recreational experiences. This was particularly the case when evaluating public perception of the level of societal impact on the marine environment where responses suggested that perceived degree of impact was significantly related to whether the individual lived by the coast.

The case study surveys did suggest a differentiation between those dependent on the coast for their livelihood, such as the fishing community, and those who view it as purely recreational. Participants felt that their awareness would be higher if they were directly impacted by the condition of the marine environment. As this study has shown, UK society has difficulties making links between their lifestyles on the condition of the marine environment. However, there was a recognition that this could potentially be different for communities who have a higher dependency on the marine environment.

7.2.5. Behaviour

One of the primary challenges to marine citizenship identified through synthesis of the data was a clear lack of public understanding regarding the impacts of their individual, and cumulative, behaviour on the marine environment. In general, practitioner interviewees were of the consensus that the majority of the UK public are unaware of the impacts of everyday life on the marine environment (Chapter Five). In addition to this, a lack of understanding and awareness of promarine environmental behavioural choices was also expected from the public rendering the availability of marine environmentally friendly products ineffective. This lack of public understanding and awareness of behavioural impacts expected by practitioner interviewees was predominantly mirrored by the observations made through the thematic case studies (Section 6.3). For example, results indicated that less than 10% of interviewees felt that their lifestyle had an impact on the marine environment, and most commonly related

impacts to the consumption of seafood products. As stated by Gillham (2000), there are often discrepancies between what people say and how they actually behave on a day to day basis. In this study this was evident in the discrepancies between the level of personal concern individuals assigned to themselves and their behaviour. In general, interviewees considered themselves to be quite well informed about issues facing the marine environment (only 17.5% indicated low levels²³ of awareness), but when pressed felt their lifestyle had no impact on the marine environment. The level of public willingness to adopt behaviour modifications into their everyday lives was considered key to the success of marine citizenship by practitioners (section 4.3.13).

Although the personal attachment interviews inferred a degree of willingness to change (Section 6.3.5), it was indicated that this would be subject to the level of convenience associated with the change. Earlier research supports this observation finding public perception of convenience relating to a particular behavioural choice to be implicated as an integral component of an individual's decision-making process (Teisl *et al.*, 2008; Hume, 2010). Although analysis of the practitioner interviews identified consumer behaviour as a relatively minor theme, it was identified as the most probable mechanism through which individuals could express a sense of marine citizenship. This assumption was further supported by observations made through the thematic case studies with interviewees most commonly linking their impact on the marine environment with their purchasing habits, in particular the consumption of fish (Section 6.3.5.2).

Given the level of importance attributed to consumer behaviour by both groups of interviewees, it can be assumed that it is an area that should be addressed by efforts to promote marine citizenship. Ecolabelling²⁴ and provision of sourcing information has been implicated in improving the levels of consumption of sustainable marine products (Jaffry *et al.*, 2004). Evidence of this approach is visible through campaigns by organisations such as the Marine Stewardship

²³ Levels of factors were identified through interviewees scoring each factor between 0 and 5. Low = 0 and 1, moderate = 2 and 3, and high = 4 and 5. (See Chapters Five and Six) ²⁴ Ecolabelling is a mechanism of providing product information to the consumer including

²⁴ Ecolabelling is a mechanism of providing product information to the consumer including where their seafood was sourced, how it was caught and the species.

Council $(MSC)^{25}$ whose eco-labelling scheme has been adopted by a number of large supermarket chains within the UK and the U.S.A (Wessells *et al.*, 1999; Jaffry *et al*, 2004) to facilitate sustainable consumer choice. The efficacy of media campaigns and ecolabelling marine derived products was recognised through the personal attachment interviews with a number of interviewees commenting that their consumption of 'dolphin friendly tuna' had increased in response to media reports and labelling campaigns (Section 6.3.5.2).

In addition to the factors contributing to public sense of willingness regarding behavioural changes, it was suggested that a proportion of society will always lack concern for the marine environment. Therefore resistance to suggested lifestyle modifications will remain a challenge to marine management. However, analysis of the personal attachment survey expressed public willingness to change behaviour. In order to facilitate these behavioural changes, it was suggested by both groups of interviewees that more marine information is required. The connection between education and behaviour is supported by earlier research suggesting that accurate information is central to the challenge of altering societal behaviour towards the marine environment (Steel *et al.*, 2005; Mamouni Limnios *et. al.*, 2009; McKinley and Fletcher, 2010). Considering this research and supporting studies, it can be inferred that in order to enable an expression of marine citizenship; a concerted effort to make marine favourable choices through accessible to a wider audience is required.

7.2.6. Public participation in marine management

Practitioner consensus that the majority of society would not be sufficiently equipped to participate meaningfully in current marine management strategies was mirrored in the observations made through case study interviews. Personal attachment interviewees stressed the importance of statutory governance bodies in the delivery of guidance and recommendations to encourage appropriate

²⁵ Marine Stewardship Council (MSC) was founded in 1996 with the aim of providing a coherent international labelling system for seafood products on the global market. Only accredited sustainable fisheries can place the MSC ecolabel on their products and a number of major supermarkets stock only certified seafood products. Other ecolabelling schemes for marine products are employed by Marine Conservation Society (MCS), World Wildlife Fund (WWF) and others.

behavioural choices. As stated by Rees *et al.* (2010) a long term commitment is required from the government in order to work towards the objectives set out by sustainable marine management.

One of the key benefits expected to be associated with the inculcation of marine citizenship is the successful implementation of more collaborative management of the marine environment. Both phases of data collection highlighted a desire among practitioners and the public for a more inclusive form of UK marine management. Practitioners were of the opinion that enhanced levels of public participation would engender a greater sense of responsibility towards the marine environment. Although the majority of interviewees promoted collaborative management between governments and society, it was acknowledged that this could be difficult to achieve owing to the current lack of public capacity for involvement in marine management. As a means of addressing this, research has found that learning through participation has been well documented as having positive impacts on individuals' behavioural choices (Hawthorne and Alabaster, 1999; UNECE, 1998). Participation is expected to encourage a higher level of public responsibility and involvement and as such is a key feature of Agenda 21 (Delgado and Strand, 2010). From the results obtained through both phases of data collection, it can be inferred that the heightened sense of public responsibility and awareness expected to coincide with educational promotion could encourage a level of marine citizenship among UK society.

7.2.7. Socio-economic factors

Research conducted on general environmental behaviour has indicated that socio-economic status and, in particular, financial constraints, can have an influence on the behavioural choices an individual makes (Hawthorne and Alabaster, 1999; Santos *et al.*, 2005; Steel *et al.*, 2005; Teisl *et al.*, 2008). The role of socio-economic status as an enabling factor in the expression of marine citizenship will be outlined in Section 7.3.

Although the relationship between socio-economic status and marine citizenship was not directly investigated, its importance was implicated throughout the research. Content analysis on both phases of the research suggested that socioeconomic status can have an influence on the priority given to the marine environment by an individual. In addition, results suggested that socio-economic status could have an influence on an individuals' capacity to engage in marine management and to express marine citizenship behaviours. Further to this, as both groups of interviewees were of the opinion that individuals could have the most impact through their consumer choices, the cost of sustainable and 'environmentally friendly' products were of particular concern, particularly in light of the current economic crisis²⁶ (Section 6.4). Early research had shown that consumers are less likely to select an environmentally friendly product over another version of the same product if they are required to pay a premium on price (Wessells *et al.*, 1999) although this has been found not to be the case in later work (Loureiro *et al.*, 2002; Brecard *et al.*, 2009).

The relationship between education and socio-economic factors was alluded to during the practitioner telephone interviews. Practitioner interviewees expressed an assumption that those individuals with a higher socio-economic status would be likely to be better educated, and therefore more aware of the marine environment than those of a lower economic status. This correlation has been observed in earlier research (Santos *et al.*, 2005; Steel *et al.*, 2005 Teisl *et al.*, 2008) and supports the suggestion that socio-economics would have an influence on the facilitation of marine citizenship.

7.2.8. Socio-demographic factors

The role of socio-demographics was considered a potential factor by practitioners (Section 4.1) who identified the younger generation as the demographic at which efforts to encourage marine citizenship should be directed. Current levels of knowledge among this generation were found to be low (identified through the education survey) and considered by teachers to be insufficient for the development of marine citizenship (Section 6.2.3).

²⁶ The personal attachment thematic case study research was conducted during the UK economic recession between July 2009 and January 2010.

Results from the personal attachment study showed a statistically significant relationship between interviewee age group and their level of personal concern for the marine environment, marine awareness and perception of the impacts of their lifestyle choices on the marine environment. Earlier research has suggested that individual perception of citizenship may vary with age, linked to an individuals' progressing maturity (Chamberlin, 1997; Teisl *et al.*, 2008; Williams, 2008). Behavioural analysis has suggested that people become increasingly sensitive to the needs of others as they get older (Chamberlin, 1997). However, in contrast to this research, practitioners identified the young generation as the key to modifying societal behaviour, a consensus supported by work done by CCW (Williams, 2008).

Hume (2010) further suggests that although for the most part younger generations exhibit low levels of awareness and concern for the marine environment (as discussed in Section 6.4), this demographic group should be considered the catalysts through which society can change. Research implies that the avenue through which the younger generations would be most able to alter behaviour is through their consumption of products (Hume, 2010). However, the results of the education thematic case study suggest that currently younger generations do not actively consider the impacts of their consumer behaviour on the marine environment, most likely due to their lack of input into household purchasing. In spite of these observations, the concept of the younger generation representing societies of the future has been expressed in previous research (Hume, 2010; Williams, 2008). This suggests that by targeting younger generations and encouraging them to behave in a certain manner, overall societal behaviour towards the marine environment could be altered and further supports their position as the initial targets for encouraging marine citizenship.

In contrast to previous studies (Hawthorne and Alabaster, 1999; Steel *et al.*, 2005; Teisl *et al.*, 2008) neither case study found gender to be an influencing factor in behavioural choices in the context of public responses to marine citizenship components. Research suggests that women are characteristically more aware and concerned about environmental issues in general than men (Wessells, *et al.*, 1999; Teisl *et al.*, 2008). However a lack of correlation

between these factors in this study (Chapter Six) suggests that education and awareness-raising efforts should be targeted at the entire population. In addition, it is suggested that it would be wise to employ a range of approaches in order to access the widest possible audience.

7.2.9. Summary of influences on marine citizenship

From the synthesis presented in Sections 7.2.1 - 7.2.8, a number of factors with the potential to influence marine citizenship have been identified and further investigated throughout this research presented in Table 7.1. A detailed analysis of the frequency of factors is presented in the grid analyses (practitioner interviews in Table 4.1 and personal attachment interviews in Appendix 13).
factors.			
Influencing factor	Summary of factor	Phase One	Phase Two
Behaviour	Related to education and associated awareness and concern for the marine environment. Well educated, aware individuals are expected to be more concerned for the marine environment engendering more favourable and responsible behaviour. A sense of willingness to change behaviour for the benefit of sustainable management and conservation of the marine environment. Consumer behaviour expected to be most common expression of marine citizenship.	`	
Education	Availability of marine specific information accessible to a wide audience through informal and formal education strategies with the aim of improving awareness, concern, sense of responsibility and other related factors.	\$	`
Awareness	Awareness of the marine environment, its management and of the impacts of individual and collective behaviour on the condition of the marine environment and its resources. Strongly linked to education, concern and behaviour and proximity to the marine environment.	\$	`
Concern	A sense of concern for the condition and management of the marine environment. Strong links to awareness, education, willingness to participation, proximity to the marine environment and related behaviour.	\$	`
Location	Location and proximity to marine environments has been shown to have an influence on the factors of marine citizenship through personal experience of the marine environment. Expect marine awareness and concern to be higher in communities where there is a direct dependency on the marine environment. Concern and awareness cannot be strictly associated with coastal areas and exceptions are possible.	>	`

289

Influencing factor	Summary of factor	Phase One	Phase Two
Participation & Collaboration	The role of a collaborative management and planning strategy for the marine environment with comprehensive guidance provided to enhance public engagement in the process. An individual and collective societal desire for participation in the processes that affect the marine environment, its resources and users. Expect participation through recreational activities to engender a sense of concern and responsibility.	`	>
Socio-economics	Most strongly related to access to education and opportunities for involvement. Also potential influence on the behavioural choices available to an individual e.g. consumption of sustainable 'green' products.	`	\$
Socio- demographics	Gender not expected to be a significantly influencing component of marine citizenship. Lifestage and age expected to have a degree of impact. Younger generation viewed as the target audience for promotion of improved social behaviour towards the marine environment	`	`

7.3 MARINE CITIZENSHIP: DEVELOPMENT OF CONCEPTUAL MODEL AND DEFINITION

7.3.1. The need for a conceptual model

One of the key objectives of the research was the development of a conceptual model of marine citizenship that could be effectively applied to the management of the marine environment. Throughout the research, an inductive mixed methods approach has been used to develop a conceptual model to describe the situations observed in the context of marine citizenship. Conceptual models are commonly used to aid in development, implementation, and standardisation of emerging concepts (Moody, 2005). In addition, they can act as a representation of a situation that enhances understanding and produces recommendations regarding a concept (Greca and Moreira, 2000). In the case of this research, the development of a conceptual model of marine citizenship serves to potentially benefit marine management in the UK through:

- Producing the first working definition and model for a marine specific form of citizenship;
- The model will provide an evaluation of the key elements specific to marine citizenship and their role in promoting the concept of marine citizenship;
- The model will identify the integral components of marine citizenship with the potential to encourage enhanced public responsibility and involvement through promotion of marine citizenship.

Through these developments, it is expected that marine citizenship will benefit marine management in the following ways:

- Through enhanced levels of public engagement in marine management and decision-making processes will result in the development of more effective and holistic management plans and policy.
- Through more easily implemented marine management strategies as a result of higher levels of public inclusion.

• Through improved individual and collective behaviour towards the marine environment which will result in easier implementation of management.

7.3.2 What is marine citizenship?

A central focus of the research is the generation of a working definition of marine citizenship that can be applied to contemporary marine management strategies in a bid to address the challenges currently facing marine environmental sustainability. Based on the available relevant literature, Table 2.4 (Chapter Two) highlighted the initial assumptions surrounding the successful promotion of marine citizenship, and examined the conditions under which an individual would be expected to exhibit a sense of marine citizenship. Taking the results of the research into consideration a refined version of this table is presented in Table 7.2 indicating the optimum levels of the key factors required to engender a sense of marine citizenship.

Component of marine	Optimum levels of personal factors for
citizenship	marine citizenship
Access to education	 Exposure to enhanced formal and informal mechanisms for marine specific education. Good level of knowledge derived from appropriate education source. High levels of education in environmental issues. High levels of marine environmental literacy. Sense of knowledge on a local, national and international scale.
Participation	 High sense of individual and cumulative societal responsibility for the marine environment. Sense of willingness to be actively involved in management and decision making process. Opportunities for involvement in marine planning and management.
Location	- Strongest when at close proximity to the marine environment.
Socio-economics	- Financially able to make marine environmentally conscientious consumer decisions.
Socio-demographics	 Families with children exposed to marine information through formal education at school. Expect sense of awareness to increase with age and maturity implying sense of marine citizenship will become stronger with age.
Awareness	 Awareness of marine environmental issues. Awareness and understanding of behavioural impacts.
Concern	 Sense of responsibility for the marine environment. Awareness and understanding of the issues facing the marine environment.
Behaviour	 Sense of willingness to change behaviour for the benefit of the marine environment. Understanding of the impacts of behaviour. High sense of personal and societal responsibility for the marine environment. High levels of awareness and concern for the marine environment.

Table 7.2: Personal factors of marine citizenship and the optimum level of

these factors at which a sense of marine citizenship would be encouraged

Component of marine	Optimum levels of personal factors for
citizenship	marine citizenship
Personal connection	 High levels of exposure and personal experience of the marine environment through holidays and recreational activities; Living within a coastal community Moderate to high levels of dependency on the marine environment for livelihood (currently or historically) Strong association with the marine environment

Based on the findings of the research outlined in Chapters Two, Four and Six a fully refined definition of marine citizenship can be established:

Marine citizenship can be defined as having understanding of the individual rights and responsibilities towards the marine environment, having an awareness and concern for the marine environment and the impacts of individual and collective behaviour, and having a desire to have a role in ensuring on-going sustainable management of the marine environment.

Given this definition, a conceptual model based on the personal factors identified as influencing components of marine citizenship was generated and is discussed in Section 7.3.3.

7.3.3. Conceptualisation of marine citizenship

In accordance with the inductive approach (detailed in Chapter Three), evolving models of marine citizenship were generated throughout the data collection. The first model (Figure 2.3) was produced based on the observations made throughout the literature review using pre-existing models of overall environmental citizenship to guide the development of an initial conceptual model of marine citizenship. This model was used to establish the direction of the research and guided the practitioner interviews (Chapters Three and Four).

Following the completion of the practitioner telephone interviews (Chapters Three and Four) a refined model for marine citizenship and its application to marine management was generated based on practitioner perception of the concept of marine citizenship (Figure 4.1). Early models of marine citizenship guided the final evaluations conducted through the thematic case studies (Chapters Five and Six) and resulted the generation of a final working model of marine citizenship, presented in Figure 7.2.

As presented in Tables 7.1 and 7.2, the conceptualisation of marine citizenship is based on a range of factors being present at optimum levels to encourage expression of marine citizenship. However, in practice, the influences on marine citizenship will be subject to variation, which will in turn be reflected in the degree of marine citizenship that an individual can be expected to exhibit. Given this, marine citizenship could be modelled as a simple linear continuum with 'successful marine citizenship' at one end and 'unsuccessful marine citizenship' represented at the other. At the 'unsuccessful' marine citizenship' end of the continuum, the elements of marine citizenship outlined in Tables 7.1 and 7.2 would not be achieved and the management measures to aid development of societal marine citizenship would be ineffectively managed or absent. In contrast, at the 'successful marine citizenship' end of the continuum, all management measures would be set in place to engender a sense of marine citizenship resulting in the presence of the factors presented in Tables 7.1 and 7.2. Position along the continuum and thus the achievable sense of marine citizenship would be determined by the presence or absence of the management measures recommended to ensure successful application of marine citizenship in a contemporary marine management context.

Whilst the continuum based model of marine citizenship provides a good illustration of the development of a conceptual model, the linear relationship does not provide a comprehensive representation of the complexities associated with the marine citizenship factors. The linear relationship between successful and unsuccessful marine citizenship can be further divided based on the presence or absence of the key elements (education and personal attachment) identified through the practitioner interviews as illustrated in Table 7.3. As discussed in Chapter 5, these broad factors were considered to encompass a wider variety of more specific components (See Table 4.1). The four potential combinations,

identified as Scenarios 1-3, represent the complexities associated with marine citizenship showing progression through the scenarios that indicates an increase in marine citizenship. Table 7.3 indicates a progression from Scenario 1 which illustrates the lowest sense of marine citizenship to Scenario 3 having the strongest sense of marine citizenship, with Scenario 3 representing a stronger sense of marine citizenship than Scenario 2. It should be noted that although the conditions of each scenario are static, it is possible for an individual to move across the boundaries of each scenario subject to changes in their own circumstances.

 Table 7.3: Conceptualisation of marine citizenship based on the presence of the key themes of education and personal attachment.

		Education	Personal
. .			attachment
Low marine citizenship	Scenario1	Х	Х
	Scenario 2a	J	Х
	Scenario 2b	Х	1
High Marine	Scenario 3	1	1
Citizenship			

7.3.3.1. Scenario 1

In Scenario 1, evidence of the factors relating to the broad themes of education and personal attachment is poor. In general, there are poor levels of marine knowledge and understanding due to a lack of access to either formal or informal education tools. As a result of a low sense of personal connection to the marine environment coupled with lack of information, this scenario exhibits low levels of concern or awareness and little or no desire to modify behaviour towards the marine environment. Under these circumstances there is little evidence to suggest a sense of marine citizenship would be evident. In addition, it is unlikely that individuals would be willing to engage in attempts to promote and improve their sense of marine citizenship. Individuals falling within Scenario1 exhibit the weakest sense of marine citizenship, as none of the factors considered to be integral components are included.

7.3.3.2. Scenario 2a

In Scenario 2a, the factors relating to the theme of education are evident with high levels of knowledge and awareness of the marine environment derived through either formal or informal education strategies. A lack of personal connection with the marine environment reduces the likelihood of an individual exhibiting marine citizenship. In Scenario 2a, an individual would be expected to exhibit some degree of marine citizenship due to moderate to high levels of awareness and knowledge. In addition, a sense of concern and responsibility towards the marine environment would be expected. However, this would be based predominantly on an awareness derived through education strategies rather than inherent awareness related to personal connection to the marine environment. Low sense of willingness for behavioural change related to personal connection is characteristic of this scenario, however, due to high levels of education it would be expected that individuals would behave in a responsible manner. Lack of connection results in a limited sense of marine citizenship, although it is likely that based on information availability and accessibility of education, individuals in this scenario exhibit some degree of marine citizenship.

7.3.3.3. Scenario 2b

In Scenario 2b, a sense of marine citizenship is predominantly based on individuals' personal connection to the marine environment, which is expected to range from moderate to high in this scenario. As such, sense of concern and responsibility linked to personal associations with the marine environment, such as livelihood dependency or proximity to the coastal are the influencing factors in Scenario 2b. In contrast with Scenario 2a, levels of knowledge and understanding associated with education would be low. In this scenario individuals would be inclined to be aware of the impacts of their behaviour and

express a willingness to modify their behaviour if necessary. Poor access to education tools and a lack of capacity for involvement may hinder the level of marine citizenship in this scenario, but it would be present in higher levels than in Scenario 2a as a result of a stronger sense of personal attachment to the marine environment.

7.3.3.4 Scenario 3

In Scenario 3, all of the integral components of marine citizenship are evident. Individuals have access to marine specific information through a variety of education tools exhibit a high level of understanding and awareness of the marine environment, its management and the role of their individual and cumulative behaviour in marine management and conservation. In addition to education based factors, individuals in this scenario would also characteristically exhibit a high sense of connection with the marine environment. This is the optimum combination of factors through which an individual could be considered to have a high sense of marine citizenship.

7.3.4. A conceptual model for marine citizenship

In addition to the conceptualisation and modelling of marine citizenship, it became apparent that the expression of marine citizenship and its application to marine management could be influenced by external circumstances. In order to conceptualise the relationship between marine citizenship and these circumstances, the idea of 'enabling factors' is now introduced. This term has been developed to describe the circumstances in which marine citizenship could be successfully applied to contemporary marine management. The concept of enabling factors suggests that in order for successful expression of marine citizenship to be enabled, presence of the appropriate factors is required as illustrated in Figure 7.1.



Figure 7.1: Progression of marine citizenship in the presence of key elements and appropriate enabling factors.

In the context of this research, synthesis of the results indicated that there are three key enabling factors required for the successful expression of marine citizenship. These are outlined below:

- Appropriate marine policy: Analysis of the practitioner interviews indicated an aspiration for more effective policy and legislation enabling public involvement in marine management and decision-making. It is therefore suggested that having marine policy of this nature in place would serve as an enabling factor for the expression of marine citizenship, through the implementation of marine management strategies guided by appropriate policies.
- Effective marine management: Synthesis of the results highlighted the need for a collaborative relationship between marine governance bodies and the public, particularly focusing on the role of marine managers as guides for improved public behaviour towards the marine environment. In the presence of suitable marine policy, it can be assumed that

appropriate management strategies (such as those highlighted in Table 7.4) would be applied to the marine environment. As such, these management strategies would act as enabling factors for the expression of marine citizenship through the provision of enhanced education and awareness-raising programmes and improved opportunities for public engagement in marine management.

• Favourable socio-economics: Throughout the research, it was apparent that although there is a sense of public willingness to change societal behaviour and perception of the marine environment, efforts to do so could be influenced by both individual and public socio-economic circumstances. Given the financial implications associated with favourable environmental behaviours, a synthesis of interviewee responses emphasised the power of socio-economics to both limit and facilitate the expression of marine citizenship. Based on these observations, it is therefore suggested that the presence or absence of suitable socio-economic circumstances acts as enabling factor for marine citizenship as it is assumed that in favourable socio-economic conditions expressions of marine citizenship would be higher.

Table 7.4 further illustrates the relationship between the factors of marine citizenship and the presence of suitable enabling factors outlined above.

Key person	al factors of	Enab	ling Factors for r	narine citizenship		
marine c	itizenship					
Education	Personal	Favourable	Favourable	Management Strategies	Influence on marine	Expressions of
	attachment	socio-economic	marine policy		management	marine
		conditions				citizenship
				Increased opportunities for wider public engagement in	Participation in management processes	Participation and collaboration in
Favour	able condi	tions of		the process.	and decision-making	marine
				Onen nrocesses with regards	will increase understanding of the	management
potn pe	rsonal raci	cors and		to marine management and	process.	
the en	abling fact	tors of		decision making		
					Improve on current	Increased
marin	e citizensh	nip are		Inclusion of the wider	statutory consultation	awareness and
•				public, in addition to the	requirements by	understanding of
requi	irea in ora	er tor		direct stakeholders, to aid the	including the wider	the marine
succes	sful expres	sion of		implementation of plans and management projects.	public.	environment
Jem	'ine citizen	chin))	Ensure consideration of	
5		2		Provision of information	all community needs.	
				regarding marine		
				management and widespread	Encourage the process	Improved public
				opportunities for	of social learning in the	desire for
				engagement with the	context of the marine	involvement and
				management and decision	environment.	willingness to
				making processes.		change behaviour.
					Engender more	

Table 7.4: Optimum conditions for successful expression of marine citizenship.

	Ensure marine citizenshin is	responsible individual	Improved
	not restricted to coastal	and societal behaviour.	understanding and
	communities,		awareness of
		Encourage greater	behavioural
	Wider awareness of the	awareness of the	impacts on the
Favourable conditions of	impacts of societal behaviour	impacts of behavioural	marine
	on the marine environment,	choices.	environment.
both personal factors and	including the impacts of both		
the enabling factors of	inland and coastal	Improved conflict	
marine citizenship are	communities.	resolution.	
required in order for	Encourage a more inherent	To empower citizens to	
	sense of awareness, concern	have more involvement	
successiul expression of	and responsibility towards	in marine management.	
marine citizenship	the marine environment.		
		Ensure transparency and	
	Provision of marine specific	accountability	
	education on a national basis	throughout the process.	
	through both formal and		
	informal methods.	Enhance understanding	
		and public capacity for	
	National awareness raising	involvement in marine	
	campaigns with regards to	management.	
	the impacts of everyday		
	behaviour, including	Easier implementation	
	identification of impacts of	of management and	
	inland communities on the	planning through more	
	marine environment.	widespread inclusion.	
	Dromote the notion of the		
	UK being an island and the		
	cultural heritage associated		
	with this.		
	:		
	Provision of financially		
	$v_{1a}v_{1c} + v_{1a}v_{1c} + v_{1a}v_{1c} + v_{1c} + v$		
	subsidised opportunities for		
	education and engagement		

Given the recognised need for collaboration between the wider public and marine management authorities (Table 7.3), a model illustrating the relationship between marine citizenship and marine management was generated and is presented in Figure 7.2. This final conceptual model of marine citizenship recognises the need for effective management strategies in order to facilitate wider public engagement and the generation of a society of 'marine citizens'. In order to facilitate the description of each of the potential relationships between the personal marine citizenship factors and the enabling factors, the outcome space of the model has been divided into four zones. The role of enabling factors in the progression and facilitation of marine citizenship is further outlined in Section 7.3.5.

Figure 7.2: Model of marine citizenship comprising the identified components and recommended management strategies to engender successful application of the concept.

	'Frustrated marine citizenship' Zone 2	'Successful marine citizenship' Zone 4
itizenship	High frequency of marine citizenship factors exhibited by individuals. Poorly managed opportunities for engagement and promotion of marine citizenship.	High frequency of marine citizenship factors exhibited by individuals. Well managed opportunities for engagement and promotion of marine citizenship.
s of marine c		ZONE MOST LIKELY TO SUCCESSFULLY ENGENDER MARINE CITIZENSHIP
ē		
of fact	'Unsuccessful marine citizenship' Zone 1	'Limited marine citizenship' Zone 3
Presence of fact	 'Unsuccessful marine citizenship' Zone 1 Low frequency of marine citizenship factors exhibited by individuals. Poorly managed opportunities for engagement and promotion of marine citizenship. 	 'Limited marine citizenship' Zone 3 Low frequency of marine citizenship factors exhibited by individuals. Well-managed opportunities for engagement and promotion of marine citizenship.
Presence of fact	 'Unsuccessful marine citizenship' Zone 1 Low frequency of marine citizenship factors exhibited by individuals. Poorly managed opportunities for engagement and promotion of marine citizenship. ZONE LEAST LIKELY TO SUCCESSFULLY ENGENDER MARINE CITIZENSHIP 	 'Limited marine citizenship' Zone 3 Low frequency of marine citizenship factors exhibited by individuals. Well-managed opportunities for engagement and promotion of marine citizenship.

Presence of management strategies necessary for the application of marine citizenship

7.3.4.1 Unsuccessful marine citizenship: Zone 1

In Zone 1, the personal factors required to engender an active sense of marine citizenship within an individual are poor or entirely absent. In addition, the recommended management strategies required to aid development of marine citizenship are poorly managed and do not facilitate public engagement or capacity building to encourage a wider level of public understanding of the marine environment. As a result, the conditions exhibited by Zone 1 represent the situation least likely to inculcate a sense of marine citizenship, as none of the factors considered to influence the concept are present.

7.3.4.2. Frustrated marine citizenship: Zone 2

In Zone 2, there is a high representation of the components of marine citizenship exhibited by individuals. Under the circumstances of Zone 2, the public willingness to adopt a sense of marine citizenship is high but is limited by a lack of facilitation and opportunity through poor management strategies. For example, within this zone, it is expected that individuals would exhibit a high level of personal connection with the marine environment resulting in an inherent sense of awareness, concern and responsibility for its condition and ongoing sustainability. The personal association with the marine environment is coupled with a high level of marine specific knowledge, access to both informal and formal education techniques, sense of willingness to change behaviour for the benefit of the marine environment and a desire to be involved in the process. However, Zone 2 is further characterised by poor management resulting in a lack of facilitation of marine citizenship. Decision-making processes would therefore be flawed due to a lack of inclusivity and poorly managed public engagement procedures. Given the recent work aimed at improving public engagement with marine management in the UK, Zone 2 represents a minority of circumstances within the UK.

7.3.4.3. Limited marine citizenship: Zone 3

Zone 3 is characterised by a low representation of factors identified as influencing an individuals' sense of marine citizenship. As such, within this zone, it is expected that there would be low levels of public connection to the marine environment, minimal concern and awareness of marine management and a lack of understanding of the role of the individual and collective society on the marine environment and its management. In addition, Zone 3 would exhibit low levels of marine specific knowledge further hindered by lack of access to effective marine education techniques. The absence of the personal factors of marine citizenship results in a lack of public willingness to engage in the marine management process and a lack of willingness to modify individual and cumulative societal behaviour for the benefit of the marine environment. In contrast to Zone 2, Zone 3 exhibits concerted efforts from management to engage with the wider public and to promote a societal sense of marine citizenship through awareness raising campaigns. In this zone, the opportunities for marine citizenship are high but are not utilised by the community due to a lack of public awareness and understanding of the role of the public in marine management.

7.3.4.4. 'Successful marine citizenship': Zone 4

In Zone 4, the expression of marine citizenship is high. Both the personal components of marine citizenship and the managerial strategies recommended for the inculcation of societal marine citizenship are present. In such circumstances, it is expected that a sense of marine citizenship would be exhibited by communities and would result in more inclusive, effective marine management. The combination of individual willingness and capacity for involvement, appropriate management strategies and the facilitation of public involvement in marine management necessary for marine citizenship is well represented in Zone 4.

7.3.4.5. Comments

Each of the four zones represented by the model in Figure 7.2 exhibits a different potential for the inculcation of a society wide sense of marine citizenship. The progression moves from zone 1 exhibiting characteristics least likely encourage a sense of marine citizenship to zone 4 representing the characteristics most likely to encourage a sense of marine citizenship among society. Zones 2 and 3 represent circumstances through which a degree of marine citizenship could be expected but it is likely that marine citizenship of this level would not engender the social change in perception and behaviour associated with successful application of marine citizenship to marine management in the UK. This model highlights the need for significant collaboration between the wider public, stakeholders, private sector and statutory governance bodies in order to reach the level of marine citizenship required to engender the level of change required to achieve the objectives set out by sustainable marine management plans.

7.3.5. Progression of marine citizenship

As outlined in Section 7.3.4 the expression of marine citizenship is dependent on the presence of the appropriate enabling factors. The presence or absence of the identified enabling factors will theoretically influence both the expression of marine citizenship and the progression between the zones illustrated in Figure 7.3. Taking this into consideration, it should be noted that an individual can be located at any position in the model of marine citizenship i.e. an individual or community would not be required to start within zone 1 ('unsuccessful marine citizenship') and gradually progress towards successful expression of marine citizenship in the presence of appropriate personal components and enabling factors. It is assumed, however, that in the presence of the appropriate enabling factors i.e. favourable policies, management strategies and socio-economic conditions an individual or community would aspire towards expression of marine citizenship behaviours and would aim to progress towards Zone 4 of the model in Figure 7.3. For an individual or community in Zone 1 (unsuccessful marine citizenship), three potential routes have been identified that would facilitate a progression towards Zone 4.



Figure 7.3: Routes towards successful expression of marine citizenship.

7.3.5.1 Route 1

The first route is to initially implement the necessary management strategies to raise public awareness, concern and sense of responsibility for the marine environment. These management strategies would be implemented to promote the personal components of marine citizenship (Figure 7.2) through awareness-raising and public capacity building. Following this, marine management strategies that provide more opportunities for public engagement would be implemented. This approach would facilitate the expression of marine citizenship by increasing information availability and opportunities for engagement in marine management.

Route 2 is to implement management strategies that enhance both personal components of marine citizenship and the opportunities for public engagement together. By providing opportunities for public engagement in marine management, the personal components of marine citizenship are promoted and more successful expression of marine citizenship behaviours is facilitated. Under these circumstances, as with Route 1, the provision of financially accessible opportunities for engagements is integral to the successful facilitation of marine citizenship. This is the most direct route to successful expression of marine citizenship from Zone 1.

7.3.5.3 Route 3

The third route is to implement management strategies aimed at promoting the personal components of marine citizenship through awareness-raising, marine education programmes and wide-scale advertisement of opportunities for public engagement in marine management. As with the other potential routes, progression of marine citizenship through this route requires favourable socio-environmental conditions through the provision of financially accessible opportunities for expression.

7.4 APPLICATION TO CONTEMPORARY MANAGEMENT STRATEGIES

7.4.1. Implications of marine citizenship for current marine management and policy

Since the beginning of this research, marine management in the UK has undergone significant changes following the ratification of the Marine and Coastal Access Act (2009) and the resultant establishment of the Marine Management Organisation (MMO). In addition to new UK legislation, the UK is a signatory of the EU Marine Strategy Framework Directive, which aims to achieve 'Good Environmental Status' (GES) for all European seas and associated resources by 2020. Central to the UKs signing of the Marine Strategy Framework Directive are five UK High Level Marine Objectives²⁷ that aim to guide statutory governance, stakeholders and the wider public in the sustainable use of the marine environment within the UK and wider European waters (Defra, 2009). There has been a concerted move towards more inclusive and holistic marine management within the UK with the promotion of marine spatial planning (MSP), development of a network of marine conservation zones (MCZ) and the generation of a national and regional marine policy statement (MPS).

This movement towards more holistic and integrated management through the MMO and the application of overarching policies is evidence of the increasing recognition that marine management in the UK has been in need of considerable reorientation. The application and promotion of marine citizenship on a national scale would aid the UK in achieving each of the five HLMOs prompted by the EU Marine Strategy Framework Directive (MSFD) and obtaining GES for our marine ecosystems. In addition to this, it would aid in the implementation of holistic and inclusive marine management strategy which are becoming increasingly common due to the rising popularity of ecosystem based management approaches to marine management (Defra, 2009). In particular the application of marine citizenship to marine management would engender a more marine educated citizenry, aware of their impacts on the marine environment and would hone the sense of individual and societal responsibility for the marine environment desired by the objectives of the MSFD.

It is a proposition of this research that the emerging concept of marine citizenship could serve as a mechanism through which societal behaviour towards the marine environment could undergo the substantial modifications required to meet the international, European and national marine management

²⁷ The UK high level objectives outlined by the Marine Strategy Framework Directive include:

[•] The achievement of a sustainable marine based economy;

[•] Assurance of a responsible and aware society with access to a healthy and safe marine environment;

[•] Establish a society living within the limits of the marine environment;

[•] Promotion of good and effective governance of the marine environment;

[•] The widespread use of responsible and sound science to support marine management and policy delivery.

objectives regarding sustainability. How and where marine citizenship could be applied to contemporary marine management and policy are outlined through a number of recommendations in Chapter Eight.

7.4.2. Limitations of the conceptual model

There are certain issues that need to be taken into consideration when evaluating the applicability of the model.

Firstly, as with any model developed through an inductive methodology and so grounded in the data collected, the model should be considered to be temporally, spatially and contextually specific to the conditions of the data collection. Through the thematic case studies, the model can be applied to the UK but not in an international context. Further research would be required to evaluate the applicability of the model in a different context.

Second, the generation of the model was grounded in data collection methodologies that did not have the capacity to incorporate all potential respondents. However, both phases of data collection obtained high response rates considered sufficient for this research. In addition, both the marine practitioner interviewees and the locations of the thematic case study sites were selected explicitly to reflect the complexities of UK marine management and the range of environments and communities represented by the UK coastline. Given these measures to ensure the incorporation of a diverse range of respondents, it can be assumed that the research is representative of UK public perceptions and can therefore be applied in a UK wide context.

Finally, since the data collected was predominantly sourced through a variety of interview techniques, it must be assumed that the participating interviewees expressed their true opinions. Data quality assurance methods were employed throughout the research in order to ensure the highest quality of data; however, this is still a relevant consideration. The high response rates obtained through the telephone interviews and both phases of the thematic case studies limited the potential for a particular interview or result to skew the observations.

The limitations of the model outlined above are characteristic of this methodological approach. Therefore, although there are some potential limitations associated with the research, the model can be viewed with confidence and considered to be applicable to marine management at all scales within the UK.

7.4.3. Strengths and weaknesses of marine citizenship

As with any emerging concept, there are a number of strengths and weaknesses that can be associated with the generated model of marine citizenship. Table 7.5 presents a SWOT (Strength, Weakness, Opportunities, and Threats) analysis conducted on the concept of marine citizenship and its implications for the wider management of the marine environment. This analysis is based on the observations resulting from a synthesis of all results. The benefits expected to be associated with marine citizenship are represented by the strengths and opportunities identified in the left column of Table 7.5, while the potential challenges resulting from and to marine citizenship are illustrated by the weaknesses and threats presented in the right column of the table.

Table 7.5: SWOT analysis of marine citizenship and its potential role in marine management in the UK (Adapted from Petts and Leach (2000))

Strengths

The application of marine citizenship could be strengthened by:

- Encouraging public participation in marine management and decision making processes;
- Improving the use of local knowledge in marine planning and management;
- Enhancing public understanding of the marine environment;
- Ensuring better evaluation of issues facing users of the marine environment as a result of more inclusive consultation and participation approaches;
- Encouraging better public knowledge of the marine management system;
- Encouraging an enhanced public sense of responsibility for their individual/ societal behaviour towards the marine environment.

Opportunities

The application of marine citizenship will encourage opportunities to:

- Improve public trust in marine governance and management bodies;
- Enhance public acceptance of marine management strategies;
- Empower the public by encouraging transparency and wider consultation in marine planning and management;
- Allow consideration of the public as stakeholders of the marine environment;
- Prevent conflicts or protests by increasing the capacity for greater inclusion;
- Expand of public understanding of local, regional, national and international marine issues;
- Improve public responsibility for behaviour towards the marine environment.

Weaknesses

The application of marine citizenship could be weakened by:

- A lack of guidelines for the inclusion of the wider public in marine management;
- Currently low levels of public awareness and knowledge of the marine environment;
- A lack of accessibility of marine specific information;
- Currently low levels of public capacity for involvement in marine management and decision-making;
- Resulting enhanced public involvement could cause tensions;
- A lack of support for effective public inclusion in marine management.

Threats

The application of marine citizenship could be threatened if:

- The public feel pressured by approaches to engender marine citizenship;
- It is not efficiently promoted by marine governance bodies and end users.
- There is a lack of acceptance among traditional stakeholders that the wider public should be included in the marine management process.
- Formal and informal mechanisms for providing marine education are not improved.
- More effort is not made to improve public awareness, responsibility and concern for the marine environment.
- Strategies for promoting marine citizenship are not made available and accessible in all areas, regardless of proximity to coast and socio-economic factors.

7.5 SUMMARY

This chapter has presented a synthesis of the main findings of this research and has provided an examination of the emergent concept of marine citizenship. Section 7.2 outlined the elements identified as being key to the successful conceptualisation of marine citizenship, and furthermore to its successful promotion as a mechanism of addressing the challenges facing contemporary management of the marine environment. Section 7.2.1 began with an identification of the role of the individual and the wider public in marine management in the UK. Throughout the Phase One interviews it became clear that marine practitioners were generally of the opinion that there is a need for further involvement of the wider public in marine management, in order to develop and implement successful sustainable marine management. The thematic case studies indicated a significant level of concern and care for the marine environment within UK society suggesting a public willingness to be involved in the process, subject to adequate guidance from governing bodies and legislators. The factors of marine citizenship were then outlined based on a synthesis of the observations made through Chapters Two, Four and Six. The main influences are summarised in Table 7.1 and were found to relate to seven main themes presented in Sections 7.2.2.-7.2.8. These themes included the role of education, the influence of location on marine citizenship, personal connections to the marine environment, behaviour and its role in marine citizenship, participation and collaboration in the context of marine management and the impacts of socio-economics and socio-demographics on an individual's sense of marine citizenship.

Section 7.3 presented an original conceptualisation of marine citizenship generated based on the synthesised results of the two phases of data collection. It mapped the factors of marine citizenship based on the two primary themes of education and personal attachment that had been investigated through the case study interviews. Following this, Section 7.3 outlined the generation of an original conceptual model for marine citizenship which mapped the influence of the personal factors identified as influencing marine citizenship against the management strategies available to facilitate development of marine citizenship.

It was apparent from the model that in order for marine citizenship to have a significant impact on marine management, certain conditions, i.e. enabling factors, must be in place to nurture the inherent levels of marine citizenship that an individual may have towards the marine environment. It was clear that in order for marine citizenship to be successfully established among society, marine management strategies must facilitate public engagement and awareness-raising on a national scale. Efficient management was identified as being integral to the elucidation of the potential held by marine citizenship to engender more responsible behaviour towards the marine environment and sustainable marine management. Sections 7.3.3 and 7.3.4 are of particular importance to the research as they identify the practical measures required to operationalise the concept of marine citizenship, examining the role of enabling factors in the facilitation of marine citizenship.

Finally, Section 7.4 outlined the role of marine citizenship in contemporary marine management by evaluating its applicability to current marine policies. It went on to examine the limitations of the conceptual model and presented a SWOT analysis of the original concept of marine citizenship.

Overall, Chapter Seven presented the first working conceptual framework of marine citizenship. It presented an evaluation of the ongoing transition from state-driven management to a successful collaboration between the wider public and traditional management bodies in the context of sustainable marine management. Chapter Eight further places these observations in the context of current marine management and legislation and considers the implications of this research for future marine management in the UK.

REFERENCES

Alessa, L., Bennett, S., M., and Kliskey, A., D. (2003). Effects of knowledge, personal attribution and perception of ecosystem health on depreciative behaviours in the intertidal zone of Pacific Rim National Park and Reserve. Journal of Environmental Management 68: 207-218.

Ananda, J. (2007). Implementing Participatory Decision Making in Forest Planning. Environmental Management 39: 534-544.

Barraza, L., and Cuaron, A., D. (2004) How values in education affect children's environmental knowledge. Journal of Biological Education 39 (1): 18-23

Berkes, F. (2004). Rethinking community-based conservation. Conservation Biology 18(3): 621-620.

Berkowitz, A., R., Ford, M., E., and Brewer, C., A. (2005). A framework for integrating ecological literacy, civics literacy, and environmental citizenship in environmental education. *In* Environmental Education and Advocacy: Perspectives of Ecology and Education. E. Johnson, A., and Mappin, M., J. Editors, Cambridge University Press. Cambridge: 227-266.

Brécard, D., Hlaimi, B., Lucas, S., Perraudeau, Y., and Salladarré, F. (2009). Determinants of demand for green products: An application to eco-label demand for fish in Europe. Ecological Economics 69(1): 115-125.

Burley, D., Jenkins, P., Laska, S. and Davis, T. (2007). Place attachment and environmental change in coastal Louisiana. Organization and Environment 20: 347-366.

Castle, Z., Fletcher, S., and McKinley, E. (2010). Coastal and marine education in schools: Constraints and opportunities created by the curriculum, schools and teachers in England. Ocean Yearbook. 24: 425-444.

Dalton, T., M. (2005). Beyond Biogeography: a framework for involving the public in planning of U.S. Marine Protected Areas. Conservation Biology: 1392-1401.

Davis, A. and J. Wagner (2006). A right to fish for a living? The case for coastal fishing people's determination of access and participation. Ocean & Coastal Management 49(7-8): 476-497.

Defra (2009). Our Seas - a shared resource. High Level Marine Objectives.

Delgado, A., and Strand, R. (2010). Looking North and South: Ideals and realities of inclusive environmental governance. Geoforum 41(1): 144-153.

Fletcher, S., Potts, J., Heeps, C., and Pike, K. (2006). Encouraging marine environmental awareness in schools: an evaluation of the Planet Ocean education pack. Proceedings of Littoral 2006, September 18-20, 2006 Gdansk, Poland.

Fletcher, S., and J. Potts (2007). Ocean Citizenship: An Emergent Geographical Concept. Coastal Management 35: 1-14.

Gillham, B. (2000). Case study Research Methods, Continuum

Greca, I. M. and M. A. Moreira (2000). Mental models, conceptual models, and modelling. International Journal of Science Education 22(1): 1-11.

Haklay, M. (2002). Public Environmental Information: understanding requirements and patterns of likely public use. Area 34(1): 17-28.

Hawthorne, M., and Alabaster, T. (1999). Citizen 2000: Development of a model of environmental citizenship. Global Environmental Change 9: 25-43.

Hay, I., and Foley, P. (1998). Ethics, Geography and Responsible Citizenship. Journal of Geography in Higher Education 22(2): 169-183.

Hume, M. Compassion without action: Examining the young consumers consumption and attitude to sustainable consumption. Journal of World Business 45(4): 385-394.

Jaffry, S., H. Pickering, et al. (2004). Consumer choices for quality and sustainability labelled seafood products in the UK. Food Policy 29(3): 215-228.

Loureiro, M., L., McCluskey, J., J., and Mittelhammer, R., C. (2002). Will consumers pay a premium for Eco-labelled apples? The Journal of Consumer Affairs 36(2): 203-219.

Mamouni Limnios, E. A., Ghadouani, A., Schilizzi, S., G., M., and Mazzarol, T. (2009). Giving the consumer the choice: A methodology for Product Ecological Footprint calculation. Ecological Economics 68(10): 2525-2534.

McKinley, E., and Fletcher, S. (2010). "Individual responsibility for the oceans? An evaluation of marine citizenship by UK marine practitioners." Ocean & Coastal Management 53(7): 379-384.

Moody, D. L. (2005). "Theoretical and practical issues in evaluating the quality of conceptual models: current state and future directions." Data & Knowledge Engineering 55(3): 243-276.

Novacek, M., J. (2008). "Engaging the public in biodiversity issues." Proceeding of National Academy of Sciences 105(1): 11571-11578.

Petts, J., and Leach, B. (2000). Evaluating methods for Public Participation: A Literature Review, Environment Agency.

Sears, A., M., and Hughes, A., S. (1996). "Citizenship Education and Current Educational Reform." Canadian Journal of Education 21(2): 123-142.

Smith, G. (2005). "Green citizenship and the social economy." Environmental Politics 14(2): 273-289.

Stedman, R., C. (2002). "Toward a social psychology of place: predicting Behaviour from place-based cognitions, attitude and identity." environment and Behaviour 34: 561.

Steel, B., S., Smith, C., Opsommer, L., Curiel, S., and Warner-Steel, R. (2005). "Public Ocean Literacy in the United States." Ocean and Coastal Management 48: 97-114.

Storrier, K., L., and McGlashan, D., J. (2006). "Development and management of a coastal litter campaign: The voluntary coastal partnership approach." Marine Policy 30: 189-196.

Teisl, M., F., Rubin, J., and Noblet, C., L. (2008) "Non-dirty dancing? Interactions between eco-labels and consumers." Journal of Economic Psychology 29: 140-159

Tytler, R., Duggan, S., and Gott, R. (2001). "Public participation in an environmental dispute: implications for science education." Public Understanding of Science 10: 343-364.

Twigger-Ross, C., L. and Uzzell, D., L. (1996). "Place and identity processes." Journal of Environmental Psychology 16: 205-220.

Uneputty, P., Evans. S. M., and Suyoso, E. (1998). "The effectiveness of a community education programme in reducing litter pollution on shores of Ambon Bay (Eastern Indonesia)." Journal of Biological Education 32(2): 143-147.

UNECE (1998) United Nations Economic Commission for Europe. Convention on Access to Information, Public participation in Decision-making and Access to Justice in Environmental Matters. Aarhus, Denmark. June 25th 1998.

UNESCO (1978). Final Report. Tbilisi, Intergovernmental Conference on Environmental Education (1977) Paris Van Dyke, J. M. (1996). "The Rio Principles and our responsibilities of ocean stewardship." Ocean and Coastal Management 31(1): 1-23.

Wessells, C., R., Johnston, R., J., and Donath, H. (1999). "Assessing consumer preferences for ecolabelled seafood: The influence of species, certifier and household attributes." American Journal of Agricultural Economics 81(5): 1084-1089.

Williams, L. B. R. (2008). Public attitudes towards the maritime environment around the Menai Strait and Conwy Bay. CCW Policy Research Report. CCW: 55.

CHAPTER EIGHT

CONCLUSION

8.1 INTRODUCTION

The aim of this research, as outlined in Section 1.3, was to 'critically evaluate of the potential role of marine citizenship in the sustainable management of the marine environment in the UK'. Through the discussion provided in this chapter, the aim will be revisited in order to draw conclusions from the research in relation to the objectives set out to achieve this aim. This chapter then highlights the contribution of this research to environment specific citizenship theories, specifically marine citizenship, and the application of this emergent concept in contemporary marine management. In addition, areas of future investigation inspired by this research will be outlined.

8.2 CONCLUDING COMMENTS

8.2.1. Evolution of marine citizenship

The first part of the thesis presented the underlying rationale for marine citizenship and its application in contemporary marine management in the UK. It was identified through the literature review that the application of traditional citizenship theories and more specifically, the evolution of environmental citizenship has been successful in developing sustainable and inclusive environmental management. However, it became clear that preceding models of environmental citizenship could not be applied to the marine environment. In addition, it was evident that traditional state-driven governance of the marine environment has been subject to a number of failures. A call for collaborative marine management was evident through the synthesis of the results, furthering acknowledging the role of the public in marine management.

suggested that a movement towards collaboration will require an improvement in current levels of public capacity for involvement.

Throughout the first phase of data collection, the current and future role of the individual and the wider public in marine management was ascertained. It was apparent that traditionally members of the public are not perceived as stakeholders of the marine environment meaning their involvement in management process has been minimal. However, a clear need to improve public capacity to deliver meaningful impact on marine management was identified.

Although there was some uncertainty over how marine citizenship could be promoted on a national scale, it was recognised that its development would benefit marine management in the UK. Benefits include a more marine aware and responsible society with the capacity for involvement in marine decisionmaking and planning processes, improved individual and collective behaviour resulting in easier implementation of management and plans as a result of enhanced public understanding. It is expected that the inculcation of marine citizenship would encourage development of more inclusive and efficient management plans resulting from higher levels of public engagement in the decision-making process. Further to this, it is suggested that marine citizenship could address the turn to citizenship experienced by wider environmental management and could act as an effective mechanism to deliver marine policy.

8.2.2. Composition of marine citizenship

A number of key influences were identified as being central to the engenderment of marine citizenship on an individual, community and national scale. The observations of the research largely confirmed that, as with environmental citizenship, marine citizenship is comprised of a complex network of interconnected components; therefore the variables with the potential to affect its expression are numerous. Taking this into consideration, two dominant categories of personal factors were identified: education and personal attachment. The first component identified as being key to the inculcation of marine citizenship among wider society was education. The application of education as a method of addressing societal behaviour towards the environment has been well documented and it can therefore be assumed that the same techniques could be applied to marine education. Improvement in the availability and accessibility of information through enhanced marine education programmes was a recurring theme, firmly placing education as one of the key components of marine citizenship. It was found that adequate education is required to prompt any sense of citizenship towards the marine environment through awareness, concern and sense of responsibility. In addition, it was recognised that improvements to the current formal and informal education systems in the UK are necessary.

In addition to education, personal attachment was found to be a contributing factor to the level of citizenship an individual could express towards the marine environment. It was found that personal attachment to the marine environment could be stimulated by a number of circumstances, including livelihood dependency, familial links to the coast and recreational involvement. Given the links between personal attachment and sense of responsibility identified, it can be concluded that concerted efforts to improve the sense of connection with the marine environment held by UK communities are required to engender the successful expression of marine citizenship.

The overall goal of engendering a societal sense of marine citizenship is to create a mechanism through which societal behaviour can undergo the necessary modifications to ensure successful implementation of sustainable marine management. By addressing other integral components of marine citizenship, such as education and personal attachment, the challenge posed by current societal behaviour towards the marine environment should be addressed with behavioural changes a natural progression. It is expected that these alterations in social behaviour would lead to an improvement in the level of public involvement in marine management allowing the development of more inclusive management plans.

8.2.3. A conceptual model of marine citizenship

Chapter Seven introduced the final working definition and conceptual model of 'Marine Citizenship'. This is an evolution of the overarching concept of environmental citizenship specifically designed to take into consideration the complexities and diversity traditionally associated with the marine environment and its management. In addition to the conceptual model and definition, the application of marine citizenship in the context of contemporary marine management was evaluated, particularly in light of the goals set out by the recent Marine Strategy Framework Directive (2008) and the Marine and Coastal Access Act (2009).

The final components of marine citizenship identified through a synthesis were mapped against specific enabling factors required to encourage the expression of marine citizenship. Of particular importance was the identification of a necessary collaboration between marine management organisations and the public in order to fully engage marine citizenship at a level that will impact the efficacy of marine management and planning. In the context of marine citizenship, three main enabling factors were identified: the presence of appropriate policy, the implementation of effective management strategies and finally, favourable socio-economic circumstances. It was concluded that in order to aid the collaboration between current marine governance arrangements and the public, appropriate marine policies and management strategies that facilitate the expression of marine citizenship are required.

It was concluded that in order for marine citizenship to be engendered, both the appropriate personal attributes and the enabling factors need to be in place. Contrasting perspectives from practitioners and the general public regarding the accessibility of information and opportunities for public involvement emphasised the need for an improved understanding of the potential role of individuals and cumulatively of the general public in establishing sustainable, long-term and effective marine management and conservation strategies. It can therefore be concluded that marine practitioners need to engage with the public and facilitate involvement through policy and management strategies that enhance public

capacity and awareness in order for the benefits of marine citizenship to be realised.

8.2.4. Key Conclusions

Based on the comments above, a number of key conclusions regarding the future application of marine citizenship to contemporary marine management can be made.

- Marine citizenship is comprised of a complex network of interconnected factors and its application to management can therefore be influence by a range of variables.
- Based on this research, marine citizenship can be defined as a level of awareness and concern for the marine environment, the rights and responsibilities of the individual, the impacts of individual and collective behaviour, and a desire to have a role in ensuring on-going sustainable management of the marine environment and its resources.
- The evolution of marine management to a more participatory management strategy would benefit from the promotion of marine citizenship.
- Successful expression of marine citizenship requires the presence of a combination of factors, namely the presence of two key groups of personal factors and enabling factors.
- In order for marine citizenship to be facilitated, appropriate marine policies and management strategies that promote mechanisms for public engagement and capacity building are required.
8.3. RECOMMENDATIONS

The final objective of this research was to determine recommendations that would allow the emergent concept of marine citizenship to be applied to contemporary marine management strategies. The recommendations are split into two main categories: 1) recommendations concerning the presence of the enabling factors required for successful expression of marine citizenship; and 2) recommendations relating to personal components of marine citizenship identified through the research.

8.3.1. Recommendations concerning enabling factors

As discussed in Chapter Seven, three enabling factors were identified for the successful expression of marine citizenship. The first recommendation applies to the overall role of the public in marine management, while the remainder are recommended in the context of the enabling factors identified in Chapter Seven:

- It is recommended that standardised 'best practice' guidelines for the inclusion of the wider public in the management of the marine environment are generated. The provision of a framework formalising the role of the public in marine management and how this can be facilitated through current marine management would enable successful expression of marine citizenship
- It is recommended that an assessment of the current capacity of contemporary governance and management bodies to facilitate marine citizenship be conducted. Both this research and previous work, for example, by Smith (2005) and Kearney *et al.* (2007), highlight the need for support from governing bodies in order for successful expression of marine citizenship to be realised. Given this, an evaluation of how governance and management bodies could currently facilitate marine citizenship is necessary. In addition, this would formalise the management strategies required to promote expression of marine

citizenship in order to establish potential areas for improvement on all scales.

- The outcomes of this research highlighted a need for greater marine practitioner acceptance of the general public in the role of stakeholders and stewards of the marine environment. As highlighted by Edwards *et al.* (1997) the role of the public in UK marine management has been acknowledged for some time, although to date the potential for this has not been realised. Therefore it is recommended that improved inclusion of the wider public as stakeholders of the marine environment through wider access to the process should be a key objective of marine managers and governance bodies. This will serve to aid public understanding of the complexities associated with marine management. Given the objectives of on-going marine planning projects, the facilitation of public participation in marine management and governance will ensure that all requirements of coastal communities are considered.
- It is the recommendation of this research that more favourable consumer behaviour be facilitated so as to enable expression of marine citizenship. This would most likely be achieved through incentives for further market innovations and campaigns to raise awareness of more sustainable options and to ensure that they are financially viable and accessible to a broad audience. By increasing accessibility with a minimal level of inconvenience to the consumer, the potential that individuals and society will make decisions that favour the sustainable management of the marine environment will be increased. Given the importance attributed to consumer behaviour, significant re-education of consumers is required if consumer behaviour is going to remain the most common method of expressing marine citizenship among the wider public. Evidence to support this can be found in work by Loureiro et al. (2002) and Teisl et al. (2008) who found that in order to facilitate pro-environmental behavioural choices, individuals must be provided with accurate and accessible product information.

It is recommended that a cost-benefit analysis be conducted with regards to the promotion of an improved sense of responsibility towards the marine environment. Previous work indicated that individuals would be prepared to pay a premium for environmentally products (Loureiro *et al.*, 2002). However, the level of public willingness to pay more is likely to have changed in recent times as a result of the global economic recession. Given the current economic climate, it will be necessary to reassure the public that any costs and inconvenience associated with marine citizenship will be minor. In addition, it should be highlighted the fact that adopting these changes will bring long-term benefits on a local, regional and national scale.

8.3.2. Recommendations concerning personal components of marine citizenship

The conceptualisation of marine citizenship identified the broad themes of education and personal attachment as the key personal components of the model. Given this, there are a number of recommendations regarding these personal factors. There are four recommendations applicable to the role of education in marine citizenship, with the first relating to formal education strategies and the second concerning informal education techniques. Further to this there are recommendations regarding raising public awareness, understanding and sense of connection with the marine environment:

• Firstly, it is recommended that there should be an increased availability of formal marine education. This would be best delivered by adapting the current National Curriculum to increase the level of mandatory teaching of marine environmental topics. As indicated by this research and by Castle *et al.* (2010) the level of marine education in UK schools is currently limited. Enhancing it would ensure that future generations will be provided with sufficient information to allow them to make informed decisions regarding the marine environment in the future. It is also recommended that improving the level of marine content in school will

encourage dissemination of marine information through families as parents become involved with their children's education.

- It is recommended that informal education techniques are proactively employed to supplement traditional classroom delivery of marine education; methods include marine environmental documentaries, the internet, the promotion of marine environment based field trips within school and increased accessibility to marine specific information at a variety of public places e.g. aquariums, museums and coastal walks. Research by Potts (2000), Storrier and McGlashan (2006) and Williams (2008) present early evidence to suggest that an approach of this nature could be effective in altering societal perception, awareness and behaviour towards the marine environment in the UK.
- One of the primary challenges to marine citizenship identified was a lack of understanding concerning the impacts of everyday life and the marine environment. As indicated by Hawthorne and Alabaster (1999)'s review of environmental citizenship, in order for environmental education to be effective, educators need to be aware of the wider context in which they operate. In the case of marine citizenship, marine education encompassing a wider societal context may further individual and collective understanding of impacts on the marine environment. In order for this to be successful, it is recommended that marine citizenship requires an integrated campaign between all sectors to emphasise the connections between society and the wider marine environment in order to promote public understanding of this relationship.
- Lack of personal attachment with the marine environment was seen to be a significant potential challenge to successful inculcation of marine citizenship. Given the importance attributed to sense of connection in the context of marine citizenship, it is the recommendation of this research that public connection with the marine environment be addressed. However, it is currently unknown as to how this could be facilitated.

Previous work has suggested that exposure to an environment can prompt an individual's sense of connection and concern (Novacek, 2008; Barbas *et al.*, 2009; Goodwin *et al.*, 2009). Therefore, it is recommended that community projects that encourage wider public engagement in marine management be made available in a bid to enhance public connection with the marine environment.

8.4 CONTRIBUTION OF THIS RESEARCH

Research into a marine specific concept of citizenship has not been undertaken on any scale with previous work dominated by terrestrial examples of environmental citizenship; therefore this research is an original contribution to the debate concerning the application of citizenship to marine management and conservation. Through effective delivery of the research objectives (stated in Section 1.3), several specific areas of contribution can be identified. These areas of contribution relate to the conceptualisation of marine citizenship, its application in marine governance and the implications of this on a national scale:

- The research has for the first time identified the key elements influencing a marine specific concept of citizenship.
- The research presents a unique insight into the perspectives of both marine practitioners and of the wider public regarding marine management and current public capacity for effective engagement in contemporary marine governance and management.
- The conceptual model of 'marine citizenship' provides an original contribution to the debate concerning the current and future role of the wider public in marine governance and management. It provides a theoretical ideal to which both marine practitioners and the wider public can aspire in the bid to develop sustainable marine management.

- Specific management measures necessary to apply marine citizenship to contemporary marine governance on a national scale have been identified. Identification of these measures aids the translation of the findings of this research from theoretical observations to being of practical benefit to the development of sustainable and effective marine governance.
- The timing of the research coincides with the publication of the High Level Marine Objectives set out by the EU Marine Strategy Framework Directive and the ongoing implementation of the UK Marine and Coastal Access Act making it possible for the research to contribute to achieving the goals set out by this legislation.

8.5 AREAS FOR FUTURE RESEARCH

As is characteristic of inductive research, several areas for further investigation were identified. These areas present a variety of potential developments of the research discussed in the thesis. The first area requiring further investigation is particularly related to the inductive nature of the research:

- In order to further assess the transferability of the research outside its developmental context (as conducted in Hawthorne and Alabaster's model (1999)), it is necessary to test the research outcomes in other locations, in areas of varying dependency on the marine environment, varying proximity to the marine environment, and of various levels of economic development and stability.
- Given the increasingly multi-national approach to marine management, the applicability and transferability of the research outside the UK requires further investigation. It would be useful to assess how the model and management recommendations can be applied to the cultural and managerial context of other countries.

 Although it was acknowledged throughout the research that socioeconomics has an integral role in the development of an individuals' sense of citizenship towards the marine environment, the evaluation conducted in this research was not sufficient to derive any significant conclusions. More comprehensive research is required in order to establish the level of impact socio-economics has on the behavioural choices made by individuals.

The following areas of investigation relate to the promotion of the influential factors of marine citizenship. In addition the relationship between efforts to enhance these components and engendering a sense of marine citizenship should be further investigated.

- Education was found to play an integral role in enhancing a sense of awareness, concern and responsibility and is therefore key to the successful delivery of marine citizenship. The content and most appropriate mechanism of delivering marine specific education to the wider public on a national scale requires further evaluation.
- Further research into the relationship between personal connection to the marine environment and individuals' sense of marine citizenship is required in order to identify strategies that could be employed to enhance a sense of connection to the marine environment.
- Given that a change in individual and collective societal behaviour towards the marine environment would be the ultimate goal of promoting marine citizenship, research is required to ascertain how this can be brought about through awareness raising and educational campaigns.

The remaining areas for further examination are focused on the facilitation of marine citizenship through the recommended management strategies and the benefits for managers, stakeholders as well as the general public:

- The ability of marine management and governance to facilitate the promotion of marine citizenship, as an aid to sustainable marine management requires further assessment. Although there is a practitioner willingness to include communities in marine management, this is commonly restricted to the involvement of traditional stakeholders and the general public have minimal opportunity for input. Methods of encouraging marine managers and governance bodies to extend their facilitation of wider engagement need to be established in order for marine citizenship to be applied successfully.
- The relationship between marine managers and governance bodies would benefit from further investigation. As outlined by the research, state driven decision making is often viewed with a lack of trust, and this perception may have a role to play in the level of public willingness to participate in the marine management process. An evaluation of this relationship would identify potential issues and challenges to management strategies aiming to promote public awareness and engagement with the marine environment and its management.
- The development of marine management to address public capacity issues on all scales was a recommendation of this research. Further work is necessary to establish the requirements of specific locations and how marine citizenship can be applied to management on all scales and in all locations.
- Given the current absence of guidelines regarding the inclusion of communities in marine management, the inclusion of recommendations of management measures (as outlined in Chapter Seven) aimed at enhancing opportunity for and efficacy of public engagement requires further investigation. In particular, the generation of specific guidelines regarding the facilitation of marine citizenship that could be adapted according to the requirements of management on a variety of scales would be benefit long term marine management.

8.6 FUTURE OF MARINE CITIZENSHIP IN UK MARINE MANAGEMENT

Traditional management of the marine environment and its associated resources has been subject to criticism as a result of fragmentation, sectorally based management and a lack of public engagement in the process. This has put into question the credibility of marine legislation and management strategies and has highlighted the need for a re-evaluation of the marine management process and how it can be improved to benefit the marine environment. Marine citizenship mirrors an international evolution in environmental management from a predominantly state-directed governance system to a more participatory, community-inclusive management regime. Although this research has found that marine citizenship would benefit marine management, the future of marine citizenship remains unclear. Based on the observations of the research, there are two dominant possibilities which are outlined below.

The first future is dependent upon the recognition of the role of the wider public in marine management and the advantages this would lend to on-going sustainable marine management. A potentially significant unknown at this time is the capacity for current marine governance to lend itself to the level of public inclusion proposed as the outcome of efforts to promote marine citizenship in the UK. One of the key goals of the UK High Level Marine Objectives is to ensure the wider public are supplied with the appropriate information to enable them to make sound decisions with regard to the marine environment. In order to achieve this goal, formal recognition of the public role in marine management is required at all levels of governance. The future efficacy of marine citizenship is dependent on the provision of facilities that will allow marine managers and planners to utilise it to develop more sustainable management. In addition, the future of marine citizenship will be dependent on marine practitioners recognising the need to facilitate the presence of appropriate enabling factors in order to encourage the expression of marine citizenship.

The second potential future is one in which there is little or no recognition of the role of marine citizenship in marine management. In this situation there would

be minimal effort to encourage public involvement in marine management with no promotion of the personal components or enabling factors of marine citizenship. In such circumstances, the likelihood of achieving the goals set out by the High Level Marine Objectives would be limited due to a continuing lack of public knowledge, understanding, and concern and ultimately a sense of willingness to be involved in the management process.

8.7. FINAL REMARKS

In conclusion, however the concept of marine citizenship is applied to existing UK marine management, it is clear that a movement towards enhanced public engagement in the process is the ideal situation. In order to facilitate this transition, capacity building measures are necessary to allow meaningful public engagement on any scale. Marine citizenship could provide a solution to a variety of capacity issues that are associated with public involvement in the management of the marine environment. While the promotion of marine citizenship will require considerable effort on the part of marine managers and governance bodies, and could carry risks, the benefits of promoting efforts to engender marine citizenship on a UK scale are likely to be significant.

REFERENCES

Barbas, T., A., Paraskevopoulos, S., and Stamou, A., G. (2009). "The effect of nature documentaries on students' environmental sensitivity: a case study." Learning, Media and Technology 34(1): 61-69.

Castle, Z., Fletcher, S., and McKinley, E. (2010). Coastal and marine education in schools: Constraints and opportunities created by the curriculum, schools and teachers in England. Ocean Yearbook. 24: 425-444.

Edwards, S. D., Jones, P. J. S. and Nowell, D. A. (1997). "Participation in coastal zone management initiatives: a review and analysis of examples from the UK." Ocean and Coastal Management 36(1-3): 143-165.

Hawthorne, M., and Alabaster, T. (1999). "Citizen 2000: Development of a model of environmental citizenship." Global Environmental Change 9: 25-43.

Kearney, J., Berkes, F., Charles, A., Pinkerton, E., and Wiber, M. (2007). "The role of participatory governance and community-based management in integrated coastal and ocean management in Canada." Coastal Management 35: 79-104.

Loureiro, M., L., McCluskey, J., J., and Mittelhammer, R., C. (2002). "Will consumers pay a premium for Eco-labelled apples?" The Journal of Consumer Affairs 36(2): 203-219.

Goodwin, M., J., Greasley, S., John, P., and Richardson, L. (2010). "Can we make environmental citizens? A randomised control trial of the effects of a school-based intervention on the attitudes and knowledge of young people." Environmental Politics 19(3): 392 - 412.

Novacek, M., J. (2008). "Engaging the public in biodiversity issues." Proceeding of National Academy of Sciences 105(1): 11571-11578.

Smith, G. (2005). "Green citizenship and the social economy." Environmental Politics 14(2): 273-289.

Teisl, M., F., Rubin, J., and Noblet, C., L. (2008). "Non-dirty dancing? Interactions between eco-labels and consumers." Journal of Economic Psychology 29: 140-159.

Appendix 1: Email sent to marine practitioners prior to interviews

Dear.....

My name is Emma McKinley. I'm currently undertaking a PhD at Bournemouth University, examining the definition and development of an international concept of marine citizenship. The initial phase of data collection is comprised of a number of telephone interviews with members of organisations currently involved in the management of the marine and coastal environments, and was wondering if you would be willing to participate. The interview should only take 25-20mins.

I would be very grateful for your participation, and look forward to hearing from you soon. If there is someone within your organisation that you feel would be more appropriate for me to speak to, I would be vert grateful if you would forward them this email and ask them to contact me

Many thanks

Emma

Emma McKinley, PhD Research Student School of Conservation Sciences Bournemouth University Talbot Campus Dorset UK email: <u>emckinley@bournemouth.ac.uk</u>



Before you print think about the ENVIRONMENT

Appendix 2: Marine practitioner introduction and rationale letters used in Phase one.



Dear

Marine Citizenship

I am writing to you to invite you to participate in a doctoral research project to examine the role of and the factors influencing marine citizenship.

The aim of this survey is:

To develop a working definition for marine citizenship, and to examine the factors that may potentially influence its promotion and development

This phase of the project will endeavour to obtain information from individuals who are already involved in the management of the marine environment and who may be involved in the promotion of a concept of marine citizenship. Your participation in this survey would therefore be greatly appreciated.

I am hoping to conduct telephone interviews throughout late September into October and November, which will follow a semi-structured interview format. It is important for me to stress that any personal opinions expressed or information given will remain completely confidential. An overview of the results will be made available to all participants following completion of research. Please find enclosed an overview of the research project, as well as an outline of the areas which will be covered during the interview.

I will telephone you within the next few days in order to arrange suitable time to conduct the interview. However, should you have any questions regarding the project or wish to discuss the research further, please do not hesitate to contact me (Tel: 07817794191 email: <u>emckinley@bournemouth.ac.uk</u>).

I look forward to speaking with you soon

Yours Sincerely

Emma McKinley Postgraduate Researcher, Bournemouth University

Practitioner Rationale information



Evaluation of the application of marine citizenship in the management of marine environment Emma McKinley

Citizenship and the Environment

The concept of environmental citizenship is based on the ideology that a higher level of environmental concern and responsibility is required within modern society, and that individuals need to be aware of the impacts they have on the environment. Following the Earth Summit, 1992, and the development of Agenda 21, it has become increasingly apparent that sustainable management and conservation of global natural resources requires cooperation between international governments, stakeholders and society as a whole.

Project Rationale and application of research

Coastal and marine environments are of massive importance globally with coastal zones exhibiting the highest level of biodiversity, whilst supporting the majority of human population with approximately 50% of the industrialised world living within 50Km of the coast. As changes in terrestrial governance occur, it has become increasingly apparent that the long term stability of the marine environment is dependent on a new form of citizenship being developed, one that highlights the need for greater sense of personal responsibility within society towards the environment as a whole.

Aim and Methodology

The general aim of this project is to examine the application of marine citizenship in the management of the marine environment, ultimately generating a working definition, whilst investigating the social, economic and environmental factors that may influence its development. This is being carried out through a series of practitioner interviews to obtain management consensus on the application of marine citizenship, followed by further analysis in case study locations in the UK.

Consensus View Telephone Survey

This initial phase of the project aims to determine the perceived role of citizenship and the factors influencing it based on the opinions of representatives

from a number of international organisations concerned with the sustainable management and development of the marine environment and its resources. Data was be collected through telephone interviews, with the discussion based primarily on current marine and coastal management, the role of the public in decision making systems, and the factors that influence citizenship.

Case study work

Case study areas have been selected to allow examination of key themes identified in the initial practitioner phase of data collection, and is guided by the practitioner interviews carried out earlier in the project. This phase of research aims to examine discrepancies between practitioner and societal perception regarding the role of the public in the management of the marine and coastal environment. This phase will also allow examination of the social, economic and cultural factors influencing the engenderment of a sense of marine citizenship in the UK and the benefits it could have for the sustainable management of the marine and coastal environment.

Contact Details

Emma McKinley en School of Conservation Sciences Bournemouth University Talbot Campus, Fern Barrow Bournemouth

email:emckinley@bournemouth.ac.uk

Appendix 3: Marine practitioner pilot interview

Name: Date of Interview: Time start: Organisation: Position:

Time End:

Interview Questions

- What is your understanding of citizenship?
- What is your understanding of citizenship in an environmental context?
- In recent times, there has been a global promotion of the concept of citizenship in areas of political development. Do you think that this should also apply to the marine and coastal environment?
- If so, to what extent?
- How do you think the concept of citizenship relates to the sustainable management of the marine environment and its associated resources?

• What factors do you think influence a sense of citizenship amongst the public towards the marine environment and its resources?

• Who do you consider responsible for the management of the marine environment and its resources - for each answer, ask candidate to explain at what level and why?

- How active a role do you think communities and stakeholders should take in the management and decision-making processes with regards to the marine environment?
- What effect do you think increasing public involvement in the management of marine resources would have?
- To what degree do you think environmental issues, in this case focusing on marine issues, relate to every day life in human society?
- Does the state of the environment and its management have an impact on social and economic issues? Explain answers
- How concerned do you think the general public are regarding the condition of the marine environment?
- How do you think public awareness can be encouraged?

• Who should be responsible for providing the resources for this?

- With regard to the previous question, what scale do you think people's awareness of the marine environment and the issues facing it are? (Local, regional, global?)
- How do you think a sense of responsibility towards the marine environment can be promoted? Whose responsibility do you think it is to do this?
- How do you think the current management of the marine environment affects its long-term sustainability?

• How do you think the sense of environmental responsibility within today's society should or could be altered and promoted

• What factors do you think influence a sense of citizenship amongst the public towards the marine environment and its resources?

Appendix 4: Actual marine practitioner interview

Name:
Date of Interview:
Time start:
Organisation:
Position:

Time End:

Interview Questions

• In your opinion, what is the role of the individual in the development of management plans and policies with regard to the marine and coastal environment?

• What is your understanding of citizenship?

• What is your understanding of citizenship in an environmental context?

• In recent times, there has been a global promotion of the concept of citizenship in areas of political development. Do you think that that this should also apply to the marine and coastal environment? If so, to what extent?

• How do you think the concept of citizenship relates to the sustainable management of the marine environment and its associated resources?

• Should citizenship be included in legislation and policy development?

- What demographic factors do you think influence a sense of citizenship amongst the public towards the marine environment and its resources?
- Who do you consider responsible for the management of the marine environment and its resources for each answer, ask candidate to explain at what level and why?

• How active a role do you think communities and stakeholders should take in the management and decision making processes with regards to the marine environment?

- What effect do you think increasing public involvement in the management of marine resources would have?
- To what degree do you think environmental issues, in this case focusing on marine issues, relate to every day life in human society? For example, does the state of the environment and its management have an impact on social and economic issues? Explain answers

• How concerned do you think the general public are regarding the condition of the marine environment?

• How do you think public awareness can be encouraged?

• Who should be responsible for providing the resources for this?

• With regard to the previous question, what scale do you think people's awareness of the marine environment and the issues facing it are? (local, regional, global?)

• How do you think a sense of responsibility towards the marine environment can be promoted? Whose responsibility do you think it is to do this?

• Do you think there is enough promotion of marine and coastal issues to result in a sense of marine citizenship being developed?

• How do you think the current management of the marine environment affects its long term sustainability?

Appendix 5: Sample marine practitioner interview transcript (including highlighted segments used to determine themes and patterns during content analysis).

Name: MP	
Date of Interview:	
Time start:	Time End:
Organisation:	
Position:	

Interview Questions

• In your opinion, what is the role of the individual in the development of management plans and policies with regard to the marine and coastal environment?

It is tricky in the sense that the general public (out with the coastal zone) would probably feel like they have a very small role – generally they may not feel an affinity towards the marine environment and its resources. In coastal communities with a strong relationship and sense of place, I think it would be completely different and their input would be significantly stronger as often their livelihoods and incomes are dependent on the condition of the sea. The role and enthusiasm to get involved would be varied, as there are always going to be some members of society who have no interest. I'm sure in Weymouth (as an example) a significant proportion of the public don't really have an interest or an understanding of the marine environment. Generally think that it would vary across the population.

• What is your understanding of citizenship?

That there is a holistic, global dimension to belonging to something.

• What is your understanding of citizenship in an environmental context?

That things are put in holistic terms with a global dimension – in terms of Marine citizenship, it would show that the seas and oceans are connected regardless of where they are. Holistic view of stewardship of the seas and oceans, balancing management at a larger scale. People tend to have a parochial views – MC involves a wider view, a holistic understanding and management of the seas.

• In recent times, there has been a global promotion of the concept of citizenship in areas of political development. Do you think that that this should also apply to the marine and coastal environment? If so, to what extent?

Yes – don't see why not, the environment cannot be a closed system, regardless of differences between the marine and terrestrial environments. There needs to

be an increased understanding by all users as there is currently a lack of understanding with regard to interconnectedness of the systems, and of the fact that often the impacts of one use can have devastating impacts on another (shipping, fishing, recreation). Marine citizenship would increase people's understanding of environments and of the connections between uses and also the ecology. There is often a tendency for people to dwell on statutory rights from out of date legislation which doesn't reflect current uses of the sea. Uses of the sea have changed and will continue to do so. There will need to be more flexibility in peoples views and use of the sea. Citizenship could also relate to ICZM which attempts to take all uses into account and balance them.

• How do you think the concept of citizenship relates to the sustainable management of the marine environment and its associated resources?

If taking citizenship as being a kind of mass stewardship then there would be an increased understanding which can only lead to better management decisions. Care needs to be taken as there can be entrenched views in everything for example within the commercial fisheries and environmental lobby groups. The balance for management is in the middle ground. Citizenship is a wider understanding, appreciation and acceptance that the sea has an environmental social and economic function... It should however play a strong role in sustainable management of the coast – if people feel close to an issue there will be greater understanding of both economic and environmental issues, and there will be more balanced views which is very important. E.g. In Lyme Bay –

• Should citizenship be included in legislation and policy development?

Not necessarily – don't think that it would be an easy thing to make statutory, we can't force people to feel a sense of stewardship, its almost a way of life or something that needs to evolve. The Marine Bill promises to involve coastal communities in future decision making in the marine environment. This should instil a sense of stewardship, so in a sense it does exist in policy or as a goal of policy. However I think it may be difficult to work into legislation – the European Marine Strategy Framework Directive will have policy e.g. will encourage education to promote citizenship but this will build a sense of citizenship rather than legislating for it.

• What demographic factors do you think influence a sense of citizenship amongst the public towards the marine environment and its resources?

Proximity to the coast – people away from the coast will be less likely to have a sense off marine citizenship.

Coastal users/ employees of the maritime sector

Social background – although it may be difficult to pinpoint, not sure if there would be a reasonable argument to link to social conditions.

Age groups – may vary depending on experiences; dependent on exposure in school curriculum (particularly in coastal communities).

Feel that a lot is down to people's individual experiences and professions as opposed to social standing.

• Who do you consider responsible for the management of the marine environment and its resources - for each answer, ask candidate to explain at what level and why?

Ideally everybody – although at a legislative level all tiers of government. However if policies and laws are going to be developed then they need to be managed – the MMO will play this role. Implementation will come from sea fisheries committees (IFCAs) and other environmental agencies. With regard to citizenship, people need to manage their own individual activities, need to understand the correct way to behave; there will always be some who do not care. Stewardship can cause increased awareness – need management by the general population to ensure they understand the general marine environment. Someone has to take responsibility for the marine environment – perhaps the proposed steering group linked to the MMO will bring all of the main sectors of the coast together; each have a responsibility to ensure that they are being responsible and not conflicting with other sectors. - national sense. International sense – IMO and other legislation. There are obviously layers local, regional, national, EU and global levels. Not sure what the best structure would be but ideally there needs to be enforcement of legislation and people need to manage their own activities.

• How active a role do you think communities and stakeholders should take in the management and decision making processes with regards to the marine environment?

Particularly active role – needs to be a mechanism to set this up. How are the local authorities going to feed it into marine planning? There seems to be disconnectedness and people often view the local authorities as a terrestrial body, often missing the notion that the constituents of coastal LAs are fishermen and other coastal workers, and the impacts of the marine environment to people in these areas are relevant and are potentially an issue. The mechanism for involvement would be through local government and it's absolutely necessary that there be input from local communities as they often know the area better than anyone else. Compliance will only be as strong as peoples input into the process, people need to feel they had a role to play and had a say in coastal and marine management initiatives.

• What effect do you think increasing public involvement in the management of marine resources would have?

More expensive, time consuming but think that the product would be better in the long run. Top down management is better for speed of development but the former bottom up approach with input will result in a building of citizenship, where people feel involved and included having obvious benefits The advantages far outweigh the disadvantages if the process is facilitated well.

• To what degree do you think environmental issues, in this case focusing on marine issues, relate to every day life in human society? For example, does the state of the environment and its management have an impact on social and economic issues? Explain answers Dependent on where they live – coastal communities should have higher awareness e.g. coastal regeneration; UK Government select committee looked at issues in coastal towns compared to inland towns, and it was found that coastal communities have different problems as a result of seasonality, retired populations etc. There are very specific issues that coastal communities have, and often they are very dependent on the marine and coastal environment – as a result coastal communities are more likely to be attuned to potential impacts on the marine environment. In contrast, people away from the coast, may not have a similar perspective and may not understand the link between the marine environment and economic well-being.

• How concerned do you think the general public are regarding the condition of the marine environment?

Varies with where people live, their background and interests. It is difficult to speculate but I think that people will have a general interest, particularly in the more charismatic marine species. Opinion will vary significantly.

• How do you think public awareness can be encouraged?

Inclusion on marine planning in coastal communities should encourage involvement, people need to be physically involved in things. Media (documentaries) encourage expansion of understanding; BBC; Blue Planet I'm sure spiked an increase in public interest in marine issues. Need to think about how people get there information these days – usually TV, internet and other media. Should also consider what the best marketing campaigns would be? There is a slow seepage of information anyway but it can depend on whatever is fashionable at the time.

Decision makers need to be seen as listening to communities as well.

• Who should be responsible for providing the resources for this?

Central government has a role to play – marine Bill, there will be a certain level of responsibility

Media – licensing fees

Everyone has to take some kind of responsibility – done through local interest groups and community level.

Maritime industry and sectors have a role to play – and should give something back to the environment they use on a day to day basis. Many coastal and marine users should take on some responsibility for educating the public about what they do, explaining their use, possible impacts and any mitigation they have in place.

• With regard to the previous question, what scale do you think people's awareness of the marine environment and the issues facing it are? (local, regional, global?)

Dependent on where you live, profession, background, education – coastal communities may generally be more aware.

• How do you think a sense of responsibility towards the marine environment can be promoted? Whose responsibility do you think it is to do this?

Comes through understanding the connections between land and sea, within the sea and connections between economic and environmental matters.

• Do you think there is enough promotion of marine and coastal issues to result in a sense of marine citizenship being developed?

Comes in peaks and troughs – at the moment, the marine bill is increasing it. Globally it varies from country to country Would imagine that people promoting citizenship are in the minority but the groups involved are good at getting messages across – some countries have much bigger problems than the marine environment. it generally varies, can't answer in a global sense, although national level there has been more promotion with the marine bill as it has become very topical.

• How do you think the current management of the marine environment affects its long term sustainability?

Think that there are pockets of the marine environment that are well managed, although certain European policies I need to be reconsidered such as the Common Fisheries Policy Nationally, long-term sustainability will be addressed through the marine bill – the picture is looking better than it was a few years ago. New legislation will be put in place that will allow for MPAs and reserves, better fisheries management, more informed licensing and more strategic planning.

Globally the credit crunch will be affecting how marine environmental issues are perceived.

Appendix 6: Education thematic case study rationale letter



C232, Christchurch House School of Conservation Sciences Bournemouth University Fern Barrow, Poole Dorset

Dear XXXXXXX

I contacted you earlier this week to enquire about the possibility of your school participating in my PhD research project. As I mentioned the project aim is to examine the application of the concept of marine citizenship in the sustainable management of the marine environment, and hopes to look at the role of education in this process.

I conducted a number of interviews with individuals currently involved in the management of the marine and coastal environment in the UK. Analysis of the information collected indicated that education is considered one of the primary influencing factors for marine citizenship. It was also suggested by a number of participants that school students would be the best audience to promote marine education about responsibility and awareness to. Taking this into consideration I propose to conduct structured interviews with students approaching the end of their compulsory school career – Stages 3 - 4 in the case of your students.

Finally, I would like to thank you and your colleagues very much for allowing me to carry this research out.

Many thanks again,

Yours truly,

Emma McKinley emckinley@bournemouth.ac.uk

Appendix 7: Rationale for school involvement sent to participating schools.



Project Rationale

Coastal and marine environments are of significant global importance with coastal zones exhibiting the highest level of biodiversity, whilst supporting the majority of human population with approximately 50% of the industrialised world living within 50Km of the coast. As changes in terrestrial governance occur, it has become increasingly apparent that the long term stability of the marine environment is dependent on a new form of citizenship being developed, one that highlights the need for greater sense of personal responsibility within society towards the environment as a whole.

The general aim of this project is to examine the application of marine citizenship and how it will impact future sustainable management of the marine and coastal environment. The first phase of research consisted of an extensive literature review which identified the key components of environmental citizenship and how they could be related to the marine environment. Following this telephone interviews were carried out with marine environmental managers and professionals in order to establish the consensus view of marine practitioners on marine citizenship, and its role in managing the marine environment. Analysis of the interviews was used to categorise three key themes. These themes encompass a variety of factors previously isolated by Hawthorne and Alabaster (1999) as having an influence on environmental citizenship. It therefore stands to reason that they will have some level of influence on the generation of marine citizenship within society,

Proposed Method of Investigation

One of the primary themes identified as having a high influence on engendering a sense of citizenship towards the marine environment is education. It was proposed that school going individuals would prove the most captive audience for the promotion of marine citizenship and are an obvious target group in which to examine the influence of education on this concept. It is proposed that students will be provided with a short questionnaire that will assess level of knowledge, awareness and sense of responsibility towards the marine environment. The questionnaires will remain anonymous with no personal information required aside from gender.

It is also proposed that the influence of teaching capacity should also be evaluated i.e. how capable teachers feel of providing accurate and correct information to students that would encourage them to become "marine citizens". Previous work has identified a number of subjects in which the marine and coastal environment is either directly or indirectly referred to. These include Geography, History, Sciences, Art and Design and Citizenship. Short, informal interviews with teachers of these subjects would be very valuable to this research and it is proposed that these are carried out in the same schools as the student surveys. These interviews will again be anonymous and will not require personal information from participants.

Appendix 8: Instructions for student questionnaire



Instructions for Student Questionnaire:

Section one:

This section is designed to examine student's knowledge about certain marine issues that are considered the biggest problems or most well known terms. For the term familiarity section, please ask students to tick the box relevant to them – i.e. whether they have heard of a term and understand it, heard of but do not understand or finally if they have never heard of term. For the short marine quiz please ask them to underline the answer that they think is correct.

Section 2:

This section is to investigate the level of marine education in school and how students think it currently influences their decision making. It is also looking at where students feel they get their information about the marine environment from and how informed they think they are. Just ask them to tick the boxes they feel are relevant to them - it's based on a sliding scale, 1 being not at all and 5 being extremely.

Section 3:

This section is intended to evaluate the students concern – again it is a case of ticking the boxes that are relevant to them, based on the same sliding scale as the previous section. There is no right or wrong answer.

Appendix 9: Student questionnaire



Student Questionnaire

Gender: Male 🗆

Female

Section 1: Marine Knowledge

Term Familiarity

Please identify which of these terms you are familiar with i.e. indicate by ticking the correct box whether they are terms you know and understand, terms you have heard of but do not know what they mean or if you have never heard of them.

	Know and	Heard of but do	Have never
	Understand	not understand	Heard term
1. Ecosystem			
2. Biodiversity			
3. Coral Bleaching			
4. Over fishing			
5. Climate change			
6. Sea Level Rise			
7. Coastal erosion			
8. Exclusive Economic			
Zone			
9. Integrated Coastal			
Zone Management			
10. Marine Bill			
11. Citizenship			
12. No-Take Zone			

Short Marine Quiz

Underline the correct answer in each sentence

a) Ocean fisheries are affected by:

climate change 🛛	red tides □	over-fishing		all of the ab	ove 🗆
b) Most sea life:					
lives in the top 500ft	of the ocean	□ lives on	the se	a floor 🛛	lives
in the great ocean b	asins 🗆	is evenly dispe	ersed t	hrough the c	ocean
depths					
c) The movement of referred to as:	f cold, nutrient	rich water to th	e surfa	ce of the oc	ean is
upwelling	southern osci	llation	trade w	vinds 🛛	
reversal tide \Box					
d) The transportation	on of sediment	along the coast	is kno	wn as:	
sediment drift	□ across	coast drift		longshore d	rift
e) By catch refers to	:				
regular fish caught	by nets	over fishing		fish that are	
harvested, but not sold or kept for personal use \Box a climate					
phenomenon					
Have you heard of the following groups?					
Marine Conservation	Society 🗆 🛛	Marine Stewards	hip Cou	uncil 🛛	
DEFRA	Crowne Estate	D WWF		une Une	SCO

Indicate which of the following designations you have heard of:

SSSI	(Site of Sp	pecial Scientific	Interest)	MPA	(Ma	rine
Protect	ted Area)		No-take Zone		AC	ONB
	(Area of C	Dutstanding Na	atural Beauty) 🗖	RAMSAR		World
Heritag	ge Site 🛛					

Section 2: Marine Education

1. How would you say you get your information about the marine environment?

School 🗆	TV 🗆 Magazines 🗖	Internet 🗆 Other	Peers □	Radio 🗆		
i vevispapei oi		Out				
2. Have you v	vatched the fol	lowing progra	mmes?			
Planet Earth	□ Blue Pl	anet 🛛	Oceans 🗆	South Pacific		
Spring watch						
3. To what extent are marine and coastal issues covered in school?						
□ 0	□ 1	□ 2	□ 3			
Not at all						
Extremely						
4. To what ex help you mak	ttent do you thi e appropriate o	nk you are prodecisions with	ovided with en regard to the	ough information to marine environment?		

$\Box 0$	\Box 1	□ 2	\Box 4	□ 5
Not at all				
Extremely	7			

5. Hov	w informed d	lo you think you	are about issue	s currently faci	ng the
marine	e and coasta	l environment?			
□ 0 Not at	□ 1 all	□ 2	□ 3	□ 4	□ 5
Extren	nely				
6. Hov	w much imp	act do you think	your day to day	v activities have	e on the
marine	e and coasta	l environment?			
□ 0 Not at Extren	□ 1 all nely	□ 2	□ 3	□ 4	□ 5
7. Do y	you consider	the marine and d or buying othe	coastal environ r goods?	ment when she	opping for
□ 0	□ 1	□ 2	□ 3	□ 4	□ 5
Not at	all				
Extren	nely				
8. Do y enviro Yes	you take par nment? □ No	t in any hobbies	linked to the m	arine and coas	tal
If yes,	please speci	ífy?			
9. From	m your own	experiences, do	you think that t	here is a threat	to the
marine	e and coasta	l environment?			
Yes					

Section 3: Concern
1. How woul environment	d you rate your a ?	wareness of p	roblems facing	g the marine	
□ 0 Not at all Extremely	□ 1	□ 2	□ 3	□ 4	□ 5
2. To what ex you?	xtent is the cons	ervation of the	marine enviro	onment importa	ant to
□ 0 Not at all Extremely	□ 1	□ 2	□ 3	□ 4	□ 5
3. To what exercise the second	xtent would you ?	say that you c	are about the r	narine	
□ 0 Not at all Extremely	□ 1	□ 2	□ 3	□ 4	□ 5
4. Who do yo coastal envir	ou think is respon onment? Tick al	nsible for the r l that apply	nanagement o	f the marine an	ıd
Individuals Government	EveryoneCoastal §	□ Non-g groups □	overnment ager	ncies 🗆	

That is the end of the survey. If you have any additional comments you would like to make about any of your answers or the questions, please add them here.

Thank you very much for your participation.

Appendix 10: Instructions for Teachers' capacity survey



Instructions for Teachers Survey

The survey consists of semi-structured questions – the idea is for teachers to answer the questions in as much details as they have time to do. If possible please give to teachers who cover marine and coastal topics in their teaching; for example, geography, science, history, citizenship, and any others that you think would be relevant.

If participants feel that other questions could be useful, please feel free to add them in at the bottom of the survey.

Thank you very much – your participation is incredibly valuable to my research.

Appendix 11: Teachers' capacity questionnaire (Chapter Five)



Teacher's Capacity Questionnaire

1. To what level is the marine/ coastal environment covered in your teaching?

2. How relevant is the marine and coastal environment to your subject?

3. To what degree is the local environment incorporated into your teaching?

4. What is your opinion on the current level of marine or coastal environment focused education in the curriculum?

5. How do you think that marine focused education could be changed to increase awareness and concern?

6. What impact do you think these alterations would have on how students awareness of marine issues? (i.e. the effect it would have on them after they have left school?)

7. How able do you think you are when it comes to delivering appropriate and coherent marine focused education?

8. What problems do you come up against when including marine related topics in your teaching?

9. How could these problems be mitigated for?

10. What would make teaching your subject easier?

11. How do you think this would affect the end result i.e. do they think better education would result in students being more marine environmentally aware adults?

12. How responsive are students to marine issues in your teaching?

13. How aware do you think your students are about marine and coastal issues?

14. What do you think would improve their awareness?

15. How concerned do you think your students are about the marine and coastal environment?

Thank you very much for your time - if you have any additional comments, or think that other questions could be included, please feel free to add them underneath.

Appendix 12: Personal attachment thematic case study community survey



Personal Attachment Survey Questions

Interviewer:	 Place

Date: _____ Start time: _____ End time: _____

Any other comments?:

OPENING: *E* nsure that you show your ID card and that your name badge is visible.

Hello,

I am conducting a short visitor survey for Bournemouth University.

Would you mind if I ask you a few questions about your visit today and your opinions of the area here?

It will only take around 5 minutes.

I must read the following statement to you before we begin:

A summary of the results of this survey may be published in the public domain. All of your answers will be treated anonymously and cannot be traced back to you. Everything you say will therefore be entirely confidential. If you wish to end the conversation at any time, you are welcome to do so.

PART 1. AWARENESS

This section of the survey asks you to rate your answer on a sliding scale from zero to 5, with zero being 'not at all' and 5 being 'extremely'. Work through the first question with interviewee as an example, if required.

1. How would you rate your awareness of problems facing the marine environment?

$\Box 0$	□ 1	□ 2	□ 3	\Box 4	□ 5
Not at all					Extremely
Comments:					2

2. How much do the problems facing the marine environment, that you know about, worry you?

□ 0 Not at all Comments:	□ 1	□ 2	□ 3	□ 4	□ 5 Extremely
3. To what to you?	extent is the con	servation of th	e marine envi	ronment	important
□ 0 Not at all Comments:	□ 1	□ 2	□ 3	□ 4	□ 5 Extremely
4: To what e environmen	extent do you fee t?	l your lifestyle	has an impact	t on the	marine
□ 0 Not at all Comments:	□ 1	□ 2	□ 3	□ 4	□ 5 Extremely
5. To what environmen	extent do you co t when you buy f	nsider the pote	ential implicat	ions for	the marine
□ 0 Not at all Comments:	□ 1	□ 2	□ 3	□ 4	□ 5 Extremely

6. To what environmen needed offer	extent do you cos t when you buy a 'cleaning products	nsider the pote any other prod s and shampoo'	ential implicati ucts for your h	ons for th ome? (if a	e marine prompt is
□ 0 Not at all Comments:	□ 1	□ 2	□ 3	□ 4	□ 5 Extremely
7. To what benefit the r	extent would you narine environm	be prepared t ent?	o change your	lifestyle i	f it would
□ 0 Not at all Comments:	□ 1	□ 2	□ 3	□ 4	☐ 5 Extremely
8. To what environmen	extent would you t?	ı say that you o	care about the	marine	
□ 0 Not at all Comments:	□ 1	□ 2	□ 3	□ 4	□ 5 Extremely
9. To what you vote at a	extent would pol an election?	icy towards th	e marine envir	onment a	ffect how
□ 0 Not at all Comments:	□ 1	□ 2	□ 3		☐ 5 Extremely
10. How res	ponsible do you t?	feel for the co	ndition of the	marine	
$\square 0$ Not at all Comments:	□ 1	□ 2	□ 3	□ 4	□ 5 Extremely

11. To what	extent do you th t should fall to th	ink responsibi	lity for	r the marine	and c	coastal
□ 0 Not at all Comments:			□ 3		4	□ 5 Extremely
12. To what	extent do vou th	ink responsibil	ity for	the marine	and c	nastal
environment	t should fall to th	ne public?	ity 101	the manne a		Gastai
$\square 0$ Not at all	□ 1	□ 2	□ 3		4	□ 5 Extremely
Comments:						
13. How effe	ective do you thi	nk current mai	nagem	ent of the m	arine	and
$\Box 0$ Not at all	\square 1	□ 2	□ 3		4	□ 5 Extremely
Comments:						
14. To what marine envir	extent do you fe conment?	el that you hav	re a pe	ersonal conne	ection	to the
□ 0 Not at all Comments:	□ 1	□ 2	□ 3		4	□ 5 Extremely
PART 2. AE	BOUT YOU					
15. Gender (by observation)			16. What is	your	home
□ Male is to compare □ Fema	e local/visitor le			first 2 letters	only:	explain this

it no	t give options	
	under 18	18. What is the make up of
your	group today?	-
	18-29	
	30-39	Adults:
	40-49	
	50-59	
	60-69	Children:
	70-79	
	80-89	
	over 90	

17. In which of the following age brackets are you? by observation if possible, if not give options

CLOSURE

19. That is the end of the survey. Is there anything you would like to add to any of your earlier answers or make any additional points?

Thank you for your time and enjoy your day.

Need for public involvement				>																						
Role of younger generations				//																						
Financial implications				>	>					>																
Other concerns																			<u> </u>							
Impact of location		~														~						ノノノノ	~			
Unaware of impacts on marine environment		/	>				//	//						^												
Convenience of changing behaviour		 																								
Concern for the marine environment		~	>	>			>							<u> </u>		~			~					>		
Role of the media		>		>			>												>							
Interviewee code	P1-P18	P19	P20	P21	P22	P23	P24	P25	P26-41	P42	P43-P50	P51	P52	P53	P54	P55	P56	P57	P58	P59	P60	P61	P62	P63-p69	P70	P71-p75

Appendix 13: Grid analysis conducted on the data collected throughout the personal attachment thematic case study

				_																												
Need for public	involvement														^																	
Role of	younger generations																															
Financial	implications																															
Other concerns																																
Impact of	location														>																	
Unaware of	impacts on marine	environment			>										>																	
Convenience	of changing behaviour		>						>						>						>											
Concern for the	marine environment								~						~						~											
Role of the	media																															
Interviewee	code	D76		LTT	P78	P79	P80	P81	P82	P83	P84	P85	P86	P86	P87	P88	P89	06d	P91	P92	P93	P94	P95	P97	P97	P98	66d	P100	P101	P102	P103	P104

Need for public involvement							>			>						< <			~		>		>	>	>			
Role of younger generations																												
Financial implications				>																								
Other concerns											>																	
Impact of location							>			>								~	~				>	\ \ \	\ \ \			
Unaware of impacts on marine environment	>			>	>								~							>		>	>		>			
Convenience of changing behaviour	>						>													>			>			>		
Concern for the marine environment	>			//	>					>						~			//	>		>	>			>		>
Role of the media																>												
Interviewee code	P105	P106	P107	G1	G2	G3	G4	GS	G6	G7	G8	69	G10	G11	G12	G13	G14	G15	G16	G17	G18	G19	G20	G21	G22	G23	G24	G25

_																													
Need for public	involvement	>		>	~	>			>	<								>	//									>	
Role of	younger generations																										//		
Financial	implications		//		//																								
Other concerns																													
Impact of	location				~				>				~					>	>		~							>	
Unaware of	impacts on marine environment				^	>							^				//		//		^					>		>	
Convenience	of changing behaviour							>		>								>											
Concern for the	marine environment		>	>		//							~					>											۲ ا
Role of the	media									>																>			
Interviewee	code	G26	G27	G28	G29	G30	G31	G32	G33	G34	G35	G36	G37	G38	G39	G40	G41	G42	B1	B2	B3	B4	B5	B5	B6	B 7	B8	B 9	B10

	_	_	_	_	_	_	_	_	_	_	_	_	_	_		_	_			_			_	_	_	_	_	
Need for public involvement		>												>						<i>▶</i>				>			>	
Role of younger generations		>											>															
Financial implications									>				>										>				>	
Other concerns							>						//															
Impact of location	>				>	\ \ \								`			>			~						>		
Unaware of impacts on marine environment							>																>	>			//	
Convenience of changing behaviour																							>					
Concern for the marine environment		~	11		~																			//			~	
Role of the media		//												>														
Interviewee code	B11	B12	B13	B14	B15	B16	B17	B18	B19	B20	B21	B22	B23	B24	B25	B26	B27	B28	B29	B 30	B31	B 32	B33	B34	B35	B36	B37	R38

Interviewe one Role of the marine environment Concern for the of charging impacts on environment Unware of impacts on environment Impact of impacts on environment Need of impacts Need of impacts B40 / / / / / / / / B41 / / / / / / / / / B43 /	public ent																											
IntervieweRole of the mediaConcern for the marineConvenience oblaviourImpact of implicationsOther concerns plantFinancial plantRole of couper839marine environmentocationlocationlocationlopationsyounger penvironment843843843843844845845845845845845845846847848849848 <t< td=""><td>Need for involven</td><td>></td><td>></td><td>></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>11</td><td>``</td></t<>	Need for involven	>	>	>																							11	``
Interviewe rode Role of the media Concern for the marine Conventience Unaware of impects on marine Impact of impects on marine Impact of of changing Impact of implications Impact of ocation Impact of ocat	Role of younger generations																											
IntervieweeRole of the mediaConcern for the of changing marineUnaware of impacts on marineImpact of locationOther concerns of changing marineB30code \circ \circ \circ \circ \circ \circ B41 \checkmark \checkmark \checkmark \checkmark \checkmark \checkmark B43 \checkmark \checkmark \checkmark \checkmark \checkmark \checkmark B44 \checkmark \checkmark \checkmark \checkmark \checkmark B45 \bullet \checkmark \checkmark \checkmark \checkmark B46 \checkmark \checkmark \checkmark \checkmark \checkmark A1 \checkmark \checkmark \checkmark \checkmark \checkmark A3 \land \checkmark \checkmark \checkmark \checkmark A4 \checkmark \checkmark \checkmark \checkmark \checkmark A5 \checkmark \checkmark \checkmark \checkmark \checkmark A6 \checkmark \checkmark \checkmark \checkmark \checkmark A1 \land \checkmark \checkmark \checkmark \checkmark A1 \checkmark \checkmark \checkmark \checkmark <td>Financial implications</td> <td></td> <td>></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>></td> <td></td> <td></td> <td></td> <td>~</td> <td></td> <td></td> <td></td>	Financial implications											>									>				~			
Interviewee Role of the concern for the convenience Unaware of Impact on location 839 media marine B40 / / B41 / / B43 / / B44 / / B43 / / B44 / / B43 / / B44 / / B45 / / B44 / / B43 / / B44 / / B45 / / B45 / / B47 / / B48 / / B47 / / B48 / / A1 / / A3 / / A4 / / A5 / / A6 / / A7 / / A8 / / A1 / <t< td=""><td>Other concerns</td><td>`</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>	Other concerns	`																										
IntervieweeRole of theConcern for theConvenienceUnaware ofcodemediamarineof changingimpacts onB39marineof changingimpacts onB40 </td <td>Impact of location</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>>></td> <td></td> <td>></td> <td></td> <td>~</td> <td></td> <td></td> <td></td> <td>></td> <td></td> <td>~ / /</td> <td></td> <td>></td> <td></td> <td></td> <td></td> <td></td> <td></td>	Impact of location										>>		>		~				>		~ / /		>					
IntervieweeRole of the mediaConcern for the marineConvenience of changingB39mediamarine environmentof changing behaviourB39 \checkmark \checkmark \land B41 \checkmark \checkmark \checkmark B43 \lor \checkmark \checkmark B44 \checkmark \checkmark \checkmark B45 \checkmark \checkmark \land B46 \checkmark \checkmark \checkmark A1 \checkmark \checkmark \land A3 A_4 \checkmark \checkmark A4 \land \checkmark \land A5 \land \checkmark \land A6 \checkmark \checkmark \checkmark A7 A_7 \checkmark \checkmark A8 A_1 \checkmark \checkmark A1 A_1 \checkmark A1 \checkmark \bullet A1 \checkmark \bullet A1 \bullet \bullet A1 \bullet \bullet	Unaware of impacts on marine environment	<u> </u>							>			>												>				
Interviewee Role of the media Concern for the marine B39 B40 B40 B41 B43 B43 B43 B44 B43 B44 B43 B44 B44 B45 A1 A3 A4 A5 A6 A7 A6 A1 A6 A1 A1 A1 A1 A1 A1 A1 A1 <td< td=""><td>Convenience of changing behaviour</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></td<>	Convenience of changing behaviour																											
Interviewee Role of the code code media B39 B40 B41 / B42 B43 B43 / B44 / B43 / B44 / B43 / B44 / B45 / A1 / A1 / A3 / A4 / A3 / A1 / A3 / A1 /	Concern for the marine environment		>	>					>					>							//							
Interviewee code B39 B40 B41 B41 B42 B43 B44 B45 B44 B45 B45 B46 B47 B48 B44 B45 A1 A1 A11 A15 A16 A17 A18 A18	Role of the media			>																					>			>
	Interviewee code	B39	B40	B41	B42	B43	B44	B45	A1	A2	A3	A4	A5	A6	A 7	A8	6 V	A10	A11	A12	A13	A14	A15	A16	A17	A18		A19

Interviewee	Role of the	Concern for the	Convenience	Unaware of	Impact of	Other concerns	Financial	Role of	Need for public
code	media	marine	of changing	impacts on	location		implications	younger	involvement
		environment	behaviour	marine environment				generations	
A22									
A23					``				>
A24		~			`				>
A25		~			>		>>		
A26		~		~	>				>
A27				~		>			
A28		~			``		>		>
A29		~					>		>
A30				~					
A31			>	//					
A32		~	>	~					>
A33				~		>			
A34			>	//					
A35		~		 					
A36									
A37			<u>ر</u>						
A38	>	~		~					>
A39				 		<i>ر</i>			/
$\mathbf{A40}$		~		 	~				
A41			>	///	>	///			
A42				//		۲			
A43									
$\mathbf{A44}$									
A45									
A46									
A47						ر			<
A48				<i>ر</i>	~				
A49									

Need for public involvement			>		//	>																		>	>	>	
Role of younger generations													>														
Financial implications						>																					
Other concerns			>																								
Impact of location					\ \ \ \														>	>	>	>			>		
Unaware of impacts on marine environment	>	//	>		>	//	>				>	>	>	//		>		>	////						>		>
Convenience of changing behaviour				>									>	>	>	>					>						
Concern for the marine environment			>		>	>	>						///		>	< <	>	>						>	>	>	> >
Role of the media					>						>>																>
Interviewee code	A50	M1	M2	M3	M4	M5	M6	M7	M8	6M	M10	M11	M12	M13	M14	M15	M16	M17	M18	M19	M20	M21	M22	M23	M24	M25	M26

Need for public involvement				~		
Role of younger	generations					
Financial implications						
Other concerns						
Impact of location						
Unaware of impacts on	marine	environment				
Convenience of changing	behaviour					
Concern for the marine	environment			///		
Role of the media						
	Interviewee	code	M27	M28	M29	M30

Improved and collaborative management						>	>	>						<i>▶</i>					>								
Island culture					>	>																					
Consumer behaviour					~	///	//	//				~		<u>^</u>		~	~ /										<u> </u>
Personal attachment to coast				~		>		>																			
Education and information				~										~													
Need for more responsibility				^		>		>				>															
Lack of trust in government				<u>ر</u>													>										
Public willing to change				^																							
Need for enhanced awareness				>		>																					~
Interviewee code	P1-P18	P19	P20	P21	P22	P23	P24	P25	P26-41	P42	P43-P50	P51	P52	P53	P54	P55	P56	P57	P58	P59	P60	P61	P62	P63-p69	P70	P71-p75	P76

Table (b) Grid analysis of the categories established through the personal attachment thematic case study.

										-			_	-	-			_	-		_	-		_	-	-		_			_		
Improved and collaborative management		>					>				/																		^				~
Island culture																																	
Consumer behaviour																																	< <
Personal attachment to coast																																	
Education and information		//				>																										 / 	
Need for more responsibility	-	~									<i>ر</i>																		<i>۲</i>			ノノ	
Lack of trust in government																													^				
Public willing to change											~																						>
Need for enhanced awareness																																~	
Interviewee code	P77	P78	P79	P80	P81	P82	P83	P84	P85	P86	P87	P88	P89	P90	P91	P92	P93	P94	P95	P97	797	P98	66d	P100	P101	P102	P103	P104	P105	P106	P107	G1	G2

Improved and collaborative management		~			~				>		۲					~	>		~						//							
Island/ heritage culture																						>			>							
Consumer behaviour	>								>		<u> </u>					ノノ				>	>	>	>	>	~ /			<u> </u>				
Personal attachment to coast																^				>			>									
Education and information																									>							
Need for more responsibility	>					///					~	>					>					///					~					>
Lack of trust in government																									>							
Public willing to change		<i>۲</i>				~																~						<i>۲</i>				
Need for enhanced awareness					~				>										~	>		>		>	\ \ \			~				>
Interviewee code	G3	G4	G5	G6	G7	G8	G8	G9	G10	G11	G12	G13	G14	G15	G16	G17	G18	G19	G20	G21	G22	G23	G24	G25	G26	G27	G28	G29	G30	G31	G32	G33

Interviewee	Need for	Public willing	Lack of trust	Need for	Education	Personal	Consumer	Island/	Improved and
COME	awareness	to that ge	government	responsibility	anu information	coast	Dellaviou	culture	management
G34									~
G35									
G36									
G37							>		
G38							>		
G39									
G40									
G41	>				>				
G42									///
B1	>					>	///		//
B2							>		
B3							~ ^		
B4							>		
BS	>			//		~			/
B6		<i>ر</i>		<i>ر</i>		^	< <		
B 7	~				~				<i>ر</i>
B8									<i>۲</i>
B 9									
B10									
B11	////		^		~ ~		~		
B12									<i>ا</i>
B13	~ ~			~	~	~	~ ^	~ ~	<pre>/ /</pre>
B14									
B15	~			1			11		/ /
B16				<i>ر</i>			///		<i>ر</i>
B17	>	~		^			~ ~		<i>۲</i>
B18						>	>		///
B19							~ / /	<u> </u>	<i>۲</i>
B20								~	
B21									
B22				>				>	~

Island Improved and culture collaborative management		、			>	>						>	~ ~			>	~	~ ~	~ ~			>	~ ~	 						~	
Consumer behaviour	//					~ /					///		>				~			~	///	~ / /				~				~	
Personal attachment to coast	~	>				>							>		~							>									
Education and information															< <				~ /			,				>					
Need for more resnonsibility											~		>				~		~			~ / /		~		< <				~	
Lack of trust in government													>																		
Public willing to change																						~		^							
Need for enhanced awareness						//										>			< <				>	///		< <					
Interviewee code	B23	B24	B25	B26	B27	B28	B29	B30	B31	B32	B33	B34	B35	B36	B37	B38	B39	B40	B41	B42	B43	B44	B45	A1	A2	A3	A4	A5	A6	A7	01

Interviewee code	Need for enhanced	Public willing to change	Lack of trust in	Need for more	Education and	Personal attachment to	Consumer behaviour	Island culture	Improved and collaborative
1	awareness)	government	responsibility	information	coast			management
$\mathbf{A10}$				~					~
A11				>			>		//
A12									
A13	>	>		//	>		>		< <
A14	>				>	//			
A15	>				>				>
A16	>				>				
A17		>					>		
A18				>			>		
A19	>	/ /	>	>					>
A20		>		///		~	>		>
A21				~		^	~ ^		~
A22			>				~		~
A23						<i>ا</i>		1	
A24									۲
A25								/	
A26									
A27		/					~		
A28				///					~
A29				 					
A30	~	/				/	~		۲
A31	~			//		<i>ا</i>	~		//
A32		~		~		<i>ر</i>	~ ^		<
A33	>			~					
A34									
A35	>					<i>ر</i>			
A36									
A37		/					~		
A38		/				<i>ر</i>	~		/ /
A39							~ /		۲
$\mathbf{A40}$	//			///				11	

Interviewee	Need for enhanced	Public willing to change	Lack of trust in	Need for more	Education	Personal attachment to	Consumer behaviour	Island culture	Improved and collaborative
	awareness	A	government	responsibility	information	coast			management
A41	>		>			>			//
A42	>	>			>				
A43									
A44	~ /		>	//	>		~ /		~
A45									
A46	~			<		^	~ /		/ /
A47									
A48		>	>	>				>	~
A49									
A50	~			~			~		ر ا
M1	>	>		`			~ /		~
M2	~ / /				~				~
M3	~ ~	~	<u> </u>	~	~				<i>ر</i>
M4				>					~
M5				//		~		/	
M6			//	<i>ا</i>					<i>ر</i>
M7		<				~	~		
M8	~								///
6M									
M10				<i>۲</i>			~		
M11				1		//			//
M12	~	~		//	<u>ر</u>	~		~ /	
M13				//					<i>ر</i>
M14									
M15	~	< <		< <		^			//
M16				~		^		~	<i>ر</i>
M17									<i>ر</i>
M18		۲	1	//		^			<i>ر</i>
M19									/ /
M20		۲		//			 	1	<i>ا</i>
M21	~	~		>	^	//		/	

Need for	improved and	collaborative	management			>	//	>			~		
Island	culture/	heritage								~			
Consumer	behaviour				~			~					
Personal	attachment to	coast		^		>		~	^		^		
Education	and	information						//			~		
Need for	more	responsibility				//		~ /		~ ~			
Lack of trust	in	government				~							
Public willing	to change							~		~	~		
Need for	enhanced	awareness				>					>		
Interviewee	code			M22	M23	M24	M25	M26	M27	M28	M29		